UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, DC 20549

FORM 8-K

CURRENT REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

Date of Report (Date of earliest event reported): April 1, 2024

Traws Pharma, Inc.

(Exact name of Registrant as specified in its charter)

Delaware

(State or Other Jurisdiction of Incorporation or Organization)

001-36020 (Commission File Number) 22-3627252 (I.R.S. Employer Identification No.)

12 Penns Trail Newtown, PA 18940 (267) 759-3680

(Address, Including Zip Code, and Telephone Number, Including Area Code, of Registrant's Principal Executive Offices)

Onconova Therapeutics, Inc.

(Former name or former address, if changed since last report)

Check the appropriate box below if the Form 8-K filing is intended to simultaneously satisfy the filing obligation of the registrant under any of the following provisions:

□ Written communications pursuant to Rule 425 under the Securities Act (17 CFR 230.425)

Soliciting material pursuant to Rule 14a-12 under the Exchange Act (17 CFR 240.14a-12)

Pre-commencement communications pursuant to Rule 14d-2(b) under the Exchange Act (17 CFR 240.14d-2(b))

Pre-commencement communications pursuant to Rule 13e-4(c) under the Exchange Act (17 CFR 240.13e-4(c))

Securities registered pursuant to Section 12(b) of the Act:

Title of each class	Trading Symbol(s)	Name of each exchange on which registered
Common Stock, par value \$.01 per share	TRAW	The Nasdaq Stock Market LLC

Indicate by check mark whether the registrant is an emerging growth company as defined in Rule 405 of the Securities Act of 1933 (§230.405 of this chapter) or Rule 12b-2 of the Securities Exchange Act of 1934 (§240.12b-2 of this chapter).

Emerging growth company \Box

Item 1.01. Entry into a Material Definitive Agreement.

Agreement and Plan of Merger

On April 1, 2024, Onconova Therapeutics, Inc., a Delaware corporation ("we," "us,", "our," or the "Company"), acquired Trawsfynydd Therapeutics, Inc., a Delaware corporation ("Trawsfynydd"), in accordance with the terms of an Agreement and Plan of Merger, dated April 1, 2024 (the "Merger Agreement"), by and among the Company, Traws Merger Sub I, Inc., a Delaware corporation ("First Merger Sub"), Traws Merger Sub II, LLC, a Delaware limited liability company ("Second Merger Sub"), and Trawsfynydd. Pursuant to the Merger Agreement, First Merger Sub merged with and into Trawsfynydd was the surviving corporation (the "First Merger"). Immediately following the First Merger, Trawsfynydd merged with and into Second Merger Sub, pursuant to which Second Merger Sub was the surviving entity and a wholly owned subsidiary of the Company (the "Second Merger" and together with the First Merger, the "Merger"). The Merger is intended to qualify as a tax-free reorganization for U.S. federal income tax purposes.

Under the terms of the Merger Agreement, upon the consummation of the Merger on April 1, 2024 (the "Closing"), in exchange for the outstanding shares of capital stock of Trawsfynydd immediately prior to the effective time of the First Merger, the Company issued to the stockholders of Trawsfynydd an aggregate of (A) 3,549,538 shares of common stock of the Company, par value \$0.01 per share (the "Common Stock") and (B) 10,359.0916 shares of Series C Preferred Stock (as defined and described below). Each share of Series C Preferred Stock is convertible into 10,000 shares of Common Stock, subject to certain conditions described below. In addition, the Company assumed all Trawsfynydd stock options immediately outstanding prior to the First Merger, each becoming an option to purchase Common Stock subject to adjustment pursuant to the terms of the Merger Agreement (the "Assumed Options"). No portion of the Assumed Options will be exercisable unless and until the Meeting Proposals (as defined below) are approved by the Company's stockholders. Once exercisable, the Assumed Options will be exercisable for an aggregate of 454,000 shares of Common Stock. Following the effective time of the Second Merger, the Company's changed its name to "Traws Pharma, Inc."

Reference is made to the discussion of the Series C Preferred Stock in Item 5.03 of this Current Report on Form 8-K, which is incorporated into this Item 1.01 by reference.

Tungsten Partners LLC ("Tungsten") acted as financial advisor to the Company in connection with the Merger. As partial compensation for services rendered by Tungsten, the Company issued to Tungsten and its affiliates and designees an aggregate of 168,601 shares of Common Stock and 535.46510 shares of Series C Preferred Stock.

Following the Closing of the Merger and the concurrent Financing (described below), the Company had 25,301,009 shares of Common Stock issued and outstanding.

Pursuant to the Merger Agreement, the Company has agreed to hold a stockholders' meeting to submit the following matters to its stockholders for their consideration: (i) the approval of the conversion of shares of Series C Preferred Stock into shares of Common Stock in accordance with the rules of the Nasdaq Stock Market LLC (the "Conversion Proposal") and (ii) if deemed necessary or appropriate by the Company or as otherwise required by applicable law or contract, the approval of an amendment to the Company's certificate of incorporation, as amended (the "Charter"), to authorize sufficient shares of Common Stock for the conversion of Series C Preferred Stock issued pursuant to the Merger Agreement (the "Share Increase Proposal" and together with the Conversion Proposal, the "Meeting Proposals"). In connection with these matters, the Company has agreed to file a proxy statement on Schedule 14A with the Securities and Exchange Commission (the "SEC").

The Board of Directors of the Company (the "Board") approved the Merger Agreement and the related transactions, and the consummation of the Merger was not subject to approval of Company stockholders.

The foregoing description of the Merger and the Merger Agreement does not purport to be complete and is qualified in its entirety by reference to the full text of the Merger Agreement, a copy of which is filed as Exhibit 2.1 to this Current Report on Form 8-K and is incorporated herein by reference.

The Merger Agreement has been filed herewith to provide investors and security holders with information regarding its terms. It is not intended to provide any other factual information about the Company or Trawsfynydd. The Merger Agreement contains representations, warranties and covenants that the Company and Trawsfynydd made to each other as of specific dates. The assertions embodied in those representations, warranties and covenants were made solely for purposes of the Merger Agreement between the Company and Trawsfynydd and may be subject to important qualifications and limitations agreed to by the Company and Trawsfynydd in connection with negotiating its terms, including being qualified by confidential disclosures exchanged between the parties in connection with the execution of the Merger Agreement. Moreover, the representations and warranties may be subject to a contractual standard of materiality that may be different from what may be viewed as material to investors or securityholders. Moreover, information concerning the subject matter of the representations and warranties may change after the date of the Merger Agreement, which subsequent information may or may not be fully reflected in the Company's public disclosures. For the foregoing reasons, no person should rely on the representations and warranties as statements of factual information at the time they were made or otherwise.

Support Agreements

In connection with the execution of the Merger Agreement, the Company and Trawsfynydd entered into stockholder support agreements (the "Company Stockholder Support Agreements") with certain of the Company's stockholders (solely in their capacity as stockholders of the Company). Pursuant to the Support Agreements, among other things, each of the Company stockholder parties thereto has agreed to vote or cause to be voted all of the shares of Common Stock owned by such stockholder in favor of the Meeting Proposals.

The foregoing description of the Company Stockholder Support Agreements does not purport to be complete and is qualified in its entirety by reference to the full text of the form of the Company Stockholder Support Agreement, a copy of which is included as Exhibit E to the Merger Agreement, which is filed as Exhibit 2.1 to this Current Report on Form 8-K and incorporated herein by reference.

In connection with the execution of the Merger Agreement, the Company and Trawsfynydd entered into stockholder support agreements (the "Trawsfynydd's stockholders (solely in their capacity as stockholders of the Company). Pursuant to the Trawsfynydd Stockholder Support Agreements, among other things, each of the Trawsfynydd stockholders has agreed to the terms and conditions of the Merger Agreement, waive any dissenters rights and release claims such stockholder may have against the Company and Trawsfynydd.

The foregoing description of the Trawsfynydd Stockholder Support Agreements does not purport to be complete and is qualified in its entirety by reference to the full text of the form of the Trawsfynydd Stockholder Support Agreement, a copy of which is included as Exhibit F to the Merger Agreement, which is filed as Exhibit 2.1 to this Current Report on Form 8-K and incorporated herein by reference.

Lock-up Agreements

Concurrently and in connection with the execution of the Merger Agreement, certain Trawsfynydd stockholders as of immediately prior to the Closing, and certain directors, officers, and stockholders of the Company as of immediately prior to the Closing entered into lock-up agreements with the Company and Trawsfynydd, pursuant to which each such stockholder agreed to be subject to a 180-day lockup on the sale or transfer of shares of the Company held by each such stockholder at the Closing, including those shares of Common Stock and Series C Preferred Stock (including the shares of Common Stock into which such Series C Preferred Stock is convertible) received by each such stockholder in the Merger (the "Lock-up Agreements").

The foregoing description of the Lock-up Agreements does not purport to be complete and is qualified in its entirety by reference to the full text of the form of the Lock-up Agreement, a copy of which is included as Exhibit B to the Merger Agreement, which is filed as Exhibit 2.1 to this Current Report on Form 8-K and incorporated herein by reference.

Contingent Value Rights Agreement

Concurrently with the Closing of the Merger, the Company entered into a contingent value rights agreement (the "CVR Agreement") with a rights agent (the "Rights Agent"), pursuant to which each holder of Common Stock as of the applicable record date (April 15, 2024), including those holders receiving shares of Common Stock in connection with the Merger, is entitled to one contractual contingent value right (each, a "CVR") issued by the Company, subject to and in accordance with the terms and conditions of the CVR Agreement, for each share of Common Stock held by such holder as of the applicable record time (5:00 p.m. ET on April 15, 2024).

When issued, each contingent value right will entitle the holder (the "Holder") thereof to distributions of the following, pro-rated on a per-CVR basis, during the CVR Term (as defined in the CVR Agreement):

(i) 43.7% of the Net Proceeds (as defined in the CVR Agreement) received by the Company in a given calendar quarter in the event of the sale, license, transfer or disposition of rights to Rigosertib or Narazaciclib (including any such sale or disposition of equity securities in any subsidiary established by the Company to hold any right, title or interest in or to Rigosertib or Narazaciclib); or

(ii) 6.24% of any Net Sales (as defined in the CVR Agreement) for Rigosertib or Narazaciclib by the Company or any of its affiliates in a given calendar quarter.

The distributions in respect of the CVRs will be made on a quarterly basis, and will be subject to a number of deductions, subject to certain exceptions or limitations, including but not limited to for certain taxes and certain out-of-pocket expenses incurred by the Company.

Under the CVR Agreement, the Rights Agent has, and Holders of at least 30% of the CVRs then-outstanding have, certain rights to audit and enforcement on behalf of all Holders of the CVRs.

The foregoing description of the CVR Agreement does not purport to be complete and is qualified in its entirety by reference to the full text of the form of the CVR Agreement, a copy of which is included as Exhibit D to the Merger Agreement, which is filed as Exhibit 2.1 to this Current Report on Form 8-K and incorporated herein by reference.

Private Placement and Securities Purchase Agreement

On April 1, 2024, the Company entered into a Securities Purchase Agreement (the "Securities Purchase Agreement") with TPAV, LLC, an affiliate of Torrey Pines, and OrbiMed Private Investments VIII, LP, an affiliate of OrbiMed Advisors (the "Investors"). Pursuant to the Securities Purchase Agreement, the Company agreed to issue and sell an aggregate of (i) 496,935 shares of Common Stock and (ii) 1,578.2120 shares of Series C Preferred Stock (the "PIPE Securities") for an aggregate purchase price of approximately \$14 million (collectively, the "Financing"). Each share of Series C Preferred Stock is convertible into 10,000 shares of Common Stock, as described below. The powers, preferences, rights, qualifications, limitations and restrictions applicable to the Series C Preferred Stock are set forth in the Certificate of Designation (as defined and described below).

The closing of the Financing occurred concurrently with the closing of the Merger on April 1, 2024 (the "Financing Closing Date").

The foregoing summary of the Securities Purchase Agreement does not purport to be complete and is qualified in its entirety by reference to the full text of the Securities Purchase Agreement, a copy of which is filed as Exhibit 10.1 to this Current Report on Form 8-K and is incorporated herein by reference.

Registration Rights Agreement

On April 1, 2024, in connection with the Securities Purchase Agreement, the Company entered into a Registration Rights Agreement (the "Registration Rights Agreement") with the holders of Common Stock and Series C Preferred Stock signatory thereto. Pursuant to the Registration Rights Agreement, the Company is required to prepare and file a resale registration statement with the SEC within 90 calendar days following the Financing Closing Date (the "Filing Deadline"), with respect to the shares of Common Stock underlying the PIPE Securities and the Common Stock and Series C Preferred Stock issued to the signatories to the Registration Rights Agreement in the Merger. The Company will use its commercially reasonable efforts to cause such registration statement to be declared effective by the SEC within 30 calendar days of the Filing Deadline (or within 60 calendar days if the SEC reviews the registration statement).

The Company has also agreed to, among other things, indemnify the holders of Common Stock and Series C Preferred Stock signatory thereto, and each of their respective officers, directors, agents, partners, members, managers, stockholders, Affiliates, investment advisers and employees of each of them, each Person who controls any such holder (within the meaning of Section 15 of the Securities Act of 1933, as amended (the "Securities Act"), or Section 20 of the Securities Exchange Act of 1934, as amended (the "Exchange Act")) and the officers, directors, partners, members, managers, stockholders, agents, investment advisers and employees of each such controlling Person under such registration statement from certain liabilities.

The foregoing summary of the Registration Rights Agreement does not purport to be complete and is qualified in its entirety by reference to the full text of the Registration Rights Agreement, a copy of which is filed as Exhibit 10.2 to this Current Report on Form 8-K and is incorporated herein by reference.

Item 2.01 Completion of Acquisition or Disposition of Assets

On April 1, 2024, the Company completed its business combination with Trawsfynydd. The information contained in Item 1.01 of this Current Report on Form 8-K is hereby incorporated by reference into this Item 2.01.

Item 3.02 Unregistered Sales of Equity Securities

The information contained in Item 1.01 and Item 2.01 of this Current Report on Form 8-K is incorporated by reference into this Item 3.02.

Pursuant to the Merger Agreement, the Company issued shares of Common Stock and Series C Preferred Stock, and pursuant to the Securities Purchase Agreement, the PIPE Securities were issued to the Investors, each of which represented that it was an "accredited investor," as defined in Regulation D, and was acquiring the PIPE Securities for investment only and not with a view towards, or for resale in connection with, the public sale or distribution thereof. Such issuances were exempt from the registration requirements of the Securities Act in reliance in Section 4(a)(2) of the Securities Act and/or Regulation D promulgated thereunder.

Neither the shares of Common Stock and Series C Preferred Stock issued in the Merger nor the PIPE Securities have been registered under the Securities Act and none of such Securities may be offered or sold in the United States absent registration or an exemption from registration under the Securities Act and any applicable state securities laws.

Neither this Current Report on Form 8-K nor any of the exhibits attached hereto will constitute an offer to sell or the solicitation of an offer to buy shares of Common Stock, Series C Preferred Stock or any other securities of the Company.

Item 3.03 Material Modification to Rights of Security Holders.

To the extent required by Item 3.03 of Form 8-K, the information contained in Item 5.03 of this Current Report on Form 8-K is incorporated by reference into this Item 3.03.

Item 5.02 Departure of Directors or Certain Officers; Election of Directors; Appointment of Certain Officers; Compensatory Arrangements of Certain Officers.

Pursuant to the Merger Agreement, on April 1, 2024, effective immediately upon the Closing, the Board, upon the recommendation of the Nominating and Corporate Governance Committee of the Board, (a) accepted the resignations of Dr. Steven M. Fruchtman, Peter Atadja, Jerome Groopman, and Viren Mehta from the Board; (b) accepted the resignations of Dr. Steven M. Fruchtman and Mark Guerin from their roles of Chief Executive Officer ("CEO") of the Company and Chief Operating Officer ("COO") of the Company, respectively; (c) appointed Iain Dukes as a director and Executive Chairman of the Company, Nikolay Savchuk as a director of the Company, and Werner Cautreels as a director of the Company; and (d) appointed Werner Cautreels as the Company's CEO and Nikolay Savchuk as the Company's COO. Dr. Steven M. Fruchtman remained as the President of the Company and was appointed Chief Scientific Officer, Oncology and Mark Guerin remained as the Chief Financial Officer of the Company.

Reference is made to the descriptions of such agreements in Item 1.01 of this Current Report on Form 8-K, which are incorporated into this Item 5.02 by reference.

In addition, effective immediately upon the Closing, C. David Pauza was appointed the Company's Chief Scientific Officer, Virology and Robert Redfield was appointed the Company's Chief Medical Officer.

Appointment of Werner Cautreels as Director and Chief Executive Officer

On April 1, 2024, the Board appointed Werner Cautreels, age 71, as a director and CEO of the Company, effective as of the Closing. Dr. Werner Cautreels is a highly accomplished biopharmaceutical executive with a core emphasis in research and development in various therapeutic areas, who brings a deep understanding of clinical and regulatory strategy. During his 40-year plus career, his work has touched on cardiovascular, autoimmune, oncology, rare disease, and vaccines. Dr. Cautreels was President and CEO of Selecta Biosciences from July 2010 until 2018. Prior to Selecta Biosciences, Dr. Cautreels served as Global CEO of Solvay Pharmaceuticals until it was acquired by Abbott Laboratories in 2010. Prior to joining Solvay, he worked at Sanofi, Sterling Winthrop and Nycomed-Amersham in a variety of research and development management positions in Europe and the United States. Dr. Cautreels was also a Director of Innogenetics NV (Gent, Belgium) and of Arqule Inc. (Woburn, Massachusetts). Until April 2019, Dr. Cautreels was Director and Chair of the Audit Committee of Galapagos NV (Mechelen, Belgium). Dr. Cautreels currently serves on the board of directors of Third Pole Therapeutics, a privately held company developing critical life-sustaining therapies for people living with cardiopulmonary and infectious diseases, and on the advisory board of Thuja Capital, an early-stage venture capital firm. Dr. Cautreels also currently serves as CEO of Cristal Therapeutics (Maastricht, The Netherlands) and Chairman of MRM Health (Gent, Belgium). Dr. Cautreels has a Ph.D. in chemistry from the University of Antwerp, Belgium, and an Executive M.B.A. from Harvard Business School.

There are no arrangements or understandings between Dr. Cautreels and any other person pursuant to which he was selected as a director and an executive officer other than in connection with the Merger as described herein, and there are no family relationships between Dr. Cautreels and any of the Company's directors or executive officers. Dr. Cautreels has no direct or indirect material interest in any existing or currently proposed transaction that would require disclosure under item 404(a) of Regulation S-K.

In connection with Dr. Cautreels appointment as CEO, Dr. Cautreels and the Company entered into an employment agreement (the "CEO Employment Agreement"), effective as of the Closing. The CEO Employment Agreement has an initial term of one year, unless terminated sooner by Dr. Cautreels or the Company, and the term renews for additional one year periods, unless either party provides written notice of termination at least 90 days prior to the end of the applicable term. The CEO Employment Agreement provides that Dr. Cautreels will receive an annual base salary of \$600,000 and will be eligible to receive discretionary annual bonuses in the amounts and in the form determined by the Board. Dr. Cautreels will be eligible to participate in the Company's health, retirement, paid time-off and other benefit plans pursuant to the term of such plans.

The CEO Employment Agreement provides for the grant of 200,000 restricted stock units, which grant is intended to be an inducement grant under Rule 5635(c)(4) of the Nasdaq Stock Market Listing Rules (the "CEO Inducement Grant"). The CEO Inducement Grant vests as to 25% of the units on each of the first four anniversaries of the date of grant, subject to Dr. Cautreels' continued employment on the appliable vesting date. The CEO Inducement Grant will be subject to the terms and conditions consistent with the CEO Employment Agreement and other terms and conditions consistent with like grants made to other senior executives of the Company.

Under the CEO Employment Agreement, if Dr. Cautreels' employment is terminated by the Company without cause or by Dr. Cautreels for good reason (as each term is defined in the CEO Employment Agreement) prior to a change in control of the Company or following the first anniversary of a change in control of the Company, Dr. Cautreels will be eligible to receive the following severance benefits, if he signs and does not revoke an effective release of claims: (i) to the extent approved, accrued, and unpaid, the annual bonus (if any) with respect to the fiscal year ended immediately prior to Dr. Cautreels' termination of employment; (ii) the sum of (x) his base salary, plus (y) any target bonus; (iii) accelerated vesting of any outstanding unvested equity awards that vest based on continued service; and (iv) Company-paid COBRA benefits for up to 12 months following termination of employment.

If Dr. Cautreels' employment is terminated by the Company without cause or by Dr. Cautreels for good reason upon or within the one-year period following a change in control of the Company, Dr. Cautreels will be eligible to receive the following severance benefits, if he signs and does not revoke an effective release of claims: (i) to the extent approved, accrued, and unpaid, the annual bonus (if any) with respect to the fiscal year ended immediately prior to Dr. Cautreels' termination of employment; (ii) one and one-half times the sum of (x) his base salary, plus (y) any target bonus; (iii) accelerated vesting of any outstanding unvested equity awards that vest based on continued service; and (iv) Company-paid COBRA benefits for up to 18 months following termination of employment.

In the event that Dr. Cautreels' employment terminates for any reason, including by the Company with or without cause, by Dr. Cautreels with or without good reason, or due to Mr. Cautreel's death or disability, Dr. Cautreels (or his spouse or estate, as applicable, in the event of his death) will be entitled to receive the balance of any accrued but unpaid base salary, unreimbursed expenses, and unused, accrued vacation time through his termination date.

Pursuant to the terms of the CEO Employment Agreement, Dr. Cautreels is subject to customary restrictive covenants, including: (i) a duty not to compete with the Company for the 12-month period following his termination of employment, (ii) an obligation not to solicit any director, officer, employee, or agent of the Company for the 12-month period following his termination of employment, and (iii) an agreement to assign his rights, title, and interest in and to any and all inventions created during his employment with the Company.

The foregoing summary of the CEO Employment Agreement does not purport to be complete and is qualified in its entirety by reference to the full text of the CEO Employment Agreement. The full text of the CEO Employment Agreement is attached as Exhibit 10.3 to this Current Report on Form 8-K and is incorporated herein by reference.

Appointment of Iain Dukes as Executive Chairman

On April 1, 2024, the Board appointed Iain Dukes, age 65, as Executive Chairman of the Company, effective as of the Closing. Dr. Dukes is a Venture Partner of OrbiMed, a global investment firm, since 2016. He previously served as Senior Vice President and Head of Business Development and Licensing for Merck Research Laboratories. Prior to joining Merck, Dr. Dukes was Vice President of External Research and Development at Amgen. He has also served as President and Chief Executive Officer, as well as a member of the Board of Directors of Essentialis Therapeutics, a clinical stage biotechnology company focused on the development of breakthrough medicines for the treatment of rare metabolic diseases. Previously, Dr. Dukes was Vice President of Scientific and Technology Licensing at GlaxoSmithKline, and he held various positions at Glaxo Wellcome, including Head of Exploratory Development for Metabolic and Urogenital Diseases and Head of Ion Channel Drug Discovery Group. Dr. Dukes is currently the chairman of the board of directors of Iovance Biotherapeutics, Inc. (Nasdaq: IOVA) and Executive Chairman of Angiex Inc. and also serves on the board of directors of Kartos Therapeutics, Feldan Therapeutics and Rathlin Therapeutics. Dr. Dukes also co-founded and serves on the board of directors of Kartos Therapeutics, and co-founded Telios Pharma, where he serves as President. He holds an M.A in Jurisprudence. and D.Phil. from the University of Oxford, an M.Sc. in Cardiovascular Studies from the University of Leeds, and a B.Sc.. in Pharmacology from the University of Bath.

There are no arrangements or understandings between Dr. Dukes and any other person pursuant to which he was selected as a director and an executive officer other than in connection with the Merger as described herein, and there are no family relationships between Dr. Dukes and any of the Company's directors or executive officers. Other than as described below, Dr. Dukes has no direct or indirect material interest in any existing or currently proposed transaction that would require disclosure under item 404(a) of Regulation S-K.

Trawsfynydd entered into that certain Master Research and Development Agreement Expert Systems, Inc. ("Expert Systems"), dated September 1, 2022, pursuant to which Expert Systems provides consulting services and preclinical drug development services to Trawsfynydd. Dr. Dukes is a stockholder of Expert Systems and a member of Expert Systems' board of directors.

Trawsfynydd entered into that certain Master Research and Development Agreement with Viriom, Inc. ("Viriom"), dated January 5, 2022, pursuant to which Viriom provides service related to the research and development in the area of virology. Trawsfynydd also entered into that certain License Agreement with Viriom, dated January 20, 2023, pursuant to which Trawsfynydd licensed certain intellectual property from Viriom. Dr. Dukes is a stockholder of Viriom and a member of its board of directors.

In connection with Dr. Dukes' appointment as Executive Chairman of the Company, he has entered into an offer letter with the Company (the "Dukes Offer Letter"), effective as of the Closing. The Dukes Offer Letter provides that Dr. Dukes will be an employee at-will. Pursuant to the Dukes Offer Letter, Dr. Dukes will receive an annual base salary of \$350,000 and will be eligible to receive a discretionary annual bonus in the amount and in the form determined by the Board. In addition, Dr. Dukes will be eligible to participate in the Company's health, retirement, and other benefit plans in accordance with the terms of such plans. Dr. Dukes, as Executive Chairman, will report to the Board.

Under the Dukes Offer Letter, Dr. Dukes received a restricted stock unit award with respect to 67,500 restricted stock units of the Company, which are intended to be inducement grants under Rule 5635(c)(4) of the Nasdaq Stock Market Listing Rules, and vests as to 25% of the units on each of the first four anniversaries of the date of grant, subject to continued employment on the applicable vesting date.

The foregoing description of the Dukes Offer Letter does not purport to be complete and is qualified in its entirety by reference to the full text of the Dukes Offer Letter, the form of which is attached as Exhibit 10.4 to this Current Report on Form 8-K and is incorporated herein by reference.

Appointment of Nikolay Savchuk as Director and Chief Operating Officer

On April 1, 2024, the Board appointed Nikolay Savchuk, age 55, as a director and COO of the Company, effective as of the Closing. Dr. Savchuk is the cofounder and Managing Member at Torrey Pines Investment LLC, a life-science investment company, since 2002. Dr. Savchuk is also the Managing General Partner of Teal Ventures, LP, a venture capital firm focused on early-stage investing in health technology, since October 2018. Prior to joining life science investment and research in the United States, Dr. Savchuk was a fond of and held various business and research management positions in the IT industry with companies in Singapore and Russia. Dr. Savchuk is currently the chairman of the board of directors of Viriom Inc., a company dedicated to advancing a pipeline of effective and affordable treatments for diseases of global interest, and chairman of the board of directors of ChemDiv Inc., a company dedicated to partnering in discovery and development of breakthrough therapies based on its unique chem-bio platforms. Dr. Savchuk obtained his M.S. degree in physics and his Ph.D. in applied mathematics from Moscow Institute of Physics and Technology.

There are no arrangements or understandings between Dr. Savchuk and any other person pursuant to which he was selected as a director and an executive officer other than in connection with the Merger as described herein, and there are no family relationships between Dr. Savchuk and any of the Company's directors or executive officers. Other than as described below, Dr. Savchuk has no direct or indirect material interest in any existing or currently proposed transaction that would require disclosure under item 404(a) of Regulation S-K.

Trawsfynydd entered into that certain Master Research And Development Agreement with ChemDiv, Inc. ("ChemDiv"), dated September 23, 2022, pursuant to which ChemDiv provides services related to preclinical drug discovery to Trawsfynydd. Dr. Savchuk is a stockholder of ChemDiv and a member of its board of directors.

Trawsfynydd entered into that certain Master Research and Development Agreement with Viriom, dated January 5, 2022, pursuant to which Viriom provides service related to the research and development in the area of virology. Trawsfynydd also entered into that certain License Agreement with Viriom, dated January 20, 2023, pursuant to which Trawsfynydd licensed certain intellectual property from Viriom. Dr. Savchuk is a stockholder of Viriom and a member of its board of directors.

Trawsfynydd also entered into the Securities Purchase Agreement with TPAV, LLC, which is managed by Dr. Savchuk. Reference is made to the description of the Securities Purchase Agreement in Item 1.01 of this Current Report on Form 8-K, which is incorporated into this Item 5.02 by reference.

In connection with Dr. Savchuk's appointment as Executive Chairman of the Company, he has entered into an offer letter with the Company (the "Savchuk Offer Letter"), effective as of the Closing. The Savchuk Offer Letter provides that Dr. Savchuk will be an employee at-will. Pursuant to the Savchuk Offer Letter, Dr. Savchuk will receive an annual base salary of \$350,000 and will be eligible to receive a discretionary annual bonus in the amount and in the form determined by the Board. In addition, Dr. Savchuk will be eligible to participate in the Company's health, retirement, and other benefit plans in accordance with the terms of such plans. Dr. Savchuk will report to the CEO of the Company.

Under the Savchuk Offer Letter, Dr. Savchuk received a restricted stock unit award with respect to 67,500 restricted stock units of the Company, which are intended to be inducement grants under Rule 5635(c)(4) of the Nasdaq Stock Market Listing Rules and vests as to 25% of the units on each of the first four anniversaries of the date of grant, subject to continued employment on the applicable vesting date.

The foregoing description of the Savchuk Offer Letter does not purport to be complete and is qualified in its entirety by reference to the full text of the Savchuk Offer Letter, the form of which is attached as Exhibit 10.4 to this Current Report on Form 8-K and is incorporated herein by reference.

Indemnification Agreements

In connection with the appointment of each of Dr. Cautreels, Dr. Dukes and Dr. Savchuk entered into the Company's standard form of indemnification agreement, a copy of which is filed as Exhibit 10.8 to the Company's Annual Report on Form 10-K for the year ended December 31, 2023.

Item 5.03. Amendments to Articles of Incorporation or Bylaws; Change in Fiscal Year.

Certificate of Designation

On April 1, 2024, the Company filed a Certificate of Designation of Preferences, Rights and Limitations of the Series C Non-Voting Convertible Preferred Stock (the "Certificate of Designation") with the Secretary of State of the State of Delaware in connection with the Merger referenced in Item 1.01 above. The Certificate of Designation provides for the designation of shares of the Company's Series C Non-Voting Convertible Preferred Stock, par value \$0.01 per share (the "Series C Preferred Stock").

Holders of Series C Preferred Stock are entitled to receive dividends on shares of Series C Preferred Stock equal to, on an as-if-converted-to-Common-Stock basis, and in the same form as dividends actually paid on shares of the Common Stock.

Except as otherwise required by law, the Series C Preferred Stock does not have voting rights. However, as long as any shares of Series C Preferred Stock are outstanding, the Company will not, without the affirmative vote of the holders of a majority of the then-outstanding shares of the Series C Preferred Stock, (i) alter or change adversely the powers, preferences or rights given to the Series C Preferred Stock or alter or amend the Certificate of Designation, amend or repeal any provision of, or add any provision to, the Charter or bylaws of the Company, or file any articles of amendment, certificate of designations, preferences, limitations and relative rights of any series of preferred stock, in each case if any such action would adversely alter or change the preferences, rights, privileges or powers of, or restrictions provided for the benefit of the Series C Preferred Stock, regardless of whether any of the foregoing actions shall be by means of amendment to the Charter or by merger, consolidation, recapitalization, reclassification, conversion or otherwise, (ii) issue further shares of Series C Preferred Stock, (iii) prior to the earlier of stockholder approval of the Conversion or the six-month anniversary of the Closing, consummate either: (A) any Fundamental Transaction (as in the Certificate of Designation) or (B) any stock sale to, or any merger, consolidation or other business combination of the Company with or into, another entity in which the stockholders of the Company immediately before such transaction do not hold at least a majority of the capital stock of the Company immediately after such transaction, or (iv) enter into any agreement with respect to any of the foregoing.

The Series C Preferred Stock does not have a preference upon any liquidation, dissolution or winding-up of the Company.

Following stockholder approval of the Conversion Proposal, each share of Series C Preferred Stock will automatically convert into 10,000 shares of Common Stock, subject to certain limitations, including that a holder of Series C Preferred Stock is prohibited from converting shares of Series C Preferred Stock into shares of Common Stock if, as a result of such conversion, such holder, together with its affiliates, would beneficially own more than a specified percentage (to be established by the holder between 4.9% and 19.9%) of the total number of shares of Common Stock issued and outstanding immediately after giving effect to such conversion.

The Series C Preferred Stock is redeemable for cash at the option of the holder thereof at any time following the date that is six months after the initial issuance of the Series C Preferred Stock or following any failure to deliver shares of Common Stock in accordance with the terms of the Series C Preferred Stock, at a price per share equal to the then-current fair value of the Series C Preferred Stock, as described in the Certificate of Designation.

The foregoing description of the Series C Preferred Stock does not purport to be complete and is qualified in its entirety by reference to the full text of the Certificate of Designation, a copy of which is filed as Exhibit 3.1 to this Current Report on Form 8-K and is incorporated herein by reference.

Certificate of Amendment

On April 2, 2024, the Company changed its name to Traws Pharma, Inc. pursuant to a certificate of amendment to its Tenth Amended and Restated Certificate of Incorporation filed with the Secretary of State of the State of Delaware (the "Name Change"). Pursuant to the Delaware General Corporation Law, a stockholder vote was not necessary to effectuate the Name Change and it does not affect the rights of the Company's stockholders.

In addition, effective at the open of market trading on April 3, 2024, the Company's Common Stock ceased trading under the ticker symbol "ONTX" and began trading on the Nasdaq Stock Market under the ticker symbol "TRAW".

A copy of the Company's Certificate of Amendment to the Tenth Amended and Restated Certificate of Incorporation is filed as Exhibit 3.2 with this Current Report on Form 8-K and is incorporated herein by reference.

Item 7.01. Regulation FD Disclosure.

Press Release

On April 2, 2024, the Company issued a press release announcing the Closing. A copy of the press release is furnished as Exhibit 99.1 to this Current Report on Form 8-K.

The information in Item 7.01 of this Current Report on Form 8-K, including the information in the press release attached as Exhibit 99.1 to this Current Report on Form 8-K, is furnished pursuant to Item 7.01 of Form 8-K and shall not be deemed "filed" for the purposes of Section 18 of the Exchange Act or otherwise subject to the liabilities of that section. Furthermore, the information in Item 7.01 of this Current Report on Form 8-K, shall not be deemed to be incorporated by reference in the filings of the Company under the Securities Act.

Item 9.01 Financial Statements and Exhibits.

(a) Financial statements of business acquired

The financial statements required by this Item, with respect to the Merger described in Item 2.01 herein, are expected to be filed by amendment as soon as practicable, and in any event not later than 71 days after the date on which this Current Report on Form 8-K is required to be filed related to Item 2.01.

(b) Pro forma financial information

The pro forma financial information required by this Item, with respect to the Merger described in Item 2.01 herein, are expected to be filed by amendment as soon as practicable, and in any event not later than 71 days after the date on which this Current Report on Form 8-K is required to be filed related to Item 2.01.

Forward Looking Statements

Any statements in this Current Report about the future expectations, plans and prospects of the Company, including without limitation, statements regarding: the Merger, stockholder approval of the conversion of the Series C Preferred Stock, and other statements containing the words "anticipate," "believe," "continue," "could," "estimate," "expect," "hypothesize," "intend," "may," "plan," "potential," "predict," "project," "should," "target," "would," and similar expressions, constitute forward-looking statements within the meaning of The Private Securities Litigation Reform Act of 1995. Actual results may differ materially from those indicated by such forward-looking statements as a result of various important factors, including, but not limited to those set forth under the caption "Risk Factors" in this Current Report on Form 8-K and in the Company's most recent Annual Report on Form 10-K filed with the SEC, as supplemented by its subsequent Quarterly Reports on Form 10-Q and in other filings that makes with the SEC. In addition, any forward-looking statements included in this Current Report represent the Company's views only as of the date hereof and should not be relied upon as representing its views as of any subsequent date. The Company specifically disclaims any intention to update any forward-looking statements included in this Current Report.

No Offer or Solicitation; Important Information About the Merger and Where to Find It

This Current Report on Form 8-K is not a proxy statement or solicitation of a proxy, consent or authorization with respect to any securities or in respect of the Merger and shall not constitute an offer to sell or a solicitation of an offer to buy the securities of the Company or Trawsfynydd, nor shall there be any sale of any such securities in any state or jurisdiction in which such offer, solicitation, or sale would be unlawful prior to registration or qualification under the securities laws of such state or jurisdiction. No offer of securities shall be made, except by means of a prospectus meeting the requirements of Section 10 of the Securities Act or an exemption therefrom.

The Company expects to file a proxy statement with the SEC relating to the Meeting Proposals. The definitive proxy statement will be sent to all Company stockholders. Before making any voting decision, investors and security holders of the Company are urged to read the proxy statement and all other relevant documents filed or that will be filed with the SEC in connection with the Meeting Proposals as they become available because they will contain important information about the Merger Agreement and related transactions and the Meeting Proposals to be voted upon. Investors and security holders will be able to obtain free copies of the proxy statement and all other relevant documents filed or that will be filed with the SEC by the Company through the website maintained by the SEC at www.sec.gov.

Participants in Solicitation

The Company, Trawsfynydd, and their respective directors, executive officers and employees may be deemed to be participants in the solicitation of proxies in respect of the Merger. Information regarding the Company's directors and executive officers is available in the Company's <u>Definitive Proxy</u> <u>Statement filed with the SEC on June 7, 2023</u> under "Proposal One – Election of Directors.". Information regarding the persons who may, under the rules of the SEC, be deemed participants in the proxy solicitation and a description of their direct and indirect interests, by security holdings or otherwise, will be contained in the proxy statement and other relevant materials to be filed with the SEC when they become available.

(d)	Exhibits
-----	----------

Exhibit	Description
Number 2.1	Description Agreement and Plan of Merger, dated April 1, 2024, by and among the Onconova Therapeutics, Inc., Traws Merger Sub I, Inc., Traws
	Merger Sub II, LLC, and Trawsfynydd Therapeutics, Inc.
<u>3.1</u>	Certificate of Designation of Series C Non-Voting Convertible Preferred Stock of Onconova Therapeutics, Inc., dated April 1, 2024.
<u>3.2</u>	Certificate of Amendment to Tenth Amended and Restated Certificate of Incorporation of Onconova Therapeutics, Inc., as amended, dated April 2, 2024.
<u>10.1*</u>	Securities Purchase Agreement, dated April 1, 2024, by and among the Onconova Therapeutics, Inc., OrbiMed and TorreyPines.
<u>10.2</u>	Registration Rights Agreement, dated April 1, 2024, by and among the Onconova Therapeutics, Inc., OrbiMed and TorreyPines.
<u>10.3</u>	Employment Agreement, dated April 1, 2024, by and between Onconova Therpeutics, Inc. and Werner Cautreels.
<u>10.4</u>	Form of Offer Letter
<u>99.1</u>	Press Release of Traws Pharma, Inc.
104	Cover Page Interactive Data File (embedded within the Inline XBRL document).

* Certain annexes, schedules and exhibits have been omitted pursuant to Item 601(a)(5) of Regulation S-K. The registrant agrees to furnish supplementally a copy of any omitted attachment to the SEC on a confidential basis upon request.

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned hereunto duly authorized.

TRAWS PHARMA, INC.

Date: April 3, 2024

By: /s/ MARK GUERIN

Mark Guerin Chief Financial Officer

AGREEMENT AND PLAN OF MERGER

by and among:

ONCONOVA THERAPEUTICS, INC.,

a Delaware corporation;

TRAWS MERGER SUB I, INC.,

a Delaware corporation;

TRAWS MERGER SUB II, LLC,

a Delaware limited liability company;

and

TRAWSFYNYDD THERAPEUTICS, INC.,

a Delaware corporation

dated as of April 1, 2024

TABLE OF CONTENTS

		Page
SECTION 1.	DESCRIPTION OF TRANSACTION	3
1.1	The Merger	3
1.2	Effects of the Merger	3
1.3	Closing; First Effective Time; Second Effective Time	3
1.4	Certificate of Designation; Certificate of Incorporation and Bylaws; Directors and Officers	4
1.5	Merger Consideration; Effect of Merger on Company Capital Stock	5
1.6	Conversion of Shares	5
1.7	Closing of the Company's Transfer Books	6
1.8	Exchange of Shares	7
1.9	Appraisal Rights	8
1.10	Company Options	8
1.11	Contingent Value Right	9
1.12	Further Action	9
1.13	Withholding	9
SECTION 2.	REPRESENTATIONS AND WARRANTIES OF THE COMPANY	10
2.1	Due Organization; Subsidiaries	10
2.2	Organizational Documents	10
2.3	Authority; Binding Nature of Agreement	10
2.4	Vote Required	11
2.5	Non-Contravention; Consents	11
2.6	Capitalization	12
2.7	Financial Statements	13
2.8	Absence of Changes	14
2.9	Absence of Undisclosed Liabilities	16
2.10	Title to Assets	16
2.11	Real Property; Leasehold	16
2.12	Intellectual Property; Privacy	17
2.13	Agreements, Contracts and Commitments	19
2.14	Compliance; Permits; Restrictions	21
2.15	Legal Proceedings; Orders	21
2.16	Tax Matters	22
2.17	Employee and Labor Matters; Benefit Plans	23

-i-

TABLE OF CONTENTS (continued)

		Page
2.18	Environmental Matters	26
2.19	Insurance	26
2.20	No Financial Advisors	27
2.21	Transactions with Affiliates	27
2.22	Anti-Bribery	27
2.23	Disclaimer of Other Representations or Warranties	27
SECTION 3.	REPRESENTATIONS AND WARRANTIES OF PARENT AND MERGER SUBS	28
3.1	Due Organization; Subsidiaries	28
3.2	Organizational Documents	29
3.3	Authority; Binding Nature of Agreement	29
3.4	Vote Required	29
3.5	Non-Contravention; Consents	29
3.6	Capitalization	30
3.7	SEC Filings; Financial Statements	32
3.8	Absence of Changes	34
3.9	Absence of Undisclosed Liabilities	36
3.10	Title to Assets	36
3.11	Real Property; Leasehold	36
3.12	Intellectual Property; Privacy	37
3.13	Agreements, Contracts and Commitments	39
3.14	Compliance; Permits	41
3.15	Legal Proceedings; Orders	41
3.16	Tax Matters	42
3.17	Employee and Labor Matters; Benefit Plans	44
3.18	Environmental Matters	46
3.19	Transactions with Affiliates	46
3.20	Insurance	46
3.21	Opinion of Financial Advisor	46
3.22	No Financial Advisors	46
3.23	Anti-Bribery	46
3.24	Valid Issuance	48
3.25	Disclaimer of Other Representations or Warranties	48

TABLE OF CONTENTS (continued)

4.1 Parent Stockholders' 14.2 Proxy Statement	EEMENTS OF THE PARTIES Meeting Common Stock; Issuance of Shares of Parent Common Stock	48 48 49 50
 4.2 Proxy Statement 4.3 Reservation of Parent 4.4 Employee Benefits 4.5 Indemnification of Of 4.6 Additional Agreement 4.7 Listing 4.8 Tax Matters 4.9 Legends 4.10 Directors and Officers 4.11 Section 16 Matters 		49
 4.3 Reservation of Parent 4.4 Employee Benefits 4.5 Indemnification of Of 4.6 Additional Agreemen 4.7 Listing 4.8 Tax Matters 4.9 Legends 4.10 Directors and Officers 4.11 Section 16 Matters 	Common Stock; Issuance of Shares of Parent Common Stock	
 4.4 Employee Benefits 4.5 Indemnification of Of 4.6 Additional Agreemen 4.7 Listing 4.8 Tax Matters 4.9 Legends 4.10 Directors and Officers 4.11 Section 16 Matters 	Common Stock; Issuance of Shares of Parent Common Stock	50
 4.5 Indemnification of Of 4.6 Additional Agreemen 4.7 Listing 4.8 Tax Matters 4.9 Legends 4.10 Directors and Officers 4.11 Section 16 Matters 		30
 4.6 Additional Agreemen 4.7 Listing 4.8 Tax Matters 4.9 Legends 4.10 Directors and Officers 4.11 Section 16 Matters 		50
 4.7 Listing 4.8 Tax Matters 4.9 Legends 4.10 Directors and Officers 4.11 Section 16 Matters 	fficers and Directors	51
 4.8 Tax Matters 4.9 Legends 4.10 Directors and Officers 4.11 Section 16 Matters 	ts	52
4.9 Legends4.10 Directors and Officers4.11 Section 16 Matters		53
4.10 Directors and Officers4.11 Section 16 Matters		53
4.11 Section 16 Matters		53
	S	53
4.12 Closing Contificator		53
4.12 Closing Certificates		54
4.13 Takeover Statutes		54
4.14 Obligations of Merge	r Subs	54
4.15 Private Placement		54
4.16 Audited Financial Sta	itements	54
SECTION 5. CONDITIONS PREC	CEDENT TO OBLIGATIONS OF EACH PARTY	55
5.1 No Restraints		55
5.2 Certificate of Designa	ation	55
5.3 Parent Financing		55
SECTION 6. CLOSING DELIVER	RIES OF THE COMPANY	55
6.1 Documents		55
6.2 FIRPTA Certificate		55
6.3 Company Lock-Up A	greements	55
6.4 Company Stockholde	r Support Agreements	56
6.5 Registration Rights A	greements	56
SECTION 7. CLOSING DELIVER	RIES OF PARENT	56
7.1 Documents		56
7.2 Parent Lock-Up Agre	ements	56
7.3 Registration Rights A		

-iii-

TABLE OF CONTENTS (continued)

		Page
SECTION 8.	MISCELLANEOUS PROVISIONS	56
8.1	Non-Survival of Representations and Warranties	56
8.2	Amendment	56
8.3	Waiver	57
8.4	Entire Agreement; Counterparts; Exchanges by Electronic Transmission	57
8.5	Applicable Law; Jurisdiction	57
8.6	Attorneys' Fees	57
8.7	Assignability	57
8.8	Notices	58
8.9	Cooperation	58
8.10	Severability	59
8.11	Other Remedies; Specific Performance	59
8.12	No Third-Party Beneficiaries	59
8.13	Construction	59
8.14	Expenses	60

-iv-

Exhibits:

Exhibit A	Definitions	A-1
Exhibit B	Form of Lock-Up Agreement	B-1
Exhibit C	Form of Certificate of Designation	C-1
Exhibit D	Form of CVR Agreement	D-1
Exhibit E	Form of Parent Stockholder Support Agreement	E-1
Exhibit F	Form of Company Stockholder Support Agreement	F-1
Exhibit G	Letter of Transmittal	G-1
Exhibit H	Registration Rights Agreement	H-1

AGREEMENT AND PLAN OF MERGER

THIS AGREEMENT AND PLAN OF MERGER is made and entered into as of April 1, 2024, by and among ONCONOVA THERAPEUTICS, INC., a Delaware corporation ("*Parent*"), TRAWS MERGER SUB I, INC., a Delaware corporation and wholly owned subsidiary of Parent ("*First Merger Sub*"), TRAWS MERGER SUB II, LLC, a Delaware limited liability company and wholly owned subsidiary of Parent ("*Second Merger Sub*" and together with First Merger Sub, "*Merger Subs*"), and TRAWSFYNYDD THERAPEUTICS, INC., a Delaware corporation (the "*Company*"). Certain capitalized terms used in this Agreement are defined in Exhibit A.

RECITALS

A. Parent and the Company desire to enter into a business combination as contemplated by this Agreement based on the mutually agreed values of each Parent and the Company.

B. Parent and the Company intend to effect a merger of First Merger Sub with and into the Company (the "*First Merger*") in accordance with this Agreement and the DGCL. Upon consummation of the First Merger, First Merger Sub will cease to exist and the Company will become a wholly owned subsidiary of Parent.

C. Immediately following the First Merger and as part of the same overall transaction as the First Merger, the Company will merge with and into Second Merger Sub (the "*Second Merger*" and, together with the First Merger, the "*Merger*"), with Second Merger Sub being the surviving entity of the Second Merger;

D. The Parties intend that, (i) the First Merger and the Second Merger, taken together, will constitute an integrated transaction described in Rev. Rul. 2001-46, 2001-2 C.B. 321 that qualifies as a "reorganization" within the meaning of Section 368(a) of the Code, and (ii) this Agreement will constitute, and is hereby adopted as, a plan of reorganization within the meaning of Treasury Regulations Sections 1.368-2(g) and 1.368-3(a).

E. The Parent Board has (i) determined that the Contemplated Transactions are fair to, advisable and in the best interests of Parent and its stockholders, (ii) approved and declared advisable this Agreement and the Contemplated Transactions, including the issuance of the Parent Common Stock Payment Shares and the Parent Preferred Stock Payment Shares to the stockholders of the Company pursuant to the terms of this Agreement, and (iii) determined to recommend, upon the terms and subject to the conditions set forth in this Agreement, that the stockholders of the Parent vote to approve the Parent Stockholder Matters at the Parent Stockholders' Meeting to be convened following the Closing.

F. The First Merger Sub Board has (i) determined that the Contemplated Transactions are fair to, advisable, and in the best interests of First Merger Sub and its sole stockholder, (ii) approved and declared advisable this Agreement and the Contemplated Transactions and (iii) determined to recommend, upon the terms and subject to the conditions set forth in this Agreement, that the stockholder of First Merger Sub votes to adopt this Agreement and thereby approve the Contemplated Transactions.

G. The sole member of the Second Merger Sub has (i) determined that the Contemplated Transactions are fair to, advisable, and in the best interests of Second Merger Sub and its sole member, (ii) approved and declared advisable this Agreement and the Contemplated Transactions and (iii) determined to recommend, upon the terms and subject to the conditions set forth in this Agreement, that the sole member of Second Merger Sub votes to adopt this Agreement and thereby approve the Contemplated Transactions.

H. The Company Board has (i) determined that the Contemplated Transactions are fair to, advisable and in the best interests of the Company and its stockholders, (ii) approved and declared advisable this Agreement and the Contemplated Transactions and (iii) recommended, upon the terms and subject to the conditions set forth in this Agreement, that the stockholders of the Company vote to approve the Company Stockholder Matters (the "*Board Approval*").

I. Subsequent to the Board Approval, but prior to the execution and delivery of this Agreement, all of the Company stockholders by written consent and in accordance with the Company's certificate of incorporation, the Company's bylaws and the DGCL (i) approved and adopted this Agreement and the Contemplated Transactions, (ii) acknowledged that the approval given thereby is irrevocable and that such stockholder is aware of its rights to demand appraisal for its shares pursuant to Section 262 of the DGCL, a true and correct copy of which was attached thereto, and that such stockholder has received and read a copy of Section 262 of the DGCL, and (iii) acknowledged that by its approval of the Merger it is not entitled to appraisal rights with respect to its shares in connection with the Merger and thereby waives any rights to receive payment of the fair value of its capital stock under the DGCL (such matters, the "*Company Stockholder Matters*" and the consent, the "*Stockholder Written Consent*"), and the Stockholder Written Consent is to become effective by its terms immediately following the execution of this Agreement by the parties hereto.

J. Concurrently with the execution and delivery of this Agreement and as a condition and inducement to each of Parent and the Company's willingness to enter into this Agreement, all of the directors, all of the officers and the stockholders of Parent listed in <u>Section A-1</u> of the Parent Disclosure Schedule (solely in their capacity as stockholders of Parent) (the "*Parent Signatories*") and the stockholders of the Company listed in <u>Section A-1</u> of the Company Disclosure Schedule (the "*Company Signatories*") (solely in their capacity as stockholders of the Company) are executing lock-up agreements in substantially the form attached as **Exhibit B** (each, a "*Lock-Up Agreement*").

K. Concurrently with the execution and delivery of this Agreement, certain investors have executed a Securities Purchase Agreement among Parent and the Persons named therein (representing an aggregate commitment no less than the Concurrent Investment Amount of \$14,000,000), pursuant to which such Persons will have agreed to purchase the number of shares of Parent Common Stock and Parent Convertible Preferred Stock set forth therein concurrently with the Closing in connection with the Parent Financing (the "*Securities Purchase Agreement*"), which is in addition to a \$1,000,000 Simple Agreement for Future Equity investment received by the Company on March 21, 2024 (the "*SAFE Investment*").

L. Concurrently with the execution and delivery of this Agreement and as a condition and inducement to the Company's willingness to enter into this Agreement, certain stockholders set forth on <u>Section A-2</u> of the Parent Disclosure Schedule (solely in their capacity as stockholders) are executing support agreements in favor of the Company in substantially the form attached hereto as **Exhibit E** (the "*Parent Stockholder Support Agreement*"), pursuant to which such Persons have, subject to the terms and conditions set forth therein, agreed to vote all of their shares of capital stock of Parent in favor of the Parent Stockholder Matters.

M. Concurrently with the execution and delivery of this Agreement and as a condition and inducement to Parent's willingness to enter into this Agreement, all of the stockholders of the Company are executing Company Stockholder Support Agreements in substantially the form attached hereto as **Exhibit F** (the "*Company Stockholder Support Agreement*").

N. Immediately following the execution and delivery of this Agreement, but prior to the filing of the Certificate of Merger, Parent will file the Certificate of Designation with the office of the Secretary of State of the State of Delaware.

O. Concurrently with the execution and delivery of this Agreement and as a condition and inducement to the Company's willingness to enter into this Agreement, Parent is executing a registration rights agreement in substantially the form attached as **Exhibit H** (the "*Registration Rights Agreement*").

AGREEMENT

The Parties, intending to be legally bound, agree as follows:

SECTION 1. DESCRIPTION OF TRANSACTION

1.1 <u>The Merger</u>. Upon the terms and subject to the conditions set forth in this Agreement, at the First Effective Time, First Merger Sub shall be merged with and into the Company, and the separate existence of First Merger Sub shall cease. As a result of the First Merger, the Company will continue as the surviving corporation in the First Merger (the "*First Step Surviving Corporation*"). Upon the terms and subject to the conditions set forth in this Agreement, at the Second Effective Time, the First Step Surviving Corporation will merge with and into Second Merger Sub, and the separate existence of the First Step Surviving Corporation shall cease. As a result of the Second Merger Sub will continue as the surviving entity in the Second Merger (the "*Surviving Entity*").

1.2 Effects of the Merger. At and after the First Effective Time, the First Merger shall have the effects set forth in this Agreement, the First Certificate of Merger and in the applicable provisions of the DGCL. As a result of the First Merger, the First Step Surviving Corporation will become a wholly owned subsidiary of Parent. Without limiting the generality of the foregoing, and subject thereto, at the First Effective Time, all the property, rights, privileges, agreements, powers and franchises, debts, liabilities, duties and obligations of the First Step Surviving Corporation. At and after the Second Effective Time, the Second Merger shall have the effects set forth in this Agreement, the Second Effective Time, the DLLCA. Without limiting the generality of the foregoing, and subject thereto, at the Second Effective Time, all the property, rights, privileges, agreements, powers and franchises, debts, liabilities, duties and obligations of Second Merger and in the applicable provisions of the DGCL and the DLLCA. Without limiting the generality of the foregoing, and subject thereto, at the Second Effective Time, all the property, rights, privileges, agreements, powers and franchises, debts, liabilities, duties and obligations of Second Merger Sub and the First Step Surviving Corporation will become the property, rights, privileges, agreements, powers and franchises, debts, liabilities, duties and obligations of Second Merger Sub and the First Step Surviving Corporation will become the property, rights, privileges, agreements, powers and franchises, debts, liabilities, duties and obligations of Second Merger Sub and the Surviving Corporation will become the property, rights, privileges, agreements, powers and franchises, debts, liabilities, duties and obligations of the Surviving Entity.

1.3 **Closing; First Effective Time; Second Effective Time**. The consummation of the Merger (the "*Closing*") is being consummated remotely via the electronic exchange of documents and signatures substantially simultaneously with the execution and delivery of this Agreement, or at such other time, date and place as Parent and the Company may mutually agree in writing. The date on which the Closing actually takes place is referred to as the "*Closing Date*." At the Closing, (i) the Parties shall cause the First Merger to be consummated by executing and filing with the Secretary of State of the State of Delaware a certificate of merger with respect to the First Merger, satisfying the applicable requirements of the DGCL and in form and substance to be agreed upon by the Parties (the "*First Certificate of Merger*") and (ii) the Parties shall cause the Second Merger, satisfying the applicable requirements of the DGCL and the DLLCA and in form and substance to be agreed upon by the Parties of Merger". The First Merger shall become effective at the time of the filing of such First Certificate of Merger"). The First Merger shall become effective at the time of the filing of such First Certificate of Merger with the Secretary of State of Delaware or at such later time as may be specified in such First *Effective Time*"). The Second Merger shall become effective at the time of the filing of such Second Merger shall become effective at the time of the filing of such Second Merger shall become effective at the time of the filing of such Second Certificate of Merger with the Secretary of State of Merger with the consent of Parent and the Company (the time as of which the First Merger becomes effective being referred to as the "*First Effective Time*"). The Second Merger becomes effective being referred to as the "*Second Certificate of Merger*".



1.4 Certificate of Designation; Certificate of Incorporation and Bylaws; Directors and Officers.

the State of Delaware.

(a)

Prior to the First Effective Time, Parent will file the Certificate of Designation with the office of the Secretary of State of

(b) At the First Effective Time:

(i) the certificate of incorporation of the First Step Surviving Corporation shall be amended and restated as set forth in an exhibit to the First Certificate of Merger, until thereafter amended as provided by the DGCL and such certificate of incorporation;

(ii) the bylaws of the First Step Surviving Corporation shall be amended and restated in their entirety to read identically to the bylaws of the Company as in effect immediately prior to the First Effective Time, until thereafter amended as provided by the DGCL and such bylaws; and

(iii) the directors and officers of the First Step Surviving Corporation, each to hold office in accordance with the certificate of incorporation and bylaws of the First Step Surviving Corporation, shall be such persons as shall be mutually agreed upon by Parent and the Company.

(c) At the Second Effective Time:

(i) the certificate of formation of the Surviving Entity shall be the certificate of formation of Second Merger Sub as in effect immediately prior to the Second Effective Time, until thereafter amended as provided by the DLLCA and such certificate of formation; *provided*, *however*, that at the Second Effective Time (as part of the Second Certificate of Merger), the certificate of formation shall be amended to (A) change the name of the Surviving Entity to "Trawsfynydd Therapeutics, LLC," (B) comply with <u>Section 4.5</u>, and (C) make such other changes as are mutually agreed to by Parent and the Company;

(ii) the limited liability company agreement of the Surviving Entity shall be amended and restated in its entirety to read identically to the limited liability company agreement of Second Merger Sub as in effect immediately prior to the Second Effective Time, until thereafter amended as provided by the DLLCA and such limited liability company agreement; *provided, however*, that following the Second Effective Time (but as soon thereafter as practicable), the limited liability company agreement shall be amended to (A) comply with <u>Section 4.5</u> and (B) change the name of the Surviving Entity to "Trawsfynydd Therapeutics, LLC";

(iii) the managers and officers of the Surviving Entity, each to hold office in accordance with the certificate of formation and limited liability company agreement of the Surviving Entity, shall be such persons as shall be mutually agreed upon by Parent and the Company; and

(iv) the certificate of incorporation of Parent shall be identical to the certificate of incorporation of Parent immediately prior to the Second Effective Time, until thereafter amended as provided by the DGCL and such certificate of incorporation, *provided*, *however*, that following the Second Effective Time, the certificate of incorporation shall be amended to (A) change the corporate name of Parent to Traws Pharma, Inc.; (B) effect the Nasdaq Reverse Split, if deemed necessary, and (C) make such other changes as are mutually agreed to by Parent and the Company.

1.5 <u>Merger Consideration; Effect of Merger on Company Capital Stock</u>. The aggregate merger consideration (the "*Merger Consideration*") to be paid by Parent for all of the outstanding shares of Company Capital Stock at the Closing shall be (a) 3,549,538 shares of Parent Common Stock ("*Parent Common Stock Payment Shares*"), which shares shall represent a number of shares equal to no more than (i) 19.9% of the outstanding shares of Parent Common Stock as of immediately before the First Effective Time *minus* (ii) the percentage of shares of Parent Common Stock issued in the Parent Financing, and (b) 10,359.09160 shares of Parent Convertible Preferred Stock ("*Parent Preferred Stock Payment Shares*"). Each Parent Preferred Stock Payment Share shall be convertible into 10,000 shares of Parent Common Stock, subject to and contingent upon the affirmative vote of a majority of the Parent Common Stock present or represented and entitled to vote at a meeting of stockholders of Parent to approve, for purposes of the Nasdaq Stock Market Rules, the issuance of shares of Parent Common Stock to the stockholders of the Company upon conversion of any and all shares of Parent Convertible Preferred Stock *Conversion Proposal*").

1.6 Conversion of Shares.

(a) At the First Effective Time, by virtue of the First Merger and without any further action on the part of Parent, Merger Subs, the Company or any stockholder of the Company or Parent:

(i) any shares of Company Common Stock held as treasury stock or held or owned by the Company or any wholly owned Subsidiary of the Company immediately prior to the First Effective Time shall be canceled and retired and shall cease to exist, and no consideration shall be delivered in exchange therefor; and

(ii) subject to Section 1.5 and Section 1.6(c), each share of Company Capital Stock outstanding immediately prior to the First Effective Time (excluding shares to be canceled pursuant to Section 1.6(a)(i) and excluding Dissenting Shares) shall be automatically converted solely into the right to receive the number of Parent Common Stock Payment Shares and Parent Preferred Stock Payment Shares as set forth on the Allocation Certificate.

(b) The Company hereby covenants and agrees that, if any shares of Company Common Stock outstanding immediately prior to the First Effective Time are subject to a repurchase option or a risk of forfeiture under any applicable restricted stock purchase agreement or other similar agreement with the Company, the Company will cause such shares of Company Common Stock to no longer be subject to any right of repurchase, risk of forfeiture or other such conditions as of the First Effective Time.

(c) No fractional shares of Parent Common Stock shall be issued in connection with the First Merger, and no certificates or scrip for any such fractional shares shall be issued and no cash shall be paid for any such fractional shares. Any fractional shares of Parent Common Stock that a holder of Company Common Stock would otherwise be entitled to receive, including without limitation any conversion of Parent Convertible Preferred Stock into Parent Common Stock, shall be aggregated with all fractional shares of Parent Common Stock issuable to such holder) or a fraction of a share of Parent Convertible Preferred Stock issuable to such holder and any remaining fractional shares shall be rounded down to the nearest whole share.

(d) At the First Effective Time, by virtue of the First Merger and without any further action on the part of Parent, Merger Subs, the Company or any member of the Company or stockholder of Parent, each share of common stock of First Merger Sub issued and outstanding immediately prior to the First Effective Time shall be converted into and exchanged for one share of common stock of the First Step Surviving Corporation. If applicable, each stock certificate of First Merger Sub evidencing ownership of any such shares shall, as of the First Effective Time, evidence ownership of such shares of common stock of the First Step Surviving Corporation.

(e) If, between the date of this Agreement and the First Effective Time, the outstanding shares of Company Common Stock or Parent Convertible Preferred Stock shall have been changed into, or exchanged for, a different number of shares or a different class, by reason of any stock dividend, subdivision, reclassification, recapitalization, split, combination or exchange of shares or other like change, the Exchange Ratio shall, to the extent necessary, be equitably adjusted to reflect such change to the extent necessary to provide the holders of Company Common Stock and Parent Convertible Preferred Stock, with the same economic effect as contemplated by this Agreement prior to such stock dividend, subdivision, reclassification, recapitalization, split, combination or exchange of shares or other like change; provided, however, that nothing herein will be construed to permit the Company or Parent to take any action with respect to Company Common Stock or Parent Convertible Preferred Stock, respectively, that is prohibited or not expressly permitted by the terms of this Agreement.

(f) At the Second Effective Time, by virtue of the Second Merger and without any action on the part of Parent, the First Step Surviving Corporation, Second Merger Sub or their respective stockholders, each share of the First Step Surviving Corporation issued and outstanding immediately prior to the Second Effective Time shall be canceled and extinguished without any conversion thereof and no payment or distribution shall be made with respect thereto.

1.7 <u>Closing of the Company's Transfer Books</u>. At the First Effective Time: (a) all holders of (i) certificates representing shares of Company Capital Stock and (ii) book-entry shares representing shares of Company Capital Stock ("*Book-Entry Shares*"), in each case, that were outstanding immediately prior to the First Effective Time shall cease to have any rights as stockholders of the Company; and (b) the stock transfer books of the Company shall be closed with respect to all shares of Company Capital Stock outstanding immediately prior to the First Effective Time. No further transfer of any such shares of Company Capital Stock shall be made on such stock transfer books after the First Effective Time. If, after the first Effective Time, a valid certificate previously representing any shares of Company Capital Stock outstanding immediately prior to the First Effective Time (a "*Company Stock Certificate*") is presented to the Exchange Agent or to the Surviving Entity, such Company Stock Certificate shall be canceled and shall be exchanged as provided in <u>Sections 1.6</u> and <u>1.8</u>.

1.8 Exchange of Shares.

(a) Parent and the Company have selected Equiniti Trust Company, LLC, a New York limited liability trust company, to act as exchange agent in the Merger (the "*Exchange Agent*"). At the First Effective Time, Parent shall deposit with the Exchange Agent certificates or evidence of book-entry shares representing the Parent Common Stock and Parent Convertible Preferred Stock issuable pursuant to <u>Section 1.6(a)</u>. The Parent Common Stock and Parent Convertible Preferred Stock issuable pursuant to section 1.6(a). The Parent Common Stock and Parent Convertible Preferred Stock so deposited with the Exchange Agent, together with any dividends or distributions received by the Exchange Agent with respect to such shares, are referred to collectively as the "*Exchange Fund*."

(b) Promptly following the First Effective Time, the Parties shall cause the Exchange Agent to mail to the Persons who are record holders of Company Stock Certificates or Book-Entry Shares that will be converted into the right to receive Merger Consideration: (i) a letter of transmittal in the form attached hereto as **Exhibit G** ("*Letter of Transmittal*") and an accredited investor questionnaire in customary form and containing such provisions as Parent may reasonably specify and (ii) instructions for effecting the surrender of the Company Stock Certificates or Book-Entry Shares in exchange for book-entry shares of Parent Common Stock and Parent Convertible Preferred Stock representing the Merger Consideration. Upon surrender of a duly executed Letter of Transmittal, accredited investor questionnaire and such other documents as may be reasonably required by the Exchange Agent or Parent, each such former holder of Company Stock Certificates or Book-Entry Shares of Parent Common Stock and Parent Convertible Preferred Stock, as applicable, representing the Merger Consideration that such holder of Company Capital Stock has the right to receive pursuant to the provisions of Section 1.6(a), and each Company Stock Certificate or Book-Entry Share shall be canceled. The Merger Consideration and any dividends or other distributions as are payable pursuant to <u>Section 1.8(d)</u> shall be deemed to have been in full satisfaction of all rights pertaining to Company Capital Stock formerly represented by such Company Stock Certificates or Book-Entry Shares.

(c) No dividends or other distributions declared or made with respect to Parent Common Stock or Parent Convertible Preferred Stock with a record date on or after the First Effective Time shall be paid to the holder of any unsurrendered Company Stock Certificate or Book-Entry Shares with respect to the shares of Parent Common Stock and/or Parent Convertible Preferred Stock that such holder has the right to receive in the Merger until such holder surrenders such Company Stock Certificate or transfers such Book-Entry Shares or provides an affidavit of loss or destruction in lieu thereof in accordance with this <u>Section 1.8</u> (at which time (or, if later, on the applicable payment date) such holder shall be entitled, subject to the effect of applicable abandoned property, escheat or similar Laws, to receive all such dividends and distributions, without interest).

(d) Any portion of the Exchange Fund that remains unclaimed by holders of shares of Company Capital Stock as of the date that is one year after the Closing Date shall be delivered to Parent upon demand, and any holders of Company Stock Certificates or Book-Entry Shares who have not theretofore surrendered their Company Stock Certificates or transferred their Book-Entry Shares in accordance with this <u>Section 1.8</u> shall thereafter look only to Parent as general creditors for satisfaction of their claims for Parent Common Stock and Parent Convertible Preferred Stock and any dividends or distributions with respect to shares of Parent Common Stock and Parent Convertible Preferred Stock.

(e) No Party shall be liable to any holder of any shares of Company Capital Stock or to any other Person with respect to any shares of Parent Common Stock or Parent Convertible Preferred Stock (or dividends or distributions with respect thereto) or for any cash amounts delivered to any public official pursuant to any applicable abandoned property Law, escheat Law or similar Law. Any portion of the Exchange Fund that remains unclaimed by holders of shares of Company Capital Stock as of the date that is two years after the Closing Date (or immediately prior to such earlier date on which the related Exchange Funds (and all dividends or other distributions in respect thereof) would otherwise escheat to or become the property of any Governmental Body) shall, to the extent permitted by applicable Law, become the property of the Surviving Entity, free and clear of all claims or interest of any Person previously entitled thereto.

1.9 Appraisal Rights.

(a) Notwithstanding any provision of this Agreement to the contrary, shares of Company Capital Stock that are outstanding immediately prior to the First Effective Time and which are held by stockholders who have exercised and perfected appraisal rights for such shares of Company Capital Stock in accordance with the DGCL (collectively, the "*Dissenting Shares*") shall not be converted into or represent the right to receive the Merger Consideration described in <u>Section 2.5</u> attributable to such Dissenting Shares. Such stockholders shall be entitled to receive payment of the appraised value of such shares of Company Capital Stock held by them in accordance with the DGCL, unless and until such stockholders fail to perfect or effectively withdraw or otherwise lose their appraisal rights under the DGCL. All Dissenting Shares held by stockholders who shall have failed to perfect or shall have effectively withdrawn or lost their right to appraisal of such shares of Company Capital Stock under the DGCL (whether occurring before, at or after the First Effective Time) shall thereupon be deemed to be converted into and to have become exchangeable for, as of the First Effective Time, the right to receive the Merger Consideration, without interest, attributable to such Dissenting Shares upon their surrender in the manner provided in <u>Section 1.7</u>.

(b) The Company shall give Parent prompt written notice of any demands by dissenting stockholders received by the Company, withdrawals of such demands and any other instruments served on the Company and any material correspondence received by the Company in connection with such demands, and Parent shall have the right to direct all negotiations and proceedings with respect to such demands; provided that the Company shall have the right to participate in such negotiations and proceedings. Neither the Parent nor the Company shall, except with the other party's prior written consent, voluntarily make any payment with respect to, or settle or offer to settle, any such demands, or approve any withdrawal of any such demands or agree to do any of the foregoing.

1.10 Company Options.

At the First Effective Time, each Company Option that is outstanding and unexercised immediately prior to the First (a) Effective Time under the Company Plan, whether or not vested, shall be converted into and become an option to purchase Parent Common Stock, and Parent shall assume the Company Plan and each such Company Option in accordance with the terms (as in effect as of the date of this Agreement) of the Company Plan and the terms of the stock option agreement by which such Company Option is evidenced (but with changes to such documents as Parent in good faith determines are necessary to reflect the substitution of the Company Options by Parent to purchase shares of Parent Common Stock, including such changes set forth in this Section 1.10(a)). All rights with respect to Company Common Stock under Company Options assumed by Parent shall thereupon be converted into rights with respect to Parent Common Stock; provided that the Company Options assumed by Parent shall not be exercisable unless and until the Preferred Stock Conversion Proposal is approved. Accordingly, from and after the First Effective Time: (i) each Company Option assumed by Parent may be exercised solely for shares of Parent Common Stock; (ii) the number of shares of Parent Common Stock subject to each Company Option assumed by Parent shall be determined by multiplying (A) the number of shares of Company Common Stock that were subject to such Company Option, as in effect immediately prior to the First Effective Time, by (B) the Exchange Ratio, and rounding the resulting number down to the nearest whole number of shares of Parent Common Stock; (iii) the per share exercise price for the Parent Common Stock issuable upon exercise of each Company Option assumed by Parent shall be determined by dividing (A) the per share exercise price of Company Common Stock subject to such Company Option, as in effect immediately prior to the First Effective Time, by (B) the Exchange Ratio and rounding the resulting exercise price up to the nearest whole cent; and (iv) any restriction on the exercise of any Company Option assumed by Parent shall continue in full force and effect and the term, exercisability, vesting schedule and other provisions of such Company Option shall otherwise remain unchanged; provided, that, (I) the exercise price, the number of shares of Parent Common Stock subject to, and the terms and conditions of exercise of each option to purchase Parent Common Stock shall also be determined in a manner consistent with the requirements of Section 409A of the Code; and (II) the Company Options assumed by Parent shall not be exercisable unless and until the Preferred Stock Conversion Proposal is approved; provided, further, that: (x) Parent may amend the terms of the Company Options and the Company Plan as may be necessary to reflect Parent's substitution of the Company Options with options to purchase Parent Common Stock (such as by making any change in control or similar definition relate to Parent and having any provision that provides for the adjustment of Company Options upon the occurrence of certain corporate events relate to corporate events that relate to Parent and/or Parent Common Stock); and (y) the Parent Board or a committee thereof shall succeed to the authority and responsibility of the Company Board or any committee thereof with respect to each Company Option assumed by Parent.

(b) Parent shall file with the SEC, after the First Effective Time, a registration statement on Form S-8 (or any successor form), if available for use by Parent, relating to the shares of Parent Common Stock issuable with respect to Company Options assumed by Parent in accordance with <u>Section 1.10(a)</u>.

1.11 Contingent Value Right.

(a) Holders of Parent Common Stock of record as of the Record Date (as defined in the CVR Agreement) shall be entitled to one contractual contingent value right (a "*CVR*") issued by Parent subject to and in accordance with the terms and conditions of the CVR Agreement, attached hereto as **Exhibit D** (the "*CVR Agreement*"), for each share of Parent Common Stock held by such holders (less applicable withholding Taxes).

(b) Prior to the First Effective Time, Parent has authorized and duly adopted, executed and delivered, and ensured that Exchange Agent executed and delivered, the CVR Agreement. Parent and the Company shall cooperate to ensure that the CVRs are not subject to registration under the Securities Act, the Exchange Act or any applicable state securities or "blue sky" laws.

1.12 **Further Action**. If, at any time after the First Effective Time, any further action is determined by the Surviving Entity to be necessary or desirable to carry out the purposes of this Agreement or to vest the Surviving Entity with full right, title and possession of and to all rights and property of the Company, then the officers and directors of the Surviving Entity shall be fully authorized, and shall use their and its reasonable best efforts (in the name of the Company, in the name of Merger Subs, in the name of the Surviving Entity and otherwise) to take such action.

1.13 <u>Withholding</u>. The Parties and the Exchange Agent (each, a "*Withholding Agent*") shall be entitled to deduct and withhold from the consideration otherwise payable pursuant to this Agreement (including the CVR Agreement) to any holder of Company Capital Stock or any other Person such amounts as such Party or the Exchange Agent is required to deduct and withhold under the Code or any other Law with respect to the making of such payment; provided, however, that if a Withholding Agent determines that any payment to any stockholder of the Company hereunder is subject to deduction and/or withholding, then, except with respect to compensatory payments, or as a result of a failure to deliver the certificate described in <u>Section 6.2</u>, such Withholding Agent shall (a) provide notice to such stockholder as soon as reasonably practicable after such determination (and no later than three (3) Business Days prior to undertaking such deduction and/or withholding, and (b) use commercially reasonable efforts to cooperate with such stockholder prior to Closing to reduce or eliminate any such deduction and/or withholding. To the extent that amounts are so withheld and paid over to the appropriate Governmental Body, such withheld amounts shall be treated for all purposes of this Agreement as having been paid to the Person in respect of whom such deduction and withholding was made.

SECTION 2. REPRESENTATIONS AND WARRANTIES OF THE COMPANY

Subject to <u>Section 8.13(h)</u>, except as set forth in the correspondingly numbered Section of the disclosure schedule delivered by the Company to Parent (the "*Company Disclosure Schedule*"), the Company represents and warrants to Parent and Merger Subs as follows:

2.1 Due Organization; Subsidiaries.

(a) The Company is a corporation duly incorporated, validly existing and in good standing under the Laws of Delaware and has all necessary corporate power and authority: (i) to conduct its business in the manner in which its business is currently being conducted; (ii) to own or lease and use its property and assets in the manner in which its property and assets are currently owned or leased and used; and (iii) to perform its obligations under all Contracts by which it is bound.

(b) The Company is duly licensed and qualified to do business, and is in good standing (to the extent applicable in such jurisdiction), under the Laws of all jurisdictions where the nature of its business requires such licensing or qualification other than in jurisdictions where the failure to be so qualified individually or in the aggregate would not be reasonably expected to have a Company Material Adverse Effect.

(c) Except as set forth in <u>Section 2.1(c)</u> of the Company Disclosure Schedule, (i) the Company has no Subsidiaries and has never had any Subsidiaries; and (ii) the Company does not own, and has never owned, any capital stock of, or any equity, ownership or profit-sharing interest of any nature in, and does not control, and has never controlled, directly or indirectly, any other Entity. No Subsidiary of the Company has conducted any clinical trial work.

(d) The Company is not nor has ever been, directly or indirectly, a party to, member of or participant in any partnership, joint venture or similar business entity. The Company has not agreed to, and is not obligated to make, nor is bound by any Contract under which it may become obligated to make, any future investment in or capital contribution to any other Entity. The Company has not, at any time, been a general partner of, or has otherwise been liable for any of the debts or other obligations of, any general partnership, limited partnership or other Entity.

2.2 <u>Organizational Documents</u>. The Company has made available to Parent accurate and complete copies of the Organizational Documents of the Company in effect as of the date of this Agreement. The Company is not in breach or violation of its respective Organizational Documents.

2.3 Authority; Binding Nature of Agreement.

(a) The Company has all necessary corporate power and authority to enter into and to perform its obligations under this Agreement and, subject to receipt of the Required Company Stockholder Vote, to consummate the Contemplated Transactions. The Company Board (at meetings duly called and held or by unanimous written consent) has (i) determined that the Contemplated Transactions are fair to, advisable and in the best interests of the Company and its stockholders, (ii) authorized, approved and declared advisable this Agreement and the Contemplated Transactions and (iii) determined to recommend, upon the terms and subject to the conditions set forth in this Agreement, that the stockholders of the Company vote in favor of the Company Stockholder Matters.

(b) This Agreement has been duly executed and delivered by the Company and, assuming the due authorization, execution and delivery by Parent and Merger Subs, constitutes the legal, valid and binding obligation of the Company, enforceable against the Company in accordance with its terms, subject to the Enforceability Exceptions.

2.4 <u>Vote Required</u>. The affirmative vote (or written consent) of the holders of a majority of the shares of Company Common Stock outstanding on the record date for the Stockholder Written Consent and entitled to vote thereon, voting as a single class (the "*Required Company Stockholder Vote*"), is the only vote (or written consent) of the holders of any class or series of Company Capital Stock necessary to adopt and approve this Agreement and approve the Contemplated Transactions. The Stockholder Written Consent became effective upon the execution of this Agreement by the parties hereto and provided the Required Company Stockholder Vote. No other corporate proceedings by the Company are necessary to authorize this Agreement or to consummate the Contemplated Transactions.

2.5 <u>Non-Contravention; Consents</u>. Subject to obtaining the required Company Stockholder Vote, the filing of the Certificates of Merger required by the DGCL, and the filing of the Certificate of Designation, neither (x) the execution, delivery or performance of this Agreement by the Company, nor (y) the consummation of the Contemplated Transactions, will directly or indirectly (with or without notice or lapse of time):

(a) contravene, conflict with or result in a violation of any of the provisions of the Company's Organizational Documents;

(b) contravene, conflict with or result in a violation of, or give any Governmental Body or other Person the right to challenge the Contemplated Transactions or to exercise any remedy or obtain any relief under, any Law or any order, writ, injunction, judgment or decree to which the Company, or any of the assets owned or used by the Company, is subject, except as would not reasonably be expected to be material to the Company or its business;

(c) contravene, conflict with or result in a violation of any of the terms or requirements of, or give any Governmental Body the right to revoke, withdraw, suspend, cancel, terminate or modify, any Governmental Authorization that is held by the Company, except as would not reasonably be expected to be material to the Company or its business;

(d) contravene, conflict with or result in a violation or breach of, or result in a default under, any provision of any Company Material Contract, or give any Person the right to: (i) declare a default or exercise any remedy under any Company Material Contract; (ii) any material payment, rebate, chargeback, penalty or change in delivery schedule under any Company Material Contract; (iii) accelerate the maturity or performance of any Company Material Contract; or (iv) cancel, terminate or modify any term of any Company Material Contract, except in the case of any non-material breach, default, penalty or modification; or

(e) result in the imposition or creation of any Encumbrance upon or with respect to any asset owned or used by the Company (except for Permitted Encumbrances).

Except for (i) any Consent set forth in <u>Section 2.5</u> of the Company Disclosure Schedule under any Company Contract, (ii) the Required Company Stockholder Vote, (iii) the filing of the Certificates of Merger with the Secretary of State of the State of Delaware pursuant to the DGCL, (iv) the filing of the Certificate of Designation with the Secretary of State of the State of Delaware pursuant to the DGCL and (v) such consents, waivers, approvals, orders, authorizations, registrations, declarations and filings as may be required under applicable federal and state securities Laws, the Company is not required to make any filing with or give any notice to, or to obtain any Consent from, any Person in connection with (A) the execution, delivery or performance of this Agreement, or (B) the consummation of the Contemplated Transactions. The Company Board has taken and will take all actions necessary to ensure that the restrictions applicable to business combinations contained in Section 203 of the DGCL are, and will be, inapplicable to the execution, delivery and performance of this Agreement, the Lock-Up Agreements and to the consummation of the Contemplated Transactions. No other state takeover statute or similar Law applies or purports to apply to the Merger, this Agreement, the Lock-Up Agreements or any of the Contemplated Transactions.

2.6 Capitalization.

(a) The authorized Company Capital Stock as of immediately prior to the Closing consists of (i) 10,000,000 shares of Company Common Stock, par value \$0.0001 per share, of which 5,322,664 shares are issued and outstanding as of immediately prior to the Closing, and (ii) 4,933,776 shares of Preferred Stock, par value \$0.0001 per share (the "*Company Preferred Stock*"), all of which have been designated Series Seed Preferred Stock and none of which are issued or outstanding as of immediately prior to the Closing. The Company does not hold any shares of its capital stock in its treasury. Section 2.6(a) of the Company Disclosure Schedule lists, as of immediately prior to the Closing, each record holder of issued and outstanding Company Capital Stock and the number and type of shares of Company Capital Stock held by such holder.

(b) All of the outstanding shares of Company Common Stock and Company Preferred Stock have been duly authorized and validly issued, and are fully paid and nonassessable. Except as set forth in the Investor Agreements, none of the outstanding shares of Company Capital Stock is entitled or subject to any preemptive right, right of participation, right of maintenance or any similar right and none of the outstanding shares of Company Capital Stock is subject to any right of first refusal in favor of the Company. Except as contemplated herein or as set forth in the Investor Agreements, there is no Company Contract relating to the voting or registration of, or restricting any Person from purchasing, selling, pledging or otherwise disposing of (or granting any option or similar right with respect to), any shares of Company Capital Stock. The Company is not under any obligation, nor is it bound by any Contract pursuant to which it may become obligated, to repurchase, redeem or otherwise acquire any outstanding shares of Company Capital Stock or other securities. Section 2.6(b) of the Company Disclosure Schedule accurately and completely lists all repurchase rights held by the Company with respect to shares of Company Capital Stock (including shares issued pursuant to the exercise of stock options) and specifies which of those repurchase rights are currently exercisable and whether the holder of such shares of Company Capital Stock is convertible into one share of Company Capital Bodies under Section 83(b) of the Code with respect to such shares. Each share of Company Preferred Stock is convertible into one share of Company Common Stock.

(c) Except for the Company's 2021 Stock Plan (the "*Company Plan*"), the Company does not have any stock option plan or any other plan, program, agreement or arrangement providing for any equity-based compensation for any Person. As of the date of this Agreement, the Company has reserved 454,000 shares of Company Common Stock for issuance under the Company Plan, all of which shares have been issued and are currently outstanding, and no shares of Company Common Stock remain available for future issuance of awards pursuant to the Company Plan. <u>Section 2.6(c)</u> of the Company Disclosure Schedule sets forth the following information with respect to each Company Option outstanding as of the date of this Agreement: (i) the name of the optionee; (ii) the number of shares of Company Option as of the date of this Agreement; (iv) the exercise price of such Company Option and rationale for the exercise price of each Company Option, including the independent valuation reports, if applicable; (v) the date on which such Company Option was granted; (vi) the applicable vesting schedule, including the number of vested and unvested shares as of the date of this Agreement and any acceleration provisions; (vii) the date on which such Company Option expires; (viii) whether such Company Option is intended to constitute an "incentive stock option" (as defined in the Code) or a non-qualified stock option; and (ix) whether such Company Option is "early exercisable." The Company has made available to Parent an accurate and complete copy of the Company Plan and a form of stock option agreement that is consistent in all material respects with the stock option agreements evidencing outstanding Company Options granted thereunder.

(d) Except for Company Options set forth in <u>Section 2.6(c)</u> of the Company Disclosure Schedule, there is no: (i) outstanding subscription, option, call, warrant or right (whether or not currently exercisable) to acquire any shares of the capital stock or other securities of the Company; (ii) outstanding security, instrument or obligation that is or may become convertible into or exchangeable for any shares of the capital stock or other securities of the Company; or (iii) condition or circumstance that could be reasonably likely to give rise to or provide a basis for the assertion of a claim by any Person to the effect that such Person is entitled to acquire or receive any shares of capital stock or other securities of the Company. There are no outstanding or authorized stock appreciation, phantom stock, profit participation or other similar rights with respect to the Company.

(e) All outstanding shares of Company Common Stock, Company Preferred Stock, Company Options, and other securities of the Company have been issued and granted in material compliance with (i) the Organizational Documents of the Company in effect as of the relevant time and all applicable securities Laws and other applicable Law, and (ii) all requirements set forth in applicable Contracts.

(f) All distributions, dividends, repurchases and redemptions of the Company Capital Stock or other equity interests of the Company were undertaken in material compliance with (i) the Organizational Documents of the Company in effect as of the relevant time and all applicable securities Laws and other applicable Laws, and (ii) all requirements set forth in applicable Contract.

2.7 Financial Statements.

(a) Concurrently with the execution hereof, the Company has provided to Parent true and complete copies of the Company Unaudited Interim Balance Sheet, together with the unaudited statements of operations and cash flows of the Company for the period reflected in the Company Unaudited Interim Balance Sheet (collectively, the "*Company Financials*"). The Company Financials were prepared in accordance with GAAP (except as may be indicated in the notes to such financial statements and except that the unaudited financial statements may not contain footnotes and are subject to normal and recurring year-end adjustments, none of which is material) and fairly present, in all material respects, the financial position and operating results of the Company as of the dates and for the periods indicated therein.

(b) The Company maintains accurate books and records reflecting its assets and liabilities and maintains a system of internal accounting controls designed to provide reasonable assurance that: (i) transactions are executed in accordance with management's general or specific authorizations; (ii) transactions are recorded as necessary to permit preparation of the financial statements of the Company in accordance with GAAP and to maintain accountability of the Company's assets; (iii) access to the Company's assets is permitted only in accordance with management's general or specific authorization; (iv) the recorded accountability for the Company's assets is compared with the existing assets at regular intervals and appropriate action is taken with respect to any differences; and (v) accounts, notes and other receivables and inventory are recorded accurately, and proper and adequate procedures are implemented which are designed to effect the collection thereof on a current and timely basis. The Company maintains internal controls consistent with the practices of similarly situated private companies over financial reporting that provide reasonable assurance regarding the reliability of financial statements for external purposes in accordance with GAAP.

(c) <u>Section 2.7(c)</u> of the Company Disclosure Schedule lists, and the Company has delivered to Parent accurate and complete copies of the documentation creating or governing, all securitization transactions and "off-balance sheet arrangements" (as defined in Item 303(c) of Regulation S-K under the Exchange Act) effected by the Company since the Company's inception.

(d) Since the Company's inception, there have been no formal internal investigations regarding financial reporting or accounting policies and practices discussed with, reviewed by or initiated at the direction of the chief executive officer, chief financial officer or general counsel of the Company, the Company Board or any committee thereof. Since the Company's inception, neither the Company nor its independent auditors have identified (i) any significant deficiency or material weakness in the design or operation of the system of internal accounting controls utilized by the Company, (ii) any fraud, whether or not material, that involves the Company, the Company or other employees who have a role in the preparation of financial statements or the internal accounting controls utilized by the Company or (iii) any claim or allegation regarding any of the foregoing.

2.8 <u>Absence of Changes</u>. Except as set forth in <u>Section 2.8</u> of the Company Disclosure Schedule, after the date of the Company Unaudited Interim Balance Sheet, the Company has conducted its business only in the Ordinary Course of Business (except for the execution and performance of this Agreement and the discussions, negotiations and transactions related thereto) and there has not been any (a) Company Material Adverse Effect and (b) the Company has not done any of the following:

(a) declared, accrued, set aside or paid any dividend or made any other distribution in respect of any shares of its capital stock; or repurchased, redeemed or otherwise reacquired any shares of its capital stock or other securities (except for shares of Company Common Stock from terminated employees, directors or consultants of the Company or in connection with the payment of the exercise price and/or withholding Taxes incurred upon the exercise, settlement or vesting of any award granted under the Company Plan);

(b) sold, issued, granted, pledged or otherwise disposed of or encumbered or authorized any of the foregoing with respect to: (i) any capital stock or other security of the Company; (ii) any option, warrant or right to acquire any capital stock or any other security, other than option grants to employees and service providers in the Ordinary Course of Business; or (iii) any instrument convertible into or exchangeable for any capital stock or other security of the Company;

(c) except as required to give effect to anything in contemplation of the Closing, amended any of its Organizational Documents, or effected or been a party to any merger, consolidation, share exchange, business combination, recapitalization, reclassification of shares, stock split, reverse stock split or similar transaction except, for the avoidance of doubt, the Contemplated Transactions;

with any other Entity;

(d) formed any Subsidiary or acquired any equity interest or other interest in any other Entity or entered into a joint venture

(e) (i) lent money to any Person (except for the advance of reasonable business expenses to employees, directors and consultants in the Ordinary Course of Business), (ii) incurred or guaranteed any indebtedness for borrowed money, or (iii) guaranteed any debt securities of others;

(f) other than as required by applicable Law or the terms of any Company Benefit Plan as in effect on the date of this Agreement and other than the Company Plan or any Company Option: (i) adopted, terminated, established or entered into any Company Benefit Plan; (ii) caused or permitted any Company Benefit Plan to be amended in any material respect; (iii) paid any material bonus or distributed any profit-sharing account balances or similar payment to, or, other than in the Ordinary Course of Business, increased the amount of the wages, salary, commissions, benefits or other compensation or remuneration payable to, any of its directors, officers or employees; (iv) increased the severance or change-of-control benefits offered to any current, former or new employees, directors or consultants or (v) hired, terminated or gave notice of termination (other than for cause) to, any (x) officer or (y) employee whose annual base salary is or is expected to be more than 100,000 per year;

(g) entered into any collective bargaining agreement or similar agreement with any labor union or similar labor organization;

(h) entered into any material transaction other than (i) in the Ordinary Course of Business or (ii) in connection with the

Contemplated Transactions;

(i) acquired any material asset or sold, leased or otherwise irrevocably disposed of any of its assets or properties, or granted any Encumbrance (other than Permitted Encumbrances) with respect to such assets or properties, except in the Ordinary Course of Business;

(j) sold, assigned, transferred, licensed, sublicensed or otherwise disposed of any material Company IP (other than pursuant to non-exclusive licenses in the Ordinary Course of Business);

(k) made, changed or revoked any material Tax election, failed to pay any income or other material Tax as such Tax becomes due and payable, filed any amendment making any material change to any Tax Return, settled or compromised any income or other material Tax liability, entered into any Tax allocation, sharing, indemnification or other similar agreement or arrangement (including any "closing agreement" described in Section 7121 of the Code (or any similar Law) with any Governmental Body, but excluding customary commercial Contracts entered into in the Ordinary Course of Business the principal subject matter of which is not Taxes), requested or consented to any extension or waiver of any limitation period with respect to any claim or assessment for any income or other material Taxes (other than pursuant to an extension of time to file any Tax Return granted in the Ordinary Course of Business of not more than six months), or adopted or changed any material accounting method in respect of Taxes;

exceed \$100,000;

(1) made any expenditures, incurred any Liabilities or discharged or satisfied any Liabilities, in each case, in amounts that

(m) other than as required by Law or GAAP, taken any action to change accounting policies or procedures;

- (n) initiated or settled any Legal Proceeding; or
- (o) agreed, resolved or committed to do any of the foregoing.

2.9 <u>Absence of Undisclosed Liabilities</u>. As of the date hereof, the Company has no liability, indebtedness, obligation or expense of any kind, whether accrued, absolute, contingent, matured or unmatured (whether or not required to be reflected in the financial statements in accordance with GAAP) (each a "*Liability*"), except for: (a) Liabilities disclosed, reflected or reserved against in the Company Unaudited Interim Balance Sheet; (b) Liabilities that have been incurred by the Company since the date of the Company Unaudited Interim Balance Sheet in the Ordinary Course of Business and which are not material to the Company, individually or in the aggregate; (c) Liabilities for performance of obligations under Company Contracts in the Ordinary Course of Business, which, in each case, are not related to any breach or default of the Company; (d) Liabilities incurred in connection with the Contemplated Transactions; and (e) Liabilities described in Section 2.9 of the Company Disclosure Schedule.

2.10 <u>Title to Assets</u>. The Company owns, and has good and valid title to, or, in the case of leased properties and assets, valid leasehold interests in, all tangible properties or tangible assets and equipment used or held for use in its business or operations or purported to be owned by it that are material to the Company or its business, including: (a) all tangible assets reflected on the Company Unaudited Interim Balance Sheet; and (b) all other tangible assets reflected in the books and records of the Company as being owned by the Company. All of such assets are owned or, in the case of leased assets, leased by the Company free and clear of any Encumbrances, other than Permitted Encumbrances.

2.11 <u>Real Property; Leasehold</u>. The Company does not own, nor has ever owned, any real property. The Company has made available to Parent (a) an accurate and complete list of all real properties with respect to which the Company directly or indirectly holds a valid leasehold interest as well as any other real estate that is in the possession of, or occupied or leased by, the Company and (b) copies of all leases under which any such real property is possessed, occupied or leased (the "*Company Real Estate Leases*"), each of which is in full force and effect, with no existing material default thereunder by the Company, or to the Knowledge of the Company, any other party thereto. The Company's possession, occupancy, lease, use and/or operation of each such leased property conforms to all applicable Laws in all material respects, and the Company has exclusive possession of each such leased property and leasehold interest is free and clear of all Encumbrances other than Permitted Encumbrances. The Company has not received any written notice from its landlords or any Governmental Body that: (i) relates to violations of building, zoning, safety or fire ordinances or regulations; (ii) claims any defect or deficiency with respect to any of such properties; or (iii) requests the performance of any repairs, alterations or other work to such properties.

2.12 Intellectual Property; Privacy.

Section 2.12(a) of the Company Disclosure Schedule identifies each item of Company Registered IP, including, with (a) respect to each application and registration: (i) the name of the applicant or registrant and any other co-owner, (ii) the jurisdiction of application or registration, (iii) the application or registration number, (iv) the date of issue, filing, or registration, as applicable, and (v) to the extent applicable, the expiration date. To the Knowledge of the Company, each of the Patents and Patent applications included in Section 2.12(a) of the Company Disclosure Schedule properly identifies by name each and every inventor of the inventions claimed therein as determined in accordance with applicable Laws of the United States and the applicable foreign jurisdiction. For all Company Registered IP owned or purported to be owned, in whole or in part, by the Company (the "Owned Company Registered IP"), and all other Company Registered IP for which the Company has responsibility for prosecution and maintenance activities, all necessary registration, maintenance, renewal and other relevant filing fees due through the Closing Date have been timely paid and all necessary documents and certificates in connection therewith have been timely filed with the relevant Patent, Trademark, Copyright, Internet domain name or other authorities in the United States or the applicable foreign jurisdiction, as the case may be, for the purpose of maintaining such Company Registered IP in full force and effect and, except as set forth on Section 2.12(a) of the Company Disclosure Schedule, there are no such filings, payments or other actions that must be made or taken on or before the three-month anniversary of the Closing Date. As of the date of this Agreement, no cancellation, interference, opposition, reissue, reexamination or other proceeding of any nature (other than office actions or similar communications issued by any Governmental Body in the ordinary course of prosecution of any pending applications for registration) is pending or, to the Knowledge of the Company, threatened in writing, in which the scope, validity, enforceability or ownership of any Company IP is being or has been contested or challenged. To the Knowledge of the Company, each item of Company IP is valid and enforceable, and with respect to the Company Registered IP, subsisting. The Company has complied with all Laws regarding the duty of disclosure, candor and good faith in connection with each Patent included in the Owned Company Registered IP and all other Company Registered IP for which the Company has responsibility for prosecution and maintenance activities.

(b) The Company (i) exclusively owns, is the sole assignee of, or has exclusively licensed all Owned Company Registered IP and all other Company IP (other than as disclosed in Section 2.12(b) of the Company Disclosure Schedule), and (ii) has valid and continuing rights, pursuant to the Company In-bound Licenses, to use all other material Intellectual Property Rights as the same are used in or necessary for the conduct of the business as presently conducted by the Company, in each case, free and clear of all Encumbrances other than Permitted Encumbrances. The Owned Company Registered IP is currently in compliance in all material respects with all Laws necessary to record and perfect the Company's interest in, and the chain of title of, the Owned Company Registered IP and to ensure the ability to claim priority in all jurisdictions. The Company IP and the Intellectual Property Rights licensed to the Company pursuant to a valid, enforceable written agreement constitute all Intellectual Property Rights used in, material to or otherwise necessary for the operation of the Company's business as currently conducted. Each Company Associate involved in the creation or development of any material Company IP, pursuant to such Company Associate's activities on behalf of the Company, has signed a written agreement containing an assignment of such Company Associate's rights in such Company IP to the Company. Each Company Associate who has or has had access to the Company's trade secrets or confidential information has signed a written agreement containing confidentiality provisions protecting the Company IP, trade secrets and confidential information. The Company has maintained copies of each such executed written agreement and, to the Knowledge of the Company, no party thereto is in default or breach of any such agreements. The Company has taken commercially reasonable steps to protect and preserve the confidentiality of its trade secrets and confidential information.

(c) No funding, facilities or personnel of any Governmental Body or any university, college, research institute or other educational institution has been used or is being used to create, in whole or in part, any Company IP that are owned or purported to be owned by Company, except for any such funding or use of facilities or personnel that does not result in such Governmental Body or institution obtaining ownership rights, license rights, or any other right to such Company IP (except for use rights during the term of the applicable agreement between the Company and such Governmental Body or educational institution), including the right to receive royalties for the practice of such Company IP (other than pursuant to any Company In-bound License disclosed on Section 2.12(d) of the Company Disclosure Schedule).

(d) <u>Section 2.12(d)</u> of the Company Disclosure Schedule sets forth each license agreement pursuant to which the Company (i) is granted a license under any material Intellectual Property Right owned by any third party that is used by the Company in its business as currently conducted (each a "*Company In-bound License*") or (ii) grants to any third party a license under any material Company IP or material Intellectual Property Right licensed to the Company under a Company In-bound License (each a "*Company Out-bound License*") (*provided*, that, Company In-bound Licenses shall not include, when entered into in the Ordinary Course of Business, material transfer agreements, clinical trial agreements, agreements with Company Associates, services agreements, non-disclosure agreements, commercially available Software-as-a-Service offerings, or off-the-shelf software licenses; and Company Out-bound Licenses shall not include, when entered into in the Ordinary Course of Business, material transfer agreements, clinical trial agreements, services agreements, non-disclosure agreements, or non-exclusive outbound licenses). All Company In-bound Licenses and Company Outbound Licenses, to the Knowledge of the Company, are in full force and effect and are valid, enforceable and binding obligations of the Company and, to the Knowledge of Company, each other party to such Company In-bound Licenses or Company Out-bound Licenses. Neither the Company, nor to the Knowledge of the Company, any other party to such Company In-bound Licenses or Company Out-bound Licenses, is in material breach under any Company In-bound Licenses or Company Out-bound Licenses. Except as set forth in <u>Section 2.12(d)</u> of the Company Disclosure Schedule, none of the terms or conditions of any Company In-bound License or any Company Out-bound License requires the Company or any of its Affiliates to maintain, develop or prosecute any Intellectual Property Rights.

(e) To the Knowledge of the Company: (i) the operation of the business of the Company as currently conducted has not infringed, misappropriated or otherwise violated and does not infringe, misappropriate or otherwise violate any Intellectual Property Rights of any other Person and (ii) no other Person is infringing, misappropriating or otherwise violating any Company IP or any Intellectual Property Rights exclusively licensed to the Company. No Legal Proceeding is pending (or, to the Knowledge of the Company, is threatened in writing) (A) against the Company alleging that the operation of the business of the Company infringes or constitutes the misappropriated or otherwise violated any of the Company IP or any Intellectual Property Rights exclusively licensed to the Company alleging that another Person has infringed, misappropriated or otherwise violated any of the Company IP or any Intellectual Property Rights exclusively licensed to the Company. Since 'December 31, 2021, the Company has not received any written notice or other written communication alleging that the operation of the business of the Company infringes or constitutes the misappropriation or other violation or other violation of any Intellectual Property Right of another Person.

(f) None of the Company IP owned by the Company or, to the Knowledge of the Company, any Company IP exclusively licensed to the Company is subject to any pending or outstanding injunction, directive, order, judgment or other disposition of dispute that adversely and materially restricts the use, transfer, registration or licensing by the Company of any such Company IP.

(g) To the Knowledge of the Company, the Company and the operation of the Company's business are and have at all times been, in material compliance with all applicable Laws and Privacy and Data Processing Requirements. To the Knowledge of the Company, the Company has at all applicable times provided all notices, and obtained and maintained all rights, consents, and authorizations, to Process Company Data as Processed by or for the Company. Since 'December 31, 2021, there have been (i) no loss or theft or security breach relating to Company Data, (ii) no violation of any security policy of the Company regarding any such Company Data, and (iii) no unauthorized access to, or unauthorized, unintended, or improper use disclosure, or other Processing of any Company Data. The Company has taken commercially reasonable steps and implemented reasonable disaster recovery and security plans and procedures to protect the information technology systems used in, material to or necessary for operation of the Company's business as currently conducted and Company Data from unauthorized use access, or other Processing, and the Company has taken commercially reasonable steps to require that any third party with access to Company Data collected by or on behalf of the Company has taken commercially reasonable steps to protect the Company Data provided to them by the Company. The Company has implemented and maintained commercially reasonable policies, procedures and systems for receiving and appropriately responding to requests from individuals concerning their Company Data where such steps are required by applicable Privacy and Data Processing Requirements.

(h) To the Knowledge of the Company, there have been no (i) material malfunctions or unauthorized intrusions or breaches of the information technology systems used in, material to or necessary for the operation of the Company's business, (ii) material unauthorized access to, or other processing of, Company Data, or (iii) material breaches, security incidents, misuse of or unauthorized access to or disclosure of any Company Data in the possession or control of the Company, and the Company has not provided or been legally required to provide any notices to any Person in connection with an unauthorized disclosure of Company Data. To the Knowledge of the Company, the Company has not been the subject of or received written notice of any complaints, claims or investigations related to their collection, use, storage or processing of Company Data or alleging any violation of applicable Privacy and Data Processing Requirements.

(i) The Company has taken reasonable security and other measures, including measures against unauthorized disclosure, to protect and maintain the secrecy, confidentiality and value of the Know-How and other confidential information included in the Company IP. No trade secret, Know-How or proprietary information material to the business of the Company as presently conducted, as of the date of this Agreement has been authorized to be disclosed or, to the knowledge of the Company, has been actually disclosed by the Company to any Person other than pursuant to a non-disclosure agreement or other agreement adequately restricting the disclosure and use of such Intellectual Property Rights or information, and excluding any Know-How or proprietary information disclosed by the Company in publications or public filings, including as required under applicable securities laws.

(j) To the Knowledge of the Company, the computer systems, including the software, firmware, hardware, networks, interfaces, platforms and related systems, owned, leased or licensed by the Company an (collectively, the "*Company Systems*") perform in all material respects as is necessary for the conduct of its business as presently conducted by Company. To the Knowledge of the Company, in the 12 months immediately prior to the date of this Agreement, (i) there have been no material failures, breakdowns or other adverse events materially affecting any such Company Systems that have caused a material disruption or interruption to the conduct of the business of the Company as currently conducted, and (ii) there have not been any material incidents of unauthorized access or other security breaches of the Company Systems.

2.13 Agreements, Contracts and Commitments.

(a) <u>Section 2.13(a)</u> of the Company Disclosure Schedule lists the following Company Contracts in effect as of the date of this Agreement other than any Company Benefit Plans (each, a "*Company Material Contract*" and collectively, the "*Company Material Contracts*"):

Course of Business;

(i) each Company Contract relating to any agreement of indemnification or guaranty not entered into in the Ordinary

(ii) each Company Contract containing (A) any covenant limiting the freedom of the Company or the Surviving Entity to engage in any line of business or compete with any Person, (B) any most-favored nation or other preferred pricing arrangement in favor of a Person other than the Company or any similar term by which any Person is or could become entitled to any benefit, right or privilege that must be at least as favorable to such Person as those offered to any other Person, (C) any exclusivity provision, right of first refusal or right of first negotiation or similar covenant in favor of a Person other than the Company, or (D) any non-solicitation provision not entered into the Ordinary Course of Business;

(iii) each Company Contract relating to capital expenditures and requiring payments after the date of this Agreement in excess of \$100,000 pursuant to its express terms and not cancelable without penalty;

(iv) each Company Contract relating to the disposition or acquisition of material assets or any ownership interest in any Entity, except as contemplated hereby;

(v) each Company Contract relating to any mortgages, indentures, loans, notes or credit agreements, security agreements or other agreements or instruments relating to the borrowing of money or extension of credit or creating any material Encumbrances with respect to any assets of the Company or any loans or debt obligations with officers or directors of the Company;

(vi) each Company Contract requiring payment by or to the Company after the date of this Agreement in excess of \$100,000 in the aggregate in the current calendar year or any future calendar year pursuant to its express terms relating to: (A) any agreement involving the development or commercialization of any pharmaceutical product (identifying any that contain exclusivity provisions); (B) any agreement involving provision of services or products with respect to any pre-clinical or clinical development activities of the Company; (C) any dealer, distributor, joint marketing, alliance, joint venture, cooperation, development or other agreement currently in force under which the Company has continuing obligations to develop or market any product, technology or service, or any agreement pursuant to which the Company has continuing obligations to develop any Intellectual Property Rights that will not be owned, in whole or in part, by the Company; or (D) any Contract with any third party providing any services relating to the manufacture or production of any product, service or technology of the Company or any Contract to sell, distribute or commercialize any products or service of the Company;

(vii) each Company Contract with any financial advisor, broker, finder, investment banker or other similar Person providing financial advisory services to the Company in connection with the Contemplated Transactions;

- (viii) each Company Real Estate Lease;
- (ix) each Company Contract with any Governmental Body;

(x) each Company Out-bound License and Company In-bound License, and each Company Contract containing a covenant not to sue or otherwise enforce any Intellectual Property Rights;

(xi) each Company Contract requiring the payment of any royalty, dividend or similar arrangement based on the revenues or profits of the Company;

(xii) each Company Contract, offer letter, employment agreement, or independent contractor agreement with any employee, independent contractor or other natural person service provider that (A) is not immediately terminable at will by the Company without notice, severance or other cost or payment, except as required under applicable Law, or (B) provides for retention payments, change of control payments, severance, accelerated vesting, or any similar payment or benefit that may or will become due as a result of the Merger;

(xiii) each Company Contract providing any option to receive a license or other right, any right of first negotiation, any right of first refusal or any similar right to any Person related to any material Company IP or material Intellectual Property Right licensed to the Company under a Company In-bound License;

(xiv) each Company Contract entered into in settlement of any Legal Proceeding or other dispute; and

(xv) any other Company Contract that is not terminable at will (with no penalty or payment or requirement for prior notice, except as required by applicable Law) by the Company and (A) which involves payment or receipt by the Company after the date of this Agreement under any such agreement, Contract or commitment of more than \$100,000 in the aggregate, or obligations after the date of this Agreement in excess of \$100,000 in the aggregate, or (B) that is material to the business or operations of the Company taken as a whole.

(b) The Company has delivered or made available to Parent accurate and complete copies of all Company Material Contracts, including all amendments thereto. There are no Company Material Contracts that are not in written form. Neither the Company nor, to the Company's Knowledge, as of the date of this Agreement any other party to a Company Material Contract, has breached, violated or defaulted under, or received notice that it breached, violated or defaulted under, any of the terms or conditions of any Company Material Contract. As to the Company, as of the date of this Agreement, each Company Material Contract is valid, binding, enforceable and in full force and effect, subject to the Enforceability Exceptions. No Person is renegotiating, or has a right pursuant to the terms of any Company Material Contract to change, any material amount paid or payable to the Company under any Company Material Contract or any other material term or provision of any Company Material Contract, and no Person has indicated to the Company that it desires to renegotiate, modify, not renew or cancel any Company Material Contract.

2.14 <u>Compliance; Permits; Restrictions</u>. The Company holds all required Governmental Authorizations which are material to the operation of the business of the Company as currently conducted (the "*Company Permits*"). <u>Section 2.14</u> of the Company Disclosure Schedule identifies each Company Permit. Each such Company Permit is valid and in full force and effect, and the Company is in material compliance with the terms of the Company Permits. No Legal Proceeding is pending or, to the Knowledge of the Company, threatened, which seeks to revoke, limit, suspend, or materially modify any Company Permit.

2.15 Legal Proceedings; Orders.

(a) Except as set forth in <u>Section 2.15(a)</u> of the Company Disclosure Schedule, as of the date of this Agreement, there is no pending Legal Proceeding and, to the Knowledge of the Company, no Person has threatened in writing to commence any Legal Proceeding: (i) that involves (A) the Company, (B) any Company Associate (in his or her capacity as such) or (C) any of the material assets owned or used by the Company; or (ii) that challenges, or that would have the effect of preventing, delaying, making illegal or otherwise interfering with, the Contemplated Transactions.

(b) Except as set forth in <u>Section 2.15(b)</u> of the Company Disclosure Schedule, since the Company's inception through the date of this Agreement, no Legal Proceeding has been pending against the Company that resulted in material liability to the Company.

(c) There is no order, writ, injunction, judgment or decree to which the Company, or any of the material assets owned or used by the Company, is subject. To the Knowledge of the Company, no officer or employees of the Company is subject to any order, writ, injunction, judgment or decree that prohibits such officer or employee from engaging in or continuing any conduct, activity or practice relating to the business of the Company or to any material assets owned or used by the Company.

2.16 Tax Matters.

(a) The Company has timely filed all income and other material Tax Returns that were required to be filed by or with respect to it under applicable Law. All such Tax Returns are correct and complete in all material respects and have been prepared in compliance with all applicable Law. No written claim has ever been made by any Governmental Body in any jurisdiction where the Company does not file a particular Tax Return or pay a particular Tax that the Company is subject to taxation by that jurisdiction.

(b) All income and other material Taxes due and owing by the Company on or before the date hereof (whether or not shown on any Tax Return) have been fully and timely paid. The unpaid Taxes of the Company did not, as of the date of the Company Unaudited Interim Balance Sheet, materially exceed the reserve for Tax liability (excluding any reserve for deferred Taxes established to reflect timing differences between book and Tax items) set forth on the face of the Company Unaudited Interim Balance Sheet. Since the date of the Company Unaudited Interim Balance Sheet, the Company has not incurred any material Liability for Taxes outside the Ordinary Course of Business.

(c) All Taxes that the Company is or was required by Law to withhold or collect have been duly and timely withheld or collected in all material respects on behalf of its respective employees, independent contractors, stockholders, lenders, customers or other third parties and have been timely paid to the proper Governmental Body or other Person or properly set aside in accounts for this purpose.

Company.

(d) There are no Encumbrances for material Taxes (other than Taxes not yet due and payable) upon any of the assets of the

company.

(e) No deficiencies for a material amount of Taxes with respect to the Company have been claimed, proposed or assessed by any Governmental Body in writing. There are no pending or ongoing and, to the Knowledge of the Company, no threatened audits, assessments or other actions for or relating to any liability in respect of a material amount of Taxes of the Company. The Company nor any of its predecessors has waived any statute of limitations or agreed to any extension of time with respect to any income or other material Tax assessment or deficiency.

(f) The Company has not been a United States real property holding corporation within the meaning of Section 897(c)(2) of the Code during the applicable period specified in Section 897(c)(1)(A)(ii) of the Code.

(g) The Company is not a party to any Tax allocation agreement, Tax sharing agreement, Tax indemnity agreement, or similar agreement or arrangement, other than customary commercial Contracts entered into in the Ordinary Course of Business the principal subject matter of which is not Taxes.

(h) The Company (nor Parent as a result of the Merger) will not be required to include any item of income in, or exclude any item of deduction from, taxable income for any Tax period (or portion thereof) ending after the Closing Date as a result of any: (i) change in method of accounting for Tax purposes for a Tax period ending on or prior to the Closing Date; (ii) use of an improper method of accounting for a Tax period ending on or prior to the Closing Date; (iii) "closing agreement" as described in Section 7121 of the Code (or any similar provision of state, local or foreign Law) executed on or prior to the Closing Date; (iv) intercompany transaction or excess loss account described in Treasury Regulations under Section 1502 of the Code (or any similar provision of state, local or foreign Law); (v) installment sale or open transaction disposition made on or prior to the Closing Date; or (vi) prepaid amount, advance payment or deferred revenue received or accrued on or prior to the Closing Date.

(i) The Company has not ever been (i) a member of a consolidated, combined or unitary Tax group (other than such a group the common parent of which is the Company) or (ii) a party to any joint venture, partnership, or other arrangement that is treated as a partnership for U.S. federal income Tax purposes. The Company has no Liability for any material Taxes of any Person (other than the Company) under Treasury Regulations Section 1.1502-6 (or any similar provision of state, local, or foreign Law), as a transferee or successor, by Contract (other than a Contract entered into in the Ordinary Course of Business the principal subject matter of which is not Taxes) or otherwise by operation of Law.

(j) The Company has not ever distributed stock of another Person, or had its stock distributed by another Person, in a transaction that was purported or intended to be governed in whole or in part by Section 355 of the Code or Section 361 of the Code (or any similar provisions of state, local or foreign Law).

(k) The Company has not had a permanent establishment (within the meaning of an applicable Tax treaty) or otherwise had an office or fixed place of business in a country other than the country in which it is organized.

 $(l) \qquad \mbox{The Company has not participated in or been a party to a transaction that, as of the date of this Agreement, constitutes a "listed transaction" within the meaning of Section 6707A(c)(2) of the Code and Treasury Regulations Section 1.6011-4(b)(2).$

(m) <u>Section 2.16(m)</u> of the Company Disclosure Schedule sets forth the entity classification of the Company for U.S. federal income tax purposes. The Company has not made an election or taken any other action to change its federal and state income tax classification from such classification.

For purposes of this <u>Section 2.16</u>, each reference to the Company shall be deemed to include any Person that was liquidated into, merged with, or is otherwise a predecessor to, the Company.

2.17 Employee and Labor Matters; Benefit Plans.

(a) <u>Section 2.17(a)</u> of the Company Disclosure Schedule is a list of all Company Benefit Plans, other than at-will employment offer letters on the Company's standard form and other than individual compensatory equity award agreements made pursuant to the Company's standard forms, in which case only representative standard forms of such agreements shall be scheduled. "*Company Benefit Plan*" means each (i) "employee benefit plan" as defined in Section 3(3) of ERISA and (ii) other pension, retirement, deferred compensation, excess benefit, profit-sharing, bonus, incentive, equity or equity-based, phantom equity, employment, consulting, severance, change-of-control, retention, health, life, disability, group insurance, paid time off, holiday, welfare and fringe benefit plan, program, agreement, Contract, or arrangement (whether written or unwritten, qualified or nonqualified, funded or unfunded, subject or not subject to ERISA and including any that have been frozen), in each case, sponsored, maintained, administered, contributed to, or required to be contributed to, by the Company for the benefit of any current or former employee, director, officer or independent contractor of the Company (or beneficiary thereof) or under which the Company has any actual or contingent liability (including, without limitation, by reason of having a Company ERISA Affiliate).

(b) As applicable with respect to each material Company Benefit Plan, the Company has made available to Parent, true and complete copies of (i) each material Company Benefit Plan, including all amendments thereto, and in the case of an unwritten material Company Benefit Plan, a written description thereof, (ii) all current trust documents, investment management Contracts, custodial agreements, administrative services agreements and insurance and annuity Contracts relating thereto, (iii) the current summary plan description and each summary of material modifications thereto, (iv) the most recently filed annual reports with any Governmental Body (*e.g.*, Form 5500 and all schedules thereto), (v) the most recent IRS determination, opinion or advisory letter, (vi) the most recent summary annual reports, nondiscrimination testing reports, actuarial reports, financial statements and trustee reports, and (vii) all notices and filings from the IRS or Department of Labor or other Governmental Body concerning audits or investigations, or "prohibited transactions" within the meaning of Section 406 of ERISA or Section 4975 of the Code.

(c) Each Company Benefit Plan has been maintained, operated and administered in compliance in all material respects with its terms and the applicable provisions of ERISA, the Code and all other Laws.

(d) The Company Benefit Plans which are "employee pension benefit plans" within the meaning of Section 3(2) of ERISA and which are intended to meet the qualification requirements of Section 401(a) of the Code have received determination or opinion letters from the IRS on which they may currently rely to the effect that such plans are qualified under Section 401(a) of the Code and the related trusts are exempt from federal income Taxes under Section 501(a) of the Code, respectively, and nothing has occurred that would reasonably be expected to materially adversely affect the qualification of such Company Benefit Plan or the tax exempt status of the related trust.

(e) Neither the Company nor any Company ERISA Affiliate maintains, contributes to, is required to contribute to, or has any actual or contingent liability with respect to, (i) any "employee pension benefit plan" (within the meaning of Section 3(2) of ERISA) that is subject to Title IV or Section 302 of ERISA or Section 412 of the Code, (ii) any "multiemployer plan" (within the meaning of Section 3(37) of ERISA), (iii) any "multiple employer plan" (within the meaning of Section 3(37) of ERISA), (iii) any "multiple employer plan" (within the meaning of Section 3(40) of ERISA).

(f) There are no pending audits or investigations by any Governmental Body involving any Company Benefit Plan, and no pending or, to the Knowledge of the Company, threatened claims (except for routine individual claims for benefits payable in the normal operation of the Company Benefit Plans), suits or proceedings involving any Company Benefit Plan, or, to the Knowledge of the Company, any fiduciary thereof or service provider thereto, in any case except as would not be reasonably expected to result in material liability to the Company. All contributions and premium payments required to have been made under any of the Company Benefit Plans or by applicable Law (without regard to any waivers granted under Section 412 of the Code), have been timely made and the Company has no material liability for any unpaid contributions with respect to any Company Benefit Plan.

(g) None of the Company, any Company ERISA Affiliate or, to the Knowledge of the Company, any fiduciary, trustee or administrator of any Company Benefit Plan, has engaged in, or in connection with the Contemplated Transactions will engage in, any transaction with respect to any Company Benefit Plan which would subject any such Company Benefit Plan, the Company, or Parent to a material Tax, material penalty or material liability for a "prohibited transaction" under Section 406 of ERISA or Section 4975 of the Code.

(h) No Company Benefit Plan provides death, medical, dental, vision, life insurance or other welfare benefits beyond termination of service or retirement other than coverage mandated by Law and, to the Knowledge of the Company, the Company has not made a written representation promising the same.

(i) Neither the execution of this Agreement, nor the performance of the Contemplated Transactions (either alone or when combined with the occurrence of any other event, including without limitation, a termination of employment), will: (i) result in any payment becoming due to any current or former employee, director, officer, or independent contractor of the Company thereof, pursuant to any Company Benefit Plan (ii) increase any amount of compensation or benefits otherwise payable under any Company Benefit Plan, (iii) result in the acceleration of the time of payment, funding or vesting of any benefits under any Company Benefit Plan, (iv) require any contribution or payment to fund any obligation under any Company Benefit Plan or (v) limit the right to merge, amend or terminate any Company Benefit Plan.

(j) Except as set forth in <u>Section 2.17(j)</u> of the Company Disclosure Schedule, neither the execution of this Agreement, nor the consummation of the Contemplated Transactions (either alone or when combined with the occurrence of any other event, including without limitation, a termination of employment) will result in the receipt or retention by any person who is a "disqualified individual" (within the meaning of Code Section 280G) with respect to the Company of any payment or benefit that is or could be characterized as a "parachute payment" (within the meaning of Code Section 280G), determined without regard to the application of Code Section 280G(b)(5).

(k) No current or former employee, officer, director or independent contractor of the Company has any "gross up" agreements with the Company or other assurance of reimbursement by the Company for any Taxes imposed under Code Section 409A or Code Section 4999.

United States.

(1) The Company does not maintain any Company Benefit Plan for the benefit of any service providers located outside of the

(m) The Company has provided to Parent a true and correct list, as of the date of this Agreement, containing the names of all current full-time, part-time or temporary employees and independent contractors (and indication as such), and, as applicable: (i) the annual dollar amount of all cash compensation in the form of wages, salary, fees, commissions, or director's fees payable to each person; (ii) dates of employment or service; (iii) title and, with respect to independent contractors, a current written description of such person's contracting services; (iv) visa status, if applicable; and (v) with respect to employees, (A) a designation of whether they are classified as exempt or non-exempt for purposes of the Fair Labor Standards Act, as amended ("*FLSA*") and any similar state law and (B) whether such an employee is on leave and, if so, the expected return date.

(n) The Company is not and has never been a party to, bound by, or has a duty to bargain under, any collective bargaining agreement or other Contract with a labor union or similar labor organization representing any of its employees, and there is no labor union or similar labor organization representing or, to the Knowledge of the Company, purporting to represent or seeking to represent any employees of the Company, including through the filing of a petition for representation election. There is not and has not been in the past five years, nor, to the Knowledge of the Company, is there or has there been since the Company's inception any threat of, any strike, slowdown, work stoppage, lockout, union election petition, demand for recognition, or any similar activity or dispute or any union organizing activity, affecting the Company.

(o) The Company is, and since the Company's inception has been, in material compliance with all applicable Laws respecting labor, employment practices, and terms and conditions of employment, including worker classification, discrimination, harassment and retaliation, equal employment opportunities, fair employment practices, meal and rest periods, immigration, employee safety and health, payment of wages (including overtime wages), unemployment and workers' compensation, leaves of absence, and hours of work. Except as would not be reasonably likely to result in a material liability to the Company, with respect to employees of the Company, the Company, since the Company's inception, has withheld and reported all amounts required by Law to be withheld and reported with respect to wages, salaries and other payments, benefits, or compensation to employees. There is no material Legal Proceeding pending or, to the Knowledge of the Company, threatened or reasonably anticipated against the Company relating to any current or former employee, applicant for employment, or consultant of the Company.

(p) Within the preceding two years, the Company has complied in all material respects with the WARN Act, and no action that could trigger the WARN Act has been implemented by the Company without advance notification to and approval of Parent.

2.18 Environmental Matters. The Company is and since inception has complied with all applicable Environmental Laws, which compliance includes the possession by the Company of all permits and other Governmental Authorizations required under applicable Environmental Laws and compliance with the terms and conditions thereof, except for any failure to be in such compliance that, either individually or in the aggregate, would not reasonably be expected to be material to the Company or its business. The Company has not received since inception any written notice or other communication (in writing or otherwise), whether from a Governmental Body or other Person, that alleges that the Company is not in compliance with or has liability pursuant to any Environmental Law and, to the Knowledge of the Company, there are no circumstances that would reasonably be expected to prevent or interfere with the Company's compliance in any material respects with any Environmental Law, except where such failure to comply would not reasonably be expected to be material to the Company or its business. No current or (during the time a prior property was leased or controlled by the Company) prior property leased or controlled by the Company has had a release of or exposure to Hazardous Materials in material violation of or as would reasonably be expected to result in any material liability of the Company pursuant to Environmental Law. No consent, approval or Governmental Authorization of or registration or filing with any Governmental Body is required by Environmental Laws in connection with the execution and delivery of this Agreement or consummation of the Contemplated Transactions by the Company. Prior to the date hereof, the Company has provided or otherwise made available to Parent true and correct copies of all material environmental reports, assessments, studies and audits in the possession or control of the Company with respect to any property leased or controlled by the Company or any business operated by it.

2.19 **Insurance**. The Company has delivered or made available to Parent accurate and complete copies of all insurance policies and all self-insurance programs and arrangements relating to the business, assets, liabilities and operations of the Company. Each of such insurance policies is in full force and effect and the Company is in compliance in all material respects with the terms thereof. Other than customary end of policy notifications from insurance carriers, since the Company's inception, the Company has not received any notice or other communication regarding any actual or possible: (a) cancellation or invalidation of any insurance policy; or (b) refusal or denial of any coverage, reservation of rights or rejection of any material claim under any insurance policy. The Company has provided timely written notice to the appropriate insurance carrier(s) of each Legal Proceeding that is currently pending against the Company for which the Company has insurance coverage, and no such carrier has issued a denial of coverage or a reservation of rights with respect to any such Legal Proceeding, or informed the Company of its intent to do so.

2.20 **No Financial Advisors**. Except as set forth in <u>Section 2.20</u> of the Company Disclosure Schedule, no broker, finder or investment banker is entitled to any brokerage fee, finder's fee, opinion fee, success fee, transaction fee or other fee or commission in connection with the Contemplated Transactions based upon arrangements made by or on behalf of the Company.

2.21 Transactions with Affiliates.

(a) <u>Section 2.21(a)</u> of the Company Disclosure Schedule describes any transactions or relationships, since the Company's inception, between, on one hand, the Company and, on the other hand, any (i) officer or director of the Company or, to the Knowledge of the Company, any of such officer's or director's immediate family members, (ii) owner of more than 5% of the voting power of the outstanding Company Capital Stock or (iii) to the Knowledge of the Company, any "related person" (within the meaning of Item 404 of Regulation S-K under the Securities Act) of any such officer, director or owner (other than the Company) in the case of each of (i), (ii) or (iii) that is of the type that would be required to be disclosed under Item 404 of Regulation S-K under the Securities Act.

(b) <u>Section 2.21(b)</u> of the Company Disclosure Schedule lists each stockholders agreement, voting agreement, registration rights agreement, co-sale agreement or other similar Contract between the Company and any holders of Company Capital Stock, including any such Contract granting any Person investor rights, rights of first refusal, rights of first offer, registration rights, director designation rights or similar rights (collectively, the "*Investor Agreements*").

2.22 <u>Anti-Bribery</u>. Neither the Company nor any of its directors, officers, employees or, to the Company's Knowledge, agents or any other Person acting on their behalf (in each in their respective capacities as such) has directly or indirectly made any bribes, rebates, payoffs, influence payments, kickbacks, illegal payments, illegal political contributions, or other payments, in the form of cash, gifts, or otherwise, or taken any other action, in violation of the Foreign Corrupt Practices Act of 1977, the UK Bribery Act of 2010 or any other anti-bribery or anti-corruption Law (collectively, the "*Anti-Bribery Laws*"). The Company is not, nor has ever been, the subject of any investigation or inquiry by any Governmental Body with respect to potential violations of Anti-Bribery Laws.

2.23 Disclaimer of Other Representations or Warranties.

(a) Except as previously set forth in this <u>Section 2</u> or in any certificate delivered by the Company to Parent and/or Merger Subs pursuant to this Agreement, the Company makes no representation or warranty, express or implied, at law or in equity, with respect to it or any of its assets, liabilities or operations, and any such other representations or warranties are hereby expressly disclaimed.

(b) The Company acknowledges and agrees that, except for the representations and warranties of Parent and Merger Subs set forth in <u>Section 3</u> or in any certificate delivered by Parent and/or Merger Subs to the Company pursuant to this Agreement, neither the Company nor any of its Representatives is relying on any other representation or warranty of Parent, Merger Subs or any other Person made outside of <u>Section 3</u> or such certificate, including regarding the accuracy or completeness of any such other representations or warranties or the omission of any material information, whether express or implied, in each case, with respect to the Contemplated Transactions.

SECTION 3. REPRESENTATIONS AND WARRANTIES OF PARENT AND MERGER SUBS

Subject to <u>Section 8.13(h)</u>, except (a) as set forth in the correspondingly numbered Section of the disclosure schedule delivered by Parent to the Company (the "*Parent Disclosure Schedule*") or (b) as disclosed in the Parent SEC Documents filed with the SEC after December 31, 2021 and at least 10 Business Days prior to the date hereof, and publicly available on the SEC's Electronic Data Gathering Analysis and Retrieval system, and that is reasonably apparent on the face of such disclosure to be applicable to the representation and warranty set forth herein (but (i) without giving effect to any amendment thereof filed with, or furnished to the SEC on or after the date hereof and (ii) excluding any disclosures contained under the heading "Risk Factors," "Quantitative and Qualitative Disclosures About Market Risk," "Forward-Looking Statements," and any disclosure of risks included in any "forward-looking statements" disclaimer or in any other section to the extent they are forward-looking statements or cautionary, predictive or forward-looking in nature), Parent and Merger Subs represent and warrant to the Company as follows:

3.1 Due Organization; Subsidiaries.

(a) Each of Parent, First Merger Sub and Second Merger Sub is a company duly incorporated or organized, validly existing and in good standing under the Laws of the State of Delaware, and has all necessary corporate or limited liability power and authority: (i) to conduct its business in the manner in which its business is currently being conducted; (ii) to own or lease and use its property and assets in the manner in which its property and assets are currently owned or leased and used; and (iii) to perform its obligations under all Contracts by which it is bound. Since their respective date of incorporation or formation, no Merger Sub has engaged in any activities other than activities incident to its formation or in connection with or as contemplated by this Agreement.

(b) Parent is duly licensed and qualified to do business, and is in good standing (to the extent applicable in such jurisdiction), under the Laws of all jurisdictions where the nature of its business requires such licensing or qualification other than in jurisdictions where the failure to be so qualified individually or in the aggregate would not be reasonably expected to have a Parent Material Adverse Effect.

(c) Parent has no Subsidiaries, except for the Entities identified in Section 3.1(c) of the Parent Disclosure Schedule; and neither Parent nor any of the Entities identified in Section 3.1(c) of the Company Disclosure Schedule owns any capital stock of, or any equity, ownership or profit-sharing interest of any nature in, or controls directly or indirectly, any other Entity other than the Entities identified in Section 3.1(c) of the Company Disclosure Schedule. Each of Parent's Subsidiaries is a corporation or other legal entity duly organized, validly existing and, if applicable, in good standing under the Laws of the jurisdiction of its organization and has all necessary corporate or other power and authority: (i) to conduct its business in the manner in which its business is currently being conducted; (ii) to own or lease and use its property and assets in the manner in which its property and used; and (iii) to perform its obligations under all Contracts by which it is bound.

(d) Neither the Parent nor any of its Subsidiaries is or has otherwise been, directly or indirectly, a party to, member of or participant in any partnership, joint venture or similar business Entity. Neither the Parent nor any of its Subsidiaries has agreed or is obligated to make, or is bound by any Contract under which it may become obligated to make, any future investment in or capital contribution to any other Entity. Neither the Parent nor any of its Subsidiaries has, at any time, been a general partner of, or has otherwise been liable for, any of the debts or other obligations of, any general partnership, limited partnership or other Entity.

3.2 **Organizational Documents**. Parent has made available to the Company accurate and complete copies of the Organizational Documents or Parent and each of its Subsidiaries in effect as of the date of this Agreement. Neither Parent nor any of its Subsidiaries is in breach or violation of its respective Organizational Documents.

3.3 Authority; Binding Nature of Agreement.

The Parent and each of its Subsidiaries (including the Merger Subs) have all necessary corporate or limited liability power (a) and authority to enter into and to perform its obligations under this Agreement, the Registration Rights Agreement, and, subject, with respect to Parent, to receipt of the Required Parent Stockholder Vote and, with respect to Merger Subs, the adoption of this Agreement by Parent in its capacity as sole stockholder of Merger Subs, to perform its obligations hereunder and to consummate the Contemplated Transactions. The Parent Board (at meetings duly called and held or by unanimous written consent) has: (i) determined that the Contemplated Transactions are fair to, advisable and in the best interests of Parent and its stockholders; (ii) authorized, approved and declared advisable this Agreement and the Contemplated Transactions, including the issuance of Parent Common Stock Payment Shares and Parent Preferred Stock Payment Shares to the stockholders of the Company pursuant to the terms of this Agreement and the treatment of the Company Options pursuant to this Agreement; and (iii) determined to recommend, upon the terms and subject to the conditions set forth in this Agreement, that the stockholders of Parent vote to approve the Parent Stockholder Matters. The First Merger Sub Board (by unanimous written consent) has: (A) determined that the Contemplated Transactions are fair to, advisable, and in the best interests of First Merger Sub and its sole stockholder; (B) authorized, approved and declared advisable this Agreement and the Contemplated Transactions; and (C) determined to recommend, upon the terms and subject to the conditions set forth in this Agreement, that the stockholder of First Merger Sub vote to adopt this Agreement and thereby approve the Contemplated Transactions. The Second Merger Sub Board (by unanimous written consent) has: (A) determined that the Contemplated Transactions are fair to, advisable, and in the best interests of Second Merger Sub and its sole member; (B) authorized, approved and declared advisable this Agreement and the Contemplated Transactions; and (C) determined to recommend, upon the terms and subject to the conditions set forth in this Agreement, that the member of Second Merger Sub vote to adopt this Agreement and thereby approve the Contemplated Transactions.

(b) This Agreement has been duly executed and delivered by Parent and each Merger Sub and, assuming the due authorization, execution and delivery by the Company, constitutes the legal, valid and binding obligation of Parent and Merger Subs, enforceable against each of Parent and Merger Subs in accordance with its terms, subject to the Enforceability Exceptions.

3.4 **<u>Vote Required</u>**. The approval of holders of Parent Common Stock is not required in order to approve this Agreement or, except with respect to Parent Stockholder Matters, the transactions contemplated hereby. The affirmative vote of a majority of the votes cast at the Parent Stockholders' Meeting by the holders of Parent Common Stock (other than the Parent Common Stock Payment Shares to be issued at Closing pursuant to this Agreement) are the only vote of the holders of any class or series of Parent's capital stock necessary to approve the proposal described in <u>Section 4.2(a)</u> (*"Required Parent Stockholder Vote"*).

3.5 <u>Non-Contravention; Consents</u>. Subject to obtaining the Required Parent Stockholder Vote, the filing of the Certificates of Merger required by the DGCL and the filing of the Certificate of Designation, neither (x) the execution, delivery or performance of this Agreement by Parent or Merger Subs, nor (y) the consummation of the Contemplated Transactions, will directly or indirectly (with or without notice or lapse of time):

Merger Subs;

(a)

contravene, conflict with or result in a violation of any of the provisions of the Organizational Documents of Parent or

(b) contravene, conflict with or result in a violation of, give any Governmental Body or other Person the right to challenge the Contemplated Transactions or to exercise any remedy or obtain any relief under, any Law or any order, writ, injunction, judgment or decree to which Parent or its Subsidiaries, or any of the assets owned or used by Parent or its Subsidiaries, is subject, except as would not reasonably be expected to be material to Parent or its business;

(c) contravene, conflict with or result in a violation of any of the terms or requirements of, or give any Governmental Body the right to revoke, withdraw, suspend, cancel, terminate or modify, any Governmental Authorization that is held by Parent, except as would not reasonably be expected to be material to Parent or its business;

(d) contravene, conflict with or result in a violation or breach of, or result in a default under, any provision of any Parent Material Contract, or give any Person the right to: (i) declare a default or exercise any remedy under any Parent Material Contract; (ii) any material payment, rebate, chargeback, penalty or change in delivery schedule under any Parent Material Contract; (iii) accelerate the maturity or performance of any Parent Material Contract; or (iv) cancel, terminate or modify any term of any Parent Material Contract, except in the case of any non-material breach, default, penalty or modification; or

(e) result in the imposition or creation of any Encumbrance upon or with respect to any asset owned or used by Parent (except for Permitted Encumbrances).

Except for (i) any Consent set forth in <u>Section 3.5</u> of the Parent Disclosure Schedule under any Parent Contract, (ii) the Required Parent Stockholder Vote, (iii) the filing of the Certificates of Merger with the Secretary of State of the State of Delaware pursuant to the DGCL, (iv) the filing of the Certificate of Designation with the Secretary of State of the State of Delaware pursuant to the DGCL and (v) such consents, waivers, approvals, orders, authorizations, registrations, declarations and filings as may be required under applicable federal and state securities Laws, neither Parent nor any of its Subsidiaries is or will be required to make any filing with or give any notice to, or to obtain any Consent from, any Person in connection with (A) the execution, delivery or performance of this Agreement, or (B) the consummation of the Contemplated Transactions. The Parent Board and the First Merger Sub Board and the Second Merger Sub Board have taken and will take all actions necessary to ensure that the restrictions applicable to business combinations contained in Section 203 of the DGCL (or analogous provisions) are, and will be, inapplicable to the execution, delivery and performance of this Agreement, the Lock-Up Agreements and to the consummation of the Contemplated Transactions. No other state takeover statute or similar Law applies or purports to apply to the Merger, this Agreement or any of the other Contemplated Transactions.

3.6 Capitalization.

(a) The authorized capital stock of Parent as of the date of this Agreement consists of 125,000,000 shares of Parent Common Stock, par value \$0.01 per share, of which 21,085,935 shares have been issued and are outstanding as of the close of business on the Reference Date and 5,000,000 shares of preferred stock of Parent, par value \$0.01 per share, of which 1,044,488 shares have been designated Series A Preferred Stock, none of which are outstanding as of the date of this Agreement and 1,796,875 shares have been designated Series B Preferred Stock, none of which are outstanding as of the date of this Agreement. Parent does not hold any shares of its capital stock in its treasury.

(b) All of the outstanding shares of Parent Common Stock have been duly authorized and validly issued, and are fully paid and nonassessable. None of the outstanding shares of Parent Common Stock are entitled or subject to any preemptive right, right of participation, right of maintenance or any similar right and none of the outstanding shares of Parent Common Stock is subject to any right of first refusal in favor of Parent. Except as contemplated herein, there is no Parent Contract relating to the voting or registration of, or restricting any Person from purchasing, selling, pledging or otherwise disposing of (or granting any option or similar right with respect to), any shares of Parent Common Stock. Parent is not under any obligation, nor is it bound by any Contract pursuant to which it may become obligated, to repurchase, redeem or otherwise acquire any outstanding shares of Parent Common Stock or other securities. <u>Section 3.6(b)</u> of the Parent Disclosure Schedule accurately and completely lists all repurchase rights held by Parent with respect to shares of Parent Common Stock (including shares issued pursuant to the exercise of stock options) and specifies which of those repurchase rights are currently exercisable and whether the holder of such shares of Parent Common Stock timely filed an election with the relevant Governmental Bodies under Section 83(b) of the Code with respect to such shares.

(c) Except for the Parent Stock Plans, and except as set forth in <u>Section 3.6(c)</u> of the Parent Disclosure Schedule, Parent does not have any stock option plan or any other plan, program, agreement or arrangement providing for any equity-based compensation for any Person. As of the close of business on the Reference Date, Parent has reserved 3,360,431 shares of Parent Common Stock for issuance under the Parent Stock Plans, of which Parent RSUs and Parent Options to purchase a total of 2,457,176 shares, in the aggregate, have been issued and are currently outstanding of, which no shares are subject to Parent's right of repurchase, of which (1) 2,292,556 shares have been reserved for issuance upon exercise of Parent Options previously granted and currently outstanding under the Parent Stock Plans, (2) 164,620 shares have been reserved for issuance upon the settlement of Parent RSUs granted under the Parent Stock Plans that are outstanding as of the close of business on the Reference Date, and (3) 903,255 shares remain available for future issuance pursuant to the Parent Stock Plans. Section 3.6(c) of the Parent Disclosure Schedule sets forth the following information with respect to each Parent Option and Parent RSU outstanding as of the Reference Date: (i) the name of the holder; (ii) the number of shares of Parent Common Stock subject to such Parent Option or Parent RSU at the time of grant; (iii) the number of shares of Parent Common Stock subject to such Parent Option or Parent RSU as of the close of business on the Reference Date; (iv) the exercise price of such Parent Option; (v) the date on which such Parent Option or Parent RSU was granted; (vi) the applicable vesting schedule, including the number of vested and unvested shares as of the close of business on the Reference Date and any acceleration provisions; (vii) the date on which such Parent Option or Parent RSU expires; (viii) whether such Parent Option is intended to constitute an "incentive stock option" (as defined in the Code) or a non-qualified stock option and (ix) whether such Parent Option is "early exercisable". Parent has made available to the Company accurate and complete copies of the Parent Stock Plans and the form of the stock option agreements and restricted stock unit agreements evidencing outstanding Parent Options and Parent RSUs granted thereunder. No vesting of Parent Options or Parent RSUs will be accelerated in connection with the closing of the Contemplated Transactions other than as set forth on such Section 3.6(c) of the Parent Disclosure Schedule.

(d) Except for the Parent Options and the Parent RSUs granted pursuant to the Parent Stock Plans, and as otherwise set forth in <u>Section 3.6(d)</u> of the Parent Disclosure Schedule, there is no: (i) outstanding subscription, option, call, warrant or right (whether or not currently exercisable) to acquire any shares of the capital stock or other securities of Parent or any of its Subsidiaries; (ii) outstanding security, instrument or obligation that is or may become convertible into or exchangeable for any shares of the capital stock or other securities of Parent or any of its Subsidiaries; or (iii) condition or circumstance that could be reasonably likely to give rise to or provide a basis for the assertion of a claim by any Person to the effect that such Person is entitled to acquire or receive any shares of capital stock or other securities of Parent or any of its Subsidiaries. There are no outstanding or authorized stock appreciation, phantom stock, profit participation or other similar rights with respect to Parent or any of its Subsidiaries. In addition, there are no stockholder rights plans (or similar plan commonly referred to as a "poison pill") or bonds, debentures, notes or other indebtedness of Parent having the right to vote (or convertible into, or exchangeable for, securities having the right to vote) on any matters on which stockholders of Parent may vote.

(e) All outstanding shares of Parent Common Stock, Parent Options, Parent RSUs and other securities of Parent have been issued and granted in material compliance with (i) the Organizational Documents of Parent in effect as of the relevant time and all applicable securities Laws and other applicable Law, and (ii) all requirements set forth in applicable Contracts.

(f) All distributions, dividends, repurchases and redemptions of Parent Common Stock or other equity interests of Parent were undertaken in material compliance with (i) the Organizational Documents of Parent in effect as of the relevant time and all applicable securities Laws and other applicable Laws, and (ii) all requirements set forth in applicable Contracts.

3.7 SEC Filings; Financial Statements.

Parent has delivered or made available to the Company accurate and complete copies of all registration statements, proxy (a) statements, Certifications (as defined below) and other statements, reports, schedules, forms and other documents filed by Parent with the SEC since December 31, 2021 (the "Parent SEC Documents"), other than such documents that can be obtained on the SEC's website at www.sec.gov. Since December 31, 2021, all material statements, reports, schedules, forms and other documents, including any exhibits thereto, required to have been filed by Parent or its officers with the SEC have been so filed on a timely basis. As of the time it was filed with the SEC (or, if amended or superseded by a filing prior to the date of this Agreement, then on the date of such filing), each of the Parent SEC Documents complied in all material respects with the applicable requirements of the Securities Act or the Exchange Act, or the Sarbanes-Oxley Act (as the case may be), and the rules and regulations thereunder, and, as of the time they were filed, or if amended or superseded by a filing prior to the date of this Agreement, on the date of the last such amendment or superseding filing prior to the date of this Agreement, none of the Parent SEC Documents contained any untrue statement of a material fact or omitted to state a material fact required to be stated therein or necessary in order to make the statements therein, in the light of the circumstances under which they were made, not misleading. The certifications and statements required by (i) Rule 13a-14 under the Exchange Act and (ii) 18 U.S.C. §1350 (Section 906 of the Sarbanes-Oxley Act) relating to the Parent SEC Documents (collectively, the "Certifications") are accurate and complete and comply as to form and content with all applicable Laws, and no current or former executive officer of Parent has failed to make the Certifications required of him or her. Parent has made available to the Company true and complete copies of all correspondence, other than transmittal correspondence or general communications by the SEC not specifically addressed to Parent, between the SEC, on the one hand, and Parent, on the other, since December 31, 2021, including all SEC comment letters and responses to such comment letters and responses to such comment letters by or on behalf of Parent except for such comment letters and responses to such comment letters that are publicly accessible through EDGAR. As of the date of this Agreement, there are no outstanding unresolved comments in comment letters received from the SEC or Nasdaq with respect to Parent SEC Documents. To the Knowledge of Parent, none of the Parent SEC Documents is the subject of ongoing SEC review and there are no inquiries or investigations by the SEC or any internal investigations pending or threatened, including with regards to any accounting practices of Parent. As used in this Section 3.7, the term "file" and variations thereof shall be broadly construed to include any manner in which a document or information is filed, furnished, supplied or otherwise made available to the SEC.

(b) The financial statements (including any related notes) contained or incorporated by reference in the Parent SEC Documents: (i) complied as to form in all material respects with the published rules and regulations of the SEC applicable thereto; (ii) were prepared in accordance with GAAP (except as may be indicated in the notes to such financial statements or, in the case of unaudited financial statements, except as permitted by Form 10-Q of the SEC, and except that the unaudited financial statements may not contain footnotes and are subject to normal and recurring year-end adjustments) applied on a consistent basis unless otherwise noted therein throughout the periods indicated; and (iii) fairly present, in all material respects, the financial position of Parent and its consolidated Subsidiaries as of the respective dates thereof and the results of operations and cash flows of Parent for the periods covered thereby. Other than as expressly disclosed in the Parent SEC Documents filed prior to the date hereof, there has been no material change in Parent's accounting methods or principles that would be required to be disclosed in Parent's financial statements in accordance with GAAP.

(c) Parent's independent registered public accounting firm has at all times since the date of enactment of the Sarbanes-Oxley Act been: (i) a registered public accounting firm (as defined in Section 2(a)(12) of the Sarbanes-Oxley Act); (ii) to the Knowledge of Parent, "independent" with respect to Parent within the meaning of Regulation S-X under the Exchange Act; and (iii) to the Knowledge of Parent, in compliance with subsections (g) through (l) of Section 10A of the Exchange Act and the rules and regulations promulgated by the SEC and the Public Company Accounting Oversight Board thereunder.

(d) Since December 31, 2021, through the date of this Agreement, Parent has not received any comment letter from the SEC or the staff thereof or any correspondence from officials of Nasdaq or the staff thereof relating to the delisting or maintenance of listing of the Parent Common Stock on Nasdaq. As of the date of this Agreement, Parent has timely responded to all comment letters of the staff of the SEC relating to the Parent SEC Documents, and the SEC has not advised Parent that any final responses are inadequate, insufficient or otherwise non-responsive. Parent has made available to the Company true, correct and complete copies or all comment letters, written inquiries and enforcement correspondences between the SEC, on the one hand, and Parent, on the other hand, occurring since December 31, 2021 and will, reasonably promptly following the receipt thereof, make available to the Company any such correspondence sent or received after the date of this Agreement. To the Knowledge of Parent, as of the date of this Agreement, none of the Parent SEC Documents is the subject of an ongoing SEC report or outstanding SEC comment.

(e) Since December 31, 2021, there have been no formal investigations regarding financial reporting or accounting policies and practices discussed with, reviewed by or initiated at the direction of the chief executive officer, chief financial officer, principal accounting officer or general counsel of Parent, the Parent Board or any committee thereof, other than ordinary course audits or reviews of accounting policies and practices or internal controls required by the Sarbanes-Oxley Act.

(f) Parent is and since its first date of listing on Nasdaq, has been, in compliance in all material respects with the applicable current listing and governance rules and regulations of Nasdaq.

Parent maintains, and at all times since December 31, 2021, has maintained, a system of internal control over financial (g) reporting (as defined in Rules 13a-15(f) and 15d-15(f) of the Exchange Act) that is designed to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with GAAP and to provide reasonable assurance (i) that transactions are recorded as necessary to permit preparation of financial statements in accordance with GAAP, (ii) that receipts and expenditures are made only in accordance with authorizations of management and the Parent Board, (iii) regarding prevention or timely detection of the unauthorized acquisition, use or disposition of Parent's assets that could have a material effect on Parent's financial statements and (iv) that Parent maintains records in reasonable detail which accurately and fairly reflect the transactions and dispositions of the assets of Parent and any of its Subsidiaries. Parent has evaluated the effectiveness of Parent's internal control over financial reporting as of December 31, 2021, and, to the extent required by applicable Law, presented in any applicable Parent SEC Document that is a report on Form 10-K or Form 10-Q (or any amendment thereto) its conclusions about the effectiveness of the internal control over financial reporting as of the end of the period covered by such report or amendment based on such evaluation. Parent has disclosed, based on its most recent evaluation of internal control over financial reporting, to Parent's auditors and audit committee (and has described in Section 3.7(g) of the Parent Disclosure Schedule) (A) all material weaknesses and all significant deficiencies, if any, in the design or operation of internal control over financial reporting that are reasonably likely to adversely affect Parent's ability to record, process, summarize and report financial information and (B) any fraud, whether or not material, that involves Parent, any of its Subsidiaries, Parent's management or other employees who have a role in the preparation of financial statements or the internal accounting controls utilized by the Parent and its Subsidiaries or (C) any claim or allegation regarding any of the foregoing. Parent has not identified, based on its most recent evaluation of internal control over financial reporting, any significant deficiencies or material weaknesses in the design or operation of Parent's internal control over financial reporting.

(h) Parent maintains "disclosure controls and procedures" (as defined in Rules 13a-15(e) and 15d-15(e) of the Exchange Act) that are reasonably designed to ensure that information required to be disclosed by Parent in the periodic reports that it files or submits under the Exchange Act is recorded, processed, summarized and reported within the required time periods, and that all such information is accumulated and communicated to Parent's management as appropriate to allow timely decisions regarding required disclosure and to make the Certifications.

(i) Parent has not been and is not currently a "shell company" as defined under Section 12b-2 of the Exchange Act.

3.8 <u>Absence of Changes</u>. Except as set forth in <u>Section 3.8</u> of the Parent Disclosure Schedule, after the date of the Parent Balance Sheet, Parent and its Subsidiaries have conducted its business only in the Ordinary Course of Business (except for the execution and performance of this Agreement and the discussions, negotiations and transactions related thereto) and there has not been any (a) Parent Material Adverse Effect and (b) neither Parent nor any of its Subsidiaries has done any of the following:

(a) declared, accrued, set aside or paid any dividend or made any other distribution in respect of any shares of its capital stock or repurchased, redeemed or otherwise reacquired any shares of its capital stock or other securities (except in connection with the payment of the exercise price and/or withholding Taxes incurred upon the exercise, settlement or vesting of any award granted under the Parent Stock Plans);

(b) sold, issued, granted, pledged or otherwise disposed of or encumbered or authorized any of the foregoing with respect to: (i) any capital stock or other security of Parent (except for Parent Common Stock issued upon the valid exercise of outstanding Parent Options); (ii) any option, warrant or right to acquire any capital stock or any other security, other than option grants to employees in the Ordinary Course of Business; or (iii) any instrument convertible into or exchangeable for any capital stock or other security of Parent;

(c) except as required to give effect to anything in contemplation of the Closing, amended any of its Organizational Documents, or effected or been a party to any merger, consolidation, share exchange, business combination, recapitalization, reclassification of shares, stock split, reverse stock split or similar transaction except, for the avoidance of doubt, the Contemplated Transactions;

with any other Entity;

(d) formed any Subsidiary or acquired any equity interest or other interest in any other Entity or entered into a joint venture

(e) (i) lent money to any Person (except for the advance of reasonable business expenses to employees, directors and consultants in the Ordinary Course of Business), (ii) incurred or guaranteed any indebtedness for borrowed money, or (iii) guaranteed any debt securities of others;

(f) other than as required by applicable Law or the terms of any Parent Benefit Plan as in effect on the date of this Agreement: (i) adopted, terminated, established or entered into any Parent Benefit Plan; (ii) caused any Parent Benefit Plan to be amended in any material respect; (iii) paid any material bonus or distributed any profit-sharing account balances or similar payment to, or, other than in the Ordinary Course of Business, increased the amount of the wages, salary, commissions, benefits or other compensation or remuneration payable to, any of its directors, officers or employees; (iv) increased the severance or change-of-control benefits offered to any current, former or new employees, directors or consultants or (v) hired, terminated or gave notice of termination (other than for cause) to any (x) officer or (y) employee whose annual base salary is or is expected to be more than \$100,000 per year;

(g) entered into any collective bargaining agreement or similar agreement with any labor union, or similar labor organization;

entered into any material transaction other than (i) in the Ordinary Course of Business or (ii) in connection with the

(h) Contemplated Transactions;

(i) acquired any material asset or sold, leased or otherwise irrevocably disposed of any of its assets or properties, or granted any Encumbrance (other than a Permitted Encumbrance) with respect to such assets or properties, except in the Ordinary Course of Business;

(j) sold, assigned, transferred, licensed, sublicensed or otherwise disposed of any material Parent IP;

(k) made, changed or revoked any material Tax election, failed to pay any income or other material Tax as such Tax becomes due and payable, filed any amendment making any material change to any Tax Return, settled or compromised any income or other material Tax liability, entered into any Tax allocation, sharing, indemnification or other similar agreement or arrangement (including any "closing agreement" described in Section 7121 of the Code (or any similar Law) with any Governmental Body, but excluding customary commercial Contracts entered into in the Ordinary Course of Business the principal subject matter of which is not Taxes), requested or consented to any extension or waiver of any limitation period with respect to any claim or assessment for any income or other material Taxes (other than pursuant to an extension of time to file any Tax Return granted in the Ordinary Course of Business of not more than six months), or adopted or changed any material accounting method in respect of Taxes;

(1) made any expenditures, incurred any Liabilities or discharged or satisfied any Liabilities, in each case, in amounts that exceed the aggregate amount of \$100,000;

- (m) other than as required by Law or GAAP, taken any action to change accounting policies or procedures;
- (n) initiated or settled any Legal Proceeding; or
- (o) agreed, resolved or committed to do any of the foregoing.

3.9 <u>Absence of Undisclosed Liabilities</u>. As of the date hereof, neither Parent nor any of its Subsidiaries has any Liability, individually or in the aggregate, except for: (a) Liabilities disclosed, reflected or reserved against in the Parent Balance Sheet; (b) Liabilities that have been incurred by Parent or its Subsidiaries since the date of the Parent Balance Sheet in the Ordinary Course of Business and which are not material to Parent or its Subsidiaries, individually or in the aggregate; (c) Liabilities for performance of obligations of Parent or any of its Subsidiaries under Parent Contracts in the Ordinary Course of Business, which, in each case, are not related to any breach or default of Parent or its Subsidiaries; (d) Liabilities incurred in connection with the Contemplated Transactions; and (e) Liabilities described in Section 3.9 of the Parent Disclosure Schedule.

3.10 <u>Title to Assets</u>. Each of Parent and its Subsidiaries owns, and has good and valid title to, or, in the case of leased properties and assets, valid leasehold interests in, all tangible properties or tangible assets and equipment used or held for use in its business or operations or purported to be owned by it that are material to Parent and its Subsidiaries or their business, including: (a) all tangible assets reflected on the Parent Balance Sheet; and (b) all other tangible assets reflected in the books and records of Parent or any of its Subsidiaries as being owned by Parent or such Subsidiary. All of such assets are owned or, in the case of leased assets, leased by Parent or its Subsidiaries free and clear of any Encumbrances, other than Permitted Encumbrances.

3.11 **Real Property; Leasehold**. Neither Parent nor any of its Subsidiaries own or ever have owned any real property. Parent has made available to the Company (a) an accurate and complete list of all real properties with respect to which Parent directly or indirectly holds a valid leasehold interest as well as any other real estate that is in the possession of, or occupied or leased by, Parent or any of its Subsidiaries, and (b) copies of all leases under which any such real property is possessed, occupied or leased (the "*Parent Real Estate Leases*"), each of which is in full force and effect, with no existing material default thereunder by Parent or any of its Subsidiaries, or to the Knowledge of Parent, any other party thereto. Parent's possession, occupancy, lease, use and/or operation of each such leased property conforms to all applicable Laws in all material respects, and Parent has exclusive possession of each such leased property and leasehold interest and has not granted any occupancy rights to tenants or licensees with respect to such leased property or leasehold interest. In addition, each such leased property and leasehold interest is free and clear of all Encumbrances other than Permitted Encumbrances. Parent has not received any written notice from its landlords or any Governmental Body that: (i) relates to violations of building, zoning, safety or fire ordinances or regulations; (ii) claims any defect or deficiency with respect to any of such properties; or (iii) requests the performance of any repairs, alterations or other work to such properties.

3.12 Intellectual Property; Privacy.

Section 3.12(a) of the Parent Disclosure Schedule identifies each item of Parent Registered IP, including, with respect to (a) each application and registration: (i) the name of the applicant or registrant and any other co-owners, (ii) the jurisdiction of application or registration, (iii) the application or registration number, (iv) the date of issue, filing, or registration, as applicable, and (v) to the extent applicable, the expiration date. To the Knowledge of Parent, each of the Patents and Patent applications included in Section 3.12(a) of the Parent Disclosure Schedule properly identifies by name each and every inventor of the inventions claimed therein as determined in accordance with applicable Laws of the United States and the applicable foreign jurisdiction. For all Parent Registered IP owned or purported to be owned, in whole or in part, by the Parent (the "Owned Parent Registered IP"), and all other Parent Registered IP for which the Parent has responsibility for prosecution and maintenance activities, all necessary registration, maintenance, renewal and other relevant filing fees due through the Closing Date have been timely paid and all necessary documents and certificates in connection therewith have been timely filed with the relevant Patent, Trademark, Copyright, Internet domain name or other authorities in the United States or the applicable foreign jurisdiction, as the case may be, for the purpose of maintaining such Parent Registered IP in full force and effect and, except as set forth on Section 3.12(a) of the Parent Disclosure Schedule, there are no such filings, payments or other actions that must be made or taken on or before the three-month anniversary of the Closing Date. As of the date of this Agreement, no cancellation, interference, opposition, reissue, reexamination or other proceeding of any nature (other than office actions or similar communications issued by any Governmental Body in the ordinary course of prosecution of any pending applications for registration) is pending or, to the Knowledge of Parent, threatened in writing, in which the scope, validity, enforceability or ownership of any Parent IP is being or has been contested or challenged. To the Knowledge of Parent, each item of Parent IP is valid and enforceable, and with respect to the material Parent Registered IP, subsisting. The Parent and its Subsidiaries have complied with all Laws regarding the duty of disclosure, candor and good faith in connection with each Patent included in the Owned Parent Registered IP and all other Parent Registered IP for which the Parent or its Subsidiaries has responsibility for prosecution and maintenance activities.

(b) Parent or its Subsidiaries (i) exclusively own, are the sole assignee of, or have exclusively licensed all Owned Parent Registered IP and all other Parent IP (other than as disclosed in <u>Section 3.12(b)</u> of the Parent Disclosure Schedule), and (ii) has valid and continuing rights, pursuant to the Parent In-bound Licenses, to use all other material Intellectual Property Rights as the same are used in or necessary for the conduct of the business as presently conducted by the Parent and its Subsidiaries, in each case, free and clear of all Encumbrances other than Permitted Encumbrances. The Owned Parent Registered IP is currently in compliance in all material respects with all Laws necessary to record and perfect Parent's or its Subsidiaries', as applicable, interest in, and the chain of title of, the Owned Parent Registered IP and to ensure the ability to claim priority in all jurisdictions. The Parent IP constitute all Intellectual Property Rights used in, material to or otherwise necessary for the operation of Parent's and any of its Subsidiaries' business as currently conducted. Each Parent Associate involved in the creation or development of any material Parent IP, pursuant to such Parent Associate's activities on behalf of Parent or any of its Subsidiaries, has signed a written agreement containing an assignment of such Parent Associate's rights in such Parent IP to Parent or its Subsidiaries. Each Parent Associate who has or has had access to Parent's or any of its Subsidiaries' trade secrets or confidential information has signed a written agreement containing confidentiality provisions protecting the Parent IP, trade secrets and confidential information. Parent has maintained copies of each such executed written agreement and, to the Knowledge of the Parent, no party thereto is in default or breach of any such agreements. Parent and its Subsidiaries have taken commercially reasonable steps to protect and preserve the confidentiality of their respective trade secrets and confidential information.

(c) No funding, facilities or personnel of any Governmental Body or any university, college, research institute or other educational institution has been used or is being used to create, in whole or in part, any Parent IP that are owned or purported to be owned by Parents or any of its Subsidiaries, except for any such funding or use of facilities or personnel that does not result in such Governmental Body or institution obtaining ownership rights, license rights, or any other right to such Parent IP (except for use rights during the term of the applicable agreement between the Parent or its Subsidiary and such Governmental Body or educational institution), including the right to receive royalties for the practice of such Parent IP (other than pursuant to any Parent In-bound License disclosed on Section 3.12(d) of the Parent Disclosure Schedule).

(d) Section 3.12(d) of Parent Disclosure Schedule sets forth each license agreement pursuant to which Parent (i) is granted a license under any material Intellectual Property Right owned by any third party that is used by Parent or its Subsidiaries in its business as currently conducted (each a "*Parent In-bound License*") or (ii) grants to any third party a license under any material Parent IP (each a "*Parent Out-bound License*") (*provided*, that, Parent In-bound Licenses shall not include, when entered into in the Ordinary Course of Business, material transfer agreements, services agreements, clinical trial agreements, agreements with Parent Associates, non-disclosure agreements, commercially available Software-as-a-Service offerings, or off-the-shelf software licenses; and Parent Out-bound Licenses shall not include, when entered into in the Ordinary Course of Business, material transfer agreements, clinical trial agreements, services agreements, non-disclosure agreements, or non-exclusive outbound licenses). All Parent In-bound Licenses and Parent Out-bound Licenses, to the Knowledge of Parent, are in full force and effect and are valid, enforceable and binding obligations of Parent or its Subsidiaries and, to the Knowledge of Parent, each other party to such Parent In-bound Licenses, is in material breach under any Parent In-bound Licenses or Parent Out-bound Licenses. Except as set forth in <u>Section 3.12(c)</u> of the Parent Disclosure Schedule, none of the terms or conditions of any Parent In-bound License or any Parent Out-bound License requires Parent or any of its Subsidiaries or any of their Affiliates to maintain, develop or prosecute any Intellectual Property Rights.

(e) To the Knowledge of Parent: (i) the operation of the business of Parent and its Subsidiaries as currently conducted has not infringed, misappropriated or otherwise violated and does not infringe, misappropriate or otherwise violate any Intellectual Property Rights of any other Person and (ii) no other Person is infringing, misappropriating or otherwise violating any Parent IP or any Intellectual Property Rights exclusively licensed to the Parent. No Legal Proceeding is pending (or, to the Knowledge of Parent, is threatened in writing) (A) against Parent or its Subsidiaries alleging that the operation of the business of Parent or its Subsidiaries infringes or constitutes the misappropriated or otherwise violated any of the Parent IP or any Intellectual Property Rights exclusively licensed to the Parent or its Subsidiaries alleging that another Person has infringed, misappropriated or otherwise violated any of the Parent IP or any Intellectual Property Rights exclusively licensed to the Parent. Since December 31, 2021, neither Parent nor its Subsidiaries have received any written notice or other written communication alleging that the operation of the business of Parent or its Subsidiaries the misappropriation or other violation of any Intellectual Property Right of another Person.

(f) None of the Parent IP owned by the Parent or, to the Knowledge of Parent, any Parent IP exclusively licensed by Parent or its Subsidiaries, is subject to any pending or outstanding injunction, directive, order, judgment or other disposition of dispute, that adversely and materially restricts the use, transfer, registration or licensing by Parent or its Subsidiaries of any such Parent IP.

(g) To the Knowledge of Parent, Parent and the operation of Parent's and its Subsidiaries' business are, and have at all times been, in material compliance with all applicable Laws and Privacy and Data Processing Requirements. To the knowledge of Parent, Parent and its Subsidiaries have at all applicable times provided all notices, and obtained and maintained all rights, consents, and authorizations, to Process Parent Data as Processed by or for Parent or its Subsidiaries. Since December 31, 2021, there have been (i) no loss or theft of, or security breach relating to. Parent Data, (ii) no violation of any security policy of Parent or its Subsidiaries regarding any such Parent Data, and (iii) no unauthorized access to, or unauthorized, unintended, or improper use, disclosure, or other Processing of, any Parent Data. Parent and its Subsidiaries have taken commercially reasonable steps and implemented reasonable disaster recovery and security plans and procedures to protect the information technology systems used in, material to or necessary for operation of Parent's or its Subsidiaries business as currently conducted and Parent Data from unauthorized use, access, or other Processing, and the Parent has taken commercially reasonable steps to require that any third party with access to Parent Data collected by or on behalf of the Parent has taken commercially reasonable steps to protect the Parent Data provided to them by the Parent. The Parent has implemented and maintained commercially reasonable policies, procedures and systems for receiving and appropriately responding to requests from individuals concerning their Parent Data where such steps are required by applicable Privacy and Data Processing Requirements.

(h) To the Knowledge of Parent, there have been no (i) material malfunctions or unauthorized intrusions or breaches of the information technology systems used in, material to or necessary for the operation of Parent's or its Subsidiaries' business, (ii) material unauthorized access to, or other processing of, Parent Data, or (iii) material breaches, security incidents, misuse of or unauthorized access to or disclosure of any Parent Data in the possession or control of the Parent and the Parent has not provided or been legally required to provide any notices to any Person in connection with an unauthorized disclosure of Parent Data. To the Knowledge of the Parent, the Parent has not been the subject of or received written notice of any complaints, claims or investigations related to their collection, use, storage or processing of Parent Data or alleging any violation of applicable Privacy and Data Processing Requirements.

(i) The Parent has taken reasonable security and other measures, including measures against unauthorized disclosure, to protect and maintain the secrecy, confidentiality and value of the Know-How and other confidential information included in the Parent IP. No trade secret, Know-How or proprietary information material to the business of the Parent as presently conducted, as of the date of this Agreement has been authorized to be disclosed or, to the Knowledge of the Parent, has been actually disclosed by the Parent or any Subsidiary to any Person other than pursuant to a non-disclosure agreement or other agreement adequately restricting the disclosure and use of such Intellectual Property Rights or information, and excluding any Know-How or proprietary information disclosed by the Parent in publications or public filings, including as required under applicable securities laws.

(j) To the Knowledge of Parent, the computer systems, including the software, firmware, hardware, networks, interfaces, platforms and related systems, owned, leased or licensed by the Parent an (collectively, the "*Parent Systems*") perform in all material respects as is necessary for the conduct of its business as presently conducted by Parent. To the Knowledge of the Parent, in the 12 months immediately prior to the date of this Agreement, (i) there have been no material failures, breakdowns or other adverse events materially affecting any such Parent Systems that have caused a material disruption or interruption to the conduct of the business of the Parent as currently conducted, and (ii) there have not been any material incidents of unauthorized access or other security breaches of the Parent Systems.

3.13 Agreements, Contracts and Commitments.

(a) <u>Section 3.13</u> of the Parent Disclosure Schedule lists the following Parent Contracts in effect as of the date of this Agreement other than any Parent Benefit Plans (each, a "*Parent Material Contract*" and collectively, the "*Parent Material Contracts*"):

(i) a material Contract as defined in Item 601(b)(10) of Regulation S-K as promulgated under the Securities Act;

Course of Business;

(ii) each Parent Contract relating to any agreement of indemnification or guaranty not entered into in the Ordinary

(iii) each Parent Contract containing (A) any covenant limiting in any material respect the freedom of Parent or its Subsidiaries to engage in any line of business or compete with any Person, (B) any most-favored nation or other preferred pricing arrangement in favor of a Person other than Parent or any similar term by which any Person is or could become entitled to any benefit, right or privilege that must be at least as favorable to such Person as those offered to any other Person, (C) any exclusivity provision, option to receive a license, right of first refusal or right of first negotiation or similar covenant in favor of a Person other than Parent, or (D) any non-solicitation provision not entered into in the Ordinary Course of Business;

(iv) each Parent Contract relating to capital expenditures and requiring payments after the date of this Agreement in excess of \$100,000 pursuant to its express terms and not cancelable without penalty;

Entity;

(v) each Parent Contract relating to the disposition or acquisition of material assets or any ownership interest in any

(vi) each Parent Contract relating to any mortgages, indentures, loans, notes or credit agreements, security agreements or other agreements or instruments relating to the borrowing of money or extension of credit or creating any material Encumbrances with respect to any assets of Parent or its Subsidiaries or any loans or debt obligations with officers or directors of Parent;

(vii) each Parent Contract requiring payment by or to Parent after the date of this Agreement in excess of \$100,000 in the aggregate in the current calendar year or any future calendar year pursuant to its express terms relating to: (A) any agreement involving the development or commercialization of any pharmaceutical product (identifying any that contain exclusivity provisions); (B) any agreement involving provision of services or products with respect to any pre-clinical or clinical development activities of Parent; (C) any dealer, distributor, joint marketing, alliance, joint venture, cooperation, development or other agreement currently in force under which Parent has continuing obligations to develop or market any product, technology or service, or any agreement pursuant to which Parent has continuing obligations to develop any Intellectual Property Rights that will not be owned, in whole or in part, by Parent; or (D) any Parent Contract with any third party providing any services relating to the manufacture or production of any product, service or technology of Parent or any Parent Contract to sell, distribute or commercialize any products or service of Parent;

(viii) each Parent Contract with any financial advisor, broker, finder, investment banker or other similar Person providing financial advisory services to Parent in connection with the Contemplated Transactions;

- (ix) each Parent Real Estate Lease;
- (x) each Parent Contract with any Governmental Body;

(xi) each Parent Out-bound License and Parent In-bound License, and each Parent Contract containing a covenant not to sue or otherwise enforce any Intellectual Property Rights;

(xii) each Parent Contract requiring the payment of any royalty, dividend or similar arrangement based on the revenues or profits of Parent or its Subsidiaries;

(xiii) each Parent Contract, offer letter, employment agreement, or independent contractor agreement with any employee, independent contractor or other natural person service provider that (A) is not immediately terminable by Parent without notice, severance, or other cost or liability, except as required under applicable Law, or (B) provides for retention payments, change-of-control payments, severance, accelerated vesting, or any similar payment or benefit that may or will become due as a result of the Merger;

(xiv) each Parent Contract providing any option to receive a license or other right, any right of first negotiation, any right of first refusal or any similar right to any Person related to any material Parent IP or material Intellectual Property Right licensed to Parent under a Parent In-bound License; and

(xv) each Parent Contract entered into in settlement of any Legal Proceeding or other dispute; and

(xvi) any other Contract that is not terminable at will (with no penalty or payment or requirement for prior notice, except as required by applicable law) by Parent or its Subsidiaries, as applicable, and (A) which involves payment or receipt by Parent or its Subsidiaries after the date of this Agreement under any such agreement, Contract or commitment of more than \$100,000 in the aggregate, or obligations after the date of this Agreement in excess of \$100,000 in the aggregate, or (B) that is material to the business or operations of Parent and its Subsidiaries, taken as a whole.

(b) Parent has delivered or made available to the Company accurate and complete copies of all Parent Material Contracts, including all amendments thereto. There are no Parent Material Contracts that are not in written form. Neither Parent nor any of its Subsidiaries has, nor, to Parent's Knowledge, as of the date of this Agreement, has any other party to a Parent Material Contract, breached, violated or defaulted under, or received notice that it breached, violated or defaulted under, any of the terms or conditions of any Parent Material Contract. As to Parent and its Subsidiaries, as of the date of this Agreement, each Parent Material Contract is valid, binding, enforceable and in full force and effect, subject to the Enforceability Exceptions. No Person is renegotiating, or has a right pursuant to the terms of any Parent Material Contract to change, any material amount paid or payable to Parent under any Parent Material Contract or any other material term or provision of any Parent Material Contract, and no Person has indicated to Parent that it desires to renegotiate, modify, not renew or cancel any Parent Material Contract.

3.14 <u>Compliance; Permits</u>. Parent or its Subsidiaries hold all required Governmental Authorizations which are material to the operation of the business of Parent or such Subsidiary as currently conducted (the "*Parent Permits*"). <u>Section 3.14</u> of the Parent Disclosure Schedule identifies each Parent Permit. Each such Parent Permit is valid and in full force and effect, and Parent is in material compliance with the terms of the Parent Permits. No Legal Proceeding is pending or, to the Knowledge of Parent, threatened, which seeks to revoke, limit, suspend, or materially modify any Parent Permit.

3.15 Legal Proceedings; Orders.

(a) As of the date of this Agreement, there is no pending Legal Proceeding and, to the Knowledge of Parent, no Person has threatened in writing to commence any Legal Proceeding: (i) that involves (A) Parent, (B) any of its Subsidiaries, (C) any Parent Associate (in his or her capacity as such) or (D) any of the material assets owned or used by Parent or its Subsidiaries; or (ii) that challenges, or that would have the effect of preventing, delaying, making illegal or otherwise interfering with, the Contemplated Transactions.

(b) Except as set forth in <u>Section 3.15(b)</u> of the Parent Disclosure Schedule, since December 31, 2021 through the date of this Agreement, no Legal Proceeding has been pending against Parent that resulted in material liability to Parent.

(c) There is no order, writ, injunction, judgment or decree to which Parent or any of its Subsidiaries, or any of the material assets owned or used by Parent or any of its Subsidiaries, is subject. To the Knowledge of Parent, no officer of Parent or any of its Subsidiaries is subject to any order, writ, injunction, judgment or decree that prohibits such officer or employee from engaging in or continuing any conduct, activity or practice relating to the business of Parent or any of its Subsidiaries or to any material assets owned or used by Parent or any of its Subsidiaries.

3.16 Tax Matters.

(a) Parent and each of its Subsidiaries have filed all income and other material Tax Returns that were required to be filed by or with respect to it under applicable Law. All such Tax Returns are correct and complete in all material respects and have been prepared in compliance with all applicable Law. No written claim has ever been made by any Governmental Body in any jurisdiction where Parent or any of its Subsidiaries does not file a particular Tax Return or pay a particular Tax that Parent or such Subsidiary is subject to taxation by that jurisdiction.

(b) All income and other material Taxes due and owing by Parent or any of its Subsidiaries on or before the date hereof (whether or not shown on any Tax Return) have been fully and timely paid. The unpaid Taxes of Parent and its Subsidiaries did not, as of the date of the Parent Balance Sheet, materially exceed the reserve for Tax liability (excluding any reserve for deferred Taxes established to reflect timing differences between book and Tax items) set forth on the face of the Parent Balance Sheet. Since the Parent Balance Sheet Date, neither Parent nor any of its Subsidiaries has incurred any material Liability for Taxes outside the Ordinary Course of Business.

(c) All Taxes that Parent and each of its Subsidiaries is or was required by Law to withhold or collect have been duly and timely withheld or collected in all material respects on behalf of its respective employees, independent contractors, stockholders, lenders, customers or other third parties and have been timely paid to the proper Governmental Body or other Person or properly set aside in accounts for this purpose.

or any of its Subsidiaries.

(d) There are no Encumbrances for material Taxes (other than Taxes not yet due and payable) upon any of the assets of Parent

(e) No deficiencies for a material amount of Taxes with respect to Parent or any of its Subsidiaries have been claimed, proposed or assessed by any Governmental Body in writing. There are no pending or ongoing and, to the Knowledge of Parent, threatened audits, assessments or other actions for or relating to any liability in respect of a material amount of Taxes of Parent. Neither Parent nor any of its Subsidiaries (or predecessors thereof) has waived any statute of limitations or agreed to any extension of time with respect to any income or other material Tax assessment or deficiency.

(f) Neither Parent nor any of its Subsidiaries has been a United States real property holding corporation within the meaning of Section 897(c)(2) of the Code during the applicable period specified in Section 897(c)(1)(A)(ii) of the Code.

(g) Neither Parent nor any of its Subsidiaries is a party to any Tax allocation agreement, Tax sharing agreement, Tax indemnity agreement, or similar agreement or arrangement, other than customary commercial Contracts entered into in the Ordinary Course of Business the principal subject matter of which is not Taxes.



(h) Neither Parent nor any of its Subsidiaries will be required to include any item of income in, or exclude any item of deduction from, taxable income for any Tax period (or portion thereof) ending after the Closing Date as a result of any: (i) change in method of accounting for Tax purposes for a Tax period ending on or prior to the Closing Date; (ii) use of an improper method of accounting for a Tax period ending on or prior to the Closing Date; (iii) "closing agreement" as described in Section 7121 of the Code (or any similar provision of state, local or foreign Law) executed on or prior to the Closing Date; (iv) intercompany transaction or excess loss account described in Treasury Regulations under Section 1502 of the Code (or any similar provision of state, local or foreign Law); (v) installment sale or open transaction disposition made on or prior to the Closing Date; (vi) prepaid amount, advance payment or deferred revenue received or accrued on or prior to the Closing Date; (vii) application of Section 367(d) of the Code to any transfer of intangible property on or prior to the Closing Date; or (viii) application of Section 951 or 951A of the Code (or any similar provision of state, local or foreign Law) to any income received or accrued on or prior to the Closing Date. Parent has not made any election under Section 965(h) of the Code.

(i) Neither Parent nor any of its Subsidiaries has ever been (i) a member of a consolidated, combined or unitary Tax group (other than such a group the common parent of which is Parent) or (ii) a party to any joint venture, partnership, or other arrangement that is treated as a partnership for U.S. federal income Tax purposes. Parent has no Liability for any material Taxes of any Person (other than Parent and any of its Subsidiaries) under Treasury Regulations Section 1.1502-6 (or any similar provision of state, local, or foreign Law), as a transferee or successor, by Contract (other than a Contract entered into in the Ordinary Course of Business the principal subject matter of which is not Taxes) or otherwise by operation of Law.

(j) Neither Parent nor any of its Subsidiaries has distributed stock of another Person, or had its stock distributed by another Person, in a transaction that was purported or intended to be governed in whole or in part by Section 355 of the Code or Section 361 of the Code (or any similar provisions of state, local or foreign Law).

(k) Parent has never had a permanent establishment (within the meaning of an applicable Tax treaty) or otherwise had an office or fixed place of business in a country other than the country in which it is organized.

(1) Neither Parent nor any of its Subsidiaries has participated in or been a party to a transaction that, as of the date of this Agreement, constitutes a "listed transaction" within the meaning of Section 6707A(c)(2) of the Code and Treasury Regulations Section 1.6011-4(b)(2).

(m) <u>Section 3.16(m)</u> of the Parent Disclosure Schedule sets forth the entity classification of Parent and each of its Subsidiaries for U.S. federal income tax purposes. Neither Parent nor any of its Subsidiaries has made an election or taken any other action to change its federal and state income tax classification from such classification.

For purposes of this <u>Section 3.16</u>, each reference to Parent or any of its Subsidiaries shall be deemed to include any Person that was liquidated into, merged with, or is otherwise a predecessor to, Parent.

3.17 Employee and Labor Matters; Benefit Plans.

(a) Section 3.17(a) of the Parent Disclosure Schedule is a list of all Parent Benefit Plans, other than employment offer letters on Parent's standard form and other than individual Parent Options or other compensatory equity award agreements made pursuant to the Parent's standard forms, in which case only representative standard forms of such agreements shall be scheduled. "*Parent Benefit Plan*" means each (i) "employee benefit plan" as defined in Section 3(3) of ERISA and (ii) other pension, retirement, deferred compensation, excess benefit, profit sharing, bonus, incentive, equity or equity-based, phantom equity, employment, consulting, severance, change-of-control, retention, health, life, disability, group insurance, paid time off, holiday, welfare and fringe benefit plan, program, agreement, Contract, or arrangement (whether written or unwritten, qualified or nonqualified, funded or unfunded, subject or not subject to ERISA and including any that have been frozen), in each case, sponsored, maintained, administered, contributed to, or required to be contributed to, by Parent or any of its Subsidiaries for the benefit of any current or former employee, director, officer or independent contractor of Parent or any of its Subsidiaries or under which Parent or any of its Subsidiaries has any actual or contingent liability (including, without limitation, by reason of having a Parent ERISA Affiliate). Section 3.17(a) of the Parent Disclosure Schedule indicates which of the Parent Benefit Plans are maintained by a professional employer organization ("*Parent PEO Plans*"). Notwithstanding anything herein to the contrary, all representations pursuant to this Section 3.17 in respect of any Parent PEO Plan shall only be made to Parent's Knowledge.

(b) As applicable with respect to each material Parent Benefit Plan, and to the extent provided to Parent from the professional employer organization, Parent has made available to the Company true and complete copies of (i) each material Parent Benefit Plan, including all amendments thereto, and in the case of an unwritten material Parent Benefit Plan, a written description thereof, (ii) all current trust documents, investment management Contracts, custodial agreements, administrative services agreements and insurance and annuity Contracts relating thereto, (iii) the current summary plan description and each summary of material modifications thereto, (iv) the most recently filed annual reports with any Governmental Body (*e.g.*, Form 5500 and all schedules thereto), (v) the most recent IRS determination, opinion or advisory letter, (vi) the most recent summary annual reports, nondiscrimination testing reports, actuarial reports, financial statements and trustee reports, and (vii) all notices and filings from the IRS or Department of Labor or other Governmental Body concerning audits or investigations, or "prohibited transactions" within the meaning of Section 406 of ERISA or Section 4975 of the Code.

(c) Each Parent Benefit Plan has been maintained, operated and administered in compliance in all material respects with its terms and the applicable provisions of ERISA, the Code and all other Laws.

(d) The Parent Benefit Plans which are "employee pension benefit plans" within the meaning of Section 3(2) of ERISA and which are intended to meet the qualification requirements of Section 401(a) of the Code have received determination or opinion letters from the IRS on which they may currently rely to the effect that such plans are qualified under Section 401(a) of the Code and the related trusts are exempt from federal income Taxes under Section 501(a) of the Code, respectively, and nothing has occurred that would reasonably be expected to materially adversely affect the qualification of such Parent Benefit Plan or the tax exempt status of the related trust.

(e) Neither Parent, any of its Subsidiaries nor any Parent ERISA Affiliate maintains, contributes to, is required to contribute to, or has any actual or contingent liability with respect to, (i) any "employee pension benefit plan" (within the meaning of Section 3(2) of ERISA) that is subject to Title IV or Section 302 of ERISA or Section 412 of the Code, (ii) any "multiemployer plan" (within the meaning of Section 3(37) of ERISA), (iii) any "multiple employer plan" (within the meaning of Section 3(37) of ERISA), employee pension of Section 3(40) of ERISA).

(f) There are no pending audits or investigations by any Governmental Body involving any Parent Benefit Plan, and no pending or, to the Knowledge of Parent, threatened claims (except for routine individual claims for benefits payable in the normal operation of the Parent Benefit Plans), suits or proceedings involving any Parent Benefit Plan, or, to the Knowledge of Parent, any fiduciary thereof or service provider thereto, in any case except as would not be reasonably expected to result in material liability to Parent or any of its Subsidiaries. All contributions and premium payments required to have been made under any of the Parent Benefit Plans or by applicable Law (without regard to any waivers granted under Section 412 of the Code), have been timely made and neither Parent nor any Parent ERISA Affiliate has any material liability for any unpaid contributions with respect to any Parent Benefit Plan.

(g) None of Parent, any of its Subsidiaries or any Parent ERISA Affiliates, or to the Knowledge of Parent, any fiduciary, trustee or administrator of any Parent Benefit Plan, has engaged in, or in connection with the Contemplated Transactions will engage in, any transaction with respect to any Parent Benefit Plan which would subject any such Parent Benefit Plan, Parent, any of its Subsidiaries or Parent ERISA Affiliates to a material Tax, material penalty or material liability for a "prohibited transaction" under Section 406 of ERISA or Section 4975 of the Code.

(h) No Parent Benefit Plan provides death, medical, dental, vision, life insurance or other welfare benefits beyond termination of service or retirement other than coverage mandated by Law and to the Knowledge of Parent, neither Parent nor any of its Subsidiaries has made a written representation promising the same.

(i) Except as set forth in <u>Section 3.17(i)</u> of the Parent Disclosure Schedule, neither the execution of this Agreement, nor the performance of the Contemplated Transactions (either alone or when combined with the occurrence of any other event, including without limitation, a termination of employment) will: (i) result in any payment becoming due to any current or former employee, director, officer, or independent contractor of Parent or any Subsidiary thereof pursuant to any Parent Benefit Plan, (ii) increase any amount of compensation or benefits otherwise payable under any Parent Benefit Plan, (iii) result in the acceleration of the time of payment, funding or vesting of any benefits under any Parent Benefit Plan, (iv) require any contribution or payment to fund any obligation under any Parent Benefit Plan or (v) limit the right to merge, amend or terminate any Parent Benefit Plan.

(j) Except as set forth in <u>Section 3.17(j)</u> of the Parent Disclosure Schedule, neither the execution of, nor the consummation of the Contemplated Transactions (either alone or when combined with the occurrence of any other event, including without limitation, a termination of employment) will result in the receipt or retention by any person who is a "disqualified individual" (within the meaning of Code Section 280G) with respect to Parent and its Subsidiaries of any payment or benefit that is or could be characterized as a "parachute payment" (within the meaning of Code Section 280G), determined without regard to the application of Code Section 280G(b)(5).

(k) No current or former employee, officer, director or independent contractor of Parent or any of its Subsidiaries has any "gross up" agreements with the Parent or any of its Subsidiaries for any Taxes imposed under Code Section 409A or Code Section 4999.

(1) To the Knowledge of Parent, each Parent Benefit Plan maintained for the benefit of service providers located outside of the United States (each, a "*Parent Foreign Plan*") has obtained from the Governmental Body having jurisdiction with respect to such plan any required determinations that such plan is in compliance with the Laws of any such Governmental Body.

(m) The assets of each of the Parent Foreign Plans that is similar to an employee pension benefit plan (as defined in Section 3(2) of ERISA (whether or not subject to ERISA)) or that otherwise provides retirement, medical or life insurance benefits following retirement or other termination of service or employment are at least equal to the liabilities of such plans.

(n) Parent has provided to the Company a true and correct list, as of the date of this Agreement, containing the names of all current full-time, part-time or temporary employees and independent contractors (and indication as such), and, as applicable: (i) the annual dollar amount of all cash compensation in the form of wages, salary, fees, commissions, or director's fees payable to each person; (ii) dates of employment or service; (iii) title and, with respect to independent contractors, a current written description of such person's contracting services; (iv) visa status, if applicable; and (v) with respect to employees, (A) a designation of whether they are classified as exempt or non-exempt for purposes of FLSA and any similar state, federal or Foreign law and (B) whether such an employee is on leave, and if so, the expected return date.

(o) Neither Parent nor any of its Subsidiaries is or has ever been a party to, bound by, or has a duty to bargain under, any collective bargaining agreement or other Contract with a labor union or similar labor organization representing any of its employees, and there is no labor union or similar labor organization representing or, to the Knowledge of Parent, purporting to represent or seeking to represent any employees of Parent or its Subsidiaries, including through the filing of a petition for representation election. There is not and has not been in the past five years, nor, to the Knowledge of Parent, is there or has there been in the past five years any threat of, any strike, slowdown, work stoppage, lockout, union election petition, demand for recognition, or any similar activity or dispute, or any union organizing activity, against Parent or any of its Subsidiaries.

(p) Parent and each of its Subsidiaries is, and since December 31, 2021 has been, in material compliance with all applicable Laws respecting labor, employment, employment practices, and terms and conditions of employment, including worker classification, discrimination, harassment and retaliation, equal employment opportunities, fair employment practices, meal and rest periods, immigration, employee safety and health, payment of wages (including overtime wages), unemployment and workers' compensation, leaves of absence, and hours of work. Except as would not be reasonably likely to result in a material liability to Parent or any of its Subsidiaries, with respect to employees of Parent and its Subsidiaries, each of Parent and its Subsidiaries, since December 31, 2021, has withheld and reported all amounts required by Law to be withheld and reported with respect to wages, salaries and other payments, benefits, or compensation to employees. There is no material Legal Proceeding pending or, to the Knowledge of Parent, threatened or reasonably anticipated against Parent or any of its Subsidiaries relating to any current or former employee, applicant for employment, or consultant of Parent.

(q) Within the preceding two years, Parent has complied in all material respects with the WARN Act.

Environmental Matters. Parent and each of its Subsidiaries are in compliance and since December 31, 2021 have complied with 3.18 all applicable Environmental Laws, which compliance includes the possession by Parent of all permits and other Governmental Authorizations required under applicable Environmental Laws and compliance with the terms and conditions thereof, except for any failure to be in such compliance that, either individually or in the aggregate, would not reasonably be expected to be material to Parent or its business. Neither Parent nor any of its Subsidiaries has received since December 31, 2021 (or prior to that time, which is pending and unresolved), any written notice or other communication (in writing or otherwise), whether from a Governmental Body or other Person, that alleges that Parent or any of its Subsidiaries is not in compliance with or has liability pursuant to any Environmental Law and, to the Knowledge of Parent, there are no circumstances that would reasonably be expected to prevent or interfere with Parent's or any of its Subsidiaries' compliance in any material respects with any Environmental Law, except where such failure to comply would not reasonably be expected to be material to Parent or its business. No current or (during the time a prior property was leased or controlled by Parent or any of its Subsidiaries) prior property leased or controlled by Parent or any of its Subsidiaries has had a release of or exposure to Hazardous Materials in material violation of or as would reasonably be expected to result in any material liability of Parent or any of its Subsidiaries pursuant to Environmental Law. No consent, approval or Governmental Authorization of or registration or filing with any Governmental Body is required by Environmental Laws in connection with the execution and delivery of this Agreement or the consummation of the Contemplated Transactions by Parent or Merger Subs. Prior to the date hereof, Parent has provided or otherwise made available to the Company true and correct copies of all material environmental reports, assessments, studies and audits in the possession or control of Parent or any of its Subsidiaries with respect to any property leased or controlled by Parent or any of its Subsidiaries or any business operated by them.

3.19 **Transactions with Affiliates**. Except as set forth in the Parent SEC Documents filed prior to the date of this Agreement, since the date of Parent's last proxy statement filed in July 2023 with the SEC, no event has occurred that would be required to be reported by Parent pursuant to Item 404 of Regulation S-K. Section 3.19 of the Parent Disclosure Schedule identifies each Person who is (or who may be deemed to be) an Affiliate of Parent as of the date of this Agreement.

3.20 **Insurance**. Parent has delivered or made available to the Company accurate and complete copies of all insurance policies and all self-insurance programs and arrangements relating to the business, assets, liabilities and operations of Parent and each of its Subsidiaries. Each of such insurance policies is in full force and effect and Parent and each of its Subsidiaries is in compliance in all material respects with the terms thereof. Other than customary end of policy notifications from insurance carriers, since January 1, 2020, neither Parent nor any of its Subsidiaries has received any notice or other communication regarding any actual or possible: (a) cancellation or invalidation of any insurance policy; or (b) refusal or denial of any coverage, reservation of rights or rejection of any material claim under any insurance policy. Parent and each of its Subsidiaries for which Parent or such Subsidiary has insurance coverage, and no such carrier has issued a denial of coverage or a reservation of rights with respect to any such Legal Proceeding, or informed Parent or any of its Subsidiaries of its intent to do so.

3.21 <u>Opinion of Financial Advisor</u>. The Parent Board has received an opinion of H.C. Wainwright & Co., LLC, dated on or about the date of the Agreement, to the effect that, as of the date of such opinion and subject to the assumptions, qualifications, limitations and other matters set forth therein, the Exchange Ratio (as specified in such opinion) is fair, from a financial point of view, to Parent. It is agreed and understood that such opinion is furnished solely for the use of the Parent Board and may not be relied upon by the Company or any other party.

3.22 **No Financial Advisors**. No broker, finder or investment banker, other than Tungsten Advisors and H.C. Wainwright & Co., LLC, is entitled to any brokerage fee, finder's fee, opinion fee, success fee, transaction fee or other fee or commission in connection with the Contemplated Transactions based upon arrangements made by or on behalf of Parent or any of its Subsidiaries.

3.23 <u>Anti-Bribery</u>. None of Parent or any of its Subsidiaries nor any of their respective directors, officers, employees or, to Parent's Knowledge, agents or any other Person acting on its behalf has directly or indirectly made any bribes, rebates, payoffs, influence payments, kickbacks, illegal payments, illegal political contributions, or other payments, in the form of cash, gifts, or otherwise, or taken any other action, in violation of Anti-Bribery Laws. Neither Parent nor any of its Subsidiaries is or has been the subject of any investigation or inquiry by any Governmental Body with respect to potential violations of Anti-Bribery Laws.

3.24 <u>Valid Issuance</u>. The Parent Common Stock and Parent Convertible Preferred Stock to be issued in the Merger will, when issued in accordance with the provisions of this Agreement, be validly issued, fully paid and nonassessable. To the Knowledge of Parent as of the date of this Agreement, no "bad actor" disqualifying event described in Rule 506(d)(1)(i)-(viii) of the Securities Act (a "*Disqualifying Event*") is applicable to Parent or, to Parent's Knowledge, any Parent Covered Person, except for a Disqualifying Event as to which Rule 506(d)(2)(ii)-(iv) or (d)(3) of the Securities Act is applicable.

3.25 Disclaimer of Other Representations or Warranties.

(a) Except as previously set forth in this <u>Section 3</u> or in any certificate delivered by Parent or Merger Subs to the Company pursuant to this Agreement, neither Parent nor any Merger Sub makes any representation or warranty, express or implied, at law or in equity, with respect to it or any of its assets, liabilities or operations, and any such other representations or warranties are hereby expressly disclaimed.

(b) Each of Parent, First Merger Sub and Second Merger Sub acknowledges and agrees that, except for the representations and warranties of the Company set forth in <u>Section 2</u> or in any certificate delivered by the Company to Parent or the Merger Subs pursuant to this Agreement, none of Parent, First Merger Sub, Second Merger Sub or any of their respective Representatives is relying on any other representation or warranty of the Company or any other Person made outside of <u>Section 2</u> or such certificates, including regarding the accuracy or completeness of any such other representations or warranties or the omission of any material information, whether express or implied, in each case, with respect to the Contemplated Transactions.

SECTION 4. ADDITIONAL AGREEMENTS OF THE PARTIES

4.1 Parent Stockholders' Meeting.

(a) As promptly as practicable following the execution of this Agreement, Parent shall take all action necessary under applicable Law to call, give notice of and hold a meeting of the holders of Parent Common Stock for the purpose of seeking:

(i) approval of the Preferred Stock Conversion Proposal;

(ii) if deemed necessary or appropriate by Parent or as otherwise required by applicable Law or Contract, to authorize the amendment of Parent's certificate of incorporation to authorize sufficient Parent Common Stock in Parent's certificate of incorporation for the conversion of the Parent Convertible Preferred Stock issued pursuant to this Agreement and/or to effectuate the Nasdaq Reverse Split (the "*Charter Amendment Proposal*") (the matters contemplated by the <u>clauses 4.1(a)(i)</u> and <u>4.1(a)(ii)</u> are referred to as the "*Parent Stockholder Matters*," and such meeting, the "*Parent Stockholders' Meeting*").

Parent agrees to use reasonable best efforts to (i) call and hold the Parent Stockholders' Meeting as soon as practicable (b) after the date hereof, and (ii) to solicit and obtain the Required Parent Stockholder Vote, including without limitation, (A) engaging a nationally recognized proxy solicitation firm and information agent, (B) actively attempting to contact and obtain votes from the Parent's stockholders (including its retail stockholders with meaningful holdings of Parent Common Stock), and (C) working with the Parent's transfer agent and inspector of elections to facilitate an appropriate and straightforward process for obtaining the Required Parent Stockholder Vote. If the approval of the Parent Stockholder Matters is not obtained at the Parent Stockholders' Meeting or if on a date preceding the Parent Stockholders' Meeting, Parent reasonably believes that (x) it will not receive proxies sufficient to obtain the Required Parent Stockholder Vote, whether or not quorum would be present or (y) it will not have sufficient shares of Parent Common Stock represented (whether in person or by proxy) to constitute a quorum necessary to conduct the business of the Parent Stockholders' Meeting, then, in each case, Parent will use its reasonable best efforts to adjourn the Parent Stockholders' Meeting one or more times to a date or dates no more than 30 days after the scheduled date for such meeting, and to obtain such approvals at such time. If the Parent Stockholders' Meeting is not so adjourned, and/or if the approval of the Parent Stockholder Matters is not then obtained, Parent will use its reasonable best efforts to obtain such approvals as soon as practicable thereafter, and in any event to obtain such approvals at the next occurring annual meeting of the stockholders of Parent or, if such annual meeting is not scheduled to be held within six months after the Parent Stockholders' Meeting, a special meeting of the stockholders of Parent to be held within six months after the Parent Stockholders' Meeting. Parent will hold an annual meeting or special meeting of its stockholders, at which a vote of the stockholders of Parent to approve the Parent Stockholder Matters will be solicited and taken, at least once every six months until Parent obtains approval of the Parent Stockholder Matters.

(c) Parent agrees that: (i) the Parent Board shall recommend that the holders of Parent Common Stock vote to approve the Parent Stockholder Matters and shall use its reasonable best efforts to solicit and obtain such approval within the time frames set forth in <u>Section 4.2(b)</u>, and (ii) the Proxy Statement shall include a statement to the effect that the Parent Board recommends that the Parent's stockholders vote to approve the Parent Stockholder Matters.

(d) The Company and Parent acknowledge that, under the Nasdaq Stock Market Rules, the Parent Common Stock Payment Shares and the Parent Preferred Stock Payment Shares will not be entitled to vote on the Preferred Stock Conversion Proposal.

4.2 Proxy Statement.

(a) As promptly as practicable after the Closing Date and subject to <u>Section 4.2(d)</u>, Parent shall prepare and file with the SEC a proxy statement relating to the Parent Stockholders' Meeting to be held in connection with the Parent Stockholder Matters (together with any amendments thereof or supplements thereto, the "*Proxy Statement*"). Parent shall use its reasonable best efforts to (i) cause the Proxy Statement to comply with applicable rules and regulations promulgated by the SEC and (ii) respond promptly to any comments or requests of the SEC or its staff related to the Proxy Statement. Parent shall not file the Proxy Statement, or any amendment or supplement thereto, or respond to SEC comments or requests, without providing the Company a reasonable opportunity to review and comment thereon (which comments shall be reasonably considered by Parent).

(b) Parent covenants and agrees that the Proxy Statement (and the letters to stockholders, notice of meeting and form of proxy included therewith) will (i) comply as to form in all material respects with the requirements of applicable U.S. federal securities Laws and the DGCL, and (ii) will not contain any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary in order to make the statements made therein, in light of the circumstances under which they were made, not misleading.

(c) Parent shall use reasonable best efforts to cause the Proxy Statement to be mailed to Parent's stockholders as promptly as practicable after the Proxy Statement has been filed with the SEC and either (i) the SEC has indicated that it does not intend to review the Proxy Statement or that its review of the Proxy Statement has been completed or (ii) at least ten (10) days shall have passed since the Proxy Statement was filed with the SEC without receiving any correspondence from the SEC commenting upon, or indicating that it intends to review, the Proxy Statement, all in compliance with applicable U.S. federal securities laws and the DGCL. If Parent, First Merger Sub, Second Merger Sub or the Surviving Entity (A) become aware of any event or information that, pursuant to the Securities Act or the Exchange Act, should be disclosed in an amendment or supplement to the Proxy Statement, (B) receives notice of any SEC request for an amendment or supplement to the Proxy Statement, or (C) receives SEC comments on the Proxy Statement, as the case may be, then such Party, as the case may be, shall promptly inform the other Parties thereof and shall cooperate and consult with such other Parties in Parent filing such amendment or supplement with the SEC and, if appropriate, in mailing such amendment or supplement to the Parent stockholders.

(d) The Parties shall reasonably cooperate and consult with each other and provide, and shall use reasonable best efforts to cause their respective Representatives to provide, the other Party and its Representatives, with all true, correct and complete information regarding such Party that is required by Law to be included in the Proxy Statement or reasonably requested by the other Party to be included in the Proxy Statement. If at any time the information provided in Proxy Statement has or will become "stale" and new information should, as determined by Parent acting reasonably, be disclosed in an amendment or supplement to the Proxy Statement, then Parent shall promptly inform the Company thereof and each such Party shall cooperate and consult with one another, and shall use reasonable best efforts to cause their accounting and other outside professionals to so cooperate and consult, (i) in providing the financial reporting necessary for such filing and (ii) in filing such amendment or supplement with the SEC (and, if related to the Proxy Statement, mailing such amendment or supplement to the Parent stockholders).

4.3 <u>Reservation of Parent Common Stock; Issuance of Shares of Parent Common Stock</u>. For as long as any Parent Preferred Stock Payment Shares remain outstanding, Parent shall at all times reserve and keep available, free from preemptive rights, out of its authorized but unissued Parent Common Stock or shares of Parent Common Stock held in treasury by Parent, for the purpose of effecting the conversion of the Parent Preferred Stock Payment Shares, the full number of shares of Parent Common Stock then issuable upon the conversion of all Parent Preferred Stock Payment Shares then outstanding. All shares of Parent Common Stock delivered upon conversion of the Parent Preferred Stock Payment Shares or shares held in treasury by Parent, shall have been duly authorized and validly issued and shall be fully paid and nonassessable, and shall be free from preemptive rights and free of any Encumbrance.

4.4 Employee Benefits.

For purposes of vesting, eligibility to participate, and level of benefits (other than for purposes of determining awards (a) under an equity incentive plan or accrued benefits under any defined benefit pension plan) under the benefit plans, programs, Contracts or arrangements of Parent or any of its Subsidiaries (including, following the Closing, the Surviving Entity and its Subsidiaries) (the "Post-Closing Plans"), Parent shall use commercially reasonable efforts to cause each employee of the Company who remains employed by Parent or the Surviving Entity, or any of their respective Subsidiaries following the Closing, (together, the "Continuing Employees") to be credited with his or her years of service with the Company or any of its predecessors; provided that the foregoing shall not apply to the extent that its application would result in a duplication of benefits. In addition, and without limiting the generality of the foregoing, for purposes of each Post-Closing Plan providing medical, dental, pharmaceutical and/or vision benefits to a Continuing Employee, Parent shall use commercially reasonable efforts to cause all pre-existing condition exclusions and actively-at-work requirements of such Post-Closing Plan to be waived for such Continuing Employee and his or her covered dependents to the extent and unless such conditions would have been waived or satisfied under the employee benefit plan whose coverage is being replaced under the Post-Closing Plan, and Parent shall use its commercially reasonable efforts to cause any eligible expenses incurred by a Continuing Employee and his or her covered dependents during the portion of such plan year in which coverage is replaced with coverage under a Post-Closing Plan to be taken into account under such Post-Closing Plan with respect to the plan year in which participation in such Post-Closing Plan begins for purposes of satisfying all deductible, coinsurance and maximum out-of-pocket requirements applicable to such Continuing Employee and his or her covered dependents for such plan year as if such amounts had been paid in accordance with such Post-Closing Plan.

(b) The provisions of this <u>Section 4.4</u> are for the sole benefit of Parent and the Company and no provision of this Agreement shall (i) create any third-party beneficiary or other rights in any Person, including rights in respect of any benefits that may be provided, directly or indirectly, under any Company Benefit Plan, Parent Benefit Plan or Post-Closing Plan or rights to continued employment or service with the Company or the Parent (or any Subsidiary thereof), (ii) be construed as an amendment, waiver or creation of or limitation on the ability to terminate any Company Benefit Plan, Parent Benefit Plan, or (iii) limit the ability of the Parent to terminate the employment of any Continuing Employee or modify the at-will status of any Continuing Employees.

4.5 Indemnification of Officers and Directors.

(a) From the First Effective Time through the sixth anniversary of the date on which the First Effective Time occurs, each of Parent and the Surviving Entity shall indemnify and hold harmless each person who is now, or has been at any time prior to the date hereof, or who becomes prior to the First Effective Time, a director or officer of Parent or the Company or any of their respective Subsidiaries, respectively (the "**D&O Indemnified Parties**"), against all claims, losses, liabilities, damages, judgments, fines and reasonable fees, costs and expenses, including attorneys' fees and disbursements, incurred in connection with any claim, action, suit, proceeding or investigation, whether civil, criminal, administrative or investigative, arising out of or pertaining to the fact that the D&O Indemnified Party is or was a director or officer of Parent or of the Company, or any Subsidiary thereof, asserted or claimed prior to the First Effective Time, in each case, to the fullest extent permitted under applicable Law. Except in the case of fraud, each D&O Indemnified Party will be entitled to advancement of expenses incurred in the defense of any such claim, action, suit, proceeding or investigation from each of Parent and the Surviving Entity, jointly and severally, upon receipt by Parent or the Surviving Entity from the D&O Indemnified Party of a request therefor; *provided* that any such person to whom expenses are advanced provides an undertaking to Parent, to the extent then required by the DGCL or DLLCA, as applicable, to repay such advances if it is ultimately determined that such person is not entitled to indemnification.

(b) The provisions of the certificate of incorporation and bylaws of Parent with respect to indemnification, advancement of expenses and exculpation of present and former directors and officers of Parent that are presently set forth in the certificate of incorporation and bylaws of Parent shall not be amended, modified or repealed for a period of six years from the First Effective Time in a manner that would adversely affect the rights thereunder of individuals who, at or prior to the First Effective Time, were officers or directors of Parent, unless such modification is required by applicable Law. The certificate of formation and limited liability company agreement of the Surviving Entity shall contain, and Parent shall cause the certificate of formation and limited liability company agreement of the Surviving Entity to so contain, provisions no less favorable with respect to indemnification, advancement of expenses and exculpation of present and former directors and officers as those presently set forth in the certificate of incorporation and bylaws of Parent.

(c) From and after the First Effective Time, (i) the Surviving Entity shall fulfill and honor in all respects the obligations of the Company to its D&O Indemnified Parties as of immediately prior to the Closing pursuant to any indemnification provisions under the Company's Organizational Documents and pursuant to any indemnification agreements between the Company and such D&O Indemnified Parties set forth on <u>Schedule 4.5(c)(i)</u> hereto, with respect to claims arising out of matters occurring at or prior to the First Effective Time and (ii) Parent shall fulfill and honor in all respects the obligations of Parent to its D&O Indemnified Parties as of immediately prior to the Closing pursuant to any indemnification provisions under Parent's Organizational Documents and pursuant to any indemnification agreements between Parent and such D&O Indemnified Parties set forth on <u>Schedule 4.5(c)(ii)</u> hereto, with respect to claims arising out of matters occurring at or prior to the First Effective Time.

(d) From and after the First Effective Time, Parent shall continue to maintain directors' and officers' liability insurance policies, with an effective date as of the Closing Date, on commercially available terms and conditions and with coverage limits customary for U.S. public companies similarly situated to Parent. From and after the First Effective Time, Parent shall pay all expenses, including reasonable attorneys' fees, that are incurred by the persons referred to in this <u>Section 4.5</u> in connection with their successful enforcement of the rights provided to such persons in this <u>Section 4.5</u>.

(e) The provisions of this <u>Section 4.5</u> are intended to be in addition to the rights otherwise available to the current and former officers and directors of Parent and the Company by Law, charter, statute, bylaw or agreement, and shall operate for the benefit of, and shall be enforceable by, each of the D&O Indemnified Parties, their heirs and their representatives.

(f) In the event Parent or the Surviving Entity or any of their respective successors or assigns (i) consolidates with or merges into any other Person and shall not be the continuing or surviving corporation or entity of such consolidation or merger, or (ii) transfers all or substantially all of its properties and assets to any Person, then, and in each such case, proper provision shall be made so that the successors and assigns of Parent or the Surviving Entity, as the case may be, shall succeed to the obligations set forth in this <u>Section 4.5</u>. Parent shall cause the Surviving Entity to perform all of the obligations of the Surviving Entity under this <u>Section 4.5</u>.

4.6 <u>Additional Agreements</u>. The Parties shall use reasonable best efforts to cause to be taken all actions necessary to consummate the Contemplated Transactions. Without limiting the generality of the foregoing, each Party to this Agreement: (a) shall make all filings and other submissions (if any) and give all notices (if any) required to be made and given by such Party in connection with the Contemplated Transactions; (b) shall use reasonable best efforts to obtain each Consent (if any) reasonably required to be obtained (pursuant to any applicable Law or Contract, or otherwise) by such Party in connection with the Contemplated Transactions or for such Contract to remain in full force and effect; (c) shall use reasonable best efforts to satisfy the conditions precedent to the consummation of this Agreement.

4.7 Listing. Parent shall use its reasonable best efforts to (a) maintain its existing listing on Nasdaq; (b) prepare and submit to Nasdaq a notification form for the listing of the shares of Parent Common Stock Payment Shares and the Parent Common Stock to be issued upon conversion of the Parent Preferred Stock Payment Shares to be issued in connection with the Contemplated Transactions, to cause such shares to be approved for listing (subject to official notice of issuance); (c) effect the Nasdaq Reverse Split and (d) to the extent required by Nasdaq rules and regulations, file an initial listing application for the Parent Common Stock on Nasdaq (the "*Nasdaq Listing Application*"), which Nasdaq Listing Application shall be prepared in cooperation with the Company, and to cause such Nasdaq Listing Application to be conditionally approved prior to the First Effective Time. The Parties will use reasonable best efforts to coordinate with respect to compliance with Nasdaq rules and regulations. Each Party will promptly inform the other Party of all verbal or written communications between Nasdaq and such Party or its representatives. The Company will cooperate with Parent as reasonably requested by Parent with respect to the Nasdaq Listing Application and promptly furnish to Parent all information concerning the Company and its stockholders that may be required or reasonably requested in connection with any action contemplated by this <u>Section 4.8</u>.

4.8 **Tax Matters**. For U.S. federal income Tax purposes, (i) the Parties intend that the First Merger and the Second Merger, taken together, constitute an integrated transaction described in Rev. Rul. 2001-46, 2001-2 C.B. 321 that qualifies as a "reorganization" within the meaning of Section 368(a) of the Code and the Treasury Regulations promulgated thereunder (the "*Intended Tax Treatment*"), and (ii) this Agreement is intended to be, and is hereby adopted as, a "plan of reorganization" for purposes of Sections 354 and 361 of the Code and Treasury Regulations Sections 1.368-2(g) and 1.368-3(a), to which the Parent, Merger Subs and the Company are parties under Section 368(b) of the Code. The Parties shall treat and shall not take any tax reporting position (including during the course of any audit, litigation or other proceeding with respect to Taxes) inconsistent with the treatment of the Merger as a reorganization" within the meaning of Section 368(a) of the Code. The Parties shall cause their Affiliates to) not take any action or cause any action to be taken, or fail to take or cause to be taken any action, which action or failure to act would reasonably be expected to prevent the Merger from qualifying, for the Intended Tax Treatment.

4.9 **Legends**. Parent shall be entitled to place appropriate legends, including the legend noted in <u>Section 4.15</u>, on the book entries and/or certificates evidencing any shares of Parent Common Stock or Parent Convertible Preferred Stock to be received in the Merger by equity holders of the Company who may be considered "affiliates" of Parent for purposes of Rules 144 and 145 under the Securities Act reflecting the restrictions set forth in Rules 144 and 145 and to issue appropriate stop transfer instructions to the transfer agent for Parent Common Stock and Parent Convertible Preferred Stock.

4.10 **Directors and Officers**. The Parties shall take all necessary action so that immediately after the Second Effective Time, (a) the Parent Board is comprised of seven members, with four such members designated by Parent, and three such members designated by the Company, as set forth on <u>Schedule 4.10(a)</u> hereto, which includes a majority of independent directors (in accordance with Nasdaq requirements) and (b) the Persons set forth on <u>Schedule 4.10(b)</u> hereto under the heading "Officers" are elected or appointed, as applicable, to the positions of officers of Parent and the Surviving Entity, as set forth therein, to serve in such positions effective as of the Second Effective Time until successors are duly appointed and qualified in accordance with applicable Law.

4.11 <u>Section 16 Matters</u>. Prior to the First Effective Time, Parent and the Company shall take all such steps as may be required (to the extent permitted under applicable Laws) to cause any acquisitions of Parent Common Stock, restricted stock awards to acquire Parent Common Stock and any Parent Options to purchase Parent Common Stock in connection with the Contemplated Transactions, by each individual who is reasonably expected to become subject to the reporting requirements of Section 16(a) of the Exchange Act with respect to Parent, to be exempt under Rule 16b-3 promulgated under the Exchange Act.

4.12 Closing Certificates.

(a) The Company will prepare and deliver to Parent prior to the Closing a certificate signed by the President of the Company in a form reasonably acceptable to Parent setting forth, as of immediately prior to the First Effective Time (i) each holder of Company Common Stock, Company Preferred Stock and Company Options, (ii) such holder's name and address, (iii) the number and type of Company Common Stock held and/or underlying the Company Options as of immediately prior to the First Effective Time for each such holder, and the number and type of Company Preferred Stock held, and (iv) the number of shares of Parent Common Stock and/or Parent Convertible Preferred Stock to be issued to such holder, or to underlie any Parent Option to be issued to such holder, pursuant to this Agreement in respect of the Company Capital Stock, Company Preferred Stock or Company Options held by such holder as of immediately prior to the First Effective Time, which shall be calculated using the Exchange Ratio (the "*Allocation Certificate*").

(b) Parent will prepare and deliver to the Company prior to the Closing a certificate signed by the Chief Financial Officer of Parent in a form reasonably acceptable to the Company, setting forth, as of immediately prior to the Reference Date (i) the number of Parent Common Stock outstanding and (ii) (A) each record holder of Parent Options or Parent RSUs, (B) such record holder's name, (C) the number of shares of Parent Common Stock underlying the Parent Options or Parent RSUs as of the First Effective Time for such holder (the "*Parent Outstanding Shares Certificate*").

4.13 **Takeover Statutes**. If any Takeover Statute is or may become applicable to the Contemplated Transactions, each of the Company, the Company Board, Parent and the Parent Board, as applicable, shall grant such approvals and take such actions as are necessary so that the Contemplated Transactions may be consummated as promptly as practicable on the terms contemplated by this Agreement and otherwise act to eliminate or minimize the effects of such statute or regulation on the Contemplated Transactions.

4.14 **Obligations of Merger Subs**. Parent will take all action necessary to cause Merger Subs to perform their obligations under this Agreement and to consummate the Merger on the terms and conditions set forth in this Agreement.

4.15 **Private Placement**. Each of the Company and Parent shall take all reasonably necessary action on its part such that the issuance of Parent Common Stock Payment Shares and Parent Preferred Stock Payment Shares pursuant to this Agreement constitutes a transaction exempt from registration under the Securities Act in compliance with Rule 506 of Regulation D promulgated thereunder. Each certificate representing Parent Common Stock Payment Shares and the Parent Preferred Stock Payment Shares comprising Merger Consideration shall, until such time that such shares are not so restricted under the Securities Act, bear a legend identical or similar in effect to the following legend (together with any other legend or legends required by applicable state securities applicable Law or otherwise, if any):

"THE SHARES REPRESENTED HEREBY HAVE NOT BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933 (THE "*ACT*") AND MAY NOT BE OFFERED, SOLD OR OTHERWISE TRANSFERRED, ASSIGNED, PLEDGED OR HYPOTHECATED UNLESS REGISTERED UNDER THE ACT OR UNLESS AN EXEMPTION FROM THE REGISTRATION REQUIREMENTS OF THE ACT IS AVAILABLE."

4.16 <u>Audited Financial Statements</u>. As promptly as practicable following the Closing, and no later than June 11, 2024, the Company shall deliver to Parent PCAOB audited financial statements of the Company prepared in accordance with GAAP.

SECTION 5. CONDITIONS PRECEDENT TO OBLIGATIONS OF EACH PARTY

The obligations of each Party to effect the Merger and otherwise consummate the Contemplated Transactions to be consummated at the Closing are subject to the satisfaction or, to the extent permitted by applicable Law, the written waiver by each of the Parties, at or prior to the Closing Date, of each of the following conditions:

5.1 <u>No Restraints</u>. No temporary restraining order, preliminary or permanent injunction or other order preventing the consummation of the Contemplated Transactions shall have been issued by any court of competent jurisdiction or other Governmental Body of competent jurisdiction and remain in effect and there shall not be any Law which has the effect of making the consummation of the Contemplated Transactions illegal.

5.2 <u>Certificate of Designation</u>. Parent shall have filed the Certificate of Designation with the Secretary of State of the State of Delaware.

5.3 <u>Parent Financing</u>. The Securities Purchase Agreement shall be in full force and effect and cash proceeds not less than the Concurrent Investment Amount shall have been received by Parent, or will be received by Parent substantially simultaneously with the Closing, in connection with the consummation of the transactions contemplated by the Securities Purchase Agreement, and Parent shall have received confirmation of the SAFE Investment.

SECTION 6. CLOSING DELIVERIES OF THE COMPANY

The obligations of Parent and Merger Subs to effect the Merger and otherwise consummate the transactions to be consummated at the Closing are subject to the satisfaction or the written waiver by Parent, at or prior to the Closing, of each of the following conditions:

6.1 **Documents**. Parent shall have received the following documents, each of which shall be in full force and effect:

(a) a written resignation, in a form reasonably satisfactory to Parent, dated as of the Closing Date and effective as of the Closing, executed by each of the directors of the Company listed on <u>Schedule 6.1(a)</u> hereto; and

(b) the Allocation Certificate.

6.2 **FIRPTA Certificate**. Parent shall have received (i) an original signed statement from the Company that the Company is not, and has not been at any time during the applicable period specified in Section 897(c)(1)(A)(ii) of the Code, a "United States real property holding corporation," as defined in Section 897(c)(2) of the Code, conforming to the requirements of Treasury Regulations Section 1.1445-2(c)(3) and 1.897-2(h), and (ii) an original signed notice to be delivered to the IRS in accordance with the provisions of Treasury Regulations Section 1.897-2(h)(2), together with written authorization for Parent to deliver such notice to the IRS on behalf of the Company following the Closing, each dated as of the Closing Date, duly executed by an authorized officer of the Company, and in form and substance reasonably acceptable to Parent; *provided*, that the Parent's sole remedy for the Company's failure to deliver such documentation shall be to withhold pursuant to Section 1.13.

6.3 <u>Company Lock-Up Agreements</u>. Parent shall have received the Lock-Up Agreements duly executed by each of the Company Signatories, each of which shall be in full force and effect.

6.4 <u>Company Stockholder Support Agreements</u>. Parent shall have received the Company Stockholder Support Agreements duly executed by each of the stockholders of the Company, each of which shall be in full force and effect.

6.5 **<u>Registration Rights Agreements</u>**. Parent shall have received the Registration Rights Agreement duly executed by each of the Purchasers under the Securities Purchase Agreement.

SECTION 7. CLOSING DELIVERIES OF PARENT

The obligations of the Company to effect the Merger and otherwise consummate the transactions to be consummated at the Closing are subject to the satisfaction or the written waiver by the Company, at or prior to the Closing, of each of the following conditions:

7.1 **Documents.** The Company shall have received the following documents, each of which shall be in full force and effect:

(a) the Parent Outstanding Shares Certificate;

(b) a written resignation, in a form reasonably satisfactory to the Company, dated as of the Closing Date and effective as of the Closing, executed by each of the officers and directors of Parent who are not to continue as officers or directors, as the case may be, of Parent after the Closing pursuant to Section 4.10 hereof; and

(c) certified copies of the resolutions duly adopted by the Parent Board and in full force and effect as of the Closing authorizing the appointment of the directors and officers set forth in Section 4.10.

7.2 Parent Lock-Up Agreements

. The Company shall have received the Lock-Up Agreements duly executed by each of the Parent Signatories, each of which shall be in full force and effect.

7.3 **<u>Registration Rights Agreements</u>**. The Company shall have received the Registration Rights Agreement duly executed by Parent.

SECTION 8. MISCELLANEOUS PROVISIONS

8.1 **Non-Survival of Representations and Warranties**. The representations and warranties of the Company, Parent and Merger Subs contained in this Agreement or any certificate or instrument delivered pursuant to this Agreement shall terminate at the First Effective Time, and only the covenants that by their terms survive the First Effective Time and this <u>Section 8</u> shall survive the First Effective Time.

8.2 <u>Amendment</u>. This Agreement may be amended with the approval of the respective boards of directors (or managers as applicable) of the Surviving Entity and Parent at any time; *provided, however*, that after any such approval of this Agreement by a Party's stockholders, no amendment shall be made which by Law requires further approval of such stockholders without the further approval of such stockholders. This Agreement may not be amended except by an instrument in writing signed on behalf of each of the Surviving Entity and Parent.

8.3 <u>Waiver</u>.

(a) No failure on the part of any Party to exercise any power, right, privilege or remedy under this Agreement, and no delay on the part of any Party in exercising any power, right, privilege or remedy under this Agreement, shall operate as a waiver of such power, right, privilege or remedy; and no single or partial exercise of any such power, right, privilege or remedy shall preclude any other or further exercise thereof or of any other power, right, privilege or remedy.

(b) No Party shall be deemed to have waived any claim arising out of this Agreement, or any power, right, privilege or remedy under this Agreement, unless the waiver of such claim, power, right, privilege or remedy is expressly set forth in a written instrument duly executed and delivered on behalf of such Party and any such waiver shall not be applicable or have any effect except in the specific instance in which it is given.

8.4 Entire Agreement; Counterparts; Exchanges by Electronic Transmission. This Agreement and the other schedules, exhibits, certificates, instruments and agreements referred to in this Agreement constitute the entire agreement and supersede all prior agreements and understandings, both written and oral, among or between any of the Parties with respect to the subject matter hereof and thereof; *provided, however*, that the Confidentiality Agreement shall not be superseded and shall remain in full force and effect in accordance with its terms. This Agreement. Counterparts may be delivered via electronic mail (including pdf or any electronic signature complying with the U.S. federal ESIGN Act of 2000, *e.g.*, www.docusign.com) or other transmission method and any counterpart so delivered shall be deemed to have been duly and validly delivered and be valid and effective for all purposes.

8.5 <u>Applicable Law; Jurisdiction</u>. This Agreement shall be governed by, and construed in accordance with, the Laws of the State of Delaware, regardless of the Laws that might otherwise govern under applicable principles of conflicts of laws. In any action or proceeding between any of the Parties arising out of or relating to this Agreement or any of the Contemplated Transactions, each of the Parties: (a) irrevocably and unconditionally consents and submits to the exclusive jurisdiction and venue of the Court of Chancery of the State of Delaware or, to the extent such court does not have subject matter jurisdiction, the United States District Court for the District of Delaware or, to the extent that neither of the foregoing courts has jurisdiction, the Superior Court of the State of Delaware; (b) agrees that all claims in respect of such action or proceeding shall be heard and determined exclusively in accordance with clause (a) of this Section 8.5; (c) waives any objection to laying venue in any such action or proceeding in such courts; (d) waives any objection that such courts are an inconvenient forum or do not have jurisdiction over any Party; (e) agrees that service of process upon such Party in any such action or proceeding shall be effective if notice is given in accordance with Section 8.8 of this Agreement; and (f) irrevocably and unconditionally waives the right to trial by jury.

8.6 <u>Attorneys' Fees</u>. In any action at law or suit in equity to enforce this Agreement or the rights of any of the Parties, the prevailing Party in such action or suit (as determined by a court of competent jurisdiction) shall be entitled to recover its reasonable out-of-pocket attorneys' fees and all other reasonable costs and expenses incurred in such action or suit.

8.7 <u>Assignability</u>. This Agreement shall be binding upon, and shall be enforceable by and inure solely to the benefit of, the Parties and their respective successors and permitted assigns; *provided, however*, that neither this Agreement nor any of a Party's rights or obligations hereunder may be assigned or delegated by such Party without the prior written consent of the other Party, and any attempted assignment or delegation of this Agreement or any of such rights or obligations by such Party without the other Party's prior written consent shall be void and of no effect.

8.8 **Notices**. All notices and other communications hereunder shall be in writing and shall be deemed to have been duly delivered and received hereunder (a) one Business Day after being sent for next Business Day delivery, fees prepaid, via a reputable international overnight courier service, (b) upon delivery in the case of delivery by hand, or (c) on the date delivered in the place of delivery if sent by email (with a written or electronic confirmation of delivery) prior to 5:00 p.m. Eastern Time, otherwise on the next succeeding Business Day, in each case to the intended recipient as set forth below:

if to Parent or Merger Subs:

Onconova Therapeutics, Inc. 12 Penns Trail Newtown, PA 18940 Attention: Steven M. Fruchtman; Mark Guerin Email Address: sfruchtman@onconova.us; mguerin@onconova.us

with a copy to (which shall not constitute notice):

Orrick, Herrington & Sutcliffe LLP Columbia Center 1152 15th Street, N.W. Washington, D.C. 20005-1706 United States Attention: David Schulman Email: dschulman@orrick.com

if to the Company:

Trawsfynydd Therapeutics, Inc. 8 The Green Suite 8664 Dover, DE 19901 Attn: Nikolay Savchuk Email: nsavchuk@torreypinesinv.com

with a copy to (which shall not constitute notice):

Snell & Wilmer L.L.P. Executive Center 12230 El Camino Real, Suite 300 San Diego, CA 92130 United States Attention: Bardia Moayedi Email: bmoayedi@swlaw.com

8.9 <u>Cooperation</u>. Each Party agrees to cooperate fully with the other Party and to execute and deliver such further documents, certificates, agreements and instruments and to take such other actions as may be reasonably requested by the other Party to evidence or reflect the Contemplated Transactions and to carry out the intent and purposes of this Agreement and the Registration Rights Agreement.

8.10 <u>Severability</u>. Any term or provision of this Agreement that is invalid or unenforceable in any situation in any jurisdiction shall not affect the validity or enforceability of the remaining terms and provisions of this Agreement or the validity or enforceability of the offending term or provision in any other situation or in any other jurisdiction. If a final judgment of a court of competent jurisdiction declares that any term or provision of this Agreement is invalid or unenforceable, the Parties agree that the court making such determination shall have the power to limit such term or provision, to delete specific words or phrases or to replace such term or provision, and this Agreement shall be valid and enforceable as so modified. In the event such court does not exercise the power granted to it in the prior sentence, the Parties agree to replace such invalid or unenforceable term or provision with a valid and enforceable term or provision that will achieve, to the extent possible, the economic, business and other purposes of such invalid or unenforceable term or provision.

8.11 <u>Other Remedies; Specific Performance</u>. Except as otherwise provided herein, any and all remedies herein expressly conferred upon a Party will be deemed cumulative with and not exclusive of any other remedy conferred hereby, or by law or equity upon such Party, and the exercise by a Party of any one remedy will not preclude the exercise of any other remedy. The Parties agree that irreparable damage for which monetary damages, even if available, would not be an adequate remedy, would occur in the event that any Party does not perform the provisions of this Agreement (including failing to take such actions as are required of it hereunder to consummate this Agreement) in accordance with its specified terms or otherwise breaches such provisions. Accordingly, the Parties acknowledge and agree that the Parties shall be entitled to seek an injunction, specific performance and other equitable relief to prevent breaches of this Agreement and to enforce specifically the terms and provisions hereof, in addition to any other remedy to which they are entitled at law or in equity. Each of the Parties agrees that it will not oppose the granting of an injunction, specific performance or other equitable relief on the basis that any other Party has an adequate remedy at law or that any award of specific performance is not an appropriate remedy for any reason at law or in equity. Any Party seeking an injunction or injunctions to prevent breaches of this Agreement shall not be required to provide any bond or other security in connection with any such order or injunction.

8.12 <u>No Third-Party Beneficiaries</u>. Nothing in this Agreement, express or implied, is intended to or shall confer upon any Person (other than the Parties and the D&O Indemnified Parties to the extent of their respective rights pursuant to <u>Section 4.5</u>) any right, benefit or remedy of any nature whatsoever under or by reason of this Agreement.

8.13 Construction.

(a) References to "cash," "dollars" or "\$" are to U.S. dollars.

(b) For purposes of this Agreement, whenever the context requires: the singular number shall include the plural, and vice versa; the masculine gender shall include the feminine and neuter genders; the feminine gender shall include the masculine and neuter genders; and the neuter gender shall include masculine and feminine genders.

(c) The Parties have participated jointly in the negotiating and drafting of this Agreement and agree that any rule of construction to the effect that ambiguities are to be resolved against the drafting Party shall not be applied in the construction or interpretation of this Agreement, and no presumption or burden of proof shall arise favoring or disfavoring any Party by virtue of the authorship of any provision of this Agreement.

(d) As used in this Agreement, the words "include" and "including," and variations thereof, shall not be deemed to be terms of limitation, but rather shall be deemed to be followed by the words "without limitation."

(e) Except as otherwise indicated, all references in this Agreement to "Sections," "Exhibits" and "Schedules" are intended to refer to Sections of this Agreement and Exhibits and Schedules to this Agreement, respectively.

(f) Any reference to legislation or to any provision of any legislation shall include any modification, amendment, reenactment thereof, any legislative provision substituted therefore and all rules, regulations, and statutory instruments issued or related to such legislations.

(g) The bold-faced headings and table of contents contained in this Agreement are for convenience of reference only, shall not be deemed to be a part of this Agreement and shall not be referred to in connection with the construction or interpretation of this Agreement.

(h) The inclusion of any information in the Company Disclosure Schedule or Parent Disclosure Schedule shall not be deemed an admission or acknowledgment to any third party, in and of itself and solely by virtue of the inclusion of such information in the Company Disclosure Schedule or Parent Disclosure Schedule, as applicable, that such information is required to be listed in the Company Disclosure Schedule or Parent Disclosure Schedule, as applicable, that such information is required to be listed in the Company Disclosure Schedule or Parent as a whole, as the case may be, or that such items have resulted in a Company Material Adverse Effect or a Parent Material Adverse Effect. The Parties agree that each of the Company Disclosure Schedule and the Parent Disclosure Schedule shall be arranged in sections and subsections corresponding to the numbered and lettered sections and subsections contained in this Agreement. The disclosures in any section or subsection of the Company Disclosure Schedule or the Parent Disclosure Schedule shall qualify other sections and subsections in this Agreement to the extent it is readily apparent on its face from a reading of the disclosure that such disclosure is applicable to such other sections and subsections.

(i) Each of "delivered" or "made available" means, with respect to any documentation, that (i) prior to 11:59 p.m. (Eastern Time) on the date that is two Business Days prior to the date of this Agreement (A) a copy of such material has been posted to and made available by a Party to the other Party and its Representatives in the electronic data room maintained by such disclosing Party or (B) such material is disclosed in the Parent SEC Documents filed with the SEC prior to the date hereof and publicly made available on the SEC's Electronic Data Gathering Analysis and Retrieval system or (ii) delivered by or on behalf of a Party or its Representatives via electronic mail or in hard copy form prior to the execution of this Agreement.

(j) Whenever the last day for the exercise of any privilege or the discharge of any duty hereunder shall fall upon a Saturday, Sunday, or any date on which banks in Philadelphia, PA, are authorized or obligated by Law to be closed, the Party having such privilege or duty may exercise such privilege or discharge such duty on the next succeeding day which is a regular Business Day.

8.14 **Expenses**. Except as otherwise expressly provided in this Agreement, all expenses incurred in connection with this Agreement and the Contemplated Transactions will be paid by the Party incurring such expenses.

(Remainder of page intentionally left blank)

IN WITNESS WHEREOF, the Parties have caused this Agreement to be executed as of the date first above written.

ONCONOVA THERAPEUTICS, INC.

By: /s/ Steven M. Fruchtman

Name: Steven M. Fruchtman Title: President and CEO

TRAWS MERGER SUB, INC.

By: <u>/s/ Steven M. Fruchtman</u> Name: Steven M. Fruchtman Title: President and CEO

TRAWS MERGER SUB II, LLC

By: Onconova Therapeutics, Inc. Its: Sole member

By: /s/ Steven M. Fruchtman Name: Steven M. Fruchtman Title: President and CEO

[Signature Page to Agreement and Plan of Merger Agreement]

IN WITNESS WHEREOF, the Parties have caused this Agreement to be executed as of the date first above written.

TRAWSFYNYDD THERAPEUTICS, INC.

By: <u>/s/ Nikolay Savchuk</u> Name: Nikolay Savchuk Title: Chief Executive Officer

[Signature Page to Agreement and Plan of Merger Agreement]

EXHIBIT A

CERTAIN DEFINITIONS

For purposes of this Agreement (including this **Exhibit A**):

"Act" has the meaning set forth in Section 4.15.

"*Affiliate*" of a Person means any other Person that directly or indirectly, through one or more intermediaries, controls, is controlled by, or is under common control with, such Person. The term "control" (including the terms "controlled by" and "under common control with") means the possession, directly or indirectly, of the power to direct or cause the direction of the management and policies of a Person, whether through the ownership of voting securities, by Contract or otherwise.

"Agreement" means the Agreement and Plan of Merger to which this Exhibit A is attached, as it may be amended from time to time.

"Allocation Certificate" has the meaning set forth in Section 4.13.

"Anti-Bribery Laws" has the meaning set forth in Section 2.22.

"Board Approval" has the meaning set forth in the Recitals.

"Book-Entry Shares" has the meaning set forth in Section 1.7.

"Business Day" means any day other than a Saturday, Sunday or other day on which banks in Philadelphia, PA, are authorized or obligated by Law to be closed.

"Certificate of Designation" means the Certificate of Designation of Preferences, Rights and Limitations of Series C Non-Voting Convertible Preferred Stock in the form attached hereto as **Exhibit C**.

"Certificates of Merger" has the meaning set forth in Section 1.3.

"Certifications" has the meaning set forth in Section 3.7(a).

"Charter Amendment Proposal" has the meaning set forth in Section 4.1(a)(ii).

"*Closing*" has the meaning set forth in <u>Section 1.3</u>.

"Closing Date" has the meaning set forth in Section 1.3.

"Code" means the Internal Revenue Code of 1986, as amended.

"Company" has the meaning set forth in the Preamble.

"Company Associate" means any current or former employee, consultant, independent contractor, officer or director of the Company.

"Company Benefit Plan" has the meaning set forth in Section 2.17(a).

"Company Board" means the board of directors of the Company.



"Company Capital Stock" means the Company Common Stock and the Company Preferred Stock.

"Company Common Stock" means the Common Stock, \$0.0001 par value per share, of the Company.

"Company Contract" means any Contract: (a) to which the Company or any of its Subsidiaries is a Party; (b) by which the Company or any of its Subsidiaries or any Company IP or any other asset of the Company or its Subsidiaries is or may become bound or under which the Company or any of its Subsidiaries has, or may become subject to, any obligation; or (c) under which the Company or any of its Subsidiaries has or may acquire any right or interest.

"Company Data" means all data and information Processed by or for the Company or any of its Subsidiaries, including any Personal

Information.

"Company Disclosure Schedule" has the meaning set forth in Section 2.

"Company ERISA Affiliate" means any corporation or trade or business (whether or not incorporated) which is (or at any relevant time was) treated with the Company as a single employer within the meaning of Section 414 of the Code.

"Company Financials" has the meaning set forth in Section 2.7(a).

"Company In-bound License" has the meaning set forth in Section 2.12(d).

"Company IP" means all Intellectual Property Rights that are owned or purported to be owned by, assigned to, or exclusively licensed by, the Company, including without limitation, all Company Registered IP.

"Company Material Adverse Effect" means any Effect that, considered together with all other Effects that have occurred prior to the date of determination of the occurrence of a Company Material Adverse Effect, has or would reasonably be expected to have a material adverse effect on the business, condition (financial or otherwise), assets, liabilities or results of operations of the Company, taken as a whole; *provided, however*, that Effects arising or resulting from the following shall not be taken into account in determining whether there has been a Company Material Adverse Effect: (a) general business or economic conditions affecting the industry in which the Company and its Subsidiaries operate, (b) acts of war, armed hostilities or terrorism, acts of God or comparable events, epidemic, pandemic or disease outbreak (including the COVID-19 virus) or any worsening of the foregoing, or any declaration of martial law, quarantine or similar directive, policy or guidance or Law or other action by any Governmental Body in response thereto, (c) changes in financial, banking or securities markets, (d) any change in, or any compliance with or action taken for the purpose of complying with, any Law or GAAP (or interpretations of any Law or GAAP), (e) resulting from the announcement of this Agreement or the pendency of the Contemplated Transactions; *provided*, that this clause (e) shall not apply to any representation or warranty (or condition to the consummation of the Merger relating to such representation or warranty) to the extent the representation and warranty expressly addresses the consequences resulting from the execution and delivery of this Agreement or the consummation of the Contemplated Transactions, or (f) resulting from the taking of any action required to be taken by this Agreement; except in each case with respect to clauses (a) through (c), to the extent disproportionately affecting the Company, taken as a whole, relative to other similarly situated companies in the industries in which the Company o

"Company Material Contract(s)" has the meaning set forth in Section 2.13(a).

"Company Options" means options or other rights to purchase shares of Company Common Stock issued by the Company.

"Company Out-bound License" has the meaning set forth in Section 2.12(d).

"Company Permits" has the meaning set forth in Section 2.14.

"Company Plan" has the meaning set forth in Section 2.6(c).

"Company Preferred Stock" has the meaning set forth in Section 2.6(a).

"Company Real Estate Leases" has the meaning set forth in Section 2.11.

"Company Registered IP" means all Registered IP owned or purported to be owned, in whole or in part, or exclusively licensed by the

Company.

"Company Signatories" has the meaning set forth in the Recitals.

"Company Stock Certificate" has the meaning set forth in Section 1.7.

"Company Stockholder Matters" has the meaning set forth in the Recitals.

"Company Stockholder Support Agreement" has the meaning set forth in the Recitals.

"Company Systems" has the meaning set forth in Section 2.12(j).

"Company Unaudited Interim Balance Sheet" means the unaudited balance sheet of the Company as of December 31, 2023 provided to Parent prior to the date of this Agreement.

"Concurrent Investment Amount" means \$14,000,000 as contemplated by the Securities Purchase Agreement.

"Confidentiality Agreement" means that certain Bilateral Confidentiality Agreement, dated as of November 21, 2023, between Parent and Viriom, Inc.

"Consent" means any approval, consent, ratification, permission, waiver or authorization (including any Governmental Authorization).

"Contemplated Transactions" means the Merger, the Nasdaq Reverse Split, Parent Stockholder Support Agreements, the CVR Agreement and the other transactions and actions contemplated by this Agreement to be consummated at or prior to the Closing (but not, for the avoidance of doubt, the actions proposed to be taken as the Parent Stockholders' Meeting following the Closing pursuant to <u>Section 4.2</u>).

"Continuing Employees" has the meaning set forth in Section 4.4(a).

"*Contract*" means, with respect to any Person, any written or oral agreement, contract, subcontract, lease (whether for real or personal property), mortgage, license, sublicense or other legally binding commitment or undertaking of any nature to which such Person is a party or by which such Person or any of its assets are bound or affected under applicable Law.

"Data Processing Policy" means each policy, statement, representation, or notice of the Company, Parent or their respective Subsidiaries relating to the Processing of Company Data or Parent Data (as applicable), privacy, data protection, or security.

"DGCL" means the General Corporation Law of the State of Delaware.

"DLLCA" means the Delaware Limited Liability Company Act.

"D&O Indemnified Parties" has the meaning set forth in Section 4.5(a).

"Disqualifying Event" has the meaning set forth in Section 3.24.

"Dissenting Shares" has the meaning set forth in Section 1.10(a).

"Effect" means any effect, change, event, circumstance, or development.

"Encumbrance" means any lien, pledge, hypothecation, charge, mortgage, security interest, lease, license, option, easement, reservation, servitude, adverse title, claim, infringement, interference, option, right of first refusal, preemptive right, community property interest or restriction or encumbrance of any nature (including any restriction on the voting of any security, any restriction on the transfer of any security or other asset, any restriction on the receipt of any income derived from any asset, any restriction on the use of any asset and any restriction on the possession, exercise or transfer of any other attribute of ownership of any asset).

"Enforceability Exceptions" means the (a) Laws of general application relating to bankruptcy, insolvency and the relief of debtors; and (b) rules of law governing specific performance, injunctive relief and other equitable remedies.

"Entity" means any corporation (including any non-profit corporation), partnership (including any general partnership, limited partnership or limited liability partnership), joint venture, estate, trust, company (including any company limited by shares, limited liability company or joint stock company), firm, society or other enterprise, association, organization or entity, and each of its successors.

"Environmental Law" means any federal, state, local or foreign Law relating to pollution or protection of human health (as it relates to exposure to Hazardous Materials) or the environment (including ambient air, surface water, ground water, land surface or subsurface strata), including any Law or regulation relating to emissions, discharges, releases or threatened releases of Hazardous Materials, or otherwise relating to the manufacture, processing, distribution, use, treatment, storage, disposal, transport or handling of Hazardous Materials.

"ERISA" means the Employee Retirement Income Security Act of 1974, as amended.

"Exchange Act" means the Securities Exchange Act of 1934.

"Exchange Agent" has the meaning set forth in Section 1.8.

"Exchange Fund" has the meaning set forth in Section 1.8.

"Exchange Ratio" means 20.1291 of Parent Common Stock for each share of Company Common Stock (each as calculated on a fully-

diluted basis).

"First Certificate of Merger" has the meaning set forth in Section 1.3.

"First Effective Time" has the meaning set forth in Section 1.3.

"First Merger" has the meaning set forth in the Recitals.

"First Merger Sub" has the meaning set forth in the Preamble.

"First Merger Sub Board" means the board of directors of First Merger Sub.

"First Step Surviving Corporation" has the meaning set forth in Section 1.1.

"FLSA" has the meaning set forth in Section 2.17(m).

"*GAAP*" means generally accepted accounting principles and practices in effect from time to time within the United States applied consistently throughout the period involved.

"Governmental Authorization" means any: (a) permit, license, certificate, franchise, permission, variance, exception, approval, exemption, order, clearance, registration, qualification or authorization issued, granted, given or otherwise made available by or under the authority of any Governmental Body or pursuant to any Law; or (b) right under any Contract with any Governmental Body.

"Governmental Body" means any: (a) nation, state, commonwealth, province, territory, county, municipality, district or other jurisdiction of any nature; (b) federal, state, local, municipal, foreign or other government; (c) governmental or quasi-governmental authority of any nature (including any governmental division, department, agency, commission, bureau, instrumentality, official, ministry, fund, foundation, center, organization, unit, body or Entity and any court or other tribunal, and for the avoidance of doubt, any taxing authority); or (d) self-regulatory organization (including Nasdaq).

"Hazardous Materials" means any pollutant, chemical, substance and any toxic, infectious, carcinogenic, reactive, corrosive, ignitable or flammable chemical, or chemical compound, or hazardous substance, material or waste, whether solid, liquid or gas, that is subject to regulation, control or remediation under any Environmental Law, including without limitation, crude oil or any fraction thereof, and petroleum products or byproducts.

"Intellectual Property Rights" means any and all of the following arising pursuant to the Laws of any jurisdiction throughout the world, including without limitation: (a) copyrights, applicable to copyrightable works, original works of authorship fixed in any tangible medium of expression, including literary works (including all forms and types of computer software, including all source code, object code, firmware, development tools, files, records and data, and all documentation related to any of the foregoing), pictorial and graphic works, database and design rights, whether or not registered or published, including all data collections, "moral" rights, mask works and copyright registrations and applications in any of the foregoing and corresponding rights in works of authorship (collectively, "Copyrights"); (b) all trademarks, service marks, trade names, service names, brand names, trade dress rights, and rights, logos, corporate names, and other source or business identifiers, together with the goodwill associated with any of the foregoing, along with all applications, registrations, renewals and extensions thereof (collectively, "Trademarks"); (c) registration right to Internet domain names, URLs, and similar rights; (d) rights under applicable trade secret Laws arising with respect to know how, inventions (including conceptions and/or reductions to practice), invention disclosures, methods, processes, protocols, specifications, techniques, discoveries and improvements, formulae, confidential and proprietary information, technical information, designs, drawings, procedures, models, formulations, manuals and systems, including all biological, chemical, biochemical, toxicological, pharmacological and metabolic material and information and data relating thereto and formulation, clinical, analytical and stability information and data, in each case which are not available in the public domain and have actual or potential commercial value that is derived, in whole or in part, from such secrecy (collectively, "Know-How"); (e) all patents, industrial property rights, patent applications, provisional patent applications and similar instruments (including any and all substitutions, revisions, divisions, continuations, continuations-in-part, divisions, reissues, renewals, re-examinations and extensions and any foreign equivalents of the foregoing (including certificates of invention and any applications therefor)) (collectively, "Patents"); and (f) all rights to prosecute and perfect any of the foregoing through administrative prosecution, registration, recordation or other administrative proceeding, and all causes of action and rights to sue or seek other remedies arising from or relating to any of the foregoing.

"Intended Tax Treatment" has the meaning set forth in Section 4.8.

"Investor Agreements" has the meaning set forth in Section 2.21(b).

"IRS" means the United States Internal Revenue Service.

"*Knowledge*" means, with respect to an individual, that such individual is actually aware of the relevant fact or such individual would reasonably be expected to know such fact in the ordinary course of the performance of such individual's employment responsibilities. Any Person that is an Entity shall have Knowledge if any officer or director of such Person as of the date such knowledge is imputed has Knowledge of such fact or other matter.

"*Law*" means any federal, state, national, foreign, material local or municipal or other law, statute, constitution, principle of common law, resolution, ordinance, code, edict, decree, rule, regulation, ruling or requirement issued, enacted, adopted, promulgated, implemented or otherwise put into effect by or under the authority of any Governmental Body (including under the authority of Nasdaq or the Financial Industry Regulatory Authority).

"*Legal Proceeding*" means any action, suit, litigation, arbitration, proceeding (including any civil, criminal, administrative, investigative or appellate proceeding), hearing, inquiry, audit, examination or investigation commenced, brought, conducted or heard by or before, or otherwise involving, any court or other Governmental Body or any arbitrator or arbitration panel.

"Letter of Transmittal" has the meaning set forth in Section 1.8(b).

"Liability" has the meaning set forth in Section 2.9.

"Lock-Up Agreement" has the meaning set forth in the Recitals.

"Merger" has the meaning set forth in the Recitals.

"Merger Consideration" has the meaning set forth in Section 1.5.

"Merger Subs" has the meaning set forth in the Preamble.

"*Nasdaq*" means the Nasdaq Stock Market, including the Nasdaq Global Select Market or such other Nasdaq market on which shares of Parent Common Stock are then listed.

"Nasdaq Listing Application" has the meaning set forth in Section 4.8.

"Nasdaq Reverse Split" means a reverse stock split of all outstanding shares of Parent Common Stock at a reverse stock split ratio to be reasonably determined by Parent for the purpose of maintaining compliance with Nasdaq listing standards.

"Ordinary Course of Business" means, in the case of each of the Company and Parent, such actions taken in the ordinary course of its normal operations and consistent with its past practices.

"Organizational Documents" means, with respect to any Person (other than an individual), (a) the certificate or articles of association or incorporation or organization or limited partnership or limited liability company, and any joint venture, limited liability company, operating or partnership agreement and other similar documents adopted or filed in connection with the creation, formation or organization of such Person and (b) all bylaws, regulations and similar documents or agreements relating to the organization or governance of such Person, in each case, as amended or supplemented.

"Owned Company Registered IP" has the meaning set forth in Section 2.12(a).

"Owned Parent Registered IP" has the meaning set forth in Section 3.12(a).

"Parent" has the meaning set forth in the Preamble.

"Parent Associate" means any current or former employee, independent contractor, officer or director of Parent.

"*Parent Balance Sheet*" means the unaudited balance sheet of Parent as of December 31, 2023 (the "*Parent Balance Sheet Date*") provided to the Company prior to the date of this Agreement.

"Parent Benefit Plan" has the meaning set forth in Section 3.17(a).

"Parent Board" means the board of directors of Parent.

"Parent Common Stock" means the Common Stock, \$0.01 par value per share, of Parent.

"Parent Common Stock Payment Shares" has the meaning set forth in Section 1.5.

"*Parent Contract*" means any Contract: (a) to which Parent or any of its Subsidiaries is a party; (b) by which Parent or any of its Subsidiaries or any Parent IP or any other asset of Parent or any of its Subsidiaries is or may become bound or under which Parent or any of its Subsidiaries has, or may become subject to, any obligation; or (c) under which Parent or any of its Subsidiaries has or may acquire any right or interest.

"Parent Convertible Preferred Stock" means Parent's non-voting convertible preferred stock, par value \$0.01 per share, with the rights, preferences, powers and privileges specified in the Certificate of Designation.

"*Parent Covered Person*" means, with respect to Parent as an "issuer" for purposes of Rule 506 promulgated under the Securities Act, any Person listed in the first paragraph of Rule 506(d)(1).

"Parent Data" means all data and information Processed by or for Parent or any of its Subsidiaries, including any Personal Information.

"Parent Disclosure Schedule" has the meaning set forth in Section 3.

"Parent ERISA Affiliate" means any corporation or trade or business (whether or not incorporated) which is (or at any relevant time was) treated with Parent or any of its Subsidiaries as a single employer within the meaning of Section 414 of the Code.

"*Parent Financing*" means an acquisition of shares of Parent Common Stock and Parent Convertible Preferred Stock to be consummated concurrently with the Closing pursuant to the Securities Purchase Agreement with aggregate gross cash proceeds to Parent of at least the Concurrent Investment Amount.

"Parent Foreign Plan" has the meaning set forth in Section 3.17(1).

"Parent In-bound License" has the meaning set forth in Section 3.12(d).

"*Parent IP*" means all Intellectual Property Rights that are owned or purported to be owned by, assigned to, or exclusively licensed by, Parent or its Subsidiaries, including without limitation, all Parent Registered IP.

"Parent Material Adverse Effect" means any Effect that, considered together with all other Effects that have occurred prior to the date of determination of the occurrence of a Parent Material Adverse Effect, has or would reasonably be expected to have a material adverse effect on the business, condition (financial or otherwise), assets, liabilities or results of operations of Parent; provided, however, that Effects arising or resulting from the following shall not be taken into account in determining whether there has been a Parent Material Adverse Effect: (a) general business or economic conditions affecting the industry in which Parent operates, (b) acts of war, armed hostilities or terrorism, acts of God or comparable events, epidemic, pandemic or disease outbreak (including the COVID-19 virus) or any worsening of the foregoing, or any declaration of martial law, quarantine or similar directive, policy or guidance or Law or other action by any Governmental Body in response thereto, (c) changes in financial, banking or securities markets, (d) the taking of any action required to be taken by this Agreement, (e) any change in the stock price or trading volume of Parent Common Stock (it being understood, however, that any Effect causing or contributing to any change in stock price or trading volume of Parent Common Stock may be taken into account in determining whether a Parent Material Adverse Effect has occurred, unless such Effects are otherwise excepted from this definition); (f) any change in, or any compliance with or action taken for the purpose of complying with, any Law or GAAP (or interpretations of any Law or GAAP); (g) resulting from the announcement of this Agreement or the pendency of the Contemplated Transactions; provided, that this clause (g) shall not apply to any representation or warranty (or condition to the consummation of the Merger relating to such representation or warranty) to the extent the representation and warranty expressly addresses the consequences resulting from the execution and delivery of this Agreement or the consummation of the Contemplated Transactions; or (h) resulting from the taking of any action or the failure to take any action, by Parent that is required to be taken by this Agreement, except in each case with respect to clauses (a) through (c), to the extent disproportionately affecting Parent relative to other similarly situated companies in the industries in which Parent operates.

"Parent Material Contract(s)" has the meaning set forth in Section 3.13(a).

"Parent Options" means options or other rights to purchase shares of Parent Common Stock issued by Parent.

"Parent Out-bound License" has the meaning set forth in Section 3.12(c).

"Parent Outstanding Shares Certificate" has the meaning set forth in Section 4.12(b)

"Parent PEO Plans" has the meaning set forth in Section 3.17(a).

"Parent Permits" has the meaning set forth in Section 3.14

"Parent Preferred Stock Payment Shares" has the meaning set forth in Section 1.5.

"Parent Real Estate Leases" has the meaning set forth in Section 3.11.

Parent.

"Parent Registered IP" means all Registered IP owned or purported to be owned, in whole or in part, or exclusively licensed by the

"Parent RSUs" means any restricted stock unit award granted pursuant to the Parent Stock Plans.

"Parent SEC Documents" has the meaning set forth in Section 3.7(a).

"Parent Signatories" has the meaning set forth in the Recitals.

"Parent Stock Plans" means collectively, the Onconova Therapeutics, Inc. 2021 Incentive Compensation Plan, as amended and restated in 2022, the Onconova Therapeutics, Inc. 2018 Omnibus Incentive Compensation Plan, and the Onconova Therapeutics, Inc. 2013 Equity Compensation Plan, each as may be amended from time to time.

"Parent Stockholder Matters" has the meaning set forth in Section 4.1(a)(ii).

"Parent Stockholders' Meeting" has the meaning set forth in Section 4.1(a)(ii).

"Parent Systems" has the meaning set forth in Section 3.12(j).

"Party" or "Parties" means the Company, First Merger Sub, Second Merger Sub and Parent.

"Permitted Encumbrance" means: (a) any Encumbrance for current Taxes not yet due and payable or for Taxes that are being contested in good faith and, in each case, for which adequate reserves have been made on the Company Unaudited Interim Balance Sheet or the Parent Balance Sheet, as applicable, in accordance with GAAP; (b) minor liens that have arisen in the Ordinary Course of Business and that do not (in any case or in the aggregate) materially detract from the value of the assets or properties subject thereto or materially impair the operations of the Company or any of its Subsidiaries or Parent, as applicable; (c) liens to secure obligations to landlords, lessors or renters under leases or rental agreements; (d) deposits or pledges made in connection with, or to secure payment of, workers' compensation, unemployment insurance or similar programs mandated by Law; (e) nonexclusive licenses of Intellectual Property Rights granted by the Company or any of its Subsidiaries or Parent, as applicable, in the Ordinary Course of Business and that do not (in any case or in the aggregate) materially detract from the value of the Intellectual Property Rights subject thereto; (f) statutory liens in favor of carriers, warehousemen, mechanics and materialmen, to secure claims for labor, materials or supplies the payment for which is not delinquent; and (g) obligations of the Company or any of its Subsidiaries or Parent under Company In-bound Licenses disclosed in <u>Section 2.12(d)</u> with respect to Company or any of its Subsidiaries or under Parent In-bound Licenses disclosed in <u>Section 3.12(d)</u> with respect to Parent, as the case may be.

"Person" means any individual, Entity or Governmental Body.

"Personal Information" means all information in any form or media that identifies, could be used to identify or is otherwise related to an individual person (including any current, prospective, or former customer, end user or employee), in addition to any definition for "personal information" or any similar term provided by applicable Law or by the Company or any of its Subsidiaries in any of its privacy policies, notices or contracts (e.g., "personal data," "personally identifiable information" or "PII")." "Post-Closing Plans" has the meaning set forth in Section 4.4(a).

"Preferred Stock Conversion Proposal" has the meaning set forth in Section 1.5.

"Privacy and Data Processing Requirements" means any applicable (i) Law (including of any applicable foreign jurisdiction) relating to privacy, data protection, security, or Personal Information, including, the Federal Trade Commission Act, California Consumer Privacy Act (CCPA), HIPAA, EU General Data Protection Regulation (GDPR), any applicable Law relating to breach notification, and any laws relating to the use of biometric identifiers, (ii) Data Processing Policy, or (iii) requirement of any self-regulatory organization, industry standard (including, as applicable, the Payment Card Industry Data Security Standard), or Contract by which the Company, Parent or their respective Subsidiaries are bound relating to the Processing of Company Data or Parent Data (as applicable), privacy, data protection, or security, including, in each case of (i) through (iii), in connection with direct marketing or the initiation, transmission, monitoring, interception, recording, or receipt of communications.

"*Process*" means, with respect to any data, information, or information technology system, any operation or set of operations performed thereon, whether or not by automated means, including access, adaptation, alignment, alteration, collection, combination, compilation, consultation, creation, derivation, destruction, disclosure, disposal, dissemination, erasure, interception, maintenance, making available, organization, recording, restriction, retention, retrieval, storage, structuring, transmission, and use, and security measures with respect thereto.

"Proxy Statement" has the meaning set forth in Section 4.2.

"Reference Date" means March 31, 2024.

"*Registered IP*" means all Intellectual Property Rights that are registered or issued under the authority of, with or by any Governmental Body or Internet domain registrar, including all Patents, registered Copyrights, registered Trademarks, Internet domain names, and all applications for registration of any of the foregoing.

"Registration Rights Agreement" has the meaning set forth in the Recitals.

"Representatives" means directors, officers, employees, agents, attorneys, accountants, investment bankers, advisors and representatives.

"Required Company Stockholder Vote" has the meaning set forth in Section 2.4.

"Required Parent Stockholder Vote" has the meaning set forth in Section 3.4.

"SAFE Investment" has the meaning sent forth in the Recitals.

"Sarbanes-Oxley Act" means the Sarbanes-Oxley Act of 2002.

"SEC" means the United States Securities and Exchange Commission.

"Second Certificate of Merger" has the meaning set forth in Section 1.3.

"Second Effective Time" has the meaning set forth in Section 1.3.

"Second Merger" has the meaning set forth in the Recitals.

"Second Merger Sub" has the meaning set forth in the Preamble.

"Second Merger Sub Board" means the board of managers of Second Merger Sub.

"Securities Act" means the Securities Act of 1933, as amended.

"Series A Preferred Stock" means shares of the Parent's Series A, par value \$0.01 per share.

"Series B Preferred Stock" means shares of the Parent's Series B, par value \$0.01 per share

"Stockholder Written Consent" has the meaning set forth in the Recitals.

"*Subsidiary*" An entity shall be deemed to be a 'subsidiary' of a Person if such Person directly or indirectly owns or purports to own, beneficially or of record, (a) an amount of voting securities or other interests in such entity that is sufficient to enable such Person to elect at least a majority of the members of such entity's board of directors or other governing body, or (b) at least 50% of the outstanding equity, voting, beneficial or financial interests in such Entity.

"Surviving Entity" has the meaning set forth in Section 1.1.

"Takeover Statute" means any "fair price," "moratorium," "control share acquisition" or other similar anti-takeover Law.

"*Tax*" means any (i) federal, state, local, foreign or other tax, including any income, capital gain, gross receipts, capital stock, profits, transfer, estimated, registration, stamp, premium, escheat, unclaimed property, customs duty, ad valorem, occupancy, occupation, alternative, add-on, windfall profits, value added, severance, property, business, production, sales, use, license, excise, franchise, employment, payroll, social security, disability, unemployment, workers' compensation, national health insurance, withholding or other taxes, duties, fees, assessments or governmental charges, surtaxes or deficiencies thereof in the nature of a tax, however denominated (whether imposed directly or through withholding and whether or not disputed), and including any fine, penalty, addition to tax, or interest or additional amount imposed by a Governmental Body with respect thereto (or attributable to the nonpayment thereof) and (ii) any liability for payment of amounts described in clause (i) whether as a result of transferee or successor liability, of being a member of an affiliated, consolidated, combined or unitary group for any period, pursuant to a Contract, through operation of Law or otherwise.

"*Tax Return*" means any return (including any information return), report, statement, declaration, claim for refund, estimate, schedule, notice, notification, form, election, certificate or other document, and any amendment or supplement to any of the foregoing, filed with or submitted to, or required to be filed with or submitted to, any Governmental Body (or provided to a payee) in connection with the determination, assessment, collection or payment of any Tax or in connection with the administration, implementation or enforcement of or compliance with any Law relating to any Tax.

"Treasury Regulations" means the United States Treasury regulations promulgated under the Code.

"WARN Act" means the Worker Adjustment Retraining and Notification Act of 1988, as amended, or any similar state or local plant closing mass layoff statute, rule or regulation.

"Withholding Agent" has the meaning set forth in Section 1.13.

EXHIBIT B

Execution Version

FORM OF LOCK-UP AGREEMENT

April 1, 2024

Onconova Therapeutics, Inc. 12 Penns Trail Newtown, PA 18940

Ladies and Gentlemen:

The undersigned signatory of this lock-up agreement (this "*Lock-Up Agreement*") understands that Onconova Therapeutics, Inc. a Delaware corporation (including any successor thereto, "*Parent*"), is entering into an Agreement and Plan of Merger, dated as of the date hereof (as the same may be amended from time to time, the "*Merger Agreement*") with Traws Merger Sub I, Inc., a Delaware corporation and a wholly owned subsidiary of Parent, Traws Merger Sub II, LLC, a Delaware limited liability company and a wholly owned subsidiary of Parent, and Trawsfynydd Therapeutics, Inc., a Delaware corporation (the "*Company*"). Capitalized terms used but not otherwise defined herein shall have the respective meanings ascribed to such terms in the Merger Agreement.

1. As a condition and material inducement to each of the parties to enter into the Merger Agreement and to consummate the Contemplated Transactions, and for other good and valuable consideration, the receipt and sufficiency of which is hereby acknowledged, the undersigned hereby irrevocably agrees that, subject to the exceptions set forth herein, without the prior written consent of Parent, the undersigned will not, during the period commencing upon the Closing and ending on the date that is 180 days after the Closing Date (the "*Restricted Period*"):

(a) offer, pledge, sell, contract to sell, sell any option or contract to purchase, purchase any option or contract to sell, grant any option, right or warrant to purchase, lend or otherwise transfer or dispose of, directly or indirectly, any shares of Parent Common Stock, Parent Convertible Preferred Stock or any securities convertible into or exercisable or exchangeable for Parent Common Stock (including without limitation, Parent Common Stock or such other securities which may be deemed to be beneficially owned by the undersigned in accordance with the rules and regulations of the SEC and securities of Parent which may be issued upon exercise of Parent Options or settlement of Parent RSUs) that are currently or hereafter owned by the undersigned (collectively, the "*Undersigned's Shares*");

(b) enter into any swap, short sale, hedge or other agreement that transfers, in whole or in part, any of the economic consequences of ownership of the Undersigned's Shares regardless of whether any such transaction described in clause (a) above or this clause (b) is to be settled by delivery of Parent Common Stock, Parent Convertible Preferred Stock or other securities, in cash or otherwise;

(c) make any demand for, or exercise any right with respect to, the registration of any shares of Parent Common Stock, Parent Convertible Preferred Stock or any security convertible into or exercisable or exchangeable for Parent Common Stock (other than such rights set forth in the Merger Agreement or the Registration Rights Agreement entered into in connection with the Securities Purchase Agreement); or

(d) publicly disclose the intention to do any of the foregoing.

2. The restrictions and obligations contemplated by this Lock-Up Agreement shall not apply to:

(a) transfers of the Undersigned's Shares:

(i) if the undersigned is a natural person, (A) to any person related to the undersigned (or to an ultimate beneficial owner of the undersigned) by blood or adoption who is an immediate family member of the undersigned, or by marriage or domestic partnership (a "*Family Member*"), or to a trust formed for the benefit of the undersigned or any of the undersigned's Family Members, (B) to the undersigned's estate, following the death of the undersigned, by will, intestacy or other operation of Law, (C) as a bona fide charitable gift or contribution, (D) by operation of Law pursuant to a qualified domestic order or in connection with a divorce settlement or (E) to any partnership, corporation or limited liability company which is controlled by or under common control with the undersigned and/or by any such Family Member(s);

(ii) if the undersigned is a corporation, partnership or other Entity, (A) to another corporation, partnership, or other Entity that is a direct or indirect affiliate (as defined under Rule 12b-2 of the Exchange Act) of the undersigned, including investment funds or other entities that control or manage, are under common control or management with, or are controlled or managed by, the undersigned (including, for the avoidance of doubt, a fund managed by the same manager, managing member, general partner or management company or by an Entity controlling, controlled by or under common control with such manager, managing member, general partner or management company of the undersigned), (B) as a distribution or dividend to equity holders, current or former general or limited partners, members or managers (or to the estates of any of the foregoing), as applicable, of the undersigned (including upon the liquidation and dissolution of the undersigned pursuant to a plan of liquidation approved by the undersigned's equity holders), (C) as a bona fide charitable gift or contribution or otherwise to a trust or other entity for the direct or indirect benefit of an immediate family member of a beneficial owner (as defined in Rule 13d-3 of the Exchange Act) of the Undersigned's Shares or (D) transfers or dispositions not involving a change in beneficial ownership, including (i) transactions involving a basket default swap or other derivative securities tied to an underlying index or basket of publicly-traded equities, corporate bonds or other assets subject to credit risk, (ii) transactions concerning an index or basket of securities (iii) a pledge or hypothecation of shares of Parent Common Stock as collateral for a loan or otherwise; and (iv) a hedge or other transaction taken with the intent of reducing the risk of loss of investment or experiencing a shortfall, in each case of (i) – (iv); or

(iii) if the undersigned is a trust, to any grantors or beneficiaries of the trust; *provided that*, in the case of any transfer or distribution pursuant to this clause (a), such transfer is not for value and each donee, heir, beneficiary or other transferee or distributee shall sign and deliver to Parent a lock-up agreement in the form of this Lock-Up Agreement with respect to the shares of Parent Common Stock, Parent Convertible Preferred Stock or such other securities that have been so transferred or distributed;

(b) the exercise of Parent Options (including a net or cashless exercise of a Parent Option), and any related transfer of shares of Parent Common Stock to Parent for the purpose of paying the exercise price of such options or for paying taxes (including estimated taxes) due as a result of the exercise of such options; *provided that*, for the avoidance of doubt, the underlying shares of Parent Common Stock shall continue to be subject to the restrictions on transfer set forth in this Lock-Up Agreement;

(c) the disposition (including a forfeiture or repurchase) to Parent of any shares of restricted stock granted pursuant to the terms of any employee benefit plan or restricted stock purchase agreement;

(d) transfers to Parent in connection with the net settlement of any Parent RSU or other equity award that represents the right to receive in the future shares of Parent Common Stock settled in Parent Common Stock to pay any tax withholding obligations; *provided that*, for the avoidance of doubt, the underlying shares of Parent Common Stock shall continue to be subject to the restrictions on transfer set forth in this Lock-Up Agreement;

(e) the establishment of a trading plan pursuant to Rule 10b5-1 under the Exchange Act for the transfer of Parent Common Stock; *provided that* such plan does not provide for any transfers of Parent Common Stock during the Restricted Period;

(f) transfers by the undersigned of shares of Parent Common Stock purchased by the undersigned on the open market or in a public offering by Parent, in each case following the Closing Date;

(g) pursuant to a bona-fide third party tender offer, merger, consolidation or other similar transaction made to all holders of Parent's capital stock involving a change of control of Parent, *provided that* in the event that such tender offer, merger, consolidation or other such transaction is not completed, the Undersigned's Shares shall remain subject to the restrictions contained in this Lock-Up Agreement;

(h) pursuant to an order of a court or regulatory agency; or

(i) any shares of Parent Convertible Preferred Stock (and shares of Parent Common Stock issuable upon conversion of such shares of Parent Convertible Preferred Stock) purchased by the undersigned in the Parent Financing.

and *provided*, *further*, that, with respect to each of (a), (b), (c), (d) and (e) above, no filing by any party (including any donor, donee, transferor, transferee, distributor or distributee) under Section 16 of the Exchange Act or other public announcement shall be made voluntarily reporting a reduction in beneficial ownership of shares of Parent Common Stock, Parent Convertible Preferred Stock or securities convertible into or exchangeable for Parent Common Stock in connection with such transfer or disposition during the Restricted Period, other than (i) any exit filings or public announcements that may be required under applicable federal and state securities Laws or (ii) in respect of a required filing under the Exchange Act reporting a reduction in beneficial ownership of such securities, including in connection with the exercise of an option to purchase Parent Common Stock or in connection with the net settlement of any RSU or other equity award that represents the right to receive in the future shares of Parent Common Stock settled in Parent Common Stock that would otherwise expire during the Restricted Period, *provided that* (1) reasonable notice shall be provided to Parent prior to any such filing and (2) such filing, report or announcement shall clearly indicate in the footnotes therein, in reasonable detail, a description of the circumstances of the transfer and that the shares remain subject to this Lock-Up Agreement.

For purposes of this Lock-Up Agreement, "change of control" shall mean the transfer (whether by tender offer, merger, consolidation or other similar transaction), in one transaction or a series of related transactions, to a person or group of affiliated persons, of Parent's voting securities if, after such transfer, Parent's stockholders as of immediately prior to such transfer do not hold a majority of the outstanding voting securities of Parent (or the surviving entity) other than the Contemplated Transactions.

3. Any attempted transfer in violation of this Lock-Up Agreement will be of no effect and null and void, regardless of whether the purported transferee has any actual or constructive knowledge of the transfer restrictions set forth in this Lock-Up Agreement, and will not be recorded on the share register of Parent. In furtherance of the foregoing, the undersigned agrees that Parent and any duly appointed transfer agent for the registration or transfer of the securities described herein are hereby authorized to decline to make any transfer of securities if such transfer would constitute a violation or breach of this Lock-Up Agreement. Parent may cause the legend set forth below, or a legend substantially equivalent thereto, to be placed upon any certificate(s) or other documents, ledgers or instruments evidencing the undersigned's ownership of Parent Common Stock, Parent Convertible Preferred Stock or any securities convertible into or exercisable or exchangeable for Parent Common Stock:

THE SHARES REPRESENTED HEREBY ARE SUBJECT TO AND MAY ONLY BE TRANSFERRED IN COMPLIANCE WITH A LOCK-UP AGREEMENT, A COPY OF WHICH IS ON FILE AT THE PRINCIPAL OFFICE OF THE COMPANY.



4. The undersigned hereby represents and warrants that the undersigned has full power and authority to enter into this Lock-Up Agreement. All authority herein conferred or agreed to be conferred and any obligations of the undersigned shall be binding upon the successors, assigns, heirs or personal representatives of the undersigned.

5. The undersigned understands that if the Merger Agreement is terminated for any reason, this Lock-Up Agreement will automatically terminate and the undersigned shall be released from all of its, his or her obligations under this Lock-Up Agreement. The undersigned understands that Parent and the Company are proceeding with the Contemplated Transactions in reliance upon this Lock-Up Agreement.

6. Any and all remedies herein expressly conferred upon Parent or the Company will be deemed cumulative with and not exclusive of any other remedy conferred hereby, or by Law or equity, and the exercise by Parent or the Company of any one remedy will not preclude the exercise of any other remedy. The undersigned agrees that irreparable damage for which monetary damages, even if available, would not be an adequate remedy, would occur to Parent and/or the Company in the event that any provision of this Lock-Up Agreement was not performed in accordance with its specific terms or was otherwise breached. It is accordingly agreed that Parent and the Company shall be entitled to an injunction or injunctions to prevent breaches of this Lock-Up Agreement and to enforce specifically the terms and provisions hereof in any court of the United States or any state having jurisdiction, this being in addition to any other remedy to which Parent or the Company is entitled at Law or in equity, and the undersigned waives any bond, surety or other security that might be required of Parent or the Company with respect thereto. Each of the parties further agrees that it will not oppose the granting of an injunction, specific performance or other equitable relief on the basis that any other party has an adequate remedy at law or that any award of specific performance is not an appropriate remedy for any reason at law or in equity.

7. In the event that any holder of securities of Parent that are subject to a substantially similar agreement entered into by such holder, other than the undersigned, is permitted by Parent to sell or otherwise transfer or dispose of shares of Parent Common Stock, Parent Convertible Preferred Stock or any securities convertible into or exercisable or exchangeable for Parent Common Stock for value other than as permitted by this or a substantially similar agreement entered into by such holder (whether in one or multiple releases or waivers), the same percentage of shares of Parent Common Stock, Parent Convertible Preferred Stock or any securities convertible into or exercisable or exchangeable for Parent Common Stock held by the undersigned on the date of such release or waiver as the percentage of the total number of outstanding shares of such securities held by such holder on the date of such release or waiver shall be immediately and fully released on the same terms from any remaining restrictions set forth herein (the "*Pro Rata Release*"); provided, however, that such Pro Rata Release shall not be applied unless and until permission has been granted by Parent to an equity holder or equity holders to sell or otherwise transfer or dispose of all or a portion of such equity holders shares of Parent Common Stock or ignally subject to a substantially similar agreement. In the event of any Pro-Rata Release, Parent shall promptly (and in any event within two (2) business days of such release) inform each relevant holder of Parent Common Stock or any securities convertible into or exercisable or exchangeable for Parent Common Stock of the terms of such Pro-Rata Release.

8. Upon the release of any of the Undersigned's Shares from this Lock-Up Agreement, Parent will cooperate with the undersigned to facilitate the timely preparation and delivery of certificates or the establishment of book-entry positions at Parent's transfer agent representing the Undersigned's Shares without the restrictive legend above or the withdrawal of any stop transfer instructions.

9. The undersigned understands that this Lock-Up Agreement is irrevocable and is binding upon the undersigned's heirs, legal representatives, successors and assigns.

10. This Lock-Up Agreement shall be governed by, and construed in accordance with, the Laws of the state of Delaware, regardless of the Laws that might otherwise govern under applicable principles of conflicts of Laws. In any action or Legal Proceeding between any of the parties arising out of or relating to this Lock-Up Agreement, each of the parties: (a) irrevocably and unconditionally consents and submits to the exclusive jurisdiction and venue of the Court of Chancery of the state of Delaware or to the extent such court does not have subject matter jurisdiction, the Superior Court of the State of Delaware or the United States District Court for the District of Delaware, (b) agrees that all claims in respect of such action or Legal Proceeding shall be heard and determined exclusively in accordance with clause (a) of this Section 10, (c) waives any objection to laying venue in any such action or Legal Proceeding in such courts, (d) waives any objection that such courts are an inconvenient forum or do not have jurisdiction over any party, and (e) agrees that service of process upon such party in any such action or Legal Proceeding shall be effective if notice is given in accordance with Section 11 of this Lock-Up Agreement. This Lock-Up Agreement constitutes the entire agreement between the parties to this Lock-Up Agreement and supersedes all other prior agreements, arrangements and understandings, both written and oral, among the parties with respect to the subject matter hereof.

THE PARTIES HERETO HEREBY WAIVE ANY RIGHT TO TRIAL BY JURY WITH RESPECT TO ANY ACTION OR LEGAL PROCEEDING RELATED TO OR ARISING OUT OF THIS LOCK-UP AGREEMENT, ANY DOCUMENT EXECUTED IN CONNECTION HEREWITH AND THE MATTERS CONTEMPLATED HEREBY AND THEREBY.

11. All notices and other communications hereunder shall be in writing and shall be deemed given if delivered personally or sent by overnight courier (providing proof of delivery), by electronic transmission (providing confirmation of transmission) to the Company or Parent, as the case may be, in accordance with Section 8.8 of the Merger Agreement and to the undersigned at his, her or its address or email address (providing confirmation of transmission) set forth on the signature page hereto (or at such other address for a party as shall be specified by like notice).

12. This Lock-Up Agreement may be executed in several counterparts, each of which shall be deemed an original and all of which shall constitute one and the same instrument. The exchange of a fully executed Lock-Up Agreement (in counterparts or otherwise) by Parent, the Company and the undersigned by electronic transmission in .pdf format shall be sufficient to bind such parties to the terms and conditions of this Lock-Up Agreement.

(Signature Page Follows)

Print Name of Stockholder:

Very truly yours, [] Signature (for individuals):

Email: Address:

Signature (for entities):

By: Name: Title: Email: Address:

SIGNATURE PAGE TO LOCK-UP AGREEMENT

Accepted and Agreed by ONCONOVA THERAPEUTICS, INC.:

By: Name: Title:

SIGNATURE PAGE TO LOCK-UP AGREEMENT

Accepted and Agreed by TRAWSFYNYDD THERAPEUTICS, INC.:

By: Name: Title:

SIGNATURE PAGE TO LOCK-UP AGREEMENT

EXHIBIT C

ONCONOVA THERAPEUTICS, INC.

CERTIFICATE OF DESIGNATION OF PREFERENCES, RIGHTS AND LIMITATIONS OF SERIES C NON-VOTING CONVERTIBLE PREFERRED STOCK

Pursuant to Section 151 of the General Corporation Law of the State of Delaware

THE UNDERSIGNED DOES HEREBY CERTIFY, on behalf of Onconova Therapeutics, Inc., a Delaware corporation (the "*Corporation*"), that the following resolution was duly adopted by the Board of Directors of the Corporation (the "*Board of Directors*"), in accordance with the provisions of Section 151 of the General Corporation Law of the State of Delaware (the "*DGCL*"), at a meeting duly called and held on April 1, 2024, which resolution provides for the creation of a series of the Corporation's Preferred Stock, par value \$0.01 per share, which is designated as "Series C Non-Voting Convertible Preferred Stock," with the preferences, rights and limitations set forth therein relating to dividends, conversion, redemption, dissolution and distribution of assets of the Corporation.

WHEREAS: the Tenth Amended and Restated Certificate of Incorporation of the Corporation, as amended (the "*Certificate of Incorporation*"), provides for a class of its authorized stock known as Preferred Stock, consisting of 5,000,000 shares, \$0.01 par value per share (the "*Preferred Stock*"), issuable from time to time in one or more series.

RESOLVED: that, pursuant to authority conferred upon the Board of Directors by the Certificate of Incorporation, (i) a series of Preferred Stock of the Corporation be, and hereby is authorized by the Board of Directors, (ii) the Board of Directors hereby authorizes the issuance of 12,472.76870 shares of "Series C Non-Voting Convertible Preferred Stock" pursuant to the terms of (a) the Securities Purchase Agreement, dated as of the date hereof, by and among the Corporation and the initial Holders (as defined below) (the "*Purchase Agreement*") and (b) the Agreement and Plan of Merger, dated as of the date hereof, by and among the Corporation, Traws Merger Sub I, Inc., a Delaware corporation and wholly owned subsidiary of the Corporation, Traws Merger Sub II, LLC, a Delaware limited liability company and wholly owned subsidiary of the Corporation and Trawsfynydd Therapeutics, Inc., a Delaware corporation (the "*Merger Agreement*"), and (iii) the Board of Directors hereby fixes the designations, powers, preferences and relative, participating, optional or other special rights, and the qualifications, limitations or restrictions thereof, of such shares of Preferred Stock, in addition to any provisions set forth in the Certificate of Incorporation that are applicable to the Preferred Stock of all classes and series, as follows:

TERMS OF SERIES C NON-VOTING CONVERTIBLE PREFERRED STOCK

1. <u>Definitions</u>. For the purposes hereof, the following terms shall have the following meanings:

"Business Day" means any day other than a Saturday, Sunday or other day on which banks in Philadelphia, PA, are authorized or obligated by Law to be closed.

"Buy-In" shall have the meaning set forth in Section 6.5.4.

"Closing Sale Price" means, for any security as of any date, the last closing trade price for such security immediately prior to 4:00 p.m., New York City time, on the principal Trading Market where such security is listed or traded, as reported by Bloomberg, L.P. (or an equivalent, reliable reporting service), or if the foregoing do not apply, the last trade price of such security in the over-the-counter market on the electronic bulletin board for such security as reported by Bloomberg, L.P., or, if no last trade price is reported for such security by Bloomberg, L.P., the average of the bid prices of any market makers for such security as reported on the OTC Pink Market by OTC Markets Group, Inc. If the Closing Sale Price cannot be calculated for a security on a particular date on any of the foregoing bases, the Closing Sale Price of such security on such date shall be the fair market value as determined in good faith by the Board of Directors of the Corporation. "Commission" means the United States Securities and Exchange Commission.

"*Common Stock*" means the Corporation's common stock, par value \$0.01 per share, and stock of any other class of securities into which such securities may hereafter be reclassified or changed.

"Conversion Shares" means, collectively, the shares of Common Stock issuable upon conversion of the shares of Series C Non-Voting Preferred Stock in accordance with the terms hereof.

"Exchange Act" means the Securities Exchange Act of 1934.

"Holder" means a holder of shares of Series C Non-Voting Preferred Stock.

"Person" means an individual or corporation, partnership, trust, incorporated or unincorporated association, joint venture, limited liability company, joint stock company, government (or an agency or subdivision thereof) or other entity of any kind.

"Trading Day" means a day on which the principal Trading Market is open for business.

"Trading Market" means any of the following markets or exchanges on which the Common Stock is listed or quoted for trading on the date in question: the NYSE American, the Nasdaq Capital Market, the Nasdaq Global Market, the Nasdaq Global Select Market, or the New York Stock Exchange (or any successors to any of the foregoing).

2. <u>Designation, Amount and Par Value</u>. The series of Preferred Stock shall be designated as the Corporation's Series C Non-Voting Convertible Preferred Stock (the "*Series C Non-Voting Preferred Stock*") and the number of shares so designated shall be 20,000. Each share of Series C Non-Voting Preferred Stock shall have a par value of \$0.01 per share.

3. <u>Dividends</u>. Holders shall be entitled to receive, and the Corporation shall pay, dividends on shares of the Series C Non-Voting Preferred Stock (on an as-if-converted-to-Common-Stock basis, without regard to the Beneficial Ownership Limitation (as defined below)) equal to and in the same form, and in the same manner, as dividends (other than dividends on shares of the Common Stock payable in the form of Common Stock) actually paid on shares of the Common Stock; provided, however, in no event shall Holders of Series C Non-Voting Preferred Stock be entitled to receive the "rights" distributed pursuant to that certain Contingent Value Rights Agreement dated as of April 1, 2024 by and between the Corporation and Equiniti Trust Company, LLC, a New York limited liability trust company, as may be amended from time to time (the "*CVR Agreement*"), or any amounts paid under the CVR Agreement. Other than as set forth in the previous sentence, no other dividends shall be paid on shares of Series C Non-Voting Preferred Stock, and the Corporation shall pay no dividends (other than dividends payable in the form of Series C Non-Voting Preferred Stock, and the Corporation shall pay no dividends (other than dividends payable in the form of Common Stock) are paid under the CVR and the previous sentence.

4. Voting Rights.

4.1 Except as otherwise provided herein or as otherwise required by the DGCL, the Series C Non-Voting Preferred Stock shall have no voting rights. However, as long as any shares of Series C Non-Voting Preferred Stock are outstanding, the Corporation shall not, without the affirmative vote of the holders of a majority of the then outstanding shares of the Series C Non-Voting Preferred Stock: (i) alter or change adversely the powers, preferences or rights given to the Series C Non-Voting Preferred Stock or alter or amend this Certificate of Designation, amend or repeal any provision of, or add any provision to, the Certificate of Incorporation or Amended and Restated Bylaws of the Corporation, or file any articles of amendment, certificate of designations, preferences, limitations and relative rights of any series of Preferred Stock, if such action would adversely alter or change the preferences, rights, privileges or powers of, or restrictions provided for the benefit of the Series C Non-Voting Preferred Stock, regardless of whether any of the foregoing actions shall be by means of amendment to the Certificate of Incorporation or by merger, consolidation, recapitalization, reclassification, conversion or otherwise, (ii) issue further shares of Series C Non-Voting Preferred Stock or increase or decrease (other than by conversion) the number of authorized shares of Series C Non-Voting Preferred Stock, (iii) prior to the Stockholder Approval (as defined below) or at any time while at least 30% of the originally issued Series C Non-Voting Preferred Stock remains issued and outstanding, consummate either: (A) any Fundamental Transaction (as defined below) or (B) any merger or consolidation of the Corporation with or into another entity or any stock sale to, or other business combination in which the stockholders of the Corporation immediately before such transaction do not hold at least a majority of the capital stock of the Corporation immediately after such transaction or (iv) enter into any agreement with respect to any of the foregoing. Holders of shares of Common Stock acquired upon the conversion of shares of Series C Non-Voting Preferred Stock shall be entitled to the same voting rights as each other holder of Common Stock, except that such holders may not vote such shares upon the proposal for Stockholder Approval in accordance with Rule 5635 of the listing rules of The Nasdaq Stock Market LLC.

4.2 Any vote required or permitted under <u>Section 4.1</u> may be taken at a meeting of the Holders or through the execution of an action by written consent in lieu of such meeting, provided that the consent is executed by Holders representing a majority of the outstanding shares of Series C Non-Voting Preferred Stock.

5. Rank; Liquidation.

5.1 The Series C Non-Voting Preferred Stock shall rank on parity with the Common Stock as to distributions of assets upon liquidation, dissolution or winding up of the Corporation, whether voluntarily or involuntarily.

5.2 Upon any liquidation, dissolution or winding-up of the Corporation, whether voluntary or involuntary (a "*Liquidation*"), each Holder shall be entitled to receive out of the assets, whether capital or surplus, of the Corporation the same amount that a holder of Common Stock would receive if the Series C Non-Voting Preferred Stock were fully converted (disregarding for such purpose any Beneficial Ownership Limitations) to Common Stock which amounts shall be paid *pari passu* with all holders of Common Stock, plus an additional amount equal to any dividends declared on but unpaid to such shares. If, upon any such Liquidation, the assets of the Corporation shall be insufficient to pay the Holders of shares of the Series C Non-Voting Preferred Stock the amount required under the preceding sentence, then all remaining assets of the Corporation shall be distributed ratably to the Holders and the holders of Common Stock in accordance with the respective amounts that would be payable on all such securities if all amounts payable thereon were paid in full. For the avoidance of any doubt, a Fundamental Transaction shall not be deemed a Liquidation unless the Corporation expressly declares that such Fundamental Transaction shall be treated as if it were a Liquidation.

6. <u>Conversion</u>.

6.1 <u>Automatic Conversion on Stockholder Approval</u>. Effective as of 5:00 p.m. Eastern time on the third Business Day after the date that the Corporation's stockholders approve the conversion of the Series C Non-Voting Preferred Stock into shares of Common Stock in accordance with the listing rules of the Nasdaq Stock Market, as set forth in Section 4.1 of the Merger Agreement (the "*Stockholder Approval*"), each share of Series C Non-Voting Preferred Stock then outstanding shall automatically convert into a number of shares of Common Stock equal to the Conversion Ratio (as defined below), subject to the Beneficial Ownership Limitation (the "*Automatic Conversion*"). In determining the application of the Beneficial Ownership Limitations solely with respect to the Automatic Conversion, the Corporation shall calculate beneficial ownership for each Holder assuming beneficial ownership by such Holder of: (x) the number of shares of Common Stock issuable to such Holder in such Automatic Conversion, plus (y) any additional shares of Common Stock for which a Holder has provided the Corporation with prior written notice of beneficial ownership within 30 days prior to the date of Stockholder Approval (a "*Beneficial Ownership Statement*") and assuming the conversion of all shares of Series C Non-Voting Preferred Stock held by all other Holders that will not convert into shares of Common Stock on account of the application of any Beneficial Ownership Limitations applicable to any such other Holders. If a Holder fails to provide the Corporation with a Beneficial Ownership Statement within 30 days prior to the date of Stockholder Approval, then the Corporation shall presume the Holder's beneficial ownership of Common Stock (excluding the Conversion Shares) to be zero. The shares of Series C Non-Voting Preferred Stock that are converted in the Automatic Conversion are referred to as the "*Converted Stock*". The Conversion Shares shall be issued as follows:

6.1.1 Converted Stock that is registered in book entry form shall be automatically cancelled upon the Automatic Conversion and converted into the corresponding Conversion Shares, which shares shall be issued in book entry form and without any action on the part of the Holders.

6.1.2 Converted Stock that is issued in certificated form shall be deemed converted into the corresponding Conversion Shares on the date of Automatic Conversion and the Holder's rights as a holder of such shares of Converted Stock shall cease and terminate on such date, excepting only the right to receive the Conversion Shares upon the Holder tendering to the Corporation (or its designated agent) the stock certificate(s) (duly endorsed) representing such certificated Converted Stock.

6.1.3 Notwithstanding the cancellation of the Converted Stock upon the Automatic Conversion, Holders of Converted Stock shall continue to have any remedies provided herein or otherwise available at law or in equity to such Holder because of a failure by the Corporation to comply with the terms of this Certificate of Designation. In all cases, the Holder shall retain all of its rights and remedies for the Corporation's failure to convert the Converted Stock.

6.2 Conversion at Option of Holder. Subject to Section 6.1, Section 6.4 and Section 6.5.3, each share of Series C Non-Voting Preferred Stock then outstanding shall be convertible, at any time and from time to time following 5:00 p.m. Eastern time on the third Business Day after the date that the Stockholder Approval is obtained by the Corporation, at the option of the Holder thereof, into a number of shares of Common Stock equal to the Conversion Ratio, subject to the Beneficial Ownership Limitation (each, an "Optional Conversion"). Holders shall effect conversions by providing the Corporation with the form of conversion notice attached hereto as <u>Annex A</u> (a "Notice of Conversion"), duly completed and executed. Provided the Corporation's transfer agent is participating in the Depository Trust Company ("DTC") Fast Automated Securities Transfer program, the Notice of Conversion may specify, at the Holder's election, whether the applicable Conversion Shares shall be credited to the account of the Holder's prime broker with DTC through its Deposit Withdrawal Agent Commission system (a "DWAC Delivery"). The date on which an Optional Conversion shall be deemed effective (the "Conversion Date") shall be the Trading Day that the Notice of Conversion, completed and executed, is sent via email to, and received during regular business hours by, the Corporation; provided, that the original certificate(s) (if any) representing such shares of Series C Non-Voting Preferred Stock being converted, duly endorsed, and the accompanying Notice of Conversion, are received by the Corporation within two (2) Trading Days thereafter. In all other cases, the Conversion Date shall be defined as the Trading Day on which the original certificate(s) (if any) representing such shares of Series C Non-Voting Preferred Stock being converted, duly endorsed, and the accompanying Notice of Conversion, are received by the Corporation. The calculations set forth in the Notice of Conversion shall control in the absence of manifest or mathematical err

6.3 <u>Conversion Ratio</u>. The "*Conversion Ratio*" for each share of Series C Non-Voting Preferred Stock shall be 10,000 shares of Common Stock issuable upon the conversion (the "*Conversion*") of each share of Series C Non-Voting Preferred Stock, subject to adjustment as provided herein.

Beneficial Ownership Limitation. Notwithstanding anything herein to the contrary, the Corporation shall not effect any conversion 6.4 of any share of Series C Non-Voting Preferred Stock, including pursuant to Section 6.1, and a Holder shall not have the right to convert any portion of the Series C Non-Voting Preferred Stock pursuant to Section 6.2, to the extent that, after giving effect to such attempted conversion set forth on an applicable Notice of Conversion (as defined in the Certificate of Designation) with respect to the Series C Preferred Stock, such Holder (or any of such Holder's affiliates or any other Person who would be a beneficial owner of Common Stock beneficially owned by the Holder for purposes of Section 13(d) or Section 16 of the Exchange Act and the applicable rules and regulations of the Commission, including any "group" of which the Holder is a member (the foregoing, "Attribution Parties")) would beneficially own a number of shares of Common Stock in excess of the Beneficial Ownership Limitation. For purposes of the foregoing sentence, the aggregate number of shares of Common Stock beneficially owned by such Holder and its Attribution Parties shall include the number of shares of Common Stock issuable upon conversion of the Series C Non-Voting Preferred Stock subject to the Notice of Conversion or the Automatic Conversion, as applicable, with respect to which such determination is being made, but shall exclude the number of shares of Common Stock which are issuable upon (A) conversion of the remaining, unconverted Series C Non-Voting Preferred Stock beneficially owned by such Holder or any of its Attribution Parties, and (B) exercise or conversion of the unexercised or unconverted portion of any other securities of the Corporation (including any warrants) beneficially owned by such Holder or any of its Attribution Parties that are subject to and would exceed a limitation on conversion or exercise similar to the limitation contained herein. Except as set forth in the preceding sentence, for purposes of this Section 6.4, beneficial ownership shall be calculated in accordance with Section 13(d) of the Exchange Act and the applicable rules and regulations of the Commission, and the terms "beneficial ownership" and "beneficially own" have the meanings ascribed to such terms therein. In addition, for purposes hereof, "group" has the meaning set forth in Section 13(d) of the Exchange Act and the applicable rules and regulations of the Commission. For purposes of this Section 6.4, in determining the number of outstanding shares of Common Stock, a Holder may rely on the number of outstanding shares of Common Stock as stated in the most recent of the following: (A) the Corporation's most recent periodic or annual filing with the Commission, as the case may be, (B) a more recent public announcement by the Corporation that is filed with the Commission, or (C) a more recent notice by the Corporation or the Corporation's transfer agent to the Holder setting forth the number of shares of Common Stock then outstanding. Upon the written request of a Holder (which may be by email), the Corporation shall, within two (2) Trading Days thereof, confirm in writing to such Holder (which may be via email) the number of shares of Common Stock then outstanding. In any case, the number of outstanding shares of Common Stock shall be determined after giving effect to any actual conversion or exercise of securities of the Corporation, including shares of Series C Non-Voting Preferred Stock, by such Holder or its Attribution Parties since the date as of which such number of outstanding shares of Common Stock was last publicly reported or confirmed to the Holder. The "Beneficial Ownership Limitation" shall initially be set at 19.9% for each Holder and its Attribution Parties and may be adjusted at the discretion of the Holder to a percentage between 4.9% and 19.9% of the number of shares of Common Stock outstanding immediately after giving effect to the issuance of shares of Common Stock pursuant to the Automatic Conversion or such Notice of Conversion (as applicable), to the extent permitted by this Section 6.3. The Corporation shall be entitled to rely on representations made to it by the Holder in any Notice of Conversion regarding its Beneficial Ownership Limitation. Notwithstanding the foregoing, by written notice to the Corporation, (i) which will not be effective until the sixty-first (61st) day after such written notice is delivered to the Corporation, the Holder may reset the Beneficial Ownership Limitation percentage to a higher percentage, not to exceed 19.9%, to the extent then applicable and (ii) which will be effective immediately after such notice is delivered to the Corporation, the Holder may reset the Beneficial Ownership Limitation percentage to a lower percentage (but in no event less than 4.9%) provided that such decrease shall not become effective until the later of (x) 5:00 p.m. Eastern time on the third Business Day after the date of the Stockholder Approval and (y) if Stockholder Approval is not obtained within six months after the initial issuance of the Series C Non-Voting Preferred Stock, the date that is three Business Days after the date that is six months after the initial issuance of the Series C Non-Voting Preferred Stock. Upon such a change by a Holder of the Beneficial Ownership Limitation, not to exceed 19.9%, the Beneficial Ownership Limitation may not be further amended by such Holder without first providing the minimum notice required by this Section 6.4. Notwithstanding the foregoing, at any time following notice of a Fundamental Transaction, the Holder may waive and/or change the Beneficial Ownership Limitation effective immediately upon written notice to the Corporation and may reinstitute a Beneficial Ownership Limitation at any time thereafter effective immediately upon written notice to the Corporation. The provisions of this Section 6.4 shall be construed, corrected and implemented in a manner so as to effectuate the intended Beneficial Ownership Limitation herein contained and the shares of Common Stock underlying the securities in excess of the Beneficial Ownership Limitation shall not be deemed to be beneficially owned by the Holder for any purpose including for purposes of Section 13(d) or Rule 16a-1(a) (1) of the Exchange Act.

6.5 Mechanics of Conversion.

6.5.1 <u>Delivery of Certificate or Electronic Issuance</u>. Upon Conversion not later than two (2) Trading Days after the applicable Conversion Date, or if the Holder requests the issuance of physical certificate(s), two (2) Trading Days after receipt by the Corporation of the original certificate(s) representing such shares of Series C Non-Voting Preferred Stock being converted, duly endorsed, and the accompanying Notice of Conversion (the "*Share Delivery Date*"), the Corporation shall either: (a) deliver, or cause to be delivered, to the converting Holder a physical certificate or certificates representing the number of Conversion Shares being acquired upon the conversion of shares of Series C Non-Voting Preferred Stock, or (b) in the case of a DWAC Delivery (if so requested by the Holder), electronically transfer such Conversion Shares by crediting the account of the Holder's prime broker with DTC through its DWAC system. If in the case of any Notice of Conversion such certificate or certificates for the Conversion Shares are not delivered to or as directed by or, in the case of a DWAC Delivery, such shares are not electronically delivered to or as directed by, the applicable Holder by the Share Delivery Date, the applicable Holder shall be entitled to elect to rescind such Notice of Conversion by written notice to the Corporation at any time on or before its receipt of such certificate or certificates for Conversion Shares or electronic receipt of such shares, as applicable, in which event the Corporation shall promptly return to such Holder any original Series C Non-Voting Preferred Stock certificate delivered to the Corporation and such Holder shall promptly return to the Corporation any Common Stock certificates or otherwise direct the return of any shares of Common Stock delivered to the Holder through the DWAC system, representing the shares of Series C Non-Voting Preferred Stock unsuccessfully tendered for conversion to the Corporation.

6.5.2 Obligation Absolute. Subject to Section 6.4 and subject to Holder's right to rescind a Notice of Conversion pursuant to Section 6.5.1, the Corporation's obligation to issue and deliver the Conversion Shares upon conversion of Series C Non-Voting Preferred Stock in accordance with the terms hereof are absolute and unconditional, irrespective of any action or inaction by a Holder to enforce the same, any waiver or consent with respect to any provision hereof, the recovery of any judgment against any Person or any action to enforce the same, or any setoff, counterclaim, recoupment, limitation or termination, or any breach or alleged breach by such Holder or any other Person of any obligation to the Corporation or any violation or alleged violation of law by such Holder or any other Person, and irrespective of any other circumstance which might otherwise limit such obligation of the Corporation to such Holder in connection with the issuance of such Conversion Shares. Subject to Section 6.4 and subject to Holder's right to rescind a Notice of Conversion pursuant to Section 6.5.1, in the event a Holder shall elect to convert any or all of its Series C Non-Voting Preferred Stock, the Corporation may not refuse conversion based on any claim that such Holder or anyone associated or affiliated with such Holder has been engaged in any violation of law, agreement or for any other reason, unless an injunction from a court, on notice to Holder, restraining and/or enjoining conversion of all or part of the Series C Non-Voting Preferred Stock of such Holder shall have been sought and obtained by the Corporation, and the Corporation posts a surety bond for the benefit of such Holder in the amount of 150% of the value of the Conversion Shares into which would be converted the Series C Non-Voting Preferred Stock which is subject to such injunction, which bond shall remain in effect until the completion of arbitration/litigation of the underlying dispute and the proceeds of which shall be payable to such Holder to the extent it obtains judgment. In the absence of such injunction, the Corporation shall, subject to Section 6.4 and subject to Holder's right to rescind a Notice of Conversion pursuant to Section 6.5.1, issue Conversion Shares upon a properly noticed conversion.

6.5.3 Cash Settlement. If, at any time after the earlier of Stockholder Approval or six months after the initial issuance of the Series C Non-Voting Preferred Stock, the Corporation fails to deliver to a Holder such certificate or certificates, or electronically deliver (or cause its transfer agent to electronically deliver) such shares in the case of a DWAC Delivery, pursuant to Section 6.5.1 on or prior to the third (3rd) Trading Day after the Share Delivery Date applicable to such conversion (other than a failure caused by (i) materially incorrect or incomplete information provided by Holder to the Corporation or (ii) the application of the Beneficial Ownership Limitation), then, unless the Holder has rescinded the applicable Notice of Conversion pursuant to Section 6.5.1, the Corporation shall, at the request of the Holder, pay an amount equal to the Fair Value (as defined below) of such undelivered shares, with such payment to be made within two Business Days from the date of request by the Holder, whereupon the Corporation's obligations to deliver such shares underlying the Notice of Conversion shall be extinguished upon payment in full of the Fair Value of such undelivered shares; provided, however that such request shall be presumed to have been made by such Holder if Stockholder Approval shall not have been obtained prior to the date on which the Notice of Conversion is delivered to the Corporation. For purposes of this Section 6.5.3, the "Fair Value" of shares shall be fixed with reference to the last reported Closing Sale Price on the principal Trading Market on which the Common Stock is listed as of the Trading Day immediately prior to, in the case of the Automatic Conversion, the date of the Stockholder Approval, and in the case of an Optional Conversion, the Conversion Date. For the avoidance of doubt, the cash settlement provisions set forth in this Section 6.5.3 shall be available irrespective of the reason for the Corporation's failure to timely deliver Conversion Shares (other than a failure caused by (i) materially incorrect or incomplete information provided by Holder to the Corporation or (ii) the application of the Beneficial Ownership Limitation), including due to limitations set forth in Section 6.5.6, the lack of obtaining Stockholder Approval, or due to applicable Trading Market rules.

Buy-In on Failure to Timely Deliver Certificates. If the Corporation fails to deliver to a Holder the applicable certificate 6.5.4 or certificates or to effect a DWAC Delivery, as applicable, by the Share Delivery Date pursuant to Section 6.5.1 (other than a failure caused by materially incorrect or incomplete information provided by Holder to the Corporation or the application of the Beneficial Ownership Limitation), and if after such Share Delivery Date such Holder is required by its brokerage firm to purchase (in an open market transaction or otherwise), or the Holder's brokerage firm otherwise purchases, shares of Common Stock to deliver in satisfaction of a sale by such Holder of the Conversion Shares which such Holder was entitled to receive upon the conversion relating to such Share Delivery Date (a "Buy-In"), then the Corporation shall (A) pay in cash to such Holder (in addition to any other remedies available to or elected by such Holder) the amount by which (x) such Holder's total purchase price (including any brokerage commissions) for the shares of Common Stock so purchased exceeds (y) the product of (1) the aggregate number of shares of Common Stock that such Holder was entitled to receive from the conversion at issue multiplied by (2) the actual sale price at which the sell order giving rise to such purchase obligation was executed (including any brokerage commissions) and (B) at the option of such Holder, either reissue (if surrendered) the shares of Series C Non-Voting Preferred Stock equal to the number of shares of Series C Non-Voting Preferred Stock submitted for conversion or deliver to such Holder the number of shares of Common Stock that would have been issued if the Corporation had timely complied with its delivery requirements under Section 6.5.1. For example, if a Holder purchases shares of Common Stock having a total purchase price of \$11,000 to cover a Buy-In with respect to an attempted conversion of shares of Series C Non-Voting Preferred Stock with respect to which the actual sale price (including any brokerage commissions) giving rise to such purchase obligation was a total of \$10,000 under clause (A) of the immediately preceding sentence, the Corporation shall be required to pay such Holder \$1,000. The Holder shall provide the Corporation written notice, within three (3) Trading Days after the occurrence of a Buy-In, indicating the amounts payable to such Holder in respect of such Buy-In together with applicable confirmations and other evidence reasonably requested by the Corporation. Nothing herein shall limit a Holder's right to pursue any other remedies available to it hereunder, at law or in equity including, without limitation, a decree of specific performance and/or injunctive relief with respect to the Corporation's failure to timely deliver certificates representing shares of Common Stock upon conversion of the shares of Series C Non-Voting Preferred Stock as required pursuant to the terms hereof or the cash settlement remedy set forth in Section 6.5.3; provided, however, that the Holder shall not be entitled to both (i) require the reissuance of the shares of Series C Non-Voting Preferred Stock submitted for conversion for which such conversion was not timely honored and (ii) receive the number of shares of Common Stock that would have been issued if the Corporation had timely complied with its delivery requirements under Section 6.5.1.

6.5.5 <u>Reservation of Shares Issuable Upon Conversion</u>. The Corporation covenants that at all times it will reserve and keep available out of its authorized and unissued shares of Common Stock for the sole purpose of issuance upon conversion of the Series C Non-Voting Preferred Stock, free from preemptive rights or any other actual contingent purchase rights of Persons other than the Holders of the Series C Non-Voting Preferred Stock, not less than such aggregate number of shares of the Common Stock as shall be issuable (taking into account the adjustments of <u>Section 7</u>) upon the conversion of all outstanding shares of Series C Non-Voting Preferred Stock. The Corporation covenants that all shares of Common Stock that shall be so issuable shall, upon issue, be duly authorized, validly issued, fully paid and non-assessable.

6.5.6 <u>Fractional Shares</u>. No fractional shares of Common Stock shall be issued upon conversion of the Series C Non-Voting Preferred Stock, no certificates or scrip for any such fractional shares shall be issued and no cash shall be paid for any such fractional shares. Any fractional shares of Common Stock that a Holder of Series C Non-Voting Preferred Stock would otherwise be entitled to receive shall be aggregated with all fractional shares of Common Stock issuable to such Holder and any remaining fractional shares shall be rounded up to the nearest whole share. Whether or not fractional shares would be issuable upon such conversion shall be determined on the basis of the total number of shares of Series C Non-Voting Preferred Stock and the aggregate number of shares of Common Stock issuable upon such conversion.

6.5.7 <u>Transfer Taxes</u>. The issuance of certificates for shares of the Common Stock upon conversion of the Series C Non-Voting Preferred Stock shall be made without charge to any Holder for any documentary stamp or similar taxes that may be payable in respect of the issue or delivery of such certificates, provided that the Corporation shall not be required to pay any tax that may be payable in respect of any transfer involved in the issuance and delivery of any such certificate upon conversion in a name other than that of the registered Holder(s) of such shares of Series C Non-Voting Preferred Stock and the Corporation shall not be required to issue or deliver such certificates unless or until the Person or Persons requesting the issuance thereof shall have paid to the Corporation the amount of such tax or shall have established to the satisfaction of the Corporation that such tax has been paid.

6.6 <u>Status as Stockholder</u>. Upon each Conversion Date, (i) the shares of Series C Non-Voting Preferred Stock being converted shall be deemed converted into shares of Common Stock and (ii) the Holder's rights as a holder of such converted shares of Series C Non-Voting Preferred Stock shall cease and terminate, excepting only the right to receive certificates for such shares of Common Stock and to any remedies provided herein or otherwise available at law or in equity to such Holder because of a failure by the Corporation to comply with the terms of this Certificate of Designation. In all cases, the Holder shall retain all of its rights and remedies for the Corporation's failure to convert Series C Non-Voting Preferred Stock. In no event shall the Series C Non-Voting Preferred Stock convert into shares of Common Stock prior to the Stockholder Approval.

7. Certain Adjustments.

7.1 Stock Dividends and Stock Splits. If the Corporation, at any time while this Series C Non-Voting Preferred Stock is outstanding: (A) pays a stock dividend or otherwise makes a distribution or distributions payable in shares of Common Stock (which, for avoidance of doubt, shall not include any shares of Common Stock issued by the Corporation upon conversion of this Series C Non-Voting Preferred Stock) with respect to the then outstanding shares of Common Stock; (B) subdivides outstanding shares of Common Stock into a larger number of shares; or (C) combines (including by way of a reverse stock split) outstanding shares of Common Stock into a smaller number of shares, then the Conversion Ratio shall be multiplied by a fraction of which the numerator shall be the number of shares of Common Stock (excluding any treasury shares of the Corporation) outstanding immediately after such event and of which the denominator shall be the number of shares of Common Stock outstanding immediately before such event (excluding any treasury shares of the Corporation). Any adjustment made pursuant to this <u>Section 7.1</u> shall become effective immediately after the effective date in the case of a subdivision or combination.

7.2 <u>Fundamental Transaction</u>. If, at any time while this Series C Non-Voting Preferred Stock is outstanding, (A) the Corporation effects any merger or consolidation of the Corporation with or into another Person or any stock sale to, or other business combination (including, without limitation, a reorganization, recapitalization, spin-off, share exchange or scheme of arrangement) with or into another Person (other than such a transaction in which the Corporation is the surviving or continuing entity and its Common Stock is not exchanged for or converted into other securities, cash or property), (B) the Corporation effects any sale, lease, transfer or exclusive license of all or substantially all of its assets in one transaction or a series of related transactions, (C) any tender offer or exchange offer (whether by the Corporation or another Person) is completed pursuant to which more than 50% of the Common Stock not held by the Corporation or such Person is exchanged for or converted into other securities, cash or property, or (D) the Corporation effects any reclassification of the Common Stock or any compulsory share exchange pursuant (other than as a result of a dividend, subdivision or combination covered by Section 7.1) to which the Common Stock is effectively converted into or exchanged for other securities, cash or property (in any such case, a "Fundamental Transaction"), then, upon any subsequent conversion of this Series C Non-Voting Preferred Stock the Holders shall have the right to receive, in lieu of the right to receive Conversion Shares, for each Conversion Share that would have been issuable upon such conversion immediately prior to the occurrence of such Fundamental Transaction, the same kind and amount of securities, cash or property as it would have been entitled to receive upon the occurrence of such Fundamental Transaction if it had been, immediately prior to such Fundamental Transaction, the holder of one share of Common Stock (the "Alternate Consideration"). For purposes of any such subsequent conversion, the determination of the Conversion Ratio shall be appropriately adjusted to apply to such Alternate Consideration based on the amount of Alternate Consideration issuable in respect of one share of Common Stock in such Fundamental Transaction, and the Corporation shall adjust the Conversion Ratio in a reasonable manner reflecting the relative value of any different components of the Alternate Consideration. If holders of Common Stock are given any choice as to the securities, cash or property to be received in a Fundamental Transaction, then the Holders shall be given the same choice as to the Alternate Consideration it receives upon any conversion of this Series C Non-Voting Preferred Stock following such Fundamental Transaction. To the extent necessary to effectuate the foregoing provisions, any successor to the Corporation or surviving entity in such Fundamental Transaction shall file a new certificate of designations with the same terms and conditions and issue to the Holders new preferred stock consistent with the foregoing provisions and evidencing the Holders' right to convert such preferred stock into Alternate Consideration. The terms of any agreement to which the Corporation is a party and pursuant to which a Fundamental Transaction is effected shall include terms requiring any such successor or surviving entity to comply with the provisions of this Section 7.2 and insuring that this Series C Non-Voting Preferred Stock (or any such replacement security) will be similarly adjusted upon any subsequent transaction analogous to a Fundamental Transaction. The Corporation shall cause to be delivered to each Holder, at its last address as it shall appear upon the stock books of the Corporation, written notice of any Fundamental Transaction at least 20 calendar days prior to the date on which such Fundamental Transaction is expected to become effective or close. Notwithstanding anything to the contrary herein, the Corporation's disposition of certain assets pursuant to the CVR Agreement shall not constitute a Fundamental Transaction.

7.3 <u>Calculations</u>. All calculations under this <u>Section 7</u> shall be made to the nearest cent or the nearest 1/100th of a share, as the case may be. For purposes of this <u>Section 7</u>, the number of shares of Common Stock deemed to be issued and outstanding as of a given date shall be the sum of the number of shares of Common Stock (excluding any treasury shares of the Corporation) issued and outstanding.

8. <u>Redemption</u>. The shares of Series C Non-Voting Preferred Stock shall not be redeemable; provided, however, that the foregoing shall not limit the ability of the Corporation to purchase or otherwise deal in such shares to the extent otherwise permitted hereby and by law, nor shall the foregoing limit the Holder's rights under <u>Section 6.5.3</u>.

9. <u>Transfer</u>. A Holder may transfer any shares of Series C Non-Voting Preferred Stock together with the accompanying rights set forth herein, held by such holder without the consent of the Corporation; provided that such transfer is in compliance with applicable securities laws. The Corporation shall in good faith (a) do and perform, or cause to be done and performed, all such further acts and things, and (b) execute and deliver all such other agreements, certificates, instruments and documents, in each case, as any holder of Series C Non-Voting Preferred Stock may reasonably request in order to carry out the intent and accomplish the purposes of this <u>Section 9</u>. The transferee of any shares of Series C Non-Voting Preferred Stock shall be subject to the Beneficial Ownership Limitation applicable to the transferor as of the time of such transfer.

10. <u>Series C Non-Voting Preferred Stock Register</u>. The Corporation shall maintain at its principal executive offices (or such other office or agency of the Corporation as it may designate by notice to the Holders in accordance with <u>Section 11</u>), a register for the Series C Non-Voting Preferred Stock, in which the Corporation shall record (a) the name, address, and electronic mail address of each holder in whose name the shares of Series C Non-Voting Preferred Stock. The Corporation may deem and treat the registered Holder of shares of Series C Non-Voting Preferred Stock as the absolute owner thereof for the purpose of any conversion thereof and for all other purposes. The Corporation shall keep the register open and available at all times during business hours for inspection by any holder of Series C Non-Voting Preferred Stock or his, her or its legal representatives.

11. <u>Notices</u>. Any notice required or permitted by the provisions of this Certificate of Designation to be given to a Holder of shares of Series C Non-Voting Preferred Stock shall be mailed, postage prepaid, to the post office address last shown on the records of the Corporation, or given by electronic communication in compliance with the provisions of the Delaware General Corporation Law, and shall be deemed sent upon such mailing or electronic transmission.

12. <u>Book-Entry; Certificates</u>. The Series C Non-Voting Preferred Stock will be issued in book-entry form; provided that, if a Holder requests that such Holder's shares of Series C Non-Voting Preferred Stock be issued in certificated form, the Corporation will instead issue a stock certificate to such Holder representing such Holder's shares of Series C Non-Voting Preferred Stock. To the extent that any shares of Series C Non-Voting Preferred Stock are issued in book-entry form, references herein to "certificates" shall instead refer to the book-entry notation relating to such shares.

13. <u>Waiver</u>. Any waiver by the Corporation or a Holder of a breach of any provision of this Certificate of Designation shall not operate as or be construed to be a waiver of any other breach of such provision or of any breach of any other provision of this Certificate of Designation or a waiver by any other Holders, other than as expressly set forth herein. The failure of the Corporation or a Holder to insist upon strict adherence to any term of this Certificate of Designation on one or more occasions shall not be considered a waiver or deprive that party (or any other Holder) of the right thereafter to insist upon strict adherence to that term or any other term of this Certificate of Designation. Any waiver by the Corporation or a Holder must be in writing. Notwithstanding any provision in this Certificate of Designation to the contrary, any provision contained herein and any right of the Holders of Series C Non-Voting Preferred Stock granted hereunder may be waived as to all shares of Series C Non-Voting Preferred Stock (and the Holders thereof) upon the written consent of the Holders of not less than a majority of the shares of Series C Non-Voting Preferred Stock then outstanding, provided, however, that the Beneficial Ownership Limitation applicable to a Holder, and any provisions contained herein that are related to such Beneficial Ownership Limitation, cannot be modified, waived or terminated without the consent of such Holder, provided further, that any proposed waiver that would, by its terms, have a disproportionate and materially adverse effect on any Holder shall require the consent of such Holder(s).

14. <u>Severability</u>. Whenever possible, each provision hereof shall be interpreted in a manner as to be effective and valid under applicable law, but if any provision hereof is held to be prohibited by or invalid under applicable law, then such provision shall be ineffective only to the extent of such prohibition or invalidity, without invalidating or otherwise adversely affecting the remaining provisions hereof.

15. <u>Status of Converted Series C Non-Voting Preferred Stock</u>. If any shares of Series C Non-Voting Preferred Stock shall be converted or redeemed by the Corporation, such shares shall, to the fullest extent permitted by applicable law, be retired and cancelled upon such acquisition, and shall not be reissued as a share of Series C Non-Voting Preferred Stock. Any share of Series C Non-Voting Preferred Stock so acquired shall, upon its retirement and cancellation, and upon the taking of any action required by applicable law, resume the status of authorized but unissued shares of preferred stock and shall no longer be designated as Series C Non-Voting Preferred Stock.

[Remainder of Page Intentionally Left Blank]

IN WITNESS WHEREOF, Onconova Therapeutics, Inc. has caused this Certificate of Designation of Preferences, Rights and Limitations of Series C Non-Voting Convertible Preferred Stock to be duly executed by its Chief Executive Officer on April 1, 2024.

ONCONOVA THERAPEUTICS, INC.

By: Name: Title:

ANNEX A

NOTICE OF CONVERSION

(TO BE EXECUTED BY THE REGISTERED HOLDER IN ORDER TO CONVERT SHARES OF SERIES C NON-VOTING CONVERTIBLE PREFERRED STOCK)

The undersigned Holder hereby irrevocably elects to convert the number of shares of Series C Non-Voting Preferred Stock indicated below, represented in book-entry form, into shares of common stock, par value \$0.01 per share (the "*Common Stock*"), of Onconova Therapeutics, Inc., a Delaware corporation (the "*Corporation*"), as of the date written below. If securities are to be issued in the name of a Person other than the undersigned, the undersigned will pay all transfer taxes payable with respect thereto. Capitalized terms utilized but not defined herein shall have the meaning ascribed to such terms in that certain Certificate of Designation of Preferences, Rights and Limitations of Series C Non-Voting Convertible Preferred Stock (the "*Certificate of Designation*") filed by the Corporation with the Secretary of State of the State of Delaware on April 1, 2024.

As of the date hereof, the number of shares of Common Stock beneficially owned by the undersigned Holder (together with such Holder's Attribution Parties), including the number of shares of Common Stock issuable upon conversion of the Series C Non-Voting Preferred Stock subject to this Notice of Conversion, but excluding the number of shares of Common Stock which are issuable upon (A) conversion of the remaining, unconverted Series C Non-Voting Preferred Stock beneficially owned by such Holder or any of its Attribution Parties, and (B) exercise or conversion of the unexercised or unconverted portion of any other securities of the Corporation (including any warrants) beneficially owned by such Holder or any of its Attribution Parties that are subject to a limitation on conversion or exercise similar to the limitation contained in <u>Section 6.4</u> of the Certificate of Designation, is ______. For purposes hereof, beneficial ownership shall be calculated in accordance with Section 13(d) of the Exchange Act and the applicable regulations of the Commission. In addition, for purposes hereof, "group" has the meaning set forth in Section 13(d) of the Exchange Act and the applicable regulations of the Commission.

CONVERSION CALCULATIONS:

Date to Effect Conversion:	
Number of shares of Series C Non-Voting Preferred Stock owned prior to Conversion:	
Number of shares of Series C Non-Voting Preferred Stock to be Converted:	
Number of shares of Common Stock to be Issued:	
Address for delivery of physical certificates:	
For DWAC Delivery, please provide the following:	
Broker No.:	
Account No.:	

[HOLDER]

By:	
Name:	
Title:	

EXHIBIT D

Execution Version

FORM OF CONTINGENT VALUE RIGHTS AGREEMENT

THIS CONTINGENT VALUE RIGHTS AGREEMENT (this "**Agreement**"), dated as of April 1, 2024, is entered into by and between Onconova Therapeutics, Inc., a Delaware corporation (the "**Company**"), and Equiniti Trust Company, LLC, a New York limited liability trust company, as Rights Agent (as defined herein).

RECITALS

WHEREAS, the Company, Traws Merger Sub I, Inc., a Delaware corporation and wholly owned subsidiary of the Company ("First Merger Sub"), Traws Merger Sub II, LLC, a Delaware limited liability company and wholly owned subsidiary of the Company ("Second Merger Sub"), and Trawsfynydd Therapeutics, Inc., a Delaware corporation ("Target"), have entered into an Agreement and Plan of Merger, dated as of April 1, 2024 (the "Merger Agreement"), pursuant to which First Merger Sub will merge with and into Target (the "First Merger"), with Target surviving the First Merger as a wholly-owned subsidiary of the Company and immediately following the First Merger and as part of the same overall transaction, Target will merge with and into Second Merger Sub (the "Second Merger" and together with the First Merger, the "Merger"), with Second Merger Sub surviving the Second Merger as a wholly owned subsidiary of the Company (the "Surviving Company");

WHEREAS, pursuant to the Merger Agreement, and in accordance with the terms and conditions thereof, the Company has agreed to provide to the Holders (as defined herein) contingent value rights as hereinafter described;

WHEREAS, the parties have done all things reasonably necessary to make the contingent value rights, when issued pursuant to the Merger Agreement and hereunder, the valid obligations of the Company and to make this Agreement a valid and binding agreement of the Company, in accordance with its terms; and

NOW, **THEREFORE**, in consideration of the premises and the consummation of the transactions referred to above, it is mutually covenanted and agreed, for the proportionate benefit of all Holders, as follows:

ARTICLE 1.

DEFINITIONS

Section 1.1 Definitions. Capitalized terms used but not otherwise defined herein have the meanings ascribed thereto in the Merger Agreement. The following terms have the meanings ascribed to them as follows:

"Acting Holders" means, at the time of determination, the Holders of at least 30% of the outstanding CVRs, as reflected on the CVR Register.

"Assignee" has the meaning set forth in Section 7.5.

"Board" means board of directors of the Company.

"Calendar Quarter" means the successive periods of three (3) consecutive calendar months ending on March 31, June 30, September 30 or December 31, for so long as this Agreement is in effect; *provided, however* that (a) the first Calendar Quarter shall commence on the date of this Agreement and shall end on the first June 30 thereafter, and (b) the last Calendar Quarter shall commence on the first day after the full Calendar Quarter immediately preceding the effective date of the termination or expiration of this Agreement and shall end on the effective date of the termination or expiration of this Agreement. "Combination Product" means a combination product as defined in 21 C.F.R. § 3.2(e), or any other Product consisting of one of the Program Assets with any other drug, biological product, or medical device whether they are formulated together, packaged together and sold for a single price, or administered to patients together according to a Product label.

"Common Stock" means the common stock, \$0.01 par value, of the Company.

"CVR" means a contingent contractual right of Holders to receive CVR Payments pursuant to the Merger Agreement and this Agreement.

"CVR Payment" means (a) 43.70% of the Net Proceeds received by the Company in a given Calendar Quarter in the event of a Disposition or (b) 6.24% of any Net Sales by the Company or any of its Affiliates in a given Calendar Quarter.

"CVR Payment Amount" means with respect to each CVR Payment and each Holder, an amount equal to such CVR Payment divided by the total number of CVRs and then multiplied by the total number of CVRs held by such Holder as reflected on the CVR Register.

"CVR Payment Period" means, as applicable on a Product-by-Product and country-by-country basis, a period equal to a Calendar Quarter ending at any time after (a) the effective date of a Disposition Agreement or (b) the First Commercial Sale with respect to a Product.

"CVR Payment Statement" means, for a given CVR Payment Period during the CVR Term, a written statement of the Company, signed on behalf of the Company, setting forth in reasonable detail the calculation of the applicable CVR Payment for such CVR Payment Period.

"CVR Register" has the meaning set forth in Section 2.3(b).

"**CVR Term**" means, as applicable on a Product-by-Product and country-by-country basis, the period beginning on the Closing and ending (a) at the end of the later of (i) the applicable term of the Disposition Agreement and (ii) the final date of payment of Net Proceeds under the applicable Disposition Agreement, in the event of a Disposition, or (b) upon the tenth (10th) anniversary of the First Commercial Sale of a Product.

"Disposition" means the sale, license, transfer or disposition of any Program Asset (including any such sale or disposition of equity securities in any Subsidiary established by the Company to hold any right, title or interest in or to any Program Asset).

"Disposition Agreement" means a definitive written agreement providing for a transaction or series of transactions between the Company or its Affiliates and any Person who is not an Affiliate of the Company regarding a Disposition.

"DTC" means The Depository Trust Company or any successor thereto.

"First Commercial Sale" means, as applicable on a Product-by-Product and country-by-country basis, the first sale that will result in any Net Sales with respect to a Product.

"Gross Proceeds" means, without duplication, the sum of all cash consideration and the value of any marketable securities actually received by the Company or its Affiliates during the CVR Term in consideration for a Disposition pursuant to a Disposition Agreement. The value of any marketable securities (whether debt or equity) shall be equal to the volume weighted average of their closing market prices for the thirty (30) trading days ending the day prior to the date of payment to, or receipt by, the Company or its relevant Affiliate.

"Holdback Amount" means, in respect of any Disposition, the amount of Gross Proceeds withheld or held in escrow pursuant to the applicable Disposition Agreement; *provided*, that upon the release of all or any portion of the Holdback Amount to the Company, such previously escrowed funds (as applicable) shall be subject to the payment procedures provided for in <u>Section 2.4</u> herein; *provided further* that such funds released from withholding or escrow shall no longer be deemed to be a Holdback Amount for purposes of Permitted Deductions.

"Holder" means, at the relevant time, a Person in whose name CVRs are registered in the CVR Register.

"Licensee or Assignee" means, with respect to a Product, a Third Party to whom any Related Party (including, for clarity, another Licensee or Assignee) has granted a written license or sublicense (other than an implied license) or assignment of rights to commercialize a Product.

"Loss" has the meaning set forth in <u>Section 3.2(g)</u>.

"**Net Proceeds**" means, for any CVR Payment Period, the Gross Proceeds minus Permitted Deductions for such CVR Payment Period, all as calculated, to the extent in accordance with GAAP, in a manner consistent with the Company's accounting practices and the most recently filed annual audited financial statements with the SEC, except as otherwise set forth herein. For clarity, to the extent Permitted Deductions exceed Gross Proceeds for any CVR Payment Period, any excess Permitted Deductions shall be applied against Gross Proceeds in subsequent CVR Payment Periods.

"Net Sales" means the gross amount invoiced, billed or otherwise recorded for sales of a Product by or on behalf of the Company or its Affiliates, or any Licensee or Assignee to a Third Party, in an arm's-length transaction, less the following amounts, to the extent actually incurred or accrued in accordance with GAAP consistently applied, and not reimbursed by such Third Party; provided, that any given amount may be taken as a permitted deduction only once:

(a) reasonable and customary rebates, chargebacks, quantity, trade and similar discounts, credits and allowances and other price reductions reasonably granted, allowed, incurred or paid in so far as they are applied to sales of a Product;

(b) discounts (including cash, quantity, trade, governmental, and similar discounts), coupons, retroactive price reductions, charge back payments and rebates granted to managed care organizations or to federal, state and local governments, or to their agencies (including payments made under the new "Medicare Part D Coverage Gap Discount Program" and the "Annual Fee for Branded Pharmaceutical Manufacturers" specific to a Product), in each case, as applied to sales of a Product and actually given to customers;

(c) reasonable and customary credits, adjustments, and allowances, including those granted on account of price adjustments, billing errors, and damage, Product otherwise not in saleable condition, and rejection, return or recall of a Product, in each case in so far as they are applied to sales of a Product;

(d) reasonable and customary freight and insurance costs incurred with respect to the shipment of a Product to customers, in each case if charged separately and invoiced to the customer;

(e) customs duties, surcharges and other similar governmental charges incurred in connection with the exportation or importation of a Product to the extent included in the gross amount invoiced;

(f) sales, use, value-added, excise, turnover, inventory and other similar Taxes (excluding income Taxes), and that portion of annual fees due under Section 9008 of the United States Patient Protection and Affordable Care Act of 2010 (Pub. L. No. 111-48) and any other fee imposed by any equivalent applicable law, in each of the foregoing cases, that the Company allocates to sales of a Product in accordance with the Company's standard policies and procedures consistently applied across its products, as adjusted for rebates and refunds, imposed in connection with the sales of a Product to any Third Party, to the extent such Taxes are not paid by the Third Party;

(g) actual copayment waiver amounts uncollected or uncollected debt amounts with respect to sales of a Product, provided that if the debt is thereafter paid, the corresponding amount shall be added to the Net Sales of the period during which it is paid;

(h) reasonable, customary and documented out of pocket amounts directly relating to co-pay programs, bridging programs or other similar patient assistance programs which may be implemented from time to time by the Company in so far as they are applied to sales of a Product; and

(i) if it is necessary for the Company to obtain a license to Intellectual Property Rights owned by a Third Party to commercialize any Product ("Third Party License") and the Company obtains such Third Party License, any license fees, royalties, or other amounts paid by the Company pursuant to such Third Party License on account of the sale of such Product.

(j) other similar or customary deductions taken in the ordinary course of business as permitted in calculating net sales or net revenue (as applicable) under GAAP in so far as they are applied to sales of a Product.

Net Sales shall be determined in U.S. dollars. Net Sales will not include (i) any arms' length sale to a Related Party, (ii) any sale for use of a Product in clinical or non-clinical development activities, or (iii) disposal or transfer of a Product for a bona fide charitable purpose, compassionate use or samples.

Notwithstanding anything to the contrary in this Agreement, Net Sales shall only be calculated once with respect to any specific unit of Product.

Notwithstanding anything to the contrary in this Agreement, if a Product is sold as part of a Combination Product, then Net Sales for such Product shall be determined by multiplying the Net Sales of the Combination Product (as calculated in accordance with analogous criteria as set forth above for the "Net Sales" definition) by the fraction, A / (A+B) where A is the weighted average sale price of such Product when sold separately in finished form, and B is the weighted average sale price of the other active product(s), compound(s) or ingredient(s) in such Combination Product sold separately in finished form (the "Other Product(s)"); provided that (i) if the weighted average sale price of a Product when sold separately in finished form can be determined but the weighted average sale price of the Other Product(s) cannot be determined, then Net Sales for such Combination Product shall be calculated by multiplying the Net Sales of such Combination Product (as calculated in accordance with analogous criteria as set forth above for the "Net Sales" definition) by the fraction A / C where A is the weighted average sale price of such Product when sold separately in finished form and C is the weighted average sale price of the Combination Product; (ii) if the weighted average sale price of the Other Product(s) can be determined but the weighted average sale price of a Product when sold separately in finished form cannot be determined. Net Sales for such Combination Product shall be calculated by multiplying the Net Sales of such Combination Product (as calculated in accordance with analogous criteria as set forth above for the "Net Sales" definition) by the following formula: one (1) minus B / C where B is the weighted average sale price of the Other Product(s) and C is the weighted average sale price of the Combination Product; and (iii) if the weighted average sale price of both a Product and the Other Product(s) cannot be determined, then Net Sales for such Combination Product shall be calculated by multiplying the Net Sales of such Combination Product (as calculated in accordance with analogous criteria as set forth above for the "Net Sales" definition) by the fraction 1 / D where D is the number of active products, compounds or ingredients in the Combination Product including a Product.

"Notice" has the meaning set forth in Section 7.1.

"Officer's Certificate" means a certificate signed by the chief executive officer and the chief financial officer of the Company, in their respective official capacities.

"Party" means the Company or the Rights Agent.

"Permitted Deductions" means the sum of, without duplication:

(a) any applicable Taxes (including any applicable value added or sales taxes) imposed on Gross Proceeds and payable by the Company or any of its Affiliates (regardless of whether the due date for such Taxes arises during or after the CVR Term) and, without duplication, any income or other similar Taxes payable by the Company or any of its Affiliates that would not have been incurred by the Company or any of its Affiliates but for the Gross Proceeds; provided that, for purposes of calculating income Taxes incurred by the Company or its Affiliates in respect of the Gross Proceeds, any such income Taxes shall be computed based on the gain recognized by the Company or its Affiliates from the Disposition after reduction for any net operating loss carryforwards or other Tax attributes of the Company or its Affiliates as of the Closing Date that are available to offset such gain after taking into account any limits of the usability of such attributes, including under Section 382 of the Code as determined by the Company's tax advisers;

(b) any reasonable out of pocket costs, fees and expenses incurred by the Company or any of its Affiliates in respect of its performance of this Agreement following the Closing Date or in respect of its performance of any Contract in connection with any Program Asset, including any costs related to the prosecution, maintenance or enforcement by the Company or any of its Subsidiaries of intellectual property rights (but excluding any costs related to a breach of this Agreement, including costs incurred in litigation in respect of the same);

(c) any reasonable out of pocket costs, fees and expenses incurred or accrued by the Company or any of its Affiliates in connection with (i) Disposition business development related efforts with respect to the relevant Program Asset(s), including technology transfer costs, contractual expenses or any costs in respect of head licenses for sublicensed technology and the development or prosecution, maintenance or enforcement by the Company or any of its Affiliates or Subsidiaries of Intellectual Property Rights, in each case primarily related to the relevant Program Asset(s), (ii) maintenance costs related to the CVRs or the Program Assets (including fees and expenses related to the Rights Agent), (iii) the Company's use of commercially reasonable efforts to negotiate or enter into any Disposition Agreement or consummate a Disposition of any Program Asset, (iv) any general administrative, legal, overhead, employee or other costs directly related to a Disposition, the CVRs or the Program Assets; and (v) any Rights Agent's fee, or in respect of any of the foregoing clauses (i)-(iii) any Rights Agent's fee, brokerage fee, finder's fee, opinion fee, success fee, transaction fee, service fee or other fee, commission or expense owed to any broker, finder, investment bank, auditor, accountant, counsel, advisor or other third party in relation thereto, but excluding (A) any costs or expenses previously deducted from Gross Proceeds, and (B) any costs related to a breach of this Agreement, including costs incurred in litigation in respect of the same;

(d) any Losses incurred by the Company or any of its Affiliates arising out of any third-party claims, demands, actions, or other proceedings relating to or in connection with any Disposition, including indemnification obligations of the Company or any of its Affiliates set forth in any Disposition Agreement;

(e) any Liabilities borne by the Company or any of its Affiliates pursuant to Contracts related to Program Assets, including costs arising from the termination thereof; and

- (f) any Holdback Amount;
- (g) any fees required to be paid pursuant to Section 3.2(h).

"**Permitted Transfer**" means a transfer of CVRs (a) upon death of a Holder by will or intestacy; (b) pursuant to a court order; (c) by operation of law (including by consolidation or merger) or without consideration in connection with the dissolution, liquidation or termination of any corporation, limited liability company, partnership or other entity; (d) in the case of CVRs held in book-entry or other similar nominee form, from a nominee to a beneficial owner and, if applicable, through an intermediary, to the extent allowable by DTC; or (e) as provided in <u>Section 2.6</u>.

"Product" means any product which contains either Narazaciclib or Rigosertib (each as more specifically defined on Annex I).

"**Program Assets**" means the tangible and intangible assets (including intellectual property) exclusively used in or exclusively related to the Company's Narazaciclib and Rigosertib programs.

"Record Date" means April 15, 2024.

"Record Time" has the meaning set forth in Section 2.1(a).

"Related Party" means each of the Company, its Affiliates, and each respective Licensee or Assignee, as applicable.

"**Rights Agent**" means the Rights Agent named in the first paragraph of this Agreement, until a successor Rights Agent will have become the Rights Agent pursuant to the applicable provisions of this Agreement, and thereafter "Rights Agent" will mean such successor Rights Agent.

"Third Party" means any Person that is not the Company or the Company's Affiliates.

ARTICLE 2.

CONTINGENT VALUE RIGHTS

Section 2.1 Holders of CVRs; Appointment of Rights Agent.

(a) The CVRs represent the rights of Holders to receive CVR Payments pursuant to this Agreement. The initial Holders will be the holders of Common Stock as of 5:00 p.m. ET on the Record Date (the "**Record Time**"). One CVR will be issued with respect to each share of Common Stock that is outstanding as of the Record Time.

(b) The Company hereby appoints the Rights Agent to act as Rights Agent for the Company in accordance with the express terms and conditions set forth in this Agreement, and the Rights Agent hereby accepts such appointment.

Section 2.2 Non-transferable. The CVRs may not be sold, assigned, transferred, pledged, encumbered or in any other manner transferred or disposed of, in whole or in part, other than through a Permitted Transfer. The CVRs will not be listed on any quotation system or traded on any securities exchange.

Section 2.3 No Certificate; Registration; Registration of Transfer; Change of Address.

(a) The CVRs will be issued in book-entry form only and will not be evidenced by a certificate or other instrument.

(b) The Rights Agent shall create and maintain a register (the "**CVR Register**") for the purpose of registering CVRs and Permitted Transfers. The CVR Register will be created, and CVRs will be distributed, pursuant to written instructions to the Rights Agent from the Company. The CVR Register will initially show one position for Cede & Co. representing shares of Common Stock held by DTC on behalf of the street holders of the shares of Common Stock held by such holders as of the Record Time. The Rights Agent will have no responsibility whatsoever directly or indirectly to the street name holders with respect to transfers of CVRs. With respect to any payments or issuances to be made under <u>Section 2.4</u> below, the Rights Agent will accomplish the payment to any former street name holders of shares of Common Stock by sending one lump-sum payment or issuance to DTC. The Rights Agent will have no responsibilities whatsoever with regard to the distribution of payments or shares of Common Stock by DTC to such street name holders.

(c) Subject to the restrictions on transferability set forth in <u>Section 2.2</u>, every request made to transfer a CVR must be in writing and accompanied by a written instrument of transfer in form reasonably satisfactory to the Rights Agent pursuant to its guidelines or procedures, including a guaranty of signature by an "eligible guarantor institution" that is a member or participant in the Securities Transfer Agents Medallion Program, duly executed by the Holder thereof, the Holder's attorney duly authorized in writing, the Holder's personal representative or the Holder's survivor, and setting forth in reasonable detail the circumstances relating to the transfer. Upon receipt of such written notice, the Rights Agent will ask the Company to confirm whether the transfer instrument is in proper form and the transfer is a Permitted Transfer and otherwise complies on its face with the other terms and conditions of this Agent may require evidence of payment of a sum sufficient to cover any stamp, documentary, registration, or other Tax or governmental charge that is imposed in connection with any such registration of transfer (or evidence that such Taxes and charges are not applicable). All duly transferred CVRs registered in the CVR Register will be the valid obligations of the Company and will entitle the transferee to the same benefits and rights under this Agreement as those held immediately prior to the transfer by the transfer. No transfer of a CVR will be valid until registered in the CVR Register.

(d) A Holder may make a written request to the Rights Agent to change such Holder's address of record in the CVR Register. The written request must be duly executed by the Holder. Upon receipt of such written notice, the Rights Agent shall, subject to its reasonable determination that the transfer instrument is in proper form, promptly record the change of address in the CVR Register. The Acting Holders may, without duplication, make a written request to the Rights Agent for a list containing the names, addresses and number of CVRs of the Holders that are registered in the CVR Register. Upon receipt of such written request from the Acting Holders, the Rights Agent shall promptly deliver a copy of such list to the Acting Holders.

(e) The Company will provide written instructions to the Rights Agent for the distribution of CVRs to holders of Common Stock as of the Record Time. Subject to the terms and conditions of this Agreement, the Rights Agent shall effect the distribution of the CVRs, less any applicable Tax withholding, to each holder of Common Stock as of the Record Time by the mailing of a statement of holding reflecting such CVRs.

Section 2.4 Payment Procedures.

(a) No later than sixty (60) days following the end of each Calendar Quarter during the CVR Term beginning with the Calendar Quarter ending on March 31, 2024, commencing with the first CVR Payment Period in which the Company or its Affiliates receives Gross Proceeds or has Net Sales, the Company shall deliver to the Rights Agent a CVR Payment Statement for such CVR Payment Period. Concurrent with the delivery of each CVR Payment Statement, on the terms and conditions of this Agreement, the Company shall pay the Rights Agent in U.S. dollars an amount equal to the CVR Payment for the applicable CVR Payment Period. Such CVR Payment will be transferred by wire transfer of immediately available funds to an account designated in writing by the Rights Agent not less than twenty (20) Business Days prior to the date of the applicable payment. Upon receipt of the wire transfer referred to in the foregoing sentence, the Rights Agent shall promptly (and in any event, within ten (10) Business Days) pay to each Holder set forth in the CVR Register at such time, an amount equal to such Holder's CVR Payment Amount. The Rights Agent shall promptly, and in any event within ten (10) Business Days after receipt of a CVR Payment Statement under this Section 2.4(a), send each Holder at its registered address a copy of such statement. For the avoidance of doubt the Company shall have no further liability in respect of the relevant CVR Payment upon delivery of such CVR Payment in accordance with this Section 2.4(a) and the satisfaction of each of the Company's obligations set forth in this Section 2.4(a).

(b) The Rights Agent shall solicit from each Holder an IRS Form W-9 or applicable IRS Form W-8 at such time or times as is necessary to permit any payment under this Agreement to be made without U.S. federal backup withholding. That notwithstanding, the Company shall be entitled to deduct and withhold, and hereby authorizes the Rights Agent to deduct and withhold, any Tax that is required to be deducted or withheld under applicable law from any amounts payable pursuant to this Agreement. To the extent the amounts are so withheld by the Company or the Rights Agent, as the case may be, and paid over to the appropriate Governmental Body, such withheld amounts shall be treated for all purposes of this Agreement as having been paid to the person in respect of whom such deduction and withholding was made.

(c) Any portion of a CVR Payment that remains undistributed to the Holders six (6) months after the applicable Calendar Quarter end (including by means of uncashed checks or invalid addresses on the CVR Register) will be delivered by the Rights Agent to the Company or a person nominated in writing by the Company (with written notice thereof from the Company to the Rights Agent), and any Holder will thereafter look only to the Company for payment of such CVR Payment (which shall be without interest).

If any CVR Payment (or portion thereof) remains unclaimed by a Holder two (2) years after the applicable Calendar Ouarter end (d) (or immediately prior to such earlier date on which such CVR Payment would otherwise escheat to or become the property of any Governmental Body). such CVR Payment (or portion thereof) will, to the extent permitted by applicable Law, become the property of the Company and will be transferred to the Company or a person nominated in writing by the Company (with written notice thereof from the Company to the Rights Agent), free and clear of all claims or interest of any Person previously entitled thereto, and no consideration or compensation shall be payable therefor. Neither the Company nor the Rights Agent will be liable to any Person in respect of a CVR Payment delivered to a public official pursuant to any applicable abandoned property, escheat or similar legal requirement under applicable Law.

Section 2.5 No Voting, Dividends or Interest; No Equity or Ownership Interest.

(a)

The CVRs will not have any voting or dividend rights, and interest will not accrue on any amounts payable in respect of CVRs to

any Holder.

(b) The CVRs will not represent any equity or ownership interest in the Company or in any constituent company to the Merger. It is hereby acknowledged and agreed that a CVR shall not constitute a security of the Company.

Nothing contained in this Agreement shall be construed as conferring upon any Holder, by virtue of the CVRs, any rights or obligations of any kind or nature whatsoever as a stockholder or member of the Company or any of its subsidiaries either at law or in equity. The rights of any Holder and the obligations of the Company and its Affiliates and their respective officers, directors and controlling Persons are contract rights limited to those expressly set forth in this Agreement.

It is hereby acknowledged and agreed that the CVRs and the possibility of any payment hereunder with respect thereto are highly (d) speculative and subject to numerous factors outside of the Company's control, and there is no assurance that Holders will receive any payments under this Agreement or in connection with the CVRs. Each Holder acknowledges that it is highly possible that no Disposition or First Commercial Sale will occur and that there will not be any Gross Proceeds or Net Sales that may be the subject of a CVR Payment Amount. It is further acknowledged and agreed that neither the Company nor its Affiliates owe, by virtue of their obligations under this Agreement, a fiduciary duty or any implied duties to the Holders and the parties hereto, and the Company and its Affiliates intend solely the express provisions of this Agreement to govern their contractual relationship with respect to the CVRs. It is acknowledged and agreed that this Section 2.5(d) is an essential and material term of this Agreement.

Section 2.6 Ability to Abandon CVR. A Holder may at any time, at such Holder's option, abandon all of such Holder's remaining rights represented by CVRs by transferring such CVRs to the Company or a Person nominated in writing by the Company (with written notice thereof from the Company to the Rights Agent) without consideration in compensation therefor, and such rights will be cancelled, with the Rights Agent being promptly notified in writing by the Holder of such transfer and cancellation. Nothing in this Agreement is intended to prohibit the Company or its Affiliates from offering to acquire or acquiring CVRs, in private transactions or otherwise, for consideration in its sole discretion.

ARTICLE 3.

THE RIGHTS AGENT

Section 3.1 Certain Duties and Responsibilities.

(a) The Rights Agent will not have any liability for any actions taken or not taken in connection with this Agreement, except to the extent such liability arises as a result of the willful misconduct, fraud, bad faith or gross negligence of the Rights Agent (in each case as determined by a final non-appealable judgment of court of competent jurisdiction). Notwithstanding anything in this Agreement to the contrary, any liability of the Rights Agent under this Agreement will be limited to the amount of annual fees paid by the Company to the Rights Agent in connection with this Agreement (but not including reimbursable expenses and other charges). Anything to the contrary notwithstanding, in no event will the Rights Agent be liable for special, punitive, indirect, incidental or consequential loss or damages of any kind whatsoever (including, without limitation, lost profits), even if the Rights Agent has been advised of the likelihood of such loss or damages, and regardless of the form of action.

(b) The Rights Agent shall not have any duty or responsibility in the case of the receipt of any written demand from any Holder with respect to any action or default by any person or entity, including, without limiting the generality of the foregoing, any duty or responsibility to initiate or attempt to initiate any proceedings at law or otherwise or to make any demand upon the Company or Target. The Rights Agent may (but shall not be required to) enforce all rights of action under this Agreement and any related claim, action, suit, audit, investigation or proceeding instituted by the Rights Agent may be brought in its name as the Rights Agent and any recovery in connection therewith will be for the proportionate benefit of all the Holders, as their respective rights or interests may appear on the CVR Register.

Section 3.2 Certain Rights of Rights Agent.

(a) The Rights Agent undertakes to perform such duties and only such duties as are specifically set forth in this Agreement, and no implied covenants or obligations will be read into this Agreement against the Rights Agent.

(b) The Rights Agent may rely and will be protected by the Company in acting or refraining from acting upon any resolution, certificate, statement, instrument, opinion, report, notice, request, direction, consent, order or other paper or document reasonably believed by it in the absence of bad faith to be genuine and to have been signed or presented by or on behalf of the Company or with respect to <u>Section 2.3(d)</u>, the Acting Holders.

(c) Whenever the Rights Agent deems it desirable that a matter be proved or established prior to taking or omitting any action hereunder, the Rights Agent may rely upon an Officer's Certificate, which certificate shall be full authorization and protection to the Rights Agent, and the Rights Agent shall, in the absence of bad faith, fraud, gross negligence or willful misconduct (each as determined by a final non-appealable judgment of a court of competent jurisdiction) on its part, incur no liability and be held harmless by the Company for or in respect of any action taken or omitted to be taken by it under the provisions of this Agreement in reliance upon such Officer's Certificate.

(d) The Rights Agent may engage and consult with counsel of its selection, and the advice or opinion of such counsel will, in the absence of bad faith, fraud, gross negligence or willful misconduct (in each case, as determined by a final, non-appealable judgment of a court of competent jurisdiction) on the part of the Rights Agent, be full and complete authorization and protection in respect of any action taken or not taken by the Rights Agent in reliance thereon.

- (e) Any permissive rights of the Rights Agent hereunder will not be construed as a duty.
- Agreement.
- (f) The Rights Agent will not be required to give any note or surety in respect of the execution of its powers or otherwise under this

(g) The Company agrees to indemnify the Rights Agent for, and to hold the Rights Agent harmless from and against, any loss, liability, damage, judgment, fine, penalty, cost or expense (each, a "Loss") suffered or incurred by the Rights Agent and arising out of or in connection with the Rights Agent's performance of its obligations under this Agreement, including the reasonable and documented costs and expenses of defending the Rights Agent against any claims, charges, demands, actions or suits arising out of or in connection in connection with the execution, acceptance, administration, exercise and performance of its duties under this Agreement, including the costs and expenses of defending against any claim of liability arising therefrom, directly or indirectly, or enforcing its rights hereunder, except to the extent such Loss has been determined by a final non-appealable decision of a court of competent jurisdiction to have resulted from the Rights Agent's fraud, gross negligence, bad faith or willful misconduct; provided that this <u>Section 3.2(g)</u> shall not apply with respect to income, receipt, franchise or similar Taxes imposed with respect to payments to the Rights Agent for its services pursuant to this Agreement.

(h) The Company agrees (i) to pay the fees of the Rights Agent in connection with the Rights Agent's performance of its obligations hereunder, as agreed upon in writing by the Rights Agent and the Company on or prior to the date of this Agreement, and (ii) to reimburse the Rights Agent for all reasonable and documented out-of-pocket expenses and other disbursements incurred in the preparation, delivery, negotiation, amendment, administration and execution of this Agreement and the exercise and performance of its duties hereunder, including all stamp and transfer Taxes (and excluding for the avoidance of doubt, any income, receipt, franchise or similar Taxes on payments to the Rights Agent for its services pursuant to this Agreement) and governmental charges, incurred by the Rights Agent in the performance of its obligations under this Agreement, except that the Company will have no obligation to pay the fees of the Rights Agent or reimburse the Rights Agent for the fees of counsel in connection with any lawsuit initiated by the Rights Agent on behalf of itself or the Holders, except in the case of any suit enforcing the provisions of <u>Section 2.4(a)</u> and <u>Section 3.2(g)</u>, if the Company is found by a court of competent jurisdiction to be liable to the Rights Agent or the Holders, as applicable in such suit.

(i) No provision of this Agreement shall require the Rights Agent to expend or risk its own funds or otherwise incur any financial liability in the performance of any of its duties hereunder or in the exercise of any of its rights or powers if it believes that repayment of such funds or adequate indemnification against such risk or liability is not reasonably assured to it.

(j) The Rights Agent shall not be subject to, nor be required to comply with, or determine if any Person has complied with, the Merger Agreement or any other agreement between or among any of the Company, Target or Holders, even though reference thereto may be made in this Agreement, or to comply with any notice, instruction, direction, request or other communication, paper or document other than as expressly set forth in this Agreement.

(k) In the event the Rights Agent reasonably believes any ambiguity or uncertainty exists hereunder or in any notice, instruction, direction, request or other communication, paper or document received by the Rights Agent hereunder, the Rights Agent shall, as soon as practicable, provide notice to the Company, and the Rights Agent may, in its sole discretion, refrain from taking any action, and shall be fully protected and shall not be liable in any way to the Company or any Holder or other Person for refraining from taking such action, unless the Rights Agent receives written instructions from the Company or such Holder or other Person which eliminate such ambiguity or uncertainty to the reasonable satisfaction of the Rights Agent;

(1) The Rights Agent may execute and exercise any of the rights or powers hereby vested in it or perform any duty hereunder either itself or by or through its attorney or agents and the Rights Agent shall not be answerable or accountable for any act, default, neglect or misconduct of any such attorney or agents or for any loss to the Company or Target resulting from any such act, default, neglect or misconduct, absent gross negligence, bad faith or willful misconduct (each as determined by a final non-appealable judgment of a court of competent jurisdiction) in the selection and continued employment thereof.

(m) The Rights Agent shall not be liable for or by reason of any of the statements of fact or recitals contained in this Agreement (except its countersignature thereof) or be required to verify the same, and all such statements and recitals are and shall be deemed to have been made by the Company only.

(n) The Rights Agent shall act hereunder solely as agent for the Company and shall not assume any obligations or relationship of agency or trust with any of the owners or holders of the CVRs. The Rights Agent shall not have any duty or responsibility in the case of the receipt of any written demand from any Holders with respect to any action or default by the Company, including, without limiting the generality of the foregoing, any duty or responsibility to initiate or attempt to initiate any proceedings at law or otherwise or to make any demand upon the Company.

(o) The Rights Agent may rely on and be fully authorized and protected in acting or failing to act upon (i) any guaranty of signature by an "eligible guarantor institution" that is a member or participant in the Securities Transfer Agents Medallion Program or other comparable "signature guarantee program" or insurance program in addition to, or in substitution for, the foregoing; or (ii) any law, act, regulation or any interpretation of the same even though such law, act, or regulation may thereafter have been altered, changed, amended or repealed.

(p) The Rights Agent shall not be liable or responsible for any failure of the Company to comply with any of its obligations relating to any registration statement filed with the Securities and Exchange Commission or this Agreement, including without limitation obligations under applicable regulation or law.

(q) The obligations of the Company and the rights of the Rights Agent under this <u>Section 3.2</u>, <u>Section 3.1</u> and <u>Section 2.4</u> shall survive the expiration of the CVRs and the termination of this Agreement and the resignation, replacement or removal of the Rights Agent.

Section 3.3 Resignation and Removal; Appointment of Successor.

(a) The Rights Agent may resign at any time by written notice to the Company. Any such resignation notice shall specify the date on which such resignation will take effect (which shall be at least thirty (30) days following the date that such resignation notice is delivered), and such resignation will be effective on the earlier of (x) the date so specified and (y) the appointment of a successor Rights Agent.

(b) The Company will have the right to remove the Rights Agent at any time by written notice to the Rights Agent, specifying the date on which such removal will take effect. Such notice will be given at least thirty (30) days prior to the date so specified (or, if earlier, the appointment of the successor Rights Agent).

(c) If the Rights Agent resigns, is removed or becomes incapable of acting, the Company will promptly appoint a qualified successor Rights Agent. Notwithstanding the foregoing, if the Company fails to make such appointment within a period of thirty (30) days after giving notice of such removal or after it has been notified in writing of such resignation or incapacity by the resigning or incapacitated Rights Agent, then the incumbent Rights Agent may apply to any court of competent jurisdiction for the appointment of a new Rights Agent. The successor Rights Agent so appointed will, upon its acceptance of such appointment in accordance with this <u>Section 3.3(c)</u> and <u>Section 3.4</u>, become the Rights Agent for all purposes hereunder.

(d) The Company will give notice to the Holders of each resignation or removal of the Rights Agent and each appointment of a successor Rights Agent in accordance with Section 7.2. Each notice will include the name and address of the successor Rights Agent. If the Company fails to send such notice within ten (10) Business Days after acceptance of appointment by a successor Rights Agent, the successor Rights Agent will cause the notice to be mailed at the expense of the Company.

(e) Notwithstanding anything to the contrary in this <u>Section 3.3</u>, unless consented to in writing by the Acting Holders, the Company will not appoint as a successor Rights Agent any Person that is not a stock transfer agent of national reputation or the corporate trust department of a commercial bank.

(f) The Rights Agent will reasonably cooperate with the Company and any successor Rights Agent in connection with the transition of the duties and responsibilities of the Rights Agent to the successor Rights Agent, including the transfer of all relevant data, including the CVR Register, to the successor Rights Agent, but such predecessor Rights Agent shall not be required to make any additional expenditure or assume any additional liability in connection with the foregoing.

Section 3.4 Acceptance of Appointment by Successor. Every successor Rights Agent appointed hereunder will, at or prior to such appointment, execute, acknowledge and deliver to the Company and to the resigning or removed Rights Agent an instrument accepting such appointment and a counterpart of this Agreement, and such successor Rights Agent, without any further act, deed or conveyance, will become vested with all the rights, powers, trusts and duties of the Rights Agent; *provided* that upon the request of the Company or the successor Rights Agent, such resigning or removed Rights Agent will execute and deliver an instrument transferring to such successor Rights Agent all the rights, powers and trusts of such resigning or removed Rights Agent.

ARTICLE 4.

COVENANTS

Section 4.1 List of Holders. The Company will furnish or cause to be furnished to the Rights Agent, in such form as the Company receives from the Company's transfer agent (or other agent performing similar services for the Company), the names and addresses of the Holders within ten (10) Business Days following the Closing Date.

Section 4.2 Limited Obligations of Public Company. Notwithstanding anything herein to the contrary, and for the avoidance of doubt, (a) during the CVR Term, the Company shall use commercially reasonable efforts to develop and commercialize or monetize the Program Assets and (b) none of the Company or any of its Affiliates (or any directors, officer, employee, or other representative of the foregoing) owes any fiduciary duty or similar duty to any Holder in respect of the Program Assets. Notwithstanding anything in this Agreement to the contrary, nothing in this Agreement shall require the Company (or any of its Affiliates) to commence, defend or otherwise pursue any litigation, regulatory action or other Legal Proceeding relating to any of the Program Assets or any other products or assets of the Company.

Section 4.3 Prohibited Actions. Unless approved by the Acting Holders, prior to the end of the CVR Term, the Company shall not grant any lien, security interest, pledge or similar interest in any Program Assets or any Net Proceeds. Unless approved by the Acting Holders, prior to end of the CVR Term, the Company shall not, and shall not permit its Affiliates to, grant, assign, transfer or otherwise convey any Program Assets (including any option to obtain rights) to any third party.

Section 4.4 Books and Records. Until the end of the CVR Term, the Company shall, and shall cause its Affiliates to, keep true, complete and accurate records in sufficient detail to enable the Rights Agent to confirm each CVR Payment payable hereunder in accordance with the terms specified in this Agreement.

Section 4.5 Audits. Until the expiration of this Agreement and for a period of ten (10) years thereafter, the Company shall keep complete and accurate records in sufficient detail to permit the Rights Agent to confirm the accuracy of the payments due hereunder. The Rights Agent or the Acting Holders, without duplication, shall each have the right to cause an independent accounting firm reasonably acceptable to the Company to audit such records for the sole purpose of confirming payments for a period covering not more than the date commencing with the first CVR Payment Period in which the Company or its Affiliates receives Gross Proceeds or Net Sales and ending on the last day of the CVR Term. The Company may require such accounting firm to execute a reasonable confidentiality agreement with the Company prior to commencing the audit. The accounting firm shall disclose to Rights Agent or the Acting Holders, as applicable, only whether the reports are correct or not and the specific details concerning any discrepancies. No other information shall be shared. Such audits may be conducted during normal business hours upon reasonable prior written notice to the Company, but no more than frequently than once per year. No accounting period of the Company shall be subject to audit more than one time by the Rights Agent or the Acting Holders, as applicable, unless after an accounting period has been audited by the Rights Agent or the Acting Holders, as applicable, the Company restates its financial results for such accounting period, in which event the Rights Agent or the Acting Holders, as applicable, may conduct a second audit of such accounting period in accordance with this Section 4.5. Adjustments (including remittances of underpayments or overpayments disclosed by such audit) shall be made by the Parties to reflect the results of such audit, which adjustments shall be paid promptly following receipt of an invoice therefor. The Rights Agent or the Acting Holders, as applicable, shall bear the full cost and expense of such audit unless such audit discloses an underpayment by the Company of twenty percent (20%) or more of the CVR Payments due under this Agreement for the audited period, in which case the Company shall bear the full cost and expense of such audit.

ARTICLE 5.

AMENDMENTS

Section 5.1 Amendments Without Consent of Holders or Rights Agent.

(a) The Company, at any time and from time to time, may (without the consent of any Person, other than the Rights Agent with such consent not to be unreasonably withheld, conditioned or delayed) enter into one or more amendments to this Agreement for any of the following purposes:

(i) to evidence the appointment of another Person as a successor Rights Agent and the assumption by any successor Rights Agent of the covenants and obligations of the Rights Agent herein in accordance with the provisions hereof;

(ii) subject to <u>Section 6.1</u>, to evidence the succession of another person to the Company and the assumption of any such successor of the covenants of the Company outlined herein in a transaction contemplated by <u>Section 6.1</u>;

(iii) to add to the covenants of the Company such further covenants, restrictions, conditions or provisions as the Company and the Rights Agent will consider to be for the protection and benefit of the Holders; *provided* that in each case, such provisions do not adversely affect the interests of the Holders;

(iv) to cure any ambiguity, to correct or supplement any provision in this Agreement that may be defective or inconsistent with any other provision in this Agreement, or to make any other provisions with respect to matters or questions arising under this Agreement; *provided* that, in each case, such provisions do not adversely affect the interests of the Holders;

(v) as may be necessary or appropriate to ensure that the CVRs are not subject to registration under the Securities Act or the Exchange Act and the rules and regulations promulgated thereunder, or any applicable state securities or "blue sky" laws;

(vi) as may be necessary or appropriate to ensure that the Company is not required to produce a prospectus or an admission document in order to comply with applicable Law;

(vii) to cancel the applicable CVRs (x) in the event that any Holder has abandoned its rights in accordance with <u>Section 2.6</u>, or (y) following a transfer of such CVRs to the Company or its Affiliates in accordance with <u>Section 2.2</u> or <u>Section 2.3</u>;

(viii) as may be necessary or appropriate to ensure that the Company complies with applicable Law; or

(ix) to effect any other amendment to this Agreement for the purpose of adding, eliminating or changing any provisions of this Agreement, *provided* that, in each case, such additions, eliminations or changes do not adversely affect the interests of the Holders.

(b) Promptly after the execution by the Company of any amendment pursuant to this <u>Section 5.1</u>, the Company will (or will cause the Rights Agent to) notify the Holders in general terms of the substance of such amendment in accordance with <u>Section 7.2</u>.

Section 5.2 Amendments with Consent of Holders.

(a) In addition to any amendments to this Agreement that may be made by the Company without the consent of any Holder pursuant to <u>Section 5.1</u>, with the consent of the Acting Holders (whether evidenced in a writing or taken at a meeting of the Holders), the Company and the Rights Agent may enter into one or more amendments to this Agreement for the purpose of adding, eliminating or amending any provisions of this Agreement, even if such addition, elimination or amendment is adverse to the interests of the Holders.

(b) Promptly after the execution by the Company and the Rights Agent of any amendment pursuant to the provisions of this <u>Section 5.2</u>, the Company will (or will cause the Rights Agent to) notify the Holders in general terms of the substance of such amendment in accordance with <u>Section 7.2</u>.

Section 5.3 Effect of Amendments. Upon the execution of any amendment under this Article 5, this Agreement will be modified in accordance therewith, such amendment will form a part of this Agreement for all purposes and every Holder will be bound thereby. Upon the delivery of a certificate from an appropriate officer of the Company which states that the proposed supplement or amendment is in compliance with the terms of this Article 5, the Rights Agent shall execute such supplement or amendment. Notwithstanding anything in this Agreement to the contrary, the Rights Agent shall not be required to execute any supplement or amendment to this Agreement that it has determined would adversely affect its own rights, duties, obligations or immunities under this Agreement. No supplement or amendment to this Agreement shall be effective unless duly executed by the Rights Agent.

ARTICLE 6.

CONSOLIDATION, MERGER, SALE OR CONVEYANCE

Section 6.1 The Company May Not Consolidate, Etc. During the CVR Term, the Company shall not consolidate with or merge into any other Person or convey, transfer or lease its properties and assets substantially as an entirety to any Person, unless:

(a) The Person formed by such consolidation or into which the Company is merged or the Person that acquires by conveyance or transfer, or that leases, the properties and assets of the Company substantially as an entirety (the "**Surviving Person**") shall expressly assume payment of amounts on all CVRs (when and as due hereunder) and the performance of every duty and covenant of this Agreement on the part of the Company to be performed or observed; and

(b) The Company has delivered to the Rights Agent an Officer's Certificate, stating that such consolidation, merger, conveyance, transfer or lease complies with this Article 6 and that all conditions precedent herein provided for relating to such transaction have been complied with.

Section 6.2 Successor Substituted. Upon any consolidation of or merger by the Company with or into any other Person, or any conveyance, transfer or lease of the properties and assets substantially as an entirety to any Person in accordance with <u>Section 6.1</u>, the Surviving Person shall succeed to, and be substituted for, and may exercise every right and power of, and shall assume all of the obligations of the Company under this Agreement with the same effect as if the Surviving Person had been named as the Company herein.

ARTICLE 7.

MISCELLANEOUS

Section 7.1 Notices to Rights Agent and to the Company. All notices and other communications hereunder shall be in writing and shall be deemed to have been duly delivered and received hereunder (a) one Business Day after being sent for next Business Day delivery, fees prepaid, via a reputable international overnight courier service, (b) upon delivery in the case of delivery by hand, or (c) on the date delivered in the place of delivery if sent by email (with a written or electronic confirmation of delivery) prior to 5:00 p.m. Eastern Time, otherwise on the next succeeding Business Day, in each case to the intended recipient as set forth below:

if to the Rights Agent, to:

Equiniti Trust Company, LLC Attn: Account Management Team 1110 Centre Pointe Curve, Suite 101 Mendota Heights, Minnesota 55120-4101 Email: <u>EQSS-AccountManagement@equiniti.com</u>

if to the Company, to:

Onconova Therapeutics, Inc. 12 Penns Trail Newtown, PA 18940 Attention: Steven M. Fruchtman; Mark Guerin Email Address: sfruchtman@onconova.us; mguerin@onconova.us

with a copy, which shall not constitute notice, to:

Orrick, Herrington & Sutcliffe LLP Columbia Center 1152 15th Street, N.W. Washington, D.C. 20005-1706 United States Attention: David Schulman Email Address: dschulman@orrick.com

Snell & Wilmer L.L.P. Executive Center 12230 El Camino Real, Suite 300 San Diego, CA 92130 United States Attention: Bardia Moayedi Email: bmoayedi@swlaw.com

or to such other address or email address as such Party may hereafter specify for the purpose by notice to the other Party.

Section 7.2 Notice to Holders. All Notices required to be given to the Holders will be given (unless otherwise herein expressly provided) in writing and mailed, first-class postage prepaid, to each Holder at such Holder's address as set forth in the CVR Register, not later than the latest date, and not earlier than the earliest date, prescribed for the sending of such Notice, if any, and will be deemed given on the date of mailing. In any case where notice to the Holders is given by mail, neither the failure to mail such Notice, nor any defect in any Notice so mailed, to any particular Holder will affect the sufficiency of such Notice with respect to other Holders.

Section 7.3 Entire Agreement. As between the Company and the Rights Agent, this Agreement constitutes the entire agreement between the parties with respect to the subject matter of this Agreement, notwithstanding the reference to any other agreement herein, and supersedes all prior agreements and understandings, both written and oral, among or between any of the parties with respect to the subject matter of this Agreement.

Section 7.4 Merger or Consolidation or Change of Name of Rights Agent. Any Person into which the Rights Agent or any successor Rights Agent may be merged or with which it may be consolidated, or Person resulting from any merger or consolidation to which the Rights Agent or any successor Rights Agent shall be a party, or any Person succeeding to the stock transfer or other shareholder services business of the Rights Agent or any successor Rights Agent, shall be the successor to the Rights Agent under this Agreement without the execution or filing of any paper or any further act on the part of any of the parties hereto, provided that such Person would be eligible for appointment as a successor Rights Agent under the provisions of <u>Section 3.3</u>. The purchase of all or substantially all of the Rights Agent's assets employed in the performance of transfer agent activities shall be deemed a merger or consolidation for purposes of this <u>Section 7.4</u>.

Section 7.5 Successors and Assigns. This Agreement will be binding upon, and will be enforceable by and inure solely to the benefit of, the Holders, the Company and the Rights Agent and their respective successors and assigns. Except for assignments pursuant to <u>Section 7.4</u>, the Rights Agent may not assign this Agreement without the Company's prior written consent. Subject to <u>Section 5.1(a)(ii)</u> and Article 6 hereof, the Company may assign, in its sole discretion and without the consent of any other party, any or all of its rights, interests and obligations hereunder to one or more of its Affiliates or to any Person with whom the Company is merged or consolidated, or any entity resulting from any merger or consolidation to which the Company shall be a party (each, an "Assignee"); *provided*, that in connection with any assignment to an Assignee, the Company shall agree to remain liable for the performance by the Company of its obligations hereunder (to the extent the Company exists following such assignment). The Company or an Assignee may not otherwise assign this Agreement without the prior consent of the Acting Holders (such consent not to be unreasonably withheld, conditioned or delayed). Any attempted assignment of this Agreement in violation of this <u>Section 7.5</u> will be void *ab initio* and of no effect.

Section 7.6 Benefits of Agreement; Action by Acting Holders. Nothing in this Agreement, express or implied, will give to any Person (other than the Company, the Rights Agent, the Holders and their respective permitted successors and assigns hereunder) any benefit or any legal or equitable right, remedy or claim under this Agreement or under any covenant or provision herein contained, all such covenants and provisions being for the sole benefit of the Company, the Rights Agent, the Holders and their permitted successors and assigns. The Holders will have no rights hereunder except as are expressly set forth herein. Except for the rights of the Rights Agent set forth herein, the Acting Holders will have the sole right, on behalf of all Holders, by virtue of or under any provision of this Agreement, to institute any action or proceeding at law or in equity with respect to this Agreement, and no individual Holder or other group of Holders will be entitled to exercise such rights.

Section 7.7 Governing Law. This Agreement and the CVRs will be governed by, and construed in accordance with, the laws of the State of Delaware, regardless of the Laws that might otherwise govern under applicable principles of conflicts of laws.

Section 7.8 Jurisdiction. In any action or proceeding between any of the parties hereto arising out of or relating to this Agreement or any of the transactions contemplated hereby, each of the parties hereto: (a) irrevocably and unconditionally consents and submits to the exclusive jurisdiction and venue of the Chancery Court of the State of Delaware, County of New Castle, or, if under applicable Law exclusive jurisdiction is vested in the Federal courts, the United States District Court for the District of Delaware (and appellate courts thereof); (b) agrees that all claims in respect of such action or proceeding shall be heard and determined exclusively in accordance with clause (a) of this Section 7.8; (c) waives any objection to laying venue in any such action or proceeding in such courts; (d) waives any objection that such courts are an inconvenient forum or do not have jurisdiction over any Party; and (e) agrees that service of process upon such Party in any such action or proceeding shall be effective if notice is given in accordance with Section 7.1 or Section 7.2 of this Agreement.

Section 7.9 WAIVER OF JURY TRIAL. EACH OF THE PARTIES HERETO HEREBY IRREVOCABLY WAIVES ANY AND ALL RIGHT TO TRIAL BY JURY IN ANY LEGAL PROCEEDING ARISING OUT OF OR RELATED TO THIS AGREEMENT OR THE TRANSACTIONS CONTEMPLATED HEREBY. EACH PARTY CERTIFIES AND ACKNOWLEDGES THAT (A)NO REPRESENTATIVE, AGENT OR ATTORNEY OF ANY OTHER PARTY HAS REPRESENTED, EXPRESSLY OR OTHERWISE, THAT SUCH OTHER PARTY WOULD NOT, IN THE EVENT OF LITIGATION, SEEK TO ENFORCE THE FOREGOING WAIVER, (B)EACH PARTY UNDERSTANDS AND HAS CONSIDERED THE IMPLICATION OF THIS WAIVER, (C)EACH PARTY MAKES THIS WAIVER VOLUNTARILY, AND (D)EACH PARTY HAS BEEN INDUCED TO ENTER INTO THIS AGREEMENT BY, AMONG OTHER THINGS, THE MUTUAL WAIVERS AND CERTIFICATIONS IN THIS <u>SECTION 7.9</u>.

Section 7.10 Severability Clause. Any term or provision of this Agreement that is invalid or unenforceable in any situation in any jurisdiction shall not affect the validity or enforceability of the remaining terms and provisions of this Agreement or the validity or enforceability of the offending term or provision in any other situation or in any other jurisdiction. If a final judgment of a court of competent jurisdiction declares that any term or provision of this Agreement is invalid or unenforceable, the parties agree that the court making such determination shall have the power to limit such term or provision, to delete specific words or phrases or to replace such term or provision, and this Agreement shall be valid and enforceable as so modified. In the event such court does not exercise the power granted to it in the prior sentence, the parties agree to replace such invalid or unenforceable term or provision with a valid and enforceable term or provision with a valid and enforceable term or provision that will achieve, to the extent possible, the economic, business and other purposes of such invalid or unenforceable term or provision.

Section 7.11 Counterparts; Effectiveness. This Agreement may be executed in several counterparts, each of which shall be deemed an original and all of which shall constitute one and the same instrument. Counterparts may be delivered via electronic mail (including pdf or any electronic signature complying with the U.S. federal ESIGN Act of 2000, e.g., www.docusign.com) or other transmission method and any counterpart so delivered shall be deemed to have been duly and validly delivered and be valid and effective for all purposes. This Agreement will become effective when each party hereto will have received a counterpart hereof signed by the other party hereto. Until and unless each party has received a counterpart hereof signed by the other party will have any right or obligation hereunder (whether by virtue of any oral or written agreement or any other communication).

Section 7.12 Termination. This Agreement will automatically terminate and be of no further force or effect and, except as provided in <u>Section 3.2</u>, the parties hereto will have no further liability hereunder, and the CVRs will expire without any consideration or compensation therefor, upon the expiration of the CVR Term. The termination of this Agreement will not affect or limit the right of Holders to receive the CVR Payments under <u>Section 2.4</u> to the extent earned prior to the termination of this Agreement, and the provisions applicable thereto will survive the expiration or termination of this Agreement until such CVR Payments have been made, if applicable.

Section 7.13 Construction.

(a) For purposes of this Agreement, whenever the context requires: the singular number shall include the plural, and vice versa; the masculine gender shall include the feminine and neuter genders; the feminine gender shall include the masculine and neuter genders; and the neuter gender shall include masculine and feminine genders.

(b) As used in this Agreement, the words "include" and "including," and variations thereof, shall not be deemed to be terms of limitation, but rather shall be deemed to be followed by the words "without limitation."

(c) The headings contained in this Agreement are for convenience of reference only, will not be deemed to be a part of this Agreement and will not be referred to in connection with the construction or interpretation of this Agreement.

(d) Unless stated otherwise, "Article" and "Section" followed by a number or letter mean and refer to the specified Article or Section of this Agreement. The term "Agreement" and any reference in this Agreement to this Agreement or any other agreement or document includes, and is a reference to, this Agreement or such other agreement or document as it may have been, or may from time to time be, amended, restated, replaced, supplemented or novated and includes all schedules to it.

(e) A period of time is to be computed as beginning on the day following the event that began the period and ending at 4:30 p.m. (Eastern Time) on the last day of the period, if the last day of the period is a Business Day, or at 4:30 p.m. (Eastern Time) on the next Business Day if the last day of the period is not a Business Day.

(f) Any reference in this Agreement to a date or time shall be deemed to be such date or time in Philadelphia, Pennsylvania, United States, unless otherwise specified. The parties hereto and the Company have participated jointly in the negotiation and drafting of this Agreement. In the event an ambiguity or question of intent or interpretation arises, this Agreement shall be construed as if drafted jointly by the parties and the Company and no presumption or burden of proof shall arise favoring or disfavoring any Person by virtue of the authorship of any provision of this Agreement.

(g) References to "cash," "dollars" or "\$" are to U.S. dollars.

[Remainder of page intentionally left blank]

IN WITNESS WHEREOF, each of the parties has caused this Agreement to be executed as of the day and year first above written.

Onconova Therapeutics, Inc.

By: Name: Title:

[Signature Page to CVR Agreement]

IN WITNESS WHEREOF, each of the parties has caused this Agreement to be executed as of the day and year first above written.

Equiniti Trust Company, LLC

By: Name: Title:

[Signature Page to CVR Agreement]

Annex I

Products

Narazaciclib (also known as ON123300) is a small molecule Cyclin D Kinase (CDK) inhibitor that targets CDK4 and CDK6, important proteins in the cell proliferation cycle, and other kinases such as CSF1R, involved in cell migration, as well as kinases (NUAK1/ARK5) involved in permitting cells to exist for some time in a hypoxic environment.

Rigosertib is a small molecule that targets multiple cell signaling pathways involved in tumorgenesis and also assists cancer cells to evade host immune surveillance.

EXHIBIT E

FORM OF PARENT STOCKHOLDER SUPPORT AGREEMENT

ONCONOVA THERAPEUTICS, INC.

SUPPORT AGREEMENT

THIS SUPPORT AGREEMENT (this "*Agreement*"), dated as of April 1, 2024, is made by and among Onconova Therapeutics, Inc., a Delaware corporation ("*Parent*"), Trawsfynydd Therapeutics, Inc., a Delaware corporation (the "*Company*"), and the undersigned holder ("*Stockholder*") of shares of capital stock (the "*Shares*") of Parent.

WHEREAS, Parent, Traws Merger Sub I, Inc., a Delaware corporation and a wholly owned subsidiary of Parent ("*First Merger Sub*"), Traws Merger Sub II, LLC, a Delaware limited liability company and a wholly owned subsidiary of Parent ("*Second Merger Sub*"), and the Company, have entered into an Agreement and Plan of Merger, dated of even date herewith (the "*Merger Agreement*"), providing for the merger of First Merger Sub with and into the Company (the "*First Merger*") and the merger of the Company with and into Second Merger Sub (the "*Second Merger*" and, together with the First Merger, the "*Merger*");

WHEREAS, Stockholder beneficially owns and has sole voting power with respect to the number of Shares, and holds options to purchase shares of Parent Common Stock ("*Parent Options*") and/or restricted stock units to acquire shares of Parent Common Stock ("*Parent RSUs*"), in each case in the number of Shares indicated opposite Stockholder's name on <u>Schedule 1</u> attached hereto;

WHEREAS, as an inducement and a condition to the willingness of Parent, First Merger Sub, Second Merger Sub and the Company to enter into the Merger Agreement, and in consideration of the substantial expenses incurred and to be incurred by them in connection therewith, Stockholder has agreed to enter into and perform this Agreement; and

WHEREAS, all capitalized terms used in this Agreement without definition herein shall have the meanings ascribed to them in the Merger Agreement.

NOW, THEREFORE, in consideration of, and as a condition to, Parent, First Merger Sub, Second Merger Sub and the Company's entering into the Merger Agreement and proceeding with the transactions contemplated thereby, and in consideration of the substantial expenses incurred and to be incurred by them in connection therewith, Stockholder, Parent and the Company agree as follows:

- 1. <u>Agreement to Vote Shares</u>. Stockholder agrees that, prior to the Expiration Date (as defined in <u>Section 2</u> below), at any meeting of the stockholders of Parent or any adjournment or postponement thereof, or in connection with any written consent of the stockholders of Parent, with respect to the Parent Stockholder Matters, Stockholder shall, or shall cause the holder of record on any applicable record date to:
 - (a) appear at such meeting or otherwise cause the Shares and any New Shares (as defined in <u>Section 3</u> below) to be counted as present thereat (in person or by proxy) for purposes of calculating a quorum;

- (b) from and after the date hereof until the Expiration Date, vote (or cause to be voted), or deliver a written consent (or cause a written consent to be delivered) covering all of the Shares and any New Shares that Stockholder shall be entitled to so vote: (i) in favor of the Parent Stockholder Matters and any matter that could reasonably be expected to facilitate the Parent Stockholder Matters; (ii) against any proposal to remove the limitation initially set at the discretion of holders of Parent Convertible Preferred Stock between 4.99% and 19.99% of the number of shares of Parent Common Stock outstanding immediately after giving effect to the issuance of shares of Parent Common Stock upon conversion (the "*Beneficial Ownership Limitation*") restricting such holders from beneficially owning a number of shares of Parent Common Stock in excess of the Beneficial Ownership Limitation or any agreement, transaction or other matter that is intended to, or would reasonably be expected to, impede, interfere with, delay, postpone, discourage or materially and adversely affect the consummation of the Parent Stockholder Matters; and (iii) to approve any proposal to adjourn or postpone the meeting to a later date, if there are not sufficient votes for the approval of the Parent Stockholder Matters on the date on which such meeting is held. Stockholder shall not take or commit or agree to take any action inconsistent with the foregoing.
- Expiration Date. As used in this Agreement, the term "Expiration Date" shall mean the earlier to occur of (a) the effective time of the approval of the Parent Stockholder Matters, (b) upon mutual written agreement of the Company, the Parent and Stockholder to terminate this Agreement or (c) nine (9) months following the date of this Agreement.
- 3. Additional Purchases. Stockholder agrees that any shares of capital stock or other equity securities of Parent that Stockholder purchases or with respect to which Stockholder otherwise acquires sole or shared voting power (including any proxy) after the execution of this Agreement and prior to the Expiration Date, whether by the exercise of any Parent Options, settlement of Parent RSUs or otherwise, including, without limitation, by gift, succession, in the event of a stock split or as a dividend or distribution of any Shares ("New Shares"), shall be subject to the terms and conditions of this Agreement to the same extent as if they constituted the Shares.
- Share Transfers. From and after the date hereof until the Expiration Date, Stockholder shall not, directly or indirectly, (a) sell, assign, transfer, 4. tender, or otherwise dispose of (including, without limitation, by the creation of any Liens (as defined in Section 5(c) below)) any Shares or any New Shares acquired, (b) deposit any Shares or New Shares into a voting trust or enter into a voting agreement or similar arrangement with respect to such Shares or New Shares or grant any proxy or power of attorney with respect thereto (other than this Agreement), (c) enter into any Contract, option, commitment or other arrangement or understanding with respect to the direct or indirect sale, transfer, assignment or other disposition of (including, without limitation, by the creation of any Liens) any Shares or New Shares, or (d) take any action that would make any representation or warranty of Stockholder contained herein untrue or incorrect or have the effect of preventing or disabling Stockholder from performing Stockholder's obligations under this Agreement. Notwithstanding the foregoing, Stockholder may make (1) transfers by will or by operation of Law or other transfers for estate-planning purposes, in which case this Agreement shall bind the transferee, (2) with respect to Stockholder's Parent Options which expire on or prior to the Expiration Date, transfers, sale, or other disposition of Shares or New Shares to Parent as payment for the (i) exercise price of Stockholder's Parent Options and (ii) taxes applicable to the exercise of Stockholder's Parent Options, (3) with respect to Stockholder's Parent RSUs, (i) transfers for the net settlement of Stockholder's Parent RSUs settled in Shares or New Shares (to pay any tax withholding obligations) or (ii) transfers for receipt upon settlement of Stockholder's Parent RSUs, and the sale of a sufficient number of such Shares acquired upon settlement of such securities as would generate sales proceeds sufficient to pay the aggregate taxes payable by Stockholder as a result of such settlement, (4) if Stockholder is a partnership or limited liability company, a transfer to one or more partners or members of Stockholder or to an Affiliated corporation, trust or other Entity under common control with Stockholder, or if Stockholder is a trust, a transfer to a beneficiary, provided that, in each such case the applicable transferee has signed a voting agreement in substantially the form hereof, (5) transfers to another holder of the capital stock of Parent that has signed a voting agreement in substantially the form hereof, and (6) transfers, sales or other dispositions as Parent may otherwise agree in writing in its sole discretion. If any voluntary or involuntary transfer of any Shares or New Shares covered hereby shall occur (including a transfer or disposition permitted by Section 4(1) through Section 4(6), sale by a Stockholder's trustee in bankruptcy, or a sale to a purchaser at any creditor's or court sale), (x) the transferee (which term, as used herein, shall include any and all transferees and subsequent transferees of the initial transferee) shall take and hold such Shares or New Shares subject to all of the restrictions, Liabilities and rights under this Agreement, which shall continue in full force and effect, and the transferee shall agree in writing to be bound by the terms and conditions of this Agreement and either the Stockholder or the transferee provides Parent with a copy of such agreement promptly upon consummation of any such transfer.

- 5. <u>Representations and Warranties of Stockholder</u>. Stockholder hereby represents and warrants to Parent and the Company as follows:
 - (a) If Stockholder is an Entity: (i) Stockholder is duly organized, validly existing and in good standing under the laws of the jurisdiction in which it is incorporated, organized or constituted, (ii) Stockholder has all necessary power and authority to execute and deliver this Agreement, to perform Stockholder's obligations hereunder and to consummate the transactions contemplated hereby, and (iii) the execution and delivery of this Agreement, performance of Stockholder's obligations hereunder and the consummation of the transactions contemplated hereby by Stockholder have been duly authorized by all necessary action on the part of Stockholder and no other proceedings on the part of Stockholder are necessary to authorize this Agreement, or to consummate the transactions contemplated hereby. If Stockholder is an individual, Stockholder has the legal capacity to execute and deliver this Agreement, to perform Stockholder's obligations hereunder and to consummate the transactions contemplated hereby;
 - (b) this Agreement has been duly executed and delivered by or on behalf of Stockholder and, assuming this Agreement constitutes a valid and binding agreement of the Company and Parent, constitutes a valid and binding agreement with respect to Stockholder, enforceable against Stockholder in accordance with its terms, except as enforcement may be limited by general principles of equity whether applied in a court of Law or a court of equity and by bankruptcy, insolvency and similar Laws affecting creditors' rights and remedies generally;
 - (c) Stockholder beneficially owns the number of Shares indicated opposite Stockholder's name on <u>Schedule 1</u>, and will own any New Shares, free and clear of any liens, claims, charges or other encumbrances or restrictions of any kind whatsoever ("*Liens*"), and has sole or shared, and otherwise unrestricted, voting power with respect to such Shares or New Shares and none of the Shares or New Shares is subject to any voting trust or other agreement, arrangement or restriction with respect to the voting of the Shares or the New Shares, except as contemplated by this Agreement;

- (d) the execution and delivery of this Agreement by Stockholder does not, and the performance by Stockholder of his, her or its obligations hereunder and the compliance by Stockholder with any provisions hereof will not, violate or conflict with, result in a material breach of or constitute a default (or an event that with notice or lapse of time or both would become a material default) under, or give to others any rights of termination, amendment, acceleration or cancellation of, or result in the creation of any Liens on any Shares or New Shares pursuant to, any agreement, instrument, note, bond, mortgage, Contract, lease, license, permit or other obligation or any order, arbitration award, judgment or decree to which Stockholder is a party or by which Stockholder is bound, or any Law, statute, rule or regulation to which Stockholder is subject or, in the event that Stockholder is a corporation, partnership, trust or other Entity, any bylaw or other Organizational Document of Stockholder; except for any of the foregoing as would not reasonably be expected to prevent or delay the performance by Stockholder of his, her or its obligations under this Agreement in any material respect;
- (e) the execution and delivery of this Agreement by Stockholder does not, and the performance of this Agreement by Stockholder does not and will not, require any consent, approval, authorization or permit of, or filing with or notification to, any Governmental Body or regulatory authority by Stockholder except for applicable requirements, if any, of the Exchange Act, and except where the failure to obtain such consents, approvals, authorizations or permits, or to make such filings or notifications, would not prevent or delay the performance by Stockholder of his, her or its obligations under this Agreement in any material respect;
- (f) no investment banker, broker, finder or other intermediary is entitled to a fee or commission from Parent or the Company in respect of this Agreement based upon any Contract made by or on behalf of Stockholder; and
- (g) as of the date of this Agreement, there is no Legal Proceeding pending or, to the knowledge of Stockholder, threatened against Stockholder that would reasonably be expected to prevent or delay the performance by Stockholder of his, her or its obligations under this Agreement in any material respect.
- 6. Irrevocable Proxy. Subject to the final sentence of this Section 6, by execution of this Agreement, Stockholder does hereby appoint Parent and any of its designees with full power of substitution and resubstitution, as Stockholder's true and lawful attorney and irrevocable proxy, to the fullest extent of Stockholder's rights with respect to the Shares or New Shares, to vote and exercise all voting and related rights, including the right to sign Stockholder's name (solely in its capacity as a stockholder) to any stockholder consent, if Stockholder is unable to perform or otherwise does not perform his, her or its obligations under this Agreement, with respect to such Shares solely with respect to the matters set forth in Section 1 hereof. Stockholder intends this proxy to be irrevocable and coupled with an interest hereunder until the Expiration Date, hereby revokes any proxy previously granted by Stockholder with respect to the Shares or New Shares and represents that none of such previously-granted proxies are irrevocable. The irrevocably proxy and power of attorney granted herein shall survive the death or incapacity of Stockholder and the obligations of Stockholder shall be binding on Stockholder's heirs, personal representatives, successors, transferees and assigns. Stockholder hereby agrees not to grant any subsequent powers of attorney or proxies with respect to any Shares or New Shares with respect to the matters set forth in Section 1 until after the Expiration Date. The Stockholder hereby affirms that the proxy set forth in this Section 6 is given in connection with and granted in consideration of and as an inducement to the Company, Parent, First Merger Sub and Second Merger Sub to enter into the Merger Agreement and that such proxy is given to secure the obligations of the Stockholder under Section 1. Notwithstanding anything contained herein to the contrary, this irrevocable proxy shall automatically terminate upon the Expiration Date.

- 7. <u>Other Remedies; Specific Performance</u>. Except as otherwise provided herein, any and all remedies herein expressly conferred upon a party will be deemed cumulative with, and not exclusive of, any other remedy conferred hereby, or by Law or equity upon such party, and the exercise by a party of any one remedy will not preclude the exercise of any other remedy. The parties hereto agree that irreparable damage would occur in the event that any of the provisions of this Agreement were not performed in accordance with their specific terms or were otherwise breached. It is accordingly agreed that the parties shall be entitled to an injunction or injunctions to prevent breaches of this Agreement and to enforce specifically the terms and provisions hereof without the need of posting bond in any court of the United States or any state having jurisdiction, this being in addition to any other remedy to which they are entitled at Law or in equity.
- 8. <u>Directors and Officers</u>. This Agreement shall apply to Stockholder solely in Stockholder's capacity as a stockholder of Parent and/or holder of Parent Options and/or Parent RSUs and not in Stockholder's capacity as a director, officer or employee of Parent or any of its Subsidiaries or in Stockholder's capacity as a trustee or fiduciary of any employee benefit plan or trust. Notwithstanding any provision of this Agreement to the contrary, nothing in this Agreement shall (or require Stockholder to attempt to) limit or restrict a director and/or officer of Parent in the exercise of his or her fiduciary duties as a director and/or officer of Parent or in his or her capacity as a trustee or fiduciary of any employee benefit plan or trust or prevent or be construed to create any obligation on the part of any director and/or officer, trustee and/or fiduciary.
- 9. <u>No Ownership Interest</u>. Nothing contained in this Agreement shall be deemed to vest in the Company any direct or indirect ownership or incidence of ownership of or with respect to any Shares or New Shares. All rights, ownership and economic benefits of and relating to the Shares or New Shares shall remain vested in and belong to Stockholder, and the Company does not have authority to manage, direct, superintend, restrict, regulate, govern, or administer any of the policies or operations of Parent or exercise any power or authority to direct Stockholder in the voting of any of the Shares or New Shares, except as otherwise provided herein.
- 10. <u>Termination</u>. This Agreement shall terminate and shall have no further force or effect as of the Expiration Date. Notwithstanding the foregoing, upon termination or expiration of this Agreement, no party shall have any further obligations or liabilities under this Agreement; *provided*, *however*, nothing set forth in this <u>Section 10</u> or elsewhere in this Agreement shall relieve any party from liability for any fraud or for any willful and material breach of this Agreement prior to termination hereof.
- 11. <u>Further Assurances</u>. Stockholder shall, from time to time, execute and deliver, or cause to be executed and delivered, such additional or further consents, documents and other instruments as the Company or Parent may reasonably request for the purpose of effectively carrying out the transactions contemplated by this Agreement and the Contemplated Transactions.
- 12. Disclosure. Stockholder hereby agrees that Parent and the Company may publish and disclose in any registration statement, any prospectus filed with any regulatory authority in connection with the Contemplated Transactions and any related documents filed with such regulatory authority and as otherwise required by Law, Stockholder's identity and ownership of Shares and the nature of Stockholder's commitments, arrangements and understandings under this Agreement and may further file this Agreement as an exhibit to any registration statement or prospectus or in any other filing made by Parent or the Company as required by Law or the terms of the Merger Agreement, including with the SEC or other regulatory authority, relating to the Contemplated Transactions, all subject to prior review and an opportunity to comment by Stockholder's counsel. Prior to the Closing, Stockholder shall not, and shall use its reasonable best efforts to cause its representatives not to, directly or indirectly, make any press release, public announcement or other public communication that criticizes or disparages this Agreement or the Merger Agreement or any of the Contemplated Transactions, without the prior written consent of Parent and the Company, *provided that*, the foregoing shall not limit or affect any actions taken by Stockholder (or any affiliated officer or director of Stockholder) that would be permitted to be taken by Stockholder, Parent or the Company pursuant to the Merger Agreement; *provided, further*, that the foregoing shall not effect any actions of Stockholder the prohibition of which would be prohibited under applicable Law.

5

- 13. <u>Notice</u>. All notices and other communications hereunder shall be in writing and shall be deemed given if delivered personally or sent by overnight courier (providing proof of delivery), by electronic transmission (providing confirmation of transmission) to the Company or Parent, as the case may be, in accordance with Section 8.8 of the Merger Agreement and to Stockholder at his, her or its address or email address (providing confirmation of transmission) set forth on <u>Schedule 1</u> attached hereto (or at such other address for a party as shall be specified by like notice).
- 14. Severability. Any term or provision of this Agreement that is invalid or unenforceable in any situation in any jurisdiction shall not affect the validity or enforceability of the remaining terms and provisions of this Agreement or the validity or enforceability of the offending term or provision in any other situation or in any other jurisdiction. If a final judgment of a court of competent jurisdiction declares that any term or provision of this Agreement is invalid or unenforceable, the parties hereto agree that the court making such determination shall have the power to limit such term or provision, to delete specific words or phrases or to replace such term or provision, and this Agreement shall be valid and enforceable and that comes closest to expressing the intention of the invalid or unenforceable term or provision, and this Agreement shall be valid and enforceable as so modified. In the event such court does not exercise the power granted to it in the prior sentence, the parties hereto agree to replace such invalid or unenforceable term or provision that will achieve, to the extent possible, the economic, business and other purposes of such invalid or unenforceable term or provision.
- 15. <u>Assignability</u>. This Agreement shall be binding upon, and shall be enforceable by and inure solely to the benefit of, the parties hereto and their respective successors and assigns; *provided*, *however*, that neither this Agreement nor any of a party's rights or obligations hereunder may be assigned or delegated by such party without the prior written consent of the other parties hereto, and any attempted assignment or delegation of this Agreement or any of such rights or obligations by such party without the other party's prior written consent shall be void and of no effect. Nothing in this Agreement, express or implied, is intended to or shall confer upon any Person (other than the parties hereto) any right, benefit or remedy of any nature whatsoever under or by reason of this Agreement.
- 16. <u>No Waivers</u>. No waivers of any breach of this Agreement extended by the Company or Parent to Stockholder shall be construed as a waiver of any rights or remedies of the Company or Parent, as applicable, with respect to any other stockholder of Parent who has executed an agreement substantially in the form of this Agreement with respect to Shares or New Shares held or subsequently held by such stockholder or with respect to any subsequent breach of Stockholder or any other stockholder of Parent. No waiver of any provisions hereof by any party shall be deemed a waiver of any other provisions hereof by any such party, nor shall any such waiver be deemed a continuing waiver of any provision hereof by such party.

- 17. <u>Applicable Law; Jurisdiction</u>. This Agreement shall be governed by, and construed in accordance with, the Laws of the state of Delaware, regardless of the Laws that might otherwise govern under applicable principles of conflicts of Laws. In any action or Legal Proceeding between any of the parties arising out of or relating to this Agreement, each of the parties: (a) irrevocably and unconditionally consents and submits to the exclusive jurisdiction and venue of the Court of Chancery of the state of Delaware or to the extent such court does not have subject matter jurisdiction, the Superior Court of the State of Delaware or the United States District Court for the District of Delaware, (b) agrees that all claims in respect of such action or Legal Proceeding shall be heard and determined exclusively in accordance with clause (a) of this <u>Section 17</u>, (c) waives any objection to laying venue in any such action or Legal Proceeding in such courts, (d) waives any objection that such courts are an inconvenient forum or do not have jurisdiction over any party, and (e) agrees that service of process upon such party in any such action or Legal Proceeding shall be effective if notice is given in accordance with <u>Section 13</u> of this Agreement.
- 18. <u>Waiver of Jury Trial</u>. THE PARTIES HERETO HEREBY WAIVE ANY RIGHT TO TRIAL BY JURY WITH RESPECT TO ANY ACTION OR LEGAL PROCEEDING RELATED TO OR ARISING OUT OF THIS AGREEMENT, ANY DOCUMENT EXECUTED IN CONNECTION HEREWITH AND THE MATTERS CONTEMPLATED HEREBY AND THEREBY.
- 19. <u>No Agreement Until Executed</u>. Irrespective of negotiations among the parties or the exchanging of drafts of this Agreement, this Agreement shall not constitute or be deemed to evidence a Contract, agreement, arrangement or understanding between the parties hereto unless and until (a) the Parent Board has approved, for purposes of any applicable anti-takeover Laws and regulations and any applicable provision of the certificate of incorporation of Parent, the Merger Agreement and the Contemplated Transactions, (b) the Merger Agreement is executed by all parties thereto, and (c) this Agreement is executed by all parties hereto.
- 20. <u>Entire Agreement; Counterparts; Exchanges by Electronic Transmission</u>. This Agreement and the other agreements referred to in this Agreement constitute the entire agreement and supersede all prior agreements and understandings, both written and oral, among or between any of the parties with respect to the subject matter hereof and thereof. This Agreement may be executed in several counterparts, each of which shall be deemed an original and all of which shall constitute one and the same instrument. The exchange of a fully executed Agreement (in counterparts or otherwise) by all parties by electronic transmission via ".pdf" shall be sufficient to bind the parties to the terms and conditions of this Agreement.
- 21. <u>Amendment</u>. This Agreement may not be amended, supplemented or modified, and no provisions hereof may be modified or waived, except by an instrument in writing signed on behalf of each party hereto; *provided*, *however*, that the rights or obligations of any Stockholder may be waived, amended or otherwise modified in a writing signed by Parent (for the avoidance of doubt, with the prior written approval required by Section 4.1 of Parent's Certificate of Designation of Preferences, Rights and Limitations of Series A Non-Voting Convertible Preferred Stock), the Company and Stockholder.
- 22. <u>Fees and Expenses</u>. Except as otherwise specifically provided herein, the Merger Agreement or any other agreement contemplated by the Merger Agreement to which a party hereto is a party, each party hereto shall bear its own expenses in connection with this Agreement and the transactions contemplated hereby.

- 23. <u>Voluntary Execution of Agreement</u>. This Agreement is executed voluntarily and without any duress or undue influence on the part or behalf of the parties. Each of the parties hereby acknowledges, represents and warrants that (a) it has read and fully understood this Agreement and the implications and consequences thereof; (b) it has been represented in the preparation, negotiation, and execution of this Agreement by legal counsel of its own choice, or it has made a voluntary and informed decision to decline to seek such counsel; and (c) it is fully aware of the legal and binding effect of this Agreement.
- 24. <u>Construction</u>.
 - (a) For purposes of this Agreement, whenever the context requires: the singular number shall include the plural, and vice versa; the masculine gender shall include the feminine and neuter genders; the feminine gender shall include the masculine and neuter genders; and the neuter gender shall include masculine and feminine genders.
 - (b) The parties hereto agree that any rule of construction to the effect that ambiguities are to be resolved against the drafting party shall not be applied in the construction or interpretation of this Agreement.
 - (c) As used in this Agreement, the words "include" and "including," and variations thereof, shall not be deemed to be terms of limitation, but rather shall be deemed to be followed by the words "without limitation."
 - (d) Except as otherwise indicated, all references in this Agreement to "Sections," and "Schedules" are intended to refer to Sections of this Agreement and Schedules to this Agreement, respectively.
 - (e) The underlined headings contained in this Agreement are for convenience of reference only, shall not be deemed to be a part of this Agreement and shall not be referred to in connection with the construction or interpretation of this Agreement.

[Remainder of Page Has Intentionally Been Left Blank]

[STOCKHOLDER]

Signature: Name (if an Entity): Title (if an Entity):

[Signature Page to Support Agreement]

ONCONOVA THERAPEUTICS, INC.

By: Name: Title:

[Signature Page to Support Agreement]

TRAWSFYNYDD THERAPEUTICS, INC.

By: Name: Title:

[Signature Page to Support Agreement]

SCHEDULE 1

Name, Address and Email Address of Stockholder

Steven M. Fruchtman 863 Riverside Drive New York, NY 10032 sfruchtman@onconova.us Shares of Parent Common Stock

Parent Options

Parent RSUs

(See attached)

Paren

EXHIBIT F

FORM OF COMPANY STOCKHOLDER SUPPORT AGREEMENT

TRAWSFYNYDD THERAPEUTICS, INC.

SUPPORT AGREEMENT

THIS SUPPORT AGREEMENT (this "*Agreement*"), dated as of April 1, 2024, is made by and among Onconova Therapeutics, Inc., a Delaware corporation ("*Parent*"), Trawsfynydd Therapeutics, Inc., a Delaware corporation (the "*Company*"), and the undersigned holder ("*Stockholder*") of shares of capital stock (the "*Shares*") of the Company.

WHEREAS, Parent, Traws Merger Sub I, Inc., a Delaware corporation and a wholly owned subsidiary of Parent ("*First Merger Sub*"), Traws Merger Sub II, LLC, a Delaware limited liability company and a wholly owned subsidiary of Parent ("*Second Merger Sub*"), and the Company, have entered into an Agreement and Plan of Merger, dated of even date herewith (the "*Merger Agreement*"), providing for the merger of First Merger Sub with and into the Company (the "*First Merger*") and the merger of the Company with and into Second Merger Sub (the "*Second Merger*" and, together with the First Merger, the "*Merger*");

WHEREAS, at the effective time of the First Merger (the "*First Effective Time*"), each issued and outstanding share in the capital stock of the Company (other than shares in respect of which dissenters or appraisal rights have been properly exercised and perfected under Delaware law and shares held in treasury) will be cancelled and cease to exist in exchange for the right to receive shares of common stock or preferred stock of Parent (the "*Merger Consideration*") as set forth on the Allocation Certificate;

WHEREAS, Stockholder beneficially owns and has sole voting power with respect to the number of Shares, and holds options to purchase shares of Company Common Stock ("*Company Options*"), in each case in the number of Shares indicated opposite Stockholder's name on <u>Schedule 1</u> attached hereto;

WHEREAS, as an inducement and a condition to the willingness of Parent, First Merger Sub, Second Merger Sub and the Company to enter into the Merger Agreement, and in consideration of the substantial expenses incurred and to be incurred by them in connection therewith, Stockholder has agreed to enter into and perform this Agreement; and

WHEREAS, all capitalized terms used in this Agreement without definition herein shall have the meanings ascribed to them in the Merger Agreement.

NOW, THEREFORE, in consideration of, and as a condition to, Parent, First Merger Sub, Second Merger Sub and the Company's entering into the Merger Agreement and proceeding with the transactions contemplated thereby, and in consideration of the substantial expenses incurred and to be incurred by them in connection therewith, Stockholder, Parent and the Company agree as follows:

- 1. <u>Preliminary Matters</u>. Stockholder understands that this Agreement is being provided to both the Company and Parent in connection with the Merger, and that the Company and Parent are relying upon the representations, warranties, covenants and agreements of Stockholder set forth in this Agreement.
- 2. Waiver of Dissenters' Rights. By signing this Agreement, Stockholder hereby irrevocably and unconditionally (a) waives and agrees to refrain from exercising any dissenter's rights, appraisal rights or similar rights (collectively, "*Dissenter's Rights*") Stockholder may have with respect to all of the Shares or other equity interests of the Company Stockholder owns that may arise with respect to the Merger or any of the transactions contemplated by the Merger Agreement, including, without limitation, Dissenter's Rights under the Delaware General Corporation Law (the "*DGCL*"), and (b) agrees that Stockholder will not bring, commence, institute, maintain, prosecute, participate in or voluntarily aid any action, in law or in equity, in any court or before any governmental entity, which (i) challenges the validity of or seeks to enjoin the operation of any provision of this Agreement, the Letter of Transmittal, the Merger Agreement or any certificate or other writing delivered by or on behalf of Stockholder at or prior to the Closing or (ii) alleges that the execution and delivery of this Agreement or the Letter of Transmittal by Stockholder, or the approval of the Merger Agreement by the Board of Directors of the Company, breaches any fiduciary duty of the Board of Directors of the Company or any member thereof. Stockholder hereby acknowledges Stockholder's right to dissent to the Merger and request an appraisal of the fair market value of the securities held by Stockholder pursuant to the DGCL, and that by signing this Agreement, Stockholder irrevocably waives the Stockholder's rights for such securities in accordance with the DGCL.

- 3. <u>Transaction Documents</u>. IN ADDITION, STOCKHOLDER HAS READ, UNDERSTANDS AND AGREES TO ALL OF THE TERMS AND CONDITIONS SET FORTH IN THE MERGER AGREEMENT, THE ANCILLARY DOCUMENTS TO WHICH THE STOCKHOLDER IS BOUND, THE LETTER OF TRANSMITTAL AND THE ACCOMPANYING INSTRUCTIONS.
- 4. <u>Representations and Warranties of Stockholder</u>. Stockholder hereby represents, warrants and covenants to the Company and Parent that as of the date of this Agreement and as of the First Effective Time:
 - (a) If Stockholder is an Entity: (i) Stockholder is duly organized, validly existing and in good standing under the laws of the jurisdiction in which it is incorporated, organized or constituted, (ii) Stockholder has all necessary power and authority to execute and deliver this Agreement, to perform Stockholder's obligations hereunder and to consummate the transactions contemplated hereby, and (iii) the execution and delivery of this Agreement, performance of Stockholder's obligations hereunder and the consummation of the transactions contemplated hereby by Stockholder have been duly authorized by all necessary action on the part of Stockholder and no other proceedings on the part of Stockholder are necessary to authorize this Agreement, or to consummate the transactions contemplated hereby. If Stockholder is an individual, Stockholder has the legal capacity to execute and deliver this Agreement, to perform Stockholder's obligations hereunder and to consummate the transactions contemplated hereby;
 - (b) this Agreement has been duly executed and delivered by or on behalf of Stockholder and, assuming this Agreement constitutes a valid and binding agreement of the Company and Parent, constitutes a valid and binding agreement with respect to Stockholder, enforceable against Stockholder in accordance with its terms, except as enforcement may be limited by general principles of equity whether applied in a court of Law or a court of equity and by bankruptcy, insolvency and similar Laws affecting creditors' rights and remedies generally;
 - (c) (i) Stockholder is the beneficial owner of the Shares and Company Options set forth in <u>Schedule 1</u> below, free and clear of any and all Encumbrances of any nature or kind whatsoever (other than those imposed by applicable securities laws or the Company's organizational documents, as in effect on the date hereof), (ii) except for the Shares and Company Options set forth on <u>Schedule 1</u>, Stockholder is not a beneficial owner or record holder of any other equity securities or rights of the Company, and (iii) Stockholder has full right, power, legal capacity and authority to sell, transfer, deliver and surrender the Shares, free and clear of all Encumbrances except pursuant to the Company's organizational documents;

- (d) the information included by or on behalf of Stockholder in the Letter of Transmittal is true, correct and complete; and
- (e) Stockholder is an "accredited investor" as defined in Rule 501(a) under the Securities Act.
- 5. <u>Merger Consideration</u>. Stockholder acknowledges that (a) each holder of Shares will be entitled to receive a portion of the Merger Consideration based on the Exchange Ratio in the Merger Agreement and as set forth on the Allocation Schedule, (b) such portion of the Merger Consideration as set forth on the Allocation Schedule represents all consideration to which Stockholder is entitled in respect of Stockholder's Shares, (c) payment of the Merger Consideration is conditioned on the closing of the Merger, and (d) to receive payment of the Merger Consideration represented by Stockholder's Shares, Stockholder must complete and sign a Letter of Transmittal, including the applicable schedules and exhibits, and deliver the Letter of Transmittal to Equiniti Trust Company, LLC (the "*Exchange Agent*").
- 6. <u>Disposition of Shares</u>. Pursuant to the Merger Agreement, effective as of the First Effective Time, Stockholder hereby surrenders, cancels and terminates Stockholder's Shares, in exchange for Stockholder's share of the Merger Consideration determined in accordance with the Merger Agreement, subject to the terms and conditions of the Merger Agreement. Stockholder hereby authorizes and instructs Parent to prepare in the name of and deliver to the address indicated in the Letter of Transmittal book-entry shares representing the portion of the Merger Consideration due to Stockholder as a result of the Merger.

7. <u>Release of Claims</u>.

- (a) Stockholder irrevocably, unconditionally and completely releases, acquits and forever discharges each of the Parent Released Parties (as defined below) and Company Released Parties (as defined below) from any Claim (as defined below), and hereby irrevocably, unconditionally and completely waives and relinquishes each and every Claim that Holder may have had in the past, may now have or may have in the future against any of the Parent Released Parties or Company Released Parties, directly or indirectly relating to or directly or indirectly arising out of: (a) any written or oral agreements or arrangements occurring, existing or entered into by Stockholder at any time up to and including the First Effective Time; and (b) any events, matters, causes, things, acts, omissions or conduct, occurring or existing at any time up to and including the First Effective Time; *provided, however*, that Stockholder is not releasing any rights available to it under the Merger Agreement or any other agreement entered into by Stockholder in connection with the closing of the transactions contemplated by the Merger Agreement.
- (b) The term "*Parent Released Parties*" means: (a) Parent, First Merger Sub, Second Merger Sub and any of their respective Affiliates and (b) the current or former equity holders, officers, directors, managers, employees and agents (in their respective capacities as such) of Parent, First Merger Sub, Second Merger Sub and any of their respective Affiliates and the successor and assigns of all of the foregoing.

- (c) The term "<u>Company Released Parties</u>" means: (a) the Company and any of its respective Affiliates and (b) the current or former equity holders, officers, directors, managers, employees and agents (in their respective capacities as such) of the Company and its respective Affiliates and the successor and assigns of all of the foregoing.
- (d) The term "<u>Claim</u>" means all past, present and future disputes, claims, controversies, demands, rights, obligations, liabilities, actions and causes of action of every kind and nature (whether matured or unmatured, absolute or contingent) that may be asserted or exercised by Stockholder in Stockholder's capacity as a holder of Shares.
- (e) Stockholder (a) represents, warrants and acknowledges that Stockholder has been fully advised by the Stockholder's attorney of the contents of Section 1542 of the Civil Code of the State of California; and (b) hereby expressly waives the benefits thereof and any rights that Holder may have thereunder. Section 1542 of the Civil Code of the State of California provides as follows:

A GENERAL RELEASE DOES NOT EXTEND TO CLAIMS WHICH THE CREDITOR OR RELEASING PARTY DOES NOT KNOW OR SUSPECT TO EXIST IN HIS OR HER FAVOR AT THE TIME OF EXECUTING THE RELEASE AND THAT, IF KNOWN BY HIM OR HER, WOULD HAVE MATERIALLY AFFECTED HIS OR HER SETTLEMENT WITH THE DEBTOR OR RELEASED PARTY.

- 8. <u>Other Remedies; Specific Performance</u>. Except as otherwise provided herein, any and all remedies herein expressly conferred upon a party will be deemed cumulative with, and not exclusive of, any other remedy conferred hereby, or by Law or equity upon such party, and the exercise by a party of any one remedy will not preclude the exercise of any other remedy. The parties hereto agree that irreparable damage would occur in the event that any of the provisions of this Agreement were not performed in accordance with their specific terms or were otherwise breached. It is accordingly agreed that the parties shall be entitled to an injunction or injunctions to prevent breaches of this Agreement and to enforce specifically the terms and provisions hereof without the need of posting bond in any court of the United States or any state having jurisdiction, this being in addition to any other remedy to which they are entitled at Law or in equity.
- 9. <u>Notice</u>. All notices and other communications hereunder shall be in writing and shall be deemed given if delivered personally or sent by overnight courier (providing proof of delivery), by electronic transmission (providing confirmation of transmission) to the Company or Parent, as the case may be, in accordance with Section 8.8 of the Merger Agreement and to Stockholder at his, her or its address or email address (providing confirmation of transmission) set forth on <u>Schedule 1</u> attached hereto (or at such other address for a party as shall be specified by like notice).
- 10. <u>Severability</u>. Any term or provision of this Agreement that is invalid or unenforceable in any situation in any jurisdiction shall not affect the validity or enforceability of the remaining terms and provisions of this Agreement or the validity or enforceability of the offending term or provision in any other situation or in any other jurisdiction. If a final judgment of a court of competent jurisdiction declares that any term or provision of this Agreement is invalid or unenforceable, the parties hereto agree that the court making such determination shall have the power to limit such term or provision, to delete specific words or phrases or to replace such term or provision, and this Agreement shall be valid and enforceable and that comes closest to expressing the intention of the invalid or unenforceable term or provision, and this Agreement shall be valid and enforceable as so modified. In the event such court does not exercise the power granted to it in the prior sentence, the parties hereto agree to replace such invalid or unenforceable term or provision that will achieve, to the extent possible, the economic, business and other purposes of such invalid or unenforceable term or provision.

- 11. <u>Assignability</u>. This Agreement shall be binding upon, and shall be enforceable by and inure solely to the benefit of, the parties hereto and their respective successors and assigns; *provided*. *however*, that neither this Agreement nor any of a party's rights or obligations hereunder may be assigned or delegated by such party without the prior written consent of the other parties hereto, and any attempted assignment or delegation of this Agreement or any of such rights or obligations by such party without the other party's prior written consent shall be void and of no effect. Nothing in this Agreement, express or implied, is intended to or shall confer upon any Person (other than the parties hereto) any right, benefit or remedy of any nature whatsoever under or by reason of this Agreement.
- 12. <u>No Waivers</u>. No waivers of any breach of this Agreement extended by the Company or Parent to Stockholder shall be construed as a waiver of any rights or remedies of the Company or Parent, as applicable, with respect to any other stockholder of the Company who has executed an agreement substantially in the form of this Agreement with respect to Shares held by such stockholder or with respect to any subsequent breach of Stockholder or any other stockholder of the Company. No waiver of any provisions hereof by any party shall be deemed a waiver of any other provisions hereof by any such party, nor shall any such waiver be deemed a continuing waiver of any provision hereof by such party.
- 13. <u>Applicable Law; Jurisdiction</u>. This Agreement shall be governed by, and construed in accordance with, the Laws of the state of Delaware, regardless of the Laws that might otherwise govern under applicable principles of conflicts of Laws. In any action or Legal Proceeding between any of the parties arising out of or relating to this Agreement, each of the parties: (a) irrevocably and unconditionally consents and submits to the exclusive jurisdiction and venue of the Court of Chancery of the state of Delaware or to the extent such court does not have subject matter jurisdiction, the Superior Court of the State of Delaware or the United States District Court for the District of Delaware, (b) agrees that all claims in respect of such action or Legal Proceeding shall be heard and determined exclusively in accordance with clause (a) of this <u>Section 17</u>, (c) waives any objection to laying venue in any such action or Legal Proceeding in such courts, (d) waives any objection that such courts are an inconvenient forum or do not have jurisdiction over any party, and (e) agrees that service of process upon such party in any such action or Legal Proceeding shall be effective if notice is given in accordance with <u>Section 13</u> of this Agreement.
- 14. <u>Waiver of Jury Trial</u>. THE PARTIES HERETO HEREBY WAIVE ANY RIGHT TO TRIAL BY JURY WITH RESPECT TO ANY ACTION OR LEGAL PROCEEDING RELATED TO OR ARISING OUT OF THIS AGREEMENT, ANY DOCUMENT EXECUTED IN CONNECTION HEREWITH AND THE MATTERS CONTEMPLATED HEREBY AND THEREBY.
- 15. Entire Agreement; Counterparts; Exchanges by Electronic Transmission. This Agreement and the other agreements referred to in this Agreement constitute the entire agreement and supersede all prior agreements and understandings, both written and oral, among or between any of the parties with respect to the subject matter hereof and thereof. This Agreement may be executed in several counterparts, each of which shall be deemed an original and all of which shall constitute one and the same instrument. The exchange of a fully executed Agreement (in counterparts or otherwise) by all parties by electronic transmission via ".pdf" shall be sufficient to bind the parties to the terms and conditions of this Agreement.

- 16. <u>Amendment</u>. This Agreement may not be amended, supplemented or modified, and no provisions hereof may be modified or waived, except by an instrument in writing signed on behalf of each party hereto.
- 17. <u>Voluntary Execution of Agreement</u>. This Agreement is executed voluntarily and without any duress or undue influence on the part or behalf of the parties. Each of the parties hereby acknowledges, represents and warrants that (a) it has read and fully understood this Agreement and the implications and consequences thereof; (b) it has been represented in the preparation, negotiation, and execution of this Agreement by legal counsel of its own choice, or it has made a voluntary and informed decision to decline to seek such counsel; and (c) it is fully aware of the legal and binding effect of this Agreement.
- 18. <u>Construction</u>.
 - (a) For purposes of this Agreement, whenever the context requires: the singular number shall include the plural, and vice versa; the masculine gender shall include the feminine and neuter genders; the feminine gender shall include the masculine and neuter genders; and the neuter gender shall include masculine and feminine genders.
 - (b) The parties hereto agree that any rule of construction to the effect that ambiguities are to be resolved against the drafting party shall not be applied in the construction or interpretation of this Agreement.
 - (c) As used in this Agreement, the words "include" and "including," and variations thereof, shall not be deemed to be terms of limitation, but rather shall be deemed to be followed by the words "without limitation."
 - (d) Except as otherwise indicated, all references in this Agreement to "Sections," and "Schedules" are intended to refer to Sections of this Agreement and Schedules to this Agreement, respectively.
 - (e) The underlined headings contained in this Agreement are for convenience of reference only, shall not be deemed to be a part of this Agreement and shall not be referred to in connection with the construction or interpretation of this Agreement.

[Remainder of Page Has Intentionally Been Left Blank]

[STOCKHOLDER]

Signature: Name (if an Entity): Title (if an Entity):

[Signature Page to Company Support Agreement]

ONCONOVA THERAPEUTICS, INC.

By: Name: Title:

[Signature Page to Company Support Agreement]

TRAWSFYNYDD THERAPEUTICS, INC.

By: Name: Title:

[Signature Page to Company Support Agreement]

SCHEDULE 1

Name, Address and Email Address of	Shares of Company	Company Options
Stockholder	Common Stock	
Nikolay Savchuk	100,000	188,925
P.O. Box 675566		
Rancho Santa Fe, CA 92067		
nsavchuk@torreypinesinv.com		
Iain Dukes	100,000	188,925
9002 Crefeld Street		
Philadelphia, PA 19118		
DukesI@OrbiMed.com		
OrbiMed Private Investments VIII, LP	1,327,888	
601 Lexington Avenue, 54th Floor		
New York, NY 10022		
Legal@OrbiMed.com		
TPAV, LLC	1,327,888	
12730 High Bluff Drive, Suite 100		
San Diego, CA 92130		
nsavchuk@torreypinesinv.com		
Viriom, Inc.	2,466,888	
12730 High Bluff Drive, Suite 100		
San Diego, CA 92130		
nsavchuk@torreypinesinv.com		

EXHIBIT G

LETTER OF TRANSMITTAL

Exchange of Common Stock of **Trawsfynydd Therapeutics**, **Inc.** for Shares of Onconova Therapeutics., **Inc.** – CUSIP # 68232V801

All shareholders must complete Boxes A, B, and G. Please also read the "General Instructions" on page 2.

FOR OFFICE USE ONLY	Approved
---------------------	----------

W-9 Completed _

BOX A – Signature of Registered Shareholders	BOX B – Certificate(s) Enclosed		
(Must be signed by all registered shareholders; include legal capacity if signing on behalf of an entity)	Certificate Number(s) (Attach additional signed list, if necessary)	Number of Shares Represented by Each Certificate	
Signature			
Signature			
Telephone Number	Total Shares Surrendered:		

□ Lost Certificates. I have lost my certificate(s) for ________ shares and require assistance in replacing them. A \$75.00 replacement fee must be sent in with the completed letter of transmittal. The check must be made payable to Shareowner Services. If additional forms and fees are needed, you will be contacted.

BOX C – New Registration Instructions	BOX D – One Time Delivery Instructions		
To be completed <i>ONLY</i> if the shares are to be issued in the name(s) of someone other than the registered holder(s) in Box E. ISSUE TO:	To be completed ONLY if the shares are to be delivered to an address other than that listed in Box E. MAIL TO:		
Name	Name		
Street Address	Street Address		
City, State and Zip Code	City, State and Zip Code		

Please remember to complete and sign the Substitute Form W-9 in Box G on the next page or, if applicable, the enclosed W-8BEN form.

BOX E – Name and Address of Registered Holder(s)	BOX F – Medallion Guarantee		
Please make any address corrections below	If you have completed Box C,D or E your signature must be <i>Medallion Guaranteed</i> by an eligible financial institution.		
□ indicates permanent address change	Note: A notarization by a notary public is not acceptable		

BOX G – Request for Taxpayer Identification Number and Certification – Substitute Form W-9 (Rev. 10/18)

Certification: Under penalties of perjury, I certify that:

- 1. The number shown on this form is my correct taxpayer identification number (or I am waiting for a number to be issued to me), and
- 2. I am not subject to backup withholding because: (a) I am exempt from backup withholding, (b) I have not been notified by the Internal Revenue Service (IRS) that I am subject to backup withholding as a result of a failure to report all interest or dividends, or (c) the IRS has notified me that I am no longer subject to backup withholding, and
- 3. I am a U.S. citizen or other U.S. person (as defined in the instructions below), and
- 4. The FATCA code entered on this form (if any) indicating that the payee is exempt from FATCA reporting is correct. (No FATCA reporting code is required for accounts maintained In the United States.)

FATCA Exemption Reporting Code: _____ (Codes are available with the official IRS Form W-9 found at www.irs.gov.)

Certification instructions. You must cross out item 2 above if you have been notified by the IRS that you are currently subject to backup withholding because you have failed to report all interest and dividends on your tax return.

Required: Check appropriate box for federal tax classification:

 \Box Individual/sole proprietor \Box C Corporation \Box S Corporation \Box Partnership \Box Trust/estate

□ Limited liability company. Enter tax classification (C=C corporation, S=S corporation, P=partnership):

The Internal Revenue Service does not require your consent to any provision of this document other than the certifications required to avoid backup withholding.

Signature:

Date:

NOTICE TO NON-RESIDENT ALIEN INDIVIDUALS OR FOREIGN ENTITIES (e.g. foreign corporation, partnership or trusts): DO NOT COMPLETE THE ABOVE SUBSTITUTE FORM W-9. NRA INDIVIDUALS SHOULD COMPLETE THE APPLICABLE FORM W-8BEN. FORMS ARE AVAILABLE AT <u>www.irs.gov</u>. FOREIGN ENTITIES NEED TO COMPLETE THE APPLICABLE TYPE OF FORM W-8 AND RETURN THAT CERTIFICATION OF FOREIGN TAX STATUS. FAILURE TO COMPLETE THE FORM WILL SUBJECT THE RECIPIENT TO THE APPLICABLE FEDERAL TAX INCOME TAX WITHHOLDING.

General Instructions

• BOX A-Signatures: All registered shareholders must sign as indicated in Box A. If you are signing on behalf of a registered shareholder or entity your signature must include your legal capacity.

YOUR GUARANTOR (BANK/BROKER) WILL REQUIRE PROOF OF YOUR AUTHORITY TO ACT. CONSULT YOUR GUARANTOR FOR THEIR SPECIFIC REQUIREMENTS. YOU OR YOUR GUARANTOR MAY ACCESS THE SECURITIES TRANSFER ASSOCIATION (STA) RECOMMENDED REQUIREMENTS ON-LINE AT www.stai.org.

- BOX B-Certificate Detail: List all certificate numbers and shares submitted in Box B. Any book-entry shares held by you will be automatically exchanged upon receipt of this properly completed Letter of Transmittal. If your certificates are lost, please check the box and complete below Box A. A \$75 lost certificate replacement fee must be mailed in with your completed Letter of Transmittal. If there are additional forms or fees needed, you will be contacted.
- BOX C-New Registration: Provide the new registration instructions (name and address) in Box C. Complete Box G to certify tax identification number for new registration of U.S. citizen, resident or entity. Signature must be that of the new registration indicated. See notice to non-resident aliens above. All changes in registration require a Medallion Signature Guarantee. Joint registrations must include the form of tenancy. Custodial registrations must include the name of the Custodian (only one). Trust account registrations must include the names of all current acting trustees and the date of the trust agreement. If this transaction results in proceeds at or above \$14,000,000.00 in value please contact EQ Shareowner Services at the number listed below.
- BOX D-One Time Delivery: Any address shown in Box D will be treated as a one-time only mailing instruction.
- BOX E-Current Name and Address of Registered Shareholder: If your permanent address should be changed on EQ Shareowner Services records, please make the necessary changes in Box E. All changes in registration require a Medallion Signature Guarantee
- BOX F-Signature Guarantee: Box F (Medallion Guarantee) needs to be completed if Box C, D or E are completed or if the name or address on the check will be different from the current registration shown in Box E. This guarantee is a form of signature verification which can be obtained through an eligible financial institution such as a commercial bank, trust company, securities broker/dealer, credit union or savings institution participating in a Medallion program approved by the Securities Transfer Association.
- BOX G-Request for Taxpayer Identification Number and Certification Substitute Form W-9: Complete Box G for registration of U.S. citizen,

Social Security Number									
			-			-			
Employer Identification Number									
		/							

resident or entity to certify tax identification number. Please provide the social security or employer identification number of the person or entity receiving payment, including signature and date for the above described shares. Completion of the Substitute Form W-9 certifies that receiver of the payment is not subject to backup withholding. Failure to complete the form will subject the recipient to the applicable federal income tax withholding from any cash payment made to them pursuant to the exchange.

- **Definition of U.S. Person:** For federal tax purposes, you are considered a U.S. person if you are (1) An individual who is a U.S. citizen or U.S. resident alien, (2) A partnership, corporation, company or association created or organized in the United States or under the laws of the United States, (3) An estate (other than a foreign estate), or (4) A domestic trust (as defined in regulation 301.7701-7).
- Deficient Presentments: If you request a registration change that is not in proper form, the required documentation will be requested.
- Returning Certificates: Return this Letter of Transmittal with the certificate(s) to be exchanged <u>only</u> to EQ Shareowner Services at the address below. The method of delivery is at your option and your risk, but it is recommended that documents be delivered via a registered method, insured for 5% of the value of your shares.

By Mail to: EQ Shareowner Services Corporate Actions Department P.O. Box 64858 St. Paul, MN 55164-0858 *By Overnight Courier or Hand-Delivery to:* EQ Shareowner Services Corporate Actions Department 1110 Centre Pointe Curve, Suite 101 Mendota Heights, MN 55120

For additional information please contact our Customer Care Center at 1-800-468-9716.

Exhibit H

REGISTRATION RIGHTS AGREEMENT

This Registration Rights Agreement (this "Agreement") is dated as of April 1, 2024, by and among Onconova Therapeutics, Inc., a Delaware corporation (the "Company"), and the several purchasers signatory hereto (each, including its successors and assigns, a "Purchaser" and collectively, the "Purchasers").

This Agreement is made pursuant to the Securities Purchase Agreement, dated as of the date hereof, between the Company and each Purchaser (the "*Purchase Agreement*").

NOW, THEREFORE, IN CONSIDERATION of the mutual covenants contained in this Agreement, and for other good and valuable consideration, the receipt and adequacy of which are hereby acknowledged, the Company and each of the Purchasers agree as follows:

1. <u>Definitions</u>. Capitalized terms used and not otherwise defined herein that are defined in the Purchase Agreement shall have the meanings given such terms in the Purchase Agreement. As used in this Agreement, the following terms shall have the following meanings:

"Advice" has the meaning set forth in Section 6(d).

"Affiliate" means any Person that, directly or indirectly through one or more intermediaries, controls or is controlled by or is under common control with a Person, as such terms are used in and construed under Rule 405 of the Securities Act.

"Agreement" has the meaning set forth in the Preamble.

"Business Day" means any day except any Saturday, any Sunday, any day which is a federal legal holiday in the United States or any day on which banking institutions in the State of New York are authorized or required by law or other governmental action to close.

"Commission" means the Securities and Exchange Commission.

"Common Stock" means the Company's common stock, par value \$0.01 per share, and stock of any other class of securities into which such securities may hereafter be reclassified or changed.

"Company" has the meaning set forth in the Preamble.

"Demand Notice" has the meaning set forth in Section 3(g).

"Demand Registration" has the meaning set forth in Section 3(g).

"Effective Date" means the date that the Registration Statement filed pursuant to Section 2(a) is first declared effective by the Commission.

"Effectiveness Deadline" means, with respect to the Initial Registration Statement or the New Registration Statement, the thirtieth (30th) calendar day following the Filing Deadline (or, in the event the Commission reviews and has written comments to the Initial Registration Statement or the New Registration Statement, the sixtieth (60th) calendar day following the Filing Deadline); provided, however, that if the Company is notified by the Commission that the Initial Registration Statement or the New Registration Statement will not be reviewed or is no longer subject to further review and comments, the Effectiveness Deadline as to such Registration Statement shall be the fifth (5th) Trading Day following the date on which the Company is so notified if such date precedes the dates otherwise required above; provided, further, that if the Effectiveness Deadline falls on a Saturday, Sunday or other day that the Commission is closed for business, the Effectiveness Deadline shall be extended to the next Business Day on which the Commission is open for business.

"Effectiveness Period" has the meaning set forth in Section 2(b).

"Exchange Act" means the Securities Exchange Act of 1934, as amended, and the rules and regulations promulgated thereunder.

"Filing Deadline" means, with respect to the Initial Registration Statement required to be filed pursuant to <u>Section 2(a)</u>, the ninetieth (90th) calendar day following the Closing Date, provided, however, that if the Filing Deadline falls on a Saturday, Sunday or other day that the Commission is closed for business, the Filing Deadline shall be extended to the next Business Day on which the Commission is open for business.

"FINRA" has the meaning set forth in Section 3(i).

"Holder" or "Holders" means the holder or holders, as the case may be, from time to time of Registrable Securities.

"Indemnified Party" has the meaning set forth in Section 5(c).

"Indemnifying Party" has the meaning set forth in Section 5(c).

"Initial Registration Statement" has the meaning set forth in Section 2(a).

"Losses" has the meaning set forth in Section 5(a).

"New Registration Statement" has the meaning set forth in Section 2(a).

"Person" means an individual or a corporation, partnership, trust, incorporated or unincorporated association, joint venture, limited liability company, joint stock company, government (or an agency or subdivision thereof) or other entity of any kind.

"Principal Market" means the Trading Market on which the Common Stock are primarily listed on and quoted for trading, which, as of the Closing Date, shall be the Nasdaq Stock Market LLC.

"Proceeding" means an action, claim, suit, investigation or proceeding (including, without limitation, an investigation or partial proceeding, such as a deposition), whether commenced or threatened.

"Prospectus" means the prospectus included in a Registration Statement (including, without limitation, a prospectus that includes any information previously omitted from a prospectus filed as part of an effective registration statement in reliance upon Rule 430B promulgated under the Securities Act), as amended or supplemented by any prospectus supplement, with respect to the terms of the offering of any portion of the Registrable Securities covered by a Registration Statement, and all other amendments and supplements to the Prospectus, including post-effective amendments, and all material incorporated by reference or deemed to be incorporated by reference in such Prospectus.

"Purchase Agreement" has the meaning set forth in the Recitals.

"Purchaser" or "Purchasers" has the meaning set forth in the Preamble.

"Registrable Securities" means all of (i) the Shares, (ii) the Series C Preferred Stock issued pursuant to the Purchase Agreement, (iii) any Common Stock issued on or around the date hereof pursuant to that certain Agreement and Plan of Merger, dated as of the date hereof, by and among the Company, Traws Merger Sub I, Inc., a Delaware corporation and wholly owned subsidiary of the Company, Traws Merger Sub II, LLC, a Delaware limited liability company and wholly owned subsidiary of the Company, and Trawsfynydd Therapeutics, Inc. (the *"Merger Agreement"*), (iv) the Series C Preferred Stock issued pursuant to the Merger Agreement, and (v) any securities issued or issuable upon any stock split, dividend or other distribution, recapitalization or similar event with respect to the foregoing, provided that, with respect to a particular Holder, such Holder's Common Stock or Series C Preferred Stock referenced in the preceding clauses (i) to (v) shall cease to be Registrable Securities upon the earlier to occur of the following: (A) a sale pursuant to a Registration Statement or Rule 144 under the Securities Act (in which case, only such security sold by the Holder shall cease to be a Registrable Security); and (B) such Common Stock or Series C Preferred Stock become eligible for resale by the Holder under Rule 144 without the requirement for the Company to be in compliance with the current public information required thereunder and without volume or manner-of-sale restrictions, pursuant to a written opinion letter of counsel for the Company to such effect, addressed, delivered and reasonably acceptable to the Transfer Agreent.

"Registration Statement(s)" means any one or more registration statements of the Company filed under the Securities Act that cover the resale of any of the Registrable Securities pursuant to the provisions of this Agreement (including without limitation the Initial Registration Statement, any New Registration Statement and any Remainder Registration Statements), amendments and supplements to such Registration Statements, including post-effective amendments, all exhibits and all material incorporated by reference or deemed to be incorporated by reference in such Registration Statements.

"Remainder Registration Statement" has the meaning set forth in Section 2(a).

"Rule 144" means Rule 144 promulgated by the Commission pursuant to the Securities Act, as such Rule may be amended from time to time, or any similar rule or regulation hereafter adopted by the Commission having substantially the same effect as such Rule.

"Rule 172" means Rule 172 promulgated by the Commission pursuant to the Securities Act, as such Rule may be amended from time to time, or any similar rule or regulation hereafter adopted by the Commission having substantially the same effect as such Rule.

"Rule 415" means Rule 415 promulgated by the Commission pursuant to the Securities Act, as such Rule may be amended from time to time, or any similar rule or regulation hereafter adopted by the Commission having substantially the same effect as such Rule.

"*Rule 424*" means Rule 424 promulgated by the Commission pursuant to the Securities Act, as such Rule may be amended from time to time, or any similar rule or regulation hereafter adopted by the Commission having substantially the same effect as such Rule.

"Rule 461" means Rule 461 promulgated by the Commission pursuant to the Securities Act, as such Rule may be amended from time to time, or any similar rule or regulation hereafter adopted by the Commission having substantially the same effect as such Rule.

"SEC Guidance" means (i) any publicly-available written or oral guidance, comments, requirements or requests of the Commission staff; provided, that any such oral guidance, comments, requirements or requests are reduced to writing by the Commission and (ii) the Securities Act.

"Securities Act" means the Securities Act of 1933, as amended, and the rules and regulations promulgated thereunder.

"Selling Shareholder Questionnaire" means a questionnaire in the form attached as Annex B hereto, or such other form of questionnaire or information provided to the Company in connection with the preparation of a Registration Statement hereunder.

"Series C Preferred Stock" means the Series C Non-Voting Convertible Preferred Stock, par value \$0.01 per share, of the Company.

"Shares" means (a) the shares of Common Stock issued to the Purchasers pursuant to the Purchase Agreement and (b) the shares of Common Stock which may be issued upon conversion of the Series C Preferred Stock issued to the Purchasers pursuant to the Purchase Agreement.

"Shelf Registration Statement" has the meaning set forth in Section 3(f).

"Takedown Request" has the meaning set forth in Section 3(f).

"Trading Day" means a day on which the Principal Market is open for business.

"Trading Market" means any of the following markets or exchanges on which the Common Stock is listed or quoted for trading on the date in question: the NYSE American, the Nasdaq Capital Market, the Nasdaq Global Market, the Nasdaq Global Select Market, the Nasdaq Stock Market LLC or the New York Stock Exchange (or any successors to any of the foregoing).

"Underwriter" means, with respect any Underwritten Offering, a securities dealer who purchases any Registrable Securities as a principal in connection with a distribution of such Registrable Securities.

"Underwritten Offering" means a public offering of securities registered under the Securities Act in which an Underwriter participates in the distribution of such securities, including on a firm commitment basis for reoffer and resale to the public, including any such offering that is a "bought deal" or a block trade.

2. <u>Registration Rights</u>.

On or prior to the Filing Deadline, the Company shall prepare and file with the Commission a Registration Statement covering (a) the resale of all of the Registrable Securities for an offering to be made on a continuous basis pursuant to Rule 415 (the "Initial Registration Statement"). The Initial Registration Statement shall be on Form S-3 (except if the Company is then ineligible to register for resale the Registrable Securities on Form S-3, in which case such registration shall be on such other form available to register for resale the Registrable Securities as a secondary offering) subject to the provisions of Section 2(c) and shall contain (except if otherwise required pursuant to written comments received from the Commission upon a review of such Registration Statement) the "Plan of Distribution" section substantially in the form attached hereto as Annex A (which may be modified to respond to comments, if any, provided by the Commission). Notwithstanding the registration obligations set forth in this Section 2, in the event the Commission informs the Company that all of the Registrable Securities cannot, as a result of the application of Rule 415, be registered for resale as a secondary offering on a single registration statement, the Company agrees to promptly (i) inform each of the Holders thereof and use its commercially reasonable efforts to file amendments to the Initial Registration Statement as required by the Commission and/or (ii) withdraw the Initial Registration Statement and file a new registration statement (a "New Registration Statement"), in either case covering the maximum number of Registrable Securities permitted to be registered by the Commission, on Form S-3 or, if the Company is ineligible to register the Registrable Securities on Form S-3, such other form available to register for resale the Registrable Securities as a secondary offering; provided, however, that prior to filing such amendment or New Registration Statement, the Company shall be obligated to use its commercially reasonable efforts to advocate with the Commission for the registration of all of the Registrable Securities in accordance with the SEC Guidance, including without limitation, the Securities Act Rules Compliance and Disclosure Interpretations Question 612.09. Notwithstanding any other provision of this Agreement, if the Commission or any SEC Guidance sets forth a limitation of the number of Registrable Securities permitted to be registered on a particular Registration Statement as a secondary offering (and notwithstanding that the Company used diligent efforts to advocate with the Commission for the registration of all or a greater number of Registrable Securities), unless otherwise directed in writing by a Holder as to its Registrable Securities, the number of Registrable Securities to be registered on such Registration Statement will first be reduced by Registrable Securities not acquired pursuant to the Purchase Agreement (whether pursuant to registration rights or otherwise), and second by Registrable Securities represented by Shares applied to the Holders on a pro rata basis based on the total number of Shares held by such Holders, subject to a determination by the Commission that certain Holders must be reduced first based on the number of Shares held by such Holders. In the event of a cutback hereunder, the Company shall give the Holder at least one (1) Trading Day prior notice along with the calculations as to such Holder's allotment. In the event the Company amends the Initial Registration Statement or files a New Registration Statement, as the case may be, in accordance with the foregoing, the Company will use its commercially reasonable efforts to file with the Commission, as promptly as allowed by Commission or SEC Guidance provided to the Company or to registrants of securities in general, one or more registration statements on Form S-3 or such other form available to register for resale those Registrable Securities that were not registered for resale on the Initial Registration Statement, as amended, or the New Registration Statement (the "Remainder Registration Statements"). No Holder shall be named as an "underwriter" in any Registration Statement without such Holder's prior written consent.

(b) The Company shall use its commercially reasonable efforts to cause each Registration Statement to be declared effective by the Commission as soon as practicable and, with respect to the Initial Registration Statement or the New Registration Statement, as applicable, no later than the Effectiveness Deadline (including filing with the Commission a request for acceleration of effectiveness in accordance with Rule 461 promulgated under the Securities Act), and shall use its commercially reasonable efforts to keep each Registration Statement continuously effective under the Securities Act until the earlier of (i) such time as all of the Registrable Securities covered by such Registration Statement have been publicly sold by the Holders; or (ii) the date that all Registrable Securities covered by such Registration Statement may be sold by non-affiliates without volume or manner-of-sale restrictions pursuant to Rule 144, without the requirement for the Company to be in compliance with the current public information requirement under Rule 144 (the "Effectiveness Period"). The Company shall request effectiveness of a Registration Statement as of 4:00 P.M. New York City time on a Trading Day. The Company shall promptly notify the Holders via e-mail of the effectiveness of a Registration Statement or any post-effective amendment thereto on the same Trading Day that the Company telephonically confirms effectiveness with the Commission, which date of confirmation shall initially be the date requested for effectiveness of such Registration Statement. The Company shall, by 9:30 A.M. New York City time on the first Trading Day after the Effective Date, file a final Prospectus with the Commission, as required by Rule 424(b) and shall provide the Purchasers with copies of the final Prospectus to be used in connection with the sale or other disposition of the securities covered thereby. The Company shall promptly inform each Holder in writing if, at any time during the Effectiveness Period, the Company does not satisfy the conditions specified in Rule 172 and, as a result thereof, the Holder is required to deliver a Prospectus in connection with any disposition of Registrable Securities.

(c) Each Holder of Registrable Securities to be sold agrees to furnish to the Company a completed Selling Shareholder Questionnaire not more than five (5) Trading Days following the date of this Agreement. At least 10 Trading Days prior to the first anticipated filing date of a Registration Statement for any registration under this Agreement, the Company will notify each Holder of the information the Company requires from that Holder for inclusion in the Registration Statement other than the information contained in the Selling Shareholder Questionnaire, if any, which shall be completed and delivered to the Company promptly upon request and, in any event, within three (3) Trading Days prior to the applicable anticipated filing date. Each Holder further agrees that it shall not be entitled to be named as a selling securityholder in the Registration Statement or use the Prospectus for offers and resales of Registrable Securities at any time, unless such Holder has provided such information to the Company and responded to any reasonable requests for further information as described in the previous sentence. Each Holder acknowledges and agrees that the information in the Selling Shareholder Questionnaire or request for further information as described in this Section 2(c) will be used by the Company in the preparation of the Registration Statement and hereby consents to the inclusion of such information in the Registration Statement (subject to such Holder's right to timely review the Registration Statement as set forth herein).

(d) In the event that Form S-3 is not available for the registration of the resale of Registrable Securities hereunder, the Company shall (i) register the resale of the Registrable Securities on another appropriate form reasonably acceptable to the Holders and (ii) undertake to register the Registrable Securities on Form S-3 promptly after such form is available, provided that the Company shall maintain the effectiveness of the Registration Statement then in effect until such time as a Registration Statement on Form S-3 covering the Registrable Securities has been declared effective by the Commission.

If (i) a Registration Statement covering the Registrable Securities is not filed with the Commission on or prior to the Filing (e) Deadline, the Company will make pro rata payments to each Purchaser then holding Registrable Securities, or (ii) following receipt of the Requisite Stockholder Approval, (A) a Registration Statement covering the Registrable Securities is not declared effective by the Commission prior to the Effectiveness Deadline or (B) after a Registration Statement has been declared effective by the Commission, (1) such Registration Statement ceases for any reason (including, without limitation, by reason of a stop order, or the Company's failure to update the Registration Statement) to remain continuously effective as to sell all Registrable Securities for which it is required to be effective, (2) the Purchasers are not permitted to utilize the Prospectus therein to resell such Registrable Securities (other than during permitted suspension under Section 6(d)), or (3) a permitted suspension of the Registration Statement under Section 6(d) exceeds the length set forth therein, (each of the foregoing events in clauses (i) and (ii), a "Registration Failure"), then, in addition to any other rights the Purchasers may have hereunder or under applicable law, the Company will make pro rata payments to each Purchaser of then outstanding Registrable Securities, as liquidated damages and not as a penalty (the "Liquidated Damages"), in an amount equal to one percent (1.0%) of the aggregate amount invested by such Purchaser for the Registrable Securities then held by such Purchaser for the initial day of a Registration Failure and for each thirty (30) day period (or pro rata portion thereof with respect to a final period, if any) thereafter until the Registration Failure is cured. The Liquidated Damages shall be paid monthly within ten (10) Business Days of the date of such Registration Failure and the end of each subsequent thirty (30) day period (or portion thereof with respect to a final period, if any) thereafter until the Registration Failure is cured. Such payments shall be made in cash to each Purchaser then holding Registrable Securities. Interest shall accrue at the rate of one percent (1%) per month on any such liquidated damages payments that shall not be paid by the applicable payment date until such amount is paid in full. Notwithstanding the foregoing, (I) no Liquidated Damages shall be payable with respect to any period after the expiration of the Effectiveness Period (as defined below) (it being understood that this sentence shall not relieve the Company of any Liquidated Damages accruing prior to the expiration of the Effectiveness Period), (II) in no event shall the aggregate amount of Liquidated Damages payable to a Purchaser exceed, in the aggregate, five percent (5.0%) of the aggregate purchase price paid by such Purchaser pursuant to the Purchase Agreement and (III) no Liquidated Damages shall accrue or be payable with respect to any reduction in the number or Registrable Securities to be included in a Registration Statement due to the application of Rule 415 as set forth in Section 2(a).

(f) At any time and from time to time, if a Registration Statement is effective that is on Form S-3 (a "Shelf Registration Statement"), then upon written request of any Holder (a "Takedown Request"), the Company will use reasonable best efforts to cooperate with the Holder and any Underwriter in effecting an Underwritten Offering pursuant to the Shelf Registration Statement as promptly as reasonably practicable following receipt of such request, provided that Holders shall not be entitled to make more than two (2) requests (whether for an Underwritten Offering or a Demand Registration) in any twelve (12)-month period. Each Takedown Request will specify the number of Registrable Securities proposed by the Holders to be included in such Underwritten Offering, the intended method of distribution and the estimated gross proceeds of such Underwritten Offering, which may not be less than \$5,000,000, other than an Underwritten Offering for all remaining Registrable Securities held by all Purchasers.

(g) If, at any time while there still remain Registrable Securities, the Company is no longer eligible to use or, notwithstanding its obligations under Sections 2(a), a Purchaser are not permitted to use the Registration Statement filed pursuant to Section 2(a) for resales, a Purchaser may deliver to the Company a written request to register a primary offering of all or a specified amount of the Registrable Securities under the Securities Act (a "Demand Notice"), and the Company shall use its commercially reasonable efforts to file, within thirty (30) Trading Days after receipt of the Demand Notice, a Registration Statement, on an appropriate form which the Company is then eligible to use, to register the offering of such Registrable Securities, which Registration Statement will (if specified in the Demand Notice) contemplate the ability of the Purchasers to effect an Underwritten Offering in accordance with Section 2(f) (each such registration, a "Demand Registration"); provided, however, that the Purchasers shall not be entitled to request more than two (2) requests (whether for an Underwritten Offering or for a Demand Registration) in any twelve-(12) month period. Each Demand Notice will specify the number of Registrable Securities proposed to be offered for sale, the intended method of distribution thereof and the estimated gross proceeds of such Demand Registration, which may not be less than \$5,000,000, other than a Demand Registration for all remaining Registrable Securities.

(h) In the event an Underwritten Offering of Registrable Securities involves one or more Underwriters, such Underwriters will be selected by the Purchasers; provided, that any Underwriter must be approved by the Company, which approval will not be unreasonably withheld. The Company shall use commercially reasonable efforts to take such actions as the lead Underwriter(s) may reasonably request in their efforts to sell Registrable Securities pursuant to such Registration Statement.

(i) Notwithstanding any other provision of this Section 2, but subject to the other terms of this Agreement, at any time and from time to time when an effective Shelf Registration Statement is on file with the Commission, if a Purchaser wishes to engage in an Underwritten Offering not involving a "roadshow," an offer commonly known as a "block trade" (a "Block Trade"), with a total offering price reasonably expected to exceed, in the aggregate, either (x) \$5,000,000.00 or (y) all remaining Registrable Securities held by the Purchaser, then such Purchaser only need to notify the Company of the Block Trade at least five (5) Trading Days prior to the day such offering is to commence and the Company shall as expeditiously as possible use its commercially reasonable efforts to facilitate such Block Trade; provided that the Purchaser wishing to engage in the Block Trade shall use commercially reasonable efforts to other offering documentation related to the Block Trade. Prior to the filing of the applicable "red herring" prospectus or prospectus supplement used in connection with a Block Trade, any Purchaser initiating such Block Trade shall have the right to submit a written notice to the Company and the Underwriter or Underwriters (if any) of their intention to withdraw from such Block Trade. Notwithstanding anything to the contrary in this Agreement, the Company shall be responsible for the registration expenses (in accordance with Section 5) incurred in connection with a Block Trade prior to its withdrawal under this subsection.

3. <u>Registration Procedures</u>

In connection with the Company's registration obligations hereunder, the Company shall:

(a) Not less than five (5) Trading Days prior to the filing of each Registration Statement and not less than two (2) Trading Days prior to the filing of any related Prospectus or any amendment or supplement thereto (except for Annual Reports on Form 10-K, and Quarterly Reports on Form 10-Q and Current Reports on Form 8-K and any similar or successor reports), (i) furnish to each Holder copies of such Registration Statement, Prospectus or amendment or supplement thereto, as proposed to be filed, which documents will be subject to the review of such Holder (it being acknowledged and agreed that if a Holder does not object to or comment on the aforementioned documents within such five (5) Trading Day or two (2) Trading Day period, as the case may be, then the Holder shall be deemed to have consented to and approved the use of such documents) and to the Underwriters, if any, (ii) to the Underwriters, if any, and to the extent that a Holder is identified in the Registration Statement as an "underwriter" (as defined in the Securities Act), use commercially reasonable efforts to cause its officers and directors, counsel and independent registered public accountants to respond to such inquiries as shall be necessary, in the reasonable opinion of respective counsel to each Holder and Underwriter, as applicable, to conduct a reasonable investigation within the meaning of the Securities Act. The Company shall not file any Registration Statement or supplement thereto in a form to which a Holder reasonably objects in good faith, provided that, such Holder notifies the Company of such objection in writing within the five (5) Trading Day or two (2) Trading Day periods described above, as applicable.

(i) Prepare and file with the Commission such amendments (including post-effective amendments) and supplements, to each (b) Registration Statement and the Prospectus used in connection therewith as may be necessary to keep such Registration Statement continuously effective as to the applicable Registrable Securities for its Effectiveness Period and prepare and file with the Commission such additional Registration Statements in order to register for resale under the Securities Act all of the Registrable Securities (and before filing with the Commission furnish to the Underwriters, if any, and to the Holders, copies of such documents) (ii) cause the related Prospectus to be amended or supplemented by any required Prospectus supplement (subject to the terms of this Agreement), and, as so supplemented or amended, to be filed pursuant to Rule 424; (iii) respond as promptly as reasonably practicable to any comments received from the Commission with respect to each Registration Statement or any amendment thereto and, as promptly as reasonably possible, provide the Holders true and complete copies of all correspondence from and to the Commission relating to such Registration Statement that pertains to the Holders as "Selling Stockholders" but not any comments that would result in the disclosure to the Holders of material and non-public information concerning the Company (unless such Holder consents to receive such material and non-public information); and (iv) comply with the provisions of the Securities Act and the Exchange Act with respect to the disposition of all Registrable Securities covered by a Registration Statement until such time as all of such Registrable Securities shall have been disposed of (subject to the terms of this Agreement) in accordance with the intended methods of disposition by the Holders thereof as set forth in such Registration Statement as so amended or in such Prospectus as so supplemented; provided, however, that each Purchaser shall be responsible for the delivery of the Prospectus to the Persons to whom such Purchaser sells any of the Registrable Securities (including in accordance with Rule 172 under the Securities Act) to the extent required under the Securities Act, and each Purchaser agrees to dispose of Registrable Securities in compliance with the "Plan of Distribution" described in the Registration Statement and otherwise in compliance with applicable federal and state securities laws. In the case of amendments and supplements to a Registration Statement which are required to be filed pursuant to this Agreement (including pursuant to this Section 3(b)) by reason of the Company filing a report on Form 10-K, Form 10-Q or Form 8-K or any analogous report under the Exchange Act, the Company shall have incorporated such report by reference into such Registration Statement, if applicable, or shall file such amendments or supplements with the Commission on the same day on which the Exchange Act report which created the requirement for the Company to amend or supplement such Registration Statement was filed.

Notify the Underwriters, if any, and the Holders of Registrable Securities to be sold (which notice shall, if given pursuant to (c) clauses (iii) through (vi) hereof, be accompanied by an instruction to suspend the use of the Prospectus until the requisite changes have been made. provided that the Company shall omit any material, non-public information relating to the Company and/or any of its subsidiaries) as promptly as reasonably practicable (and, in the case of (i)(A) below, not less than one (1) Trading Day prior to such filing) and (if requested by any such Person) confirm such notice in writing no later than one (1) Trading Day following the day: (i)(A) when a Prospectus or any Prospectus supplement or posteffective amendment to a Registration Statement is proposed to be filed; (B) when the Commission notifies the Company whether there will be a "review" of such Registration Statement and whenever the Commission comments in writing on any Registration Statement (in which case the Company shall provide to each of the Holders and Underwriters, if any, true and complete copies of all comments that pertain to the Holders as a "Selling Stockholder" or to the "Plan of Distribution" and all written responses thereto, but not information that the Company believes would constitute material and non-public information); and (C) with respect to each Registration Statement or any post-effective amendment, when the same has become effective; (ii) of any request by the Commission or any other federal or state governmental authority for amendments or supplements to a Registration Statement or Prospectus or for additional information that pertains to the Holders as "Selling Stockholders" or the "Plan of Distribution"; (iii) of the issuance by the Commission or any other federal or state governmental authority of any stop order suspending the effectiveness of a Registration Statement covering any or all of the Registrable Securities or the initiation of any Proceedings for that purpose; (iv) of the receipt by the Company of any notification with respect to the suspension of the qualification or exemption from qualification of any of the Registrable Securities for sale in any jurisdiction, or the initiation or threatening of any Proceeding for such purpose; (v) of the occurrence of any event or passage of time that makes the financial statements included or incorporated by reference in a Registration Statement ineligible for inclusion or incorporation by reference therein or any statement made in such Registration Statement or Prospectus or any document incorporated or deemed to be incorporated therein by reference untrue in any material respect or that requires any revisions to such Registration Statement, Prospectus or other documents so that, in the case of such Registration Statement or the Prospectus, as the case may be, it will not contain any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary to make the statements therein (in the case of any Prospectus, form of prospectus or supplement thereto, in light of the circumstances under which they were made), not misleading and (vi) of the occurrence or existence of any pending corporate development with respect to the Company that, upon the advice of legal counsel, the Company's board of directors reasonably believes may be material and that, in the reasonable determination of the Company's board of directors, makes it not in the best interest of the Company to allow continued availability of a Registration Statement or Prospectus, provided that, in no event shall any such notice contain any information that would constitute material, non-public information regarding the Company or any of its subsidiaries and the Company agrees that the Holders shall not have any duty of confidentiality to the Company or any of its subsidiaries and shall not have any duty to the Company or any of its subsidiaries not to trade on the basis of such information.

(d) Use commercially reasonable efforts to avoid the issuance of, or, if issued, obtain the withdrawal of (i) any order stopping or suspending the effectiveness of a Registration Statement, or (ii) any suspension of the qualification (or exemption from qualification) of any of the Registrable Securities for sale in any jurisdiction, as soon as practicable.

(e) If requested by a Holder or an Underwriter (if any), furnish to such Holder or Underwriter, without charge, at least one conformed copy of each Registration Statement and each amendment thereto, including financial statements and schedules, all documents incorporated or deemed to be incorporated therein by reference to the extent requested by such Person, and all exhibits to the extent requested by such Person (including those previously furnished or incorporated by reference) promptly after the filing of such documents with the Commission; provided, that the Company shall have no obligation to provide any document pursuant to this clause that is available on the Commission's EDGAR system.

(f) Prior to any resale of Registrable Securities by a Holder, use its commercially reasonable efforts to register or qualify or cooperate with the selling Holders and Underwriter, if any, in connection with the registration or qualification (or exemption from the registration or qualification) of the resale of such Registrable Securities (or, in the case of qualification, of such Registrable Securities for the resale) by the Holder under the securities or blue sky laws of such jurisdictions within the United States as any Holder reasonably requests in writing, to keep each registration or qualification (or exemption therefrom) effective during the Effectiveness Period and to do any and all other acts or things reasonably necessary to enable the disposition in such jurisdictions of the Registrable Securities covered by each Registration Statement; provided, that the Company shall not be required to qualify generally to do business in any jurisdiction where it is not then so qualified, subject the Company to any material tax in any such jurisdiction where it is not then so subject or file a general consent to service of process in any such jurisdiction.

(g) Cooperate with such Holder or Underwriter, if any, to facilitate the timely preparation and delivery of certificates or book entry statements, as applicable, representing Registrable Securities to be delivered to a transferee pursuant to the Registration Statement, which certificates or statements shall be free, to the extent permitted by the Purchase Agreement and under law, of all restrictive legends, and to enable such Registrable Securities to be in such denominations and registered in such names as any such Holders or Underwriter may reasonably request.

Following the occurrence of any event contemplated by Section 3(c), as promptly as reasonably practicable (taking into (h) account the Company's good faith assessment of any adverse consequences to the Company and its shareholders of the premature disclosure of such event), prepare a supplement or amendment, including a post-effective amendment, to the affected Registration Statements or a supplement to the related Prospectus or any document incorporated or deemed to be incorporated therein by reference, and file any other required document so that, as thereafter delivered, no Registration Statement nor any Prospectus will contain an untrue statement of a material fact or omit to state a material fact required to be stated therein or necessary to make the statements therein (in the case of any Prospectus, form of prospectus or supplement thereto, in light of the circumstances under which they were made), not misleading. If the Company notifies the Holders and Underwriter, if any, in accordance with clauses (iii) through (vi) of Section 3(c) above to suspend the use of any Prospectus until the requisite changes to such Prospectus have been made, then the Holders shall suspend use of such Prospectus. The Company will use its commercially reasonable efforts to ensure that the use of the Prospectus may be resumed as promptly as is practicable. The Company shall be entitled to exercise its right under this Section 3(h) to suspend the availability of a Registration Statement and Prospectus, subject to the payment of Liquidated Damages otherwise required pursuant to Section 2(e), in accordance with the time periods set forth in Section 6(d), which may be extended only in accordance with Section 6(f). For the avoidance of doubt, the Company's rights under this Section 3(h) shall include suspensions of availability arising from the filing of a post-effective amendment to a Registration Statement to update the Prospectus therein to include the information contained in the Company's Annual Report on Form 10-K, which suspensions may extend for the amount of time reasonably required to respond to any comments of the staff of the Commission on such amendment.

(i) The Company may require each selling Holder to furnish to the Company a certified statement as to (i) the number of shares of Common Stock beneficially owned by such Holder and any Affiliate thereof, (ii) any Financial Industry Regulatory Authority (*"FINRA"*) affiliations, (iii) any natural persons who have the power to vote or dispose of the Common Stock and (iv) any other information as may be requested by the Commission, FINRA or any state securities commission.

(j) The Company shall cooperate with any registered broker through which a Holder proposes to resell its Registrable Securities in effecting a filing with FINRA pursuant to FINRA Rule 5110 as requested by any such Holder and the Company shall pay the filing fee required for the first such filing within two (2) Business Days of the request therefor.

(k) If at any time the Commission takes the position that the offering of some or all of the Registrable Securities in a Registration Statement is not eligible to be made on a delayed or continuous basis under the provisions of Rule 415 under the Securities Act or requires any Holder to be named as an "underwriter," the Company shall use commercially reasonable efforts to persuade the Commission that the offering contemplated by such Registration Statement is a valid secondary offering and not an offering "by or on behalf of the issuer" as defined in Rule 415 and that none of the Holders is an "underwriter."

(1) With respect to an Underwritten Offering, obtain from the Company's independent public accountants any comfort letter reasonably requested by any Underwriter and furnish to each Holder of Registrable Securities included in any Registration Statement a signed counterpart of such comfort letter.

4 Registration Expenses. All fees and expenses incident to the Company's performance of or compliance with its obligations under this Agreement (excluding any underwriting discounts and selling commissions and all legal fees and expenses of legal counsel for any Holder) shall be borne by the Company whether or not any Registrable Securities are sold pursuant to a Registration Statement. The fees and expenses referred to in the foregoing sentence shall include, without limitation, (i) all registration and filing fees (including, without limitation, fees and expenses of the Company's counsel and independent registered public accountants (A) with respect to filings required to be made with any Trading Market on which the Common Stock are then listed for trading, (B) with respect to compliance with applicable state securities or blue sky laws (including, without limitation, fees and disbursements of counsel for the Company in connection with blue sky qualifications or exemptions of the Registrable Securities and determination of the eligibility of the Registrable Securities for investment under the laws of such jurisdictions as requested by the Holders) and (C) if not previously paid by the Company in connection with Section 3(j) above, with respect to any filing that may be required to be made by any broker through which a Holder intends to make sales of Registrable Securities with FINRA pursuant to the FINRA Rule 5110, so long as the broker is receiving no more than a customary brokerage commission in connection with such sale), (ii) printing expenses (including, without limitation, expenses of printing certificates for Registrable Securities and of printing prospectuses if the printing of prospectuses is reasonably requested by the Holders of a majority of the Registrable Securities included in the Registration Statement), (iii) messenger, telephone and delivery expenses, (iv) fees, expenses and disbursements of counsel for the Company, (v) Securities Act liability insurance, if the Company so desires such insurance, and (vi) fees and expenses of all other Persons retained by the Company in connection with the registrations and consummation of the transactions contemplated by this Agreement. In addition, the Company shall be responsible for all of its internal expenses incurred in connection with the consummation of the transactions contemplated by this Agreement (including, without limitation, all salaries and expenses of its officers and employees performing legal or accounting duties), the expense of any annual audit and the fees and expenses incurred in connection with the listing of the Registrable Securities on any securities exchange as required hereunder. In no event shall the Company be responsible for any underwriting, broker or similar fees or commissions of any Holder or, except to the extent provided for in the Transaction Documents, any legal fees or other costs of the Holders.

5. <u>Indemnification</u>.

Indemnification by the Company. The Company shall, notwithstanding any termination of this Agreement, indemnify, defend (a) and hold harmless each Holder and each of their respective officers, directors, agents, partners, members, managers, stockholders, Affiliates, investment advisers, brokers (including brokers who offer and sell Registrable Securities as principal as a result of a pledge or any failure to perform under a margin call of Common Stock), and employees (and any other Persons with a functionally equivalent role of a Person holding such titles, notwithstanding a lack of such title or any other title) of each of them, each Person who controls any such Holder (within the meaning of Section 15 of the Securities Act or Section 20 of the Exchange Act) and the officers, directors, partners, members, managers, stockholders, agents, investment advisers and employees (and any Persons with a functionally equivalent role of a Person holding such titles, notwithstanding a lack of such title or any other title) of each such controlling Person, to the fullest extent permitted by applicable law, from and against any and all losses, claims, damages, liabilities, costs (including, without limitation, reasonable costs of preparation and investigation and reasonable attorneys' fees), expenses and disbursements (collectively, "Losses"), as incurred, that arise out of or relate to (i) any untrue or alleged untrue statement of a material fact contained in any Registration Statement, any Prospectus or any form of prospectus or in any amendment or supplement thereto or in any preliminary prospectus, or arising out of or relating to any omission or alleged omission to state a material fact required to be stated therein or necessary to make the statements therein (in the case of any Prospectus or form of prospectus or supplement thereto, in light of the circumstances under which they were made) not misleading, or (ii) any violation or alleged violation by the Company or its agents of the Securities Act, the Exchange Act or any state securities law or any rule or regulation thereunder, in connection with the performance of its obligations under this Agreement or any action or inaction required of the Company in connection with any registration, except to the extent, but only to the extent, that (A) such untrue statements, alleged untrue statements, omissions or alleged omissions are based solely upon information regarding such Holder furnished in writing to the Company by such Holder expressly for use therein, or to the extent that such information relates to such Holder or such Holder's proposed method of distribution of Registrable Securities and was reviewed and expressly approved in writing by such Holder expressly for use in the Registration Statement, such Prospectus or such form of Prospectus or in any amendment or supplement thereto (it being understood that each Holder has approved Annex A hereto for this purpose), (B) in the case of an occurrence of an event of the type specified in Section 3(c)(iii)-(vi), related to the use by a Holder of an outdated or defective Prospectus after the Company has notified such Holder in writing that the Prospectus is outdated or defective and prior to the receipt by such Holder of the Advice contemplated and defined in Section 6(d) below, to the extent that following the receipt of the Advice the misstatement or omission giving rise to such Loss would have been corrected or (C) to the extent that any such Losses arise out of the Purchaser's (or any other indemnified Person's) failure to send or give a copy of the Prospectus or supplement (as then amended or supplemented), if required, pursuant to Rule 172 under the Securities Act (or any successor rule) to the Persons asserting an untrue statement or alleged untrue statement or alleged untrue statement or omission or alleged omission at or prior to the written confirmation of the sale of Registrable Securities to such Person if such statement or omission was corrected in such Prospectus or supplement. The Company shall notify the Holders promptly of the institution, threat or assertion of any Proceeding arising from or in connection with the transactions contemplated by this Agreement of which the Company is aware. Such indemnity shall remain in full force and effect regardless of any investigation made by or on behalf of an Indemnified Party (as defined in Section 5(c)) and shall survive the transfer of the Registrable Securities by the Holders.

Indemnification by Holders. Each Holder shall, severally and not jointly, indemnify and hold harmless the Company, its (b) directors, officers, agents and employees, each Person who controls the Company (within the meaning of Section 15 of the Securities Act and Section 20 of the Exchange Act), and the directors, officers, agents or employees of such controlling Persons (and any other Persons with a functionally equivalent role of a Person holding such titles, notwithstanding a lack of such title or any other title), to the fullest extent permitted by applicable law, from and against all Losses, as incurred, arising out of or are based solely upon any untrue or alleged untrue statement of a material fact contained in any Registration Statement, any Prospectus, or any form of prospectus, or in any amendment or supplement thereto or in any preliminary prospectus, or arising out of or relating to any omission or alleged omission of a material fact required to be stated therein or necessary to make the statements therein (in the case of any Prospectus, or any form of prospectus or supplement thereto, in light of the circumstances under which they were made) not misleading (i) to the extent that such untrue statements or omissions are based solely upon information regarding such Holder furnished in writing to the Company by such Holder expressly for use therein or (ii) to the extent that such information relates to such Holder or such Holder's proposed method of distribution of Registrable Securities and was reviewed and approved in writing by such Holder expressly for use in a Registration Statement (it being understood that the Holder has approved Annex A hereto for this purpose), such Prospectus or such form of Prospectus or in any amendment or supplement thereto or (iii) in the case of an occurrence of an event of the type specified in Section 3(c)(iii)-(vi), to the extent related to the use by such Holder of an outdated or defective Prospectus after the Company has notified such Holder in writing that the Prospectus is outdated or defective and prior to the receipt by such Holder of the Advice contemplated in Section 6(d). In no event shall the liability of any selling Holder hereunder be greater in amount than the dollar amount of the net proceeds received by such Holder upon the sale of the Registrable Securities giving rise to such indemnification obligation.

(c) <u>Conduct of Indemnification Proceedings</u>. If any Proceeding shall be brought or asserted against any Person entitled to indemnity hereunder (an "*Indemnified Party*"), such Indemnified Party shall promptly notify the Person from whom indemnity is sought (the "*Indemnifying Party*") in writing, and the Indemnifying Party shall have the right to assume the defense thereof, including the employment of counsel reasonably satisfactory to the Indemnified Party and the payment of all reasonable fees, expenses and disbursements incurred in connection with defense thereof; provided, that the failure of any Indemnified Party to give such notice shall not relieve the Indemnifying Party of its obligations or liabilities pursuant to this Agreement, except (and only) to the extent that it shall be finally determined by a court of competent jurisdiction (which determination is not subject to appeal or further review) that such failure shall have materially and adversely prejudiced the Indemnifying Party.

An Indemnified Party shall have the right to employ separate counsel in any such Proceeding and to participate in the defense thereof, but the fees, expenses and disbursements of such counsel shall be at the expense of such Indemnified Party or Indemnified Parties unless: (1) the Indemnifying Party has agreed in writing to pay such fees, expenses and amounts; (2) the Indemnifying Party shall have failed promptly to assume the defense of such Proceeding and to employ counsel reasonably satisfactory to such Indemnified Party in any such Proceeding; or (3) the named parties to any such Proceeding (including any impleaded parties) include both such Indemnified Party and the Indemnifying Party, and such Indemnified Party shall have been advised by counsel that a conflict of interest exists if the same counsel were to represent such Indemnified Party and the Indemnifying Party (in which case, if such Indemnifying Party shall not have the right to assume the defense thereof and such counsel shall be at the expense of the Indemnifying Party); provided, that the Indemnifying Party shall not be liable for the fees, expenses and disbursements of more than one separate firm of attorneys at any time for all Indemnified Parties. The Indemnifying Party shall not be liable for any settlement of any such Proceeding effected without its written consent, which consent shall not be unreasonably withheld, delayed or conditioned. No Indemnifying Party shall, without the prior written consent of the Indemnified Party, effect any settlement of any pending Proceeding in respect of which any Indemnified Party is a party, unless such settlement includes an unconditional release of such Indemnified Party from all liability on claims that are the subject matter of such Proceeding.

Subject to the terms of this Agreement, all fees, expenses and disbursements of the Indemnified Party (including reasonable fees, expenses and disbursements to the extent incurred in connection with investigating or preparing to defend such Proceeding in a manner not inconsistent with this <u>Section 5</u>) shall be paid to the Indemnified Party, as incurred, within 20 Trading Days of written notice thereof to the Indemnifying Party; provided, that the Indemnified Party shall promptly reimburse the Indemnifying Party for that portion of such fees, expenses and disbursements applicable to such actions for which such Indemnified Party is finally judicially determined to not be entitled to indemnification hereunder. The failure to deliver written notice to the Indemnifying Party within a reasonable time of the commencement of any such action shall not relieve such Indemnifying Party of any liability to the Indemnified Party under this <u>Section 5</u>, except to the extent that the Indemnifying Party is materially and adversely prejudiced in its ability to defend such action.

(d) <u>Contribution</u>. If a claim for indemnification under Section 5(a) or 5(b) is unavailable to an Indemnified Party or insufficient to hold an Indemnified Party harmless for any Losses, then each Indemnifying Party, in lieu of indemnifying such Indemnified Party, shall contribute to the amount paid or payable by such Indemnified Party as a result of such Losses, in such proportion as is appropriate to reflect the relative fault of the Indemnifying Party and Indemnified Party in connection with the actions, statements or omissions that resulted in such Losses as well as any other relevant equitable considerations. The relative fault of such Indemnifying Party and Indemnified Party shall be determined by reference to, among other things, whether any action in question, including any untrue or alleged untrue statement of a material fact or omission or alleged omission of a material fact, has been taken or made by, or relates to information supplied by, such Indemnifying Party or Indemnified Party, and the parties' relative intent, knowledge, access to information and opportunity to correct or prevent such action, statement or omission. The amount paid or payable by a party as a result of any Losses shall be deemed to include, subject to the limitations set forth in this Agreement, any reasonable attorneys' or other reasonable fees, expenses or disbursements incurred by such party in connection with any Proceeding to the extent such party would have been indemnified for such fees, expenses or disbursements if the indemnification provided for in this Section 5 was available to such party would have been indemnified for such fees, expenses or disbursements if the indemnification provided for in this Section 5 was available to such party in accordance with its terms.

The parties hereto agree that it would not be just and equitable if contribution pursuant to this <u>Section 5(d)</u> were determined by pro rata allocation or by any other method of allocation that does not take into account the equitable considerations referred to in the immediately preceding paragraph. Notwithstanding the provisions of this <u>Section 5(d)</u>, (A) no Holder shall be required to contribute, in the aggregate, any amount in excess of the amount by which the net proceeds actually received by such Holder from the sale of the Registrable Securities subject to the Proceeding exceeds the amount of any damages that such Holder has otherwise been required to pay by reason of such untrue or alleged untrue statement or omission or alleged omission and (B) no contribution will be made under circumstances where the maker of such contribution would not have been required to indemnify the Indemnified Party under the fault standards set forth in this <u>Section 5</u>. No person guilty of fraudulent misrepresentation (within the meaning of Section 11(f) of the Securities Act) shall be entitled to contribution from any Person who was not guilty of such fraudulent misrepresentation.

The indemnity and contribution agreements contained in this <u>Section 5</u> are in addition to any liability that the Indemnifying Parties may have to the Indemnified Parties and are not in diminution or limitation of the indemnification provisions under the Purchase Agreement.

6. <u>Miscellaneous</u>.

(a) <u>Remedies</u>. In the event of a breach by the Company or by a Holder of any of their obligations under this Agreement, each Holder or the Company, as the case may be, in addition to being entitled to exercise all rights granted by law and under this Agreement, including recovery of damages, will be entitled to seek specific performance of its rights under this Agreement, without the requirement of posting a bond. The Company and each Holder agree that monetary damages would not provide adequate compensation for any losses incurred by reason of a breach by it of any of the provisions of this Agreement and hereby further agrees that, in the event of any action for specific performance in respect of such breach, it shall waive the defense that a remedy at law would be adequate.

(b) <u>No Piggyback on Registrations; Prohibition on Filing Other Registration Statements</u>. Neither the Company nor any of its security holders (other than the Holders in such capacity pursuant hereto) may include securities of the Company in a Registration Statement other than the Registrable Securities and the Company shall not prior to the Effective Date enter into any agreement providing any such right to any of its security holders. For the avoidance of doubt, the provisions of this Agreement shall not impact the terms of any lock-up agreement entered into by Purchaser for the benefit of the Company on or about the date hereof.

(c) <u>Compliance</u>. Each Holder covenants and agrees that it will comply with the prospectus delivery requirements of the Securities Act as applicable to it (unless an exemption therefrom is available) in connection with sales of Registrable Securities pursuant to the Registration Statement and shall sell the Registrable Securities only in accordance with a method of distribution described in the Registration Statement, or pursuant to an exemption from, or in a transaction not subject to, the registration requirements of the Securities Act.

(d) <u>Discontinued Disposition</u>. By its acquisition of Registrable Securities, each Holder agrees that, upon receipt of a notice from the Company of the occurrence of any event of the kind described in <u>Section 3(c)(iii)-(vi)</u>, such Holder will forthwith discontinue disposition of such Registrable Securities under a Registration Statement until it is advised in writing (the "*Advice*") by the Company that the use of the applicable Prospectus (as it may have been supplemented or amended) may be resumed. The Company will use its commercially reasonable efforts to ensure that the use of the Prospectus may be resumed as promptly as is practicable. Notwithstanding anything herein to the contrary, no Holder shall be required to discontinue disposition of Registrable Securities under a Registration Statement by virtue of the delivery by the Company of a notice of the occurrence of any event of the kind described in <u>Section 3(c)(vi)</u> on more than two occasions or for more than 90 total calendar days, in each case during any twelve-month period, or for more than 45 calendar days during any 90-day period.

(e) <u>Piggy-Back Registrations</u>. If, at any time during the Effectiveness Period, there is not an effective Registration Statement covering all of the Registrable Securities and the Company shall determine to prepare and file with the Commission a registration statement relating to an offering for its own account or the account of others under the Securities Act of any of its equity securities, other than on Form S-4 or Form S-8 (each as promulgated under the Securities Act) or their then equivalents relating to equity securities to be issued solely in connection with any acquisition of any entity or business or equity securities issuable in connection with the Company's stock option or other employee benefit plans, then the Company shall deliver to each Holder a written notice of such determination and, if within fifteen days after the date of the delivery of such notice, any such Holder shall so request in writing, the Company shall include in such registration statement all or any part of such Registrable Securities such Holder requests to be registered; provided, however, that the Company shall not be required to register any Registrable Securities pursuant to this Section 6(e) that are eligible for resale pursuant to Rule 144 (without volume restrictions or current public information requirements) promulgated by the Commission pursuant to the Securities Act or that are the subject of a then effective Registration Statement.

(f) <u>No Inconsistent Agreements</u>. Neither the Company nor any of its subsidiaries has entered, as of the date hereof, nor shall the Company or any of its subsidiaries, on or after the date hereof, enter into any agreement with respect to its securities, that would have the effect of impairing the rights granted to the Holders in this Agreement or otherwise conflicts with the provisions hereof.

(g) <u>Amendments and Waivers</u>. The provisions of this Agreement, including the provisions of this sentence, may not be amended, modified or supplemented, or waived unless the same shall be in writing and signed by the Company and Holders holding no less than a majority of the then outstanding Registrable Securities, provided that (i) any party may give a waiver as to itself and (ii) any proposed amendment that would, by its terms, have a disproportionate and materially adverse effect on any Holder shall require the consent of such Holder(s). Notwithstanding the foregoing, (1) a waiver or consent to depart from the provisions hereof with respect to a matter that relates exclusively to the rights of Holders and that does not directly or indirectly affect the rights of other Holders may be given by Holders of all of the Registrable Securities to which such waiver or consent relates and (2) none of the definitions of Filing Deadline, Effectiveness Deadline or Effectiveness Period, Section 2(e), Section 3(c), Section 5, Section 6(d), or the provisions of this sentence, may be amended, modified, or supplemented except with the consent of each Holder.

(h) <u>Notices</u>. Any and all notices or other communications or deliveries required or permitted to be provided hereunder shall be delivered as set forth in the Purchase Agreement.

(i) <u>Successors and Assigns</u>. This Agreement shall inure to the benefit of and be binding upon the successors and permitted assigns of each of the parties and shall inure to the benefit of each Holder. Nothing in this Agreement, express or implied, is intended to confer upon any party other than the parties hereto or their respective successors and assigns any rights, remedies, obligations, or liabilities under or by reason of this Agreement, except as expressly provided in this Agreement. The Company may not assign its rights (except by merger or in connection with another entity acquiring all or substantially all of the Company's assets) or obligations hereunder without the prior written consent of all the Holders of the then outstanding Registrable Securities. Each Holder may assign its respective rights hereunder in the manner and to the Persons as permitted under the Purchase Agreement; provided in each case that (i) the Holder agrees in writing with the transferee or assignee to assign such rights and related obligations under this Agreement, and for the transferee or assignee to assume such obligations, and a copy of such agreement is furnished to the Company within a reasonable time after such transfere or assignee and the securities with respect to which such registration rights are being transferred or assigned, (iii) at or before the time the Company received the written notice contemplated by clause (ii) of this sentence, the transferee or assignee agrees in writing with the Company to be bound by all of the provisions contained herein and (iv) the transferee is an "accredited investor," as that term is defined in Rule 501 of Regulation D.

(j) Execution and Counterparts. This Agreement may be executed in two or more counterparts, each of which when so executed shall be deemed to be an original and, all of which taken together shall constitute one and the same Agreement and shall become effective when counterparts have been signed by each party and delivered to the other party, it being understood that both parties need not sign the same counterpart. Counterparts may be delivered via facsimile, electronic mail (including without limitation any electronic signature covered by the U.S. federal ESIGN Act of 2000) or other transmission method and any counterpart so delivered shall be deemed to have been duly and validly delivered and be valid and effective for all purposes.

(k) <u>Governing Law</u>. All questions concerning the construction, validity, enforcement and interpretation of this Agreement shall be determined in accordance with the provisions of the Purchase Agreement.

(I) <u>Cumulative Remedies</u>. The remedies provided herein are cumulative and not exclusive of any other remedies provided by law.



(m) <u>Severability</u>. If any term, provision, covenant or restriction of this Agreement is held by a court of competent jurisdiction to be invalid, illegal, void or unenforceable, the remainder of the terms, provisions, covenants and restrictions set forth herein shall remain in full force and effect and shall in no way be affected, impaired or invalidated, and the parties hereto shall use their good faith reasonable efforts to find and employ an alternative means to achieve the same or substantially the same result as that contemplated by such term, provision, covenant or restriction. It is hereby stipulated and declared to be the intention of the parties that they would have executed the remaining terms, provisions, covenants and restrictions without including any of such that may be hereafter declared invalid, illegal, void or unenforceable.

(n) <u>Headings</u>. The headings in this Agreement are for convenience only and shall not limit or otherwise affect the meaning hereof.

(0) Independent Nature of Purchasers' Obligations and Rights. The obligations of each Purchaser under this Agreement are several and not joint with the obligations of any other Purchaser hereunder, and no Purchaser shall be responsible in any way for the performance of the obligations of any other Purchaser hereunder. The decision of each Purchaser to purchase Securities pursuant to the Transaction Documents has been made by such Purchaser independently of any other Purchaser and independently of any information, materials, statements or opinions as to the business, affairs, operations, assets, properties, liabilities, results of operations, condition (financial or otherwise) or prospects of the Company or any Subsidiary which may have been made or given by any other Purchaser or by any agent or employee of any other Purchaser, and no Purchaser and any of its agents or employees shall have any liability to any other Purchaser (or any other Person) relating to or arising from any such information, materials, statement or opinions. Nothing contained herein or in any Transaction Document, and no action taken by any Purchaser pursuant hereto or thereto, shall be deemed to constitute the Purchasers as a partnership, an association, a joint venture or any other kind of entity, or create a presumption that the Purchasers are in any way acting in concert or as a group (including, without limitation, a "group" within the meaning of Section 13(d)(3) of the Exchange Act) with respect to such obligations or the transactions contemplated by the Transaction Documents. Each Purchaser acknowledges that no other Purchaser has acted as agent for such Purchaser in connection with making its investment hereunder and that no Purchaser will be acting as agent of such Purchaser in connection with monitoring its investment in the securities or enforcing its rights under the Transaction Documents. Each Purchaser shall be entitled to independently protect and enforce its rights, including, without limitation, the rights arising out of this Agreement or out of the other Transaction Documents, and it shall not be necessary for any other Purchaser to be joined as an additional party in any Proceeding for such purpose. The Company acknowledges that each of the Purchasers has been provided with the same Registration Rights Agreement for the purpose of closing a transaction with multiple Purchasers and not because it was required or requested to do so by any Purchaser. It is expressly understood that each provision contained in this Agreement is between the Company and a Purchaser, solely, and not between the Company and the Purchasers collectively and not between and among the Purchasers.

(p) <u>Current Public Information</u>. With a view to making available to the Holders the benefits of Rule 144 (or its successor rule) and any other rule or regulation of the Commission that may at any time permit the Holders to sell shares of Common Stock to the public without registration, the Company covenants and agrees to use commercially reasonable efforts to: (i) make and keep adequate current public information available, as those terms are understood and defined in Rule 144, until the earlier of (A) six months after such date as all of the Registrable Securities may be sold without restriction by the holders thereof pursuant to Rule 144 or any other rule of similar effect or (B) such date as there are no longer Registrable Securities; and (ii) file with the SEC in a timely manner all reports and other documents required of the Company under the Exchange Act.

17

IN WITNESS WHEREOF, the parties have executed this Registration Rights Agreement as of the date first written above.

ONCONOVA THERAPEUTICS, INC.

By: Name: Title: IN WITNESS WHEREOF, the parties have executed this Registration Rights Agreement as of the date first written above.

[NAME OF INVESTING ENTITY]

By: Name: Title:

PLAN OF DISTRIBUTION¹

We are registering the shares of common stock of Onconova Therapeutics, Inc., par value of \$0.01 per share, or the Common Stock, which we refer to herein as "Shares," issued to the selling stockholders to permit the sale, transfer or other disposition of the Shares by the selling stockholders or their donees, pledgees, transferees or other successors-in-interest from time to time after the date of this prospectus. We will not receive any of the proceeds from the sale by the selling stockholders of the Shares. We will, or will procure to, bear all fees and expenses incident to our obligation to register the Shares.

The selling stockholders may sell all or a portion of the Shares beneficially owned by them and offered hereby from time to time directly or through one or more underwriters, broker-dealers or agents. If the Shares are sold through underwriters or broker-dealers, the selling stockholders will be responsible for underwriting discounts (it being understood that the selling stockholders shall not be deemed to be underwriters solely as a result of their participation in this offering) or commissions or agent's commissions. The Shares may be sold on any national securities exchange or quotation service on which the securities may be listed or quoted at the time of sale, in the over-the-counter market or in transactions otherwise than on these exchanges or systems or in the over-the-counter market and in one or more transactions at fixed prices, at prevailing market prices at the time of the sale, at varying prices determined at the time of sale, or at negotiated prices. These sales may be effected in transactions, which may involve crosses or block transactions. The selling stockholders may use any one or more of the following methods when selling Shares:

- ordinary brokerage transactions and transactions in which the broker-dealer solicits purchasers;
- block trades in which the broker-dealer will attempt to sell the shares as agent but may position and resell a portion of the block as principal to facilitate the transaction;
- to or through underwriters or purchases by a broker-dealer as principal and resale by the broker-dealer for its account;
- an exchange distribution in accordance with the rules of the applicable exchange;
- · privately negotiated transactions;
- settlement of short sales entered into after the effective date of the registration statement of which this prospectus is a part;
- broker-dealers may agree with the selling stockholders to sell a specified number of such Shares at a stipulated price per Share;
- through the writing or settlement of options or other hedging transactions, whether such options are listed on an options exchange or otherwise;
- a combination of any such methods of sale; and
- any other method permitted pursuant to applicable law.

¹ Note to Draft: Definitions to be conformed to resale S-3.

The selling stockholders also may resell all or a portion of the Shares in open market transactions in reliance upon Rule 144 under the Securities Act, as amended, or the Securities Act, as permitted by that rule, or Section 4(a)(1) under the Securities Act, if available, rather than under this prospectus, provided that they meet the criteria and conform to the requirements of those provisions.

Broker-dealers engaged by the selling stockholders may arrange for other broker-dealers to participate in sales. If the selling stockholders effect such transactions by selling Shares to or through underwriters, broker-dealers or agents, such underwriters, broker-dealers or agents may receive commissions in the form of discounts, concessions or commissions from the selling stockholders or commissions from purchasers of the Shares for whom they may act as agent or to whom they may sell as principal. Such commissions will be in amounts to be negotiated, but, except as set forth in a supplement to this Prospectus, in the case of an agency transaction will not be in excess of a customary brokerage commission in compliance with FINRA Rule 2121; and in the case of a principal transaction a markup or markdown in compliance with FINRA IM-2121.01.

In connection with sales of the Shares or otherwise, the selling stockholders may enter into hedging transactions with broker-dealers or other financial institutions, which may in turn engage in short sales of the Shares in the course of hedging in positions they assume. The selling stockholders may also sell Shares short and if such short sale takes place after the date that this Registration Statement is declared effective by the Commission, the selling stockholders may deliver Shares covered by this prospectus to close out short positions and to return borrowed Shares in connection with such short sales. The selling stockholders may also loan or pledge Shares to broker-dealers that in turn may sell such Shares, to the extent permitted by applicable law. The selling stockholders may also enter into option or other transactions with broker-dealers or other financial institutions or the creation of one or more derivative securities which require the delivery to such broker-dealer or other financial institution of shares offered by this prospectus, which shares such broker-dealer or other financial institution may resell pursuant to this prospectus (as supplemented or amended to reflect such transaction). Notwithstanding the foregoing, the selling stockholders have been advised that they may not use Shares the resale of which has been registered on this registration statement to cover short sales of our Common Stock made prior to the date the registration statement, of which this prospectus forms a part, has been declared effective by the SEC.

The selling stockholders may, from time to time, pledge or grant a security interest in some or all of the Shares owned by them and, if they default in the performance of their secured obligations, the pledgees or secured parties may offer and sell the Shares from time to time pursuant to this prospectus or any amendment to this prospectus under Rule 424(b)(3) or other applicable provision of the Securities Act of 1933, amending, if necessary, the list of selling stockholders to include the pledgee, transferee or other successors in interest as selling stockholders under this prospectus. The selling stockholders also may transfer and donate the Shares in other circumstances in which case the transferees, donees, pledgees or other successors in interest will be the selling beneficial owners for purposes of this prospectus.

The selling stockholders and any broker-dealer or agents participating in the distribution of the Shares may be deemed to be "underwriters" within the meaning of Section 2(11) of the Securities Act in connection with such sales. In such event, any commissions paid, or any discounts or concessions allowed to, any such broker-dealer or agent and any profit on the resale of the shares purchased by them may be deemed to be underwriting commissions or discounts under the Securities Act. Selling Stockholders who are "underwriters" within the meaning of Section 2(11) of the Securities Act will be subject to the applicable prospectus delivery requirements of the Securities Act including Rule 172 thereunder and may be subject to certain statutory liabilities of, including but not limited to, Sections 11, 12 and 17 of the Securities Act and Rule 10b-5 under the Securities Exchange Act of 1934, as amended, or the Exchange Act. Each selling stockholder has informed the Company that it is not a registered broker-dealer and does not have any written or oral agreement or understanding, directly or indirectly, with any person to distribute the Shares. Upon the Company being notified in writing by a selling stockholder that any material arrangement has been entered into with a broker-dealer for the sale of Common Stock through a block trade, special offering, exchange distribution or secondary distribution or a purchase by a broker or dealer, a supplement to this prospectus will be filed, if required, pursuant to Rule 424(b) under the Securities Act, disclosing (i) the name of each such selling stockholder and of the participating broker-dealer(s), (ii) the number of Shares involved, (iii) the price at which such the Shares were sold, (iv) the commissions paid or discounts or concessions allowed to such broker-dealer(s), where applicable, (v) that such broker-dealer(s) did not conduct any investigation to verify the information set out or incorporated by reference in this prospectus, and (vi) other facts material to the transaction. In no event shall any broker-dealer receive fees, commissions and markups, which, in the aggregate, would exceed eight percent (8.0%).

Under the securities laws of some U.S. states, the Shares may be sold in such states only through registered or licensed brokers or dealers. In addition, in some U.S. states the Shares may not be sold unless such shares have been registered or qualified for sale in such state or an exemption from registration or qualification is available and is complied with.

There can be no assurance that any selling stockholder will sell any or all of the Shares registered pursuant to the shelf registration statement, of which this prospectus forms a part.

Each selling stockholder and any other person participating in such distribution will be subject to applicable provisions of the Exchange Act and the rules and regulations thereunder, including, without limitation, to the extent applicable, Regulation M of the Exchange Act, which may limit the timing of purchases and sales of any of the Shares by the selling stockholder and any other participating person. To the extent applicable, Regulation M may also restrict the ability of any person engaged in the distribution of the Shares to engage in market-making activities with respect to the Shares. All of the foregoing may affect the marketability of the Shares and the ability of any person or entity to engage in market-making activities with respect to the Shares.

We will pay all expenses of the registration of the Shares pursuant to the registration rights agreement, including, without limitation, Securities and Exchange Commission filing fees and expenses of compliance with state securities or "blue sky" laws; provided, however, that each selling stockholder will pay all underwriting discounts and selling commissions, if any and any related legal expenses incurred by it. We will indemnify the selling stockholders against certain liabilities, including some liabilities under the Securities Act, in accordance with the registration rights agreement, or the selling stockholders will be entitled to contribution. We may be indemnified by the selling stockholders against certain civil liabilities under the Securities Act, that may arise from any written information furnished to us by the selling stockholders specifically for use in this prospectus, in accordance with the registration rights agreements, or we may be entitled to contribution.

SELLING STOCKHOLDER NOTICE AND QUESTIONNAIRE

The undersigned holder of shares of the (i) common stock, par value \$0.01 per share, of Onconova Therapeutics, Inc. (the "*Company*") and/or (ii) Series C Non-Voting Convertible Preferred Shares, par value \$0.01 per share, of the Company issued pursuant to a certain Securities Purchase Agreement by and among the Company and the Purchasers named therein, dated as of April 1, 2024 (the "*Agreement*"), understands that the Company intends to file with the Securities and Exchange Commission a registration statement on Form S-3 (the "*Resale Registration Statement*") for the registration and the resale under Rule 415 of the Securities Act of 1933, as amended (the "*Securities Act*"), of the Registrable Securities (as defined in the Registration Rights Agreement) in accordance with the terms of the Agreement. All capitalized terms not otherwise defined herein shall have the meanings ascribed thereto in the Agreement.

In order to sell or otherwise dispose of any Registrable Securities pursuant to the Resale Registration Statement, a holder of Registrable Securities generally will be required to be named as a selling stockholder in the related prospectus or a supplement thereto (as so supplemented, the "*Prospectus*"), deliver the Prospectus to purchasers of Registrable Securities (including pursuant to Rule 172 under the Securities Act) and be bound by the provisions of the Agreement (including certain indemnification provisions, as described below). Holders must complete and deliver this Notice and Questionnaire in order to be named as selling stockholders in the Prospectus. Holders of Registrable Securities who do not complete, execute and return this Notice and Questionnaire within five (5) Trading Days following the date of the Agreement (1) will not be named as selling stockholders in the Resale Registration Statement or the Prospectus and (2) may not use the Prospectus for resales of Registrable Securities.

Certain legal consequences arise from being named as a selling stockholder in the Resale Registration Statement and the Prospectus. Holders of Registrable Securities are advised to consult their own securities law counsel regarding the consequences of being named or not named as a selling stockholder in the Resale Registration Statement and the Prospectus.

NOTICE

The undersigned holder (the "Selling Stockholder") of Registrable Securities hereby gives notice to the Company of its intention to sell or otherwise dispose of Registrable Securities owned by it and listed below in Item (3), unless otherwise specified in Item (3), pursuant to the Resale Registration Statement. The undersigned, by signing and returning this Notice and Questionnaire, understands and agrees that it will be bound by the terms and conditions of this Notice and Questionnaire and the Agreement.

The undersigned hereby provides the following information to the Company and represents and warrants that such information is accurate and complete:

QUESTIONNAIRE

1. Name.

- (a) Full Legal Name of Selling Stockholder:
- (b) Full Legal Name of Registered Holder (if not the same as (a) above) through which Registrable Securities Listed in Item 3 below are held:
- (c) Full Legal Name of Natural Control Person (which means a natural person who directly or indirectly alone or with others has power to vote or dispose of the securities covered by the questionnaire):

2. Address for Notices to Selling Stockholder:

Telephone: Fax: Contact Person: E-mail address of Contact Person:

3. Beneficial Ownership of Registrable Securities Issuable Pursuant to the Agreement:

- (a) Type and Number of Registrable Securities beneficially owned and issued pursuant to the Agreement:
- (b) Number of Registrable Securities to be registered pursuant to this Notice for resale:

4. Broker-Dealer Status:

(a) Are you a broker-dealer?

Yes \Box No \Box

(b) If "yes" to Section 4(a), did you receive your Registrable Securities as compensation for investment banking services to the Company?

Yes 🗆 No 🗆

Note: If no, the Commission's staff has indicated that you should be identified as an underwriter in the Registration Statement.

(c) Are you an affiliate of a broker-dealer?

Yes \Box No \Box

Note: If yes, provide a narrative explanation below:

(d) If you are an affiliate of a broker-dealer, do you certify that you bought the Registrable Securities in the ordinary course of business, and at the time of the purchase of the Registrable Securities to be resold, you had no agreements or understandings, directly or indirectly, with any person to distribute the Registrable Securities?

Yes \Box No \Box

Note: If no, the Commission's staff has indicated that you should be identified as an underwriter in the Registration Statement.

5. Beneficial Ownership of Other Securities of the Company Owned by the Selling Stockholder.

Except as set forth below in this Item 5, the undersigned is not the beneficial or registered owner of any securities of the Company other than the Registrable Securities listed above in Item 3.

Type and amount of other securities beneficially owned:

6. Relationships with the Company:

Except as set forth below, neither the undersigned nor any of its affiliates, officers, directors or principal equity holders (owners of 5% of more of the equity securities of the undersigned) has held any position or office or has had any other material relationship with the Company (or its predecessors or affiliates) during the past three years.

State any exceptions here:

7. Plan of Distribution:

The undersigned has reviewed the form of Plan of Distribution attached as Annex A to the Registration Rights Agreement, and hereby confirms that, except as set forth below, the information contained therein regarding the undersigned and its plan of distribution is correct and complete.

State any exceptions here:

The undersigned agrees to promptly notify the Company of any inaccuracies or changes in the information provided herein that may occur subsequent to the date hereof and prior to the effective date of any applicable Resale Registration Statement. All notices hereunder and pursuant to the Agreement shall be made in writing, by hand delivery, confirmed or facsimile transmission, first-class mail or air courier guaranteeing overnight delivery at the address set forth below. In the absence of any such notification, the Company shall be entitled to continue to rely on the accuracy of the information in this Notice and Questionnaire.

By signing below, the undersigned consents to the disclosure of the information contained herein in its answers to Items (1) through (7) above and the inclusion of such information in the Resale Registration Statement and the Prospectus. The undersigned understands that such information will be relied upon by the Company in connection with the preparation or amendment of any such Registration Statement and the Prospectus.

By signing below, the undersigned acknowledges that it understands its obligation to comply, and agrees that it will comply, with the provisions of the Exchange Act and the rules and regulations thereunder, particularly Regulation M in connection with any offering of Registrable Securities pursuant to the Resale Registration Statement. The undersigned also acknowledges that it understands that the answers to this Questionnaire are furnished for use in connection with Registration Statements filed pursuant to the Registration Rights Agreement and any amendments or supplements thereto filed with the Commission pursuant to the Securities Act.

The undersigned hereby acknowledges and is advised of the following Question 239.10 of the Securities Act Rules Compliance and Disclosure Interpretations regarding short selling:

"An Issuer filed a Form S-3 registration statement for a secondary offering of common stock which is not yet effective. One of the selling stockholders wanted to do a short sale of common stock "against the box" and cover the short sale with registered shares after the effective date. The issuer was advised that the short sale could not be made before the registration statement become effective, because the shares underlying the short sale are deemed to be sold at the time such sale is made. There would, therefore, be a violation of Section 5 if the shares were effectively sold prior to the effective date."

By returning this Questionnaire, the undersigned will be deemed to be aware of the foregoing interpretation.

I confirm that, to the best of my knowledge and belief, the foregoing statements (including without limitation the answers to this Questionnaire) are correct.

IN WITNESS WHEREOF the undersigned, by authority duly given, has caused this Questionnaire to be executed and delivered either in person or by its duly authorized agent.

Dated:

Beneficial Owner:

By: Name: Title:

PLEASE FAX A COPY OF THE COMPLETED AND EXECUTED NOTICE AND QUESTIONNAIRE, AND RETURN THE ORIGINAL BY OVERNIGHT MAIL, TO:

Bryan S. Keighery One Federal Street, Boston, MA 02110 Telephone No.: (617) 341-7269 Attention: Bryan S. Keighery E-mail: bryan.keighery@morganlewis.com

ONCONOVA THERAPEUTICS, INC.

CERTIFICATE OF DESIGNATION OF PREFERENCES, RIGHTS AND LIMITATIONS OF SERIES C NON-VOTING CONVERTIBLE PREFERRED STOCK

Pursuant to Section 151 of the General Corporation Law of the State of Delaware

THE UNDERSIGNED DOES HEREBY CERTIFY, on behalf of Onconova Therapeutics, Inc., a Delaware corporation (the "*Corporation*"), that the following resolution was duly adopted by the Board of Directors of the Corporation (the "*Board of Directors*"), in accordance with the provisions of Section 151 of the General Corporation Law of the State of Delaware (the "*DGCL*"), at a meeting duly called and held on April 1, 2024, which resolution provides for the creation of a series of the Corporation's Preferred Stock, par value \$0.01 per share, which is designated as "Series C Non-Voting Convertible Preferred Stock," with the preferences, rights and limitations set forth therein relating to dividends, conversion, redemption, dissolution and distribution of assets of the Corporation.

WHEREAS: the Tenth Amended and Restated Certificate of Incorporation of the Corporation, as amended (the "*Certificate of Incorporation*"), provides for a class of its authorized stock known as Preferred Stock, consisting of 5,000,000 shares, \$0.01 par value per share (the "*Preferred Stock*"), issuable from time to time in one or more series.

RESOLVED: that, pursuant to authority conferred upon the Board of Directors by the Certificate of Incorporation, (i) a series of Preferred Stock of the Corporation be, and hereby is authorized by the Board of Directors, (ii) the Board of Directors hereby authorizes the issuance of 12,472.76870 shares of "Series C Non-Voting Convertible Preferred Stock" pursuant to the terms of (a) the Securities Purchase Agreement, dated as of the date hereof, by and among the Corporation and the initial Holders (as defined below) (the "*Purchase Agreement*") and (b) the Agreement and Plan of Merger, dated as of the date hereof, by and among the Corporation, Traws Merger Sub I, Inc., a Delaware corporation and wholly owned subsidiary of the Corporation, Traws Merger Sub II, LLC, a Delaware limited liability company and wholly owned subsidiary of the Corporation and Trawsfynydd Therapeutics, Inc., a Delaware corporation (the "*Merger Agreement*"), and (iii) the Board of Directors hereby fixes the designations, powers, preferences and relative, participating, optional or other special rights, and the qualifications, limitations or restrictions thereof, of such shares of Preferred Stock, in addition to any provisions set forth in the Certificate of Incorporation that are applicable to the Preferred Stock of all classes and series, as follows:

TERMS OF SERIES C NON-VOTING CONVERTIBLE PREFERRED STOCK

1. <u>Definitions</u>. For the purposes hereof, the following terms shall have the following meanings:

"Business Day" means any day other than a Saturday, Sunday or other day on which banks in Philadelphia, PA, are authorized or obligated by Law to be closed.

"Buy-In" shall have the meaning set forth in Section 6.5.4.

"Closing Sale Price" means, for any security as of any date, the last closing trade price for such security immediately prior to 4:00 p.m., New York City time, on the principal Trading Market where such security is listed or traded, as reported by Bloomberg, L.P. (or an equivalent, reliable reporting service), or if the foregoing do not apply, the last trade price of such security in the over-the-counter market on the electronic bulletin board for such security as reported by Bloomberg, L.P., or, if no last trade price is reported for such security by Bloomberg, L.P., the average of the bid prices of any market makers for such security as reported on the OTC Pink Market by OTC Markets Group, Inc. If the Closing Sale Price cannot be calculated for a security on a particular date on any of the foregoing bases, the Closing Sale Price of such security on such date shall be the fair market value as determined in good faith by the Board of Directors of the Corporation.

"Commission" means the United States Securities and Exchange Commission.

"Common Stock" means the Corporation's common stock, par value \$0.01 per share, and stock of any other class of securities into which such securities may hereafter be reclassified or changed.

"Conversion Shares" means, collectively, the shares of Common Stock issuable upon conversion of the shares of Series C Non-Voting Preferred Stock in accordance with the terms hereof.

"Exchange Act" means the Securities Exchange Act of 1934.

"Holder" means a holder of shares of Series C Non-Voting Preferred Stock.

"*Person*" means an individual or corporation, partnership, trust, incorporated or unincorporated association, joint venture, limited liability company, joint stock company, government (or an agency or subdivision thereof) or other entity of any kind.

"Trading Day" means a day on which the principal Trading Market is open for business.

"*Trading Market*" means any of the following markets or exchanges on which the Common Stock is listed or quoted for trading on the date in question: the NYSE American, the Nasdaq Capital Market, the Nasdaq Global Market, the Nasdaq Global Select Market, or the New York Stock Exchange (or any successors to any of the foregoing).

2. <u>Designation, Amount and Par Value</u>. The series of Preferred Stock shall be designated as the Corporation's Series C Non-Voting Convertible Preferred Stock (the "*Series C Non-Voting Preferred Stock*") and the number of shares so designated shall be 20,000. Each share of Series C Non-Voting Preferred Stock shall have a par value of \$0.01 per share.

3. Dividends. Holders shall be entitled to receive, and the Corporation shall pay, dividends on shares of the Series C Non-Voting Preferred Stock (on an as-if-converted-to-Common-Stock basis, without regard to the Beneficial Ownership Limitation (as defined below)) equal to and in the same form, and in the same manner, as dividends (other than dividends on shares of the Common Stock payable in the form of Common Stock) actually paid on shares of the Common Stock; provided, however, in no event shall Holders of Series C Non-Voting Preferred Stock be entitled to receive the "rights" distributed pursuant to that certain Contingent Value Rights Agreement dated as of April 1, 2024 by and between the Corporation and Equiniti Trust Company, LLC, a New York limited liability trust company, as may be amended from time to time (the "*CVR Agreement*"), or any amounts paid under the CVR Agreement. Other than as set forth in the previous sentence, no other dividends shall be paid on shares of Series C Non-Voting Preferred Stock, and the Corporation shall pay no dividends (other than dividends payable in the form of Common Stock unless it simultaneously complies with the previous sentence.

4. Voting Rights.

Except as otherwise provided herein or as otherwise required by the DGCL, the Series C Non-Voting Preferred Stock shall have no 41 voting rights. However, as long as any shares of Series C Non-Voting Preferred Stock are outstanding, the Corporation shall not, without the affirmative vote of the holders of a majority of the then outstanding shares of the Series C Non-Voting Preferred Stock: (i) alter or change adversely the powers, preferences or rights given to the Series C Non-Voting Preferred Stock or alter or amend this Certificate of Designation, amend or repeal any provision of, or add any provision to, the Certificate of Incorporation or Amended and Restated Bylaws of the Corporation, or file any articles of amendment, certificate of designations, preferences, limitations and relative rights of any series of Preferred Stock, if such action would adversely alter or change the preferences, rights, privileges or powers of, or restrictions provided for the benefit of the Series C Non-Voting Preferred Stock, regardless of whether any of the foregoing actions shall be by means of amendment to the Certificate of Incorporation or by merger, consolidation, recapitalization, reclassification, conversion or otherwise, (ii) issue further shares of Series C Non-Voting Preferred Stock or increase or decrease (other than by conversion) the number of authorized shares of Series C Non-Voting Preferred Stock, (iii) prior to the Stockholder Approval (as defined below) or at any time while at least 30% of the originally issued Series C Non-Voting Preferred Stock remains issued and outstanding, consummate either: (A) any Fundamental Transaction (as defined below) or (B) any merger or consolidation of the Corporation with or into another entity or any stock sale to, or other business combination in which the stockholders of the Corporation immediately before such transaction do not hold at least a majority of the capital stock of the Corporation immediately after such transaction or (iv) enter into any agreement with respect to any of the foregoing. Holders of shares of Common Stock acquired upon the conversion of shares of Series C Non-Voting Preferred Stock shall be entitled to the same voting rights as each other holder of Common Stock, except that such holders may not vote such shares upon the proposal for Stockholder Approval in accordance with Rule 5635 of the listing rules of The Nasdaq Stock Market LLC.

4.2 Any vote required or permitted under <u>Section 4.1</u> may be taken at a meeting of the Holders or through the execution of an action by written consent in lieu of such meeting, provided that the consent is executed by Holders representing a majority of the outstanding shares of Series C Non-Voting Preferred Stock.

5. Rank; Liquidation.

5.1 The Series C Non-Voting Preferred Stock shall rank on parity with the Common Stock as to distributions of assets upon liquidation, dissolution or winding up of the Corporation, whether voluntarily or involuntarily.

5.2 Upon any liquidation, dissolution or winding-up of the Corporation, whether voluntary or involuntary (a "*Liquidation*"), each Holder shall be entitled to receive out of the assets, whether capital or surplus, of the Corporation the same amount that a holder of Common Stock would receive if the Series C Non-Voting Preferred Stock were fully converted (disregarding for such purpose any Beneficial Ownership Limitations) to Common Stock which amounts shall be paid *pari passu* with all holders of Common Stock, plus an additional amount equal to any dividends declared on but unpaid to such shares. If, upon any such Liquidation, the assets of the Corporation shall be insufficient to pay the Holders of shares of the Series C Non-Voting Preferred Stock the amount required under the preceding sentence, then all remaining assets of the Corporation shall be distributed ratably to the Holders and the holders of Common Stock in accordance with the respective amounts that would be payable on all such securities if all amounts payable thereon were paid in full. For the avoidance of any doubt, a Fundamental Transaction shall not be deemed a Liquidation unless the Corporation expressly declares that such Fundamental Transaction shall be treated as if it were a Liquidation.

6. <u>Conversion</u>.

6.1 <u>Automatic Conversion on Stockholder Approval</u>. Effective as of 5:00 p.m. Eastern time on the third Business Day after the date that the Corporation's stockholders approve the conversion of the Series C Non-Voting Preferred Stock into shares of Common Stock in accordance with the listing rules of the Nasdaq Stock Market, as set forth in Section 4.1 of the Merger Agreement (the "*Stockholder Approval*"), each share of Series C Non-Voting Preferred Stock then outstanding shall automatically convert into a number of shares of Common Stock equal to the Conversion Ratio (as defined below), subject to the Beneficial Ownership Limitation (the "*Automatic Conversion*"). In determining the application of the Beneficial Ownership Limitations solely with respect to the Automatic Conversion, the Corporation shall calculate beneficial ownership for each Holder assuming beneficial ownership by such Holder of: (x) the number of shares of Common Stock issuable to such Holder in such Automatic Conversion, plus (y) any additional shares of Common Stock for which a Holder has provided the Corporation with prior written notice of beneficial ownership within 30 days prior to the date of Stockholder Approval (a "*Beneficial Ownership Statement*") and assuming the conversion of all shares of Series C Non-Voting Preferred Stock held by all other Holders that will not convert into shares of Common Stock on account of the application of any Beneficial Ownership Limitations applicable to any such other Holders. If a Holder fails to provide the Corporation with a Beneficial Ownership Statement within 30 days prior to the date of Stockholder Approval, then the Corporation shall presume the Holder's beneficial ownership of Common Stock (excluding the Conversion Shares) to be zero. The shares of Series C Non-Voting Preferred Stock that are converted in the Automatic Conversion are referred to as the "*Converted Stock*". The Conversion Shares shall be issued as follows:

6.1.1 Converted Stock that is registered in book entry form shall be automatically cancelled upon the Automatic Conversion and converted into the corresponding Conversion Shares, which shares shall be issued in book entry form and without any action on the part of the Holders.

6.1.2 Converted Stock that is issued in certificated form shall be deemed converted into the corresponding Conversion Shares on the date of Automatic Conversion and the Holder's rights as a holder of such shares of Converted Stock shall cease and terminate on such date, excepting only the right to receive the Conversion Shares upon the Holder tendering to the Corporation (or its designated agent) the stock certificate(s) (duly endorsed) representing such certificated Converted Stock.

6.1.3 Notwithstanding the cancellation of the Converted Stock upon the Automatic Conversion, Holders of Converted Stock shall continue to have any remedies provided herein or otherwise available at law or in equity to such Holder because of a failure by the Corporation to comply with the terms of this Certificate of Designation. In all cases, the Holder shall retain all of its rights and remedies for the Corporation's failure to convert the Converted Stock.

6.2 <u>Conversion at Option of Holder</u>. Subject to <u>Section 6.1</u>, <u>Section 6.4</u> and <u>Section 6.5.3</u>, each share of Series C Non-Voting Preferred Stock then outstanding shall be convertible, at any time and from time to time following 5:00 p.m. Eastern time on the third Business Day after the date that the Stockholder Approval is obtained by the Corporation, at the option of the Holder thereof, into a number of shares of Common Stock equal to the Conversion Ratio, subject to the Beneficial Ownership Limitation (each, an "*Optional Conversion*"). Holders shall effect conversions by providing the Corporation with the form of conversion notice attached hereto as <u>Annex A</u> (a "*Notice of Conversion*"), duly completed and executed. Provided the Corporation's transfer agent is participating in the Depository Trust Company ("*DTC*") Fast Automated Securities Transfer program, the Notice of Conversion may specify, at the Holder's election, whether the applicable Conversion Shares shall be credited to the account of the Holder's prime broker with DTC through its Deposit Withdrawal Agent Commission system (a "*DWAC Delivery*"). The date on which an Optional Conversion shall be deemed effective (the "*Conversion Date*") shall be the Trading Day that the Notice of Conversion, completed and executed, is sent via email to, and received during regular business hours by, the Corporation; provided, that the original certificate(s) (if any) representing such shares of Series C Non-Voting Preferred Stock being converted, duly endorsed, and the accompanying Notice of Conversion, are received by the Corporation within two (2) Trading Days thereafter. In all other cases, the Conversion Date shall be defined as the Trading Day on which the original certificate(s) (if any) representing such shares of Series C Non-Voting Preferred Stock being converted, duly endorsed, and the accompanying Notice of Conversion, are received by the Corporation. The calculations set forth in the Notice of Conversion shall control in the absence of manifest o

6.3 <u>Conversion Ratio</u>. The "*Conversion Ratio*" for each share of Series C Non-Voting Preferred Stock shall be 10,000 shares of Common Stock issuable upon the conversion (the "*Conversion*") of each share of Series C Non-Voting Preferred Stock, subject to adjustment as provided herein.

Beneficial Ownership Limitation. Notwithstanding anything herein to the contrary, the Corporation shall not effect any conversion 6.4 of any share of Series C Non-Voting Preferred Stock, including pursuant to Section 6.1, and a Holder shall not have the right to convert any portion of the Series C Non-Voting Preferred Stock pursuant to Section 6.2, to the extent that, after giving effect to such attempted conversion set forth on an applicable Notice of Conversion (as defined in the Certificate of Designation) with respect to the Series C Preferred Stock, such Holder (or any of such Holder's affiliates or any other Person who would be a beneficial owner of Common Stock beneficially owned by the Holder for purposes of Section 13(d) or Section 16 of the Exchange Act and the applicable rules and regulations of the Commission, including any "group" of which the Holder is a member (the foregoing, "Attribution Parties")) would beneficially own a number of shares of Common Stock in excess of the Beneficial Ownership Limitation. For purposes of the foregoing sentence, the aggregate number of shares of Common Stock beneficially owned by such Holder and its Attribution Parties shall include the number of shares of Common Stock issuable upon conversion of the Series C Non-Voting Preferred Stock subject to the Notice of Conversion or the Automatic Conversion, as applicable, with respect to which such determination is being made, but shall exclude the number of shares of Common Stock which are issuable upon (A) conversion of the remaining, unconverted Series C Non-Voting Preferred Stock beneficially owned by such Holder or any of its Attribution Parties, and (B) exercise or conversion of the unexercised or unconverted portion of any other securities of the Corporation (including any warrants) beneficially owned by such Holder or any of its Attribution Parties that are subject to and would exceed a limitation on conversion or exercise similar to the limitation contained herein. Except as set forth in the preceding sentence, for purposes of this Section 6.4, beneficial ownership shall be calculated in accordance with Section 13(d) of the Exchange Act and the applicable rules and regulations of the Commission, and the terms "beneficial ownership" and "beneficially own" have the meanings ascribed to such terms therein. In addition, for purposes hereof, "group" has the meaning set forth in Section 13(d) of the Exchange Act and the applicable rules and regulations of the Commission. For purposes of this Section 6.4, in determining the number of outstanding shares of Common Stock, a Holder may rely on the number of outstanding shares of Common Stock as stated in the most recent of the following: (A) the Corporation's most recent periodic or annual filing with the Commission, as the case may be, (B) a more recent public announcement by the Corporation that is filed with the Commission, or (C) a more recent notice by the Corporation or the Corporation's transfer agent to the Holder setting forth the number of shares of Common Stock then outstanding. Upon the written request of a Holder (which may be by email), the Corporation shall, within two (2) Trading Days thereof, confirm in writing to such Holder (which may be via email) the number of shares of Common Stock then outstanding. In any case, the number of outstanding shares of Common Stock shall be determined after giving effect to any actual conversion or exercise of securities of the Corporation, including shares of Series C Non-Voting Preferred Stock, by such Holder or its Attribution Parties since the date as of which such number of outstanding shares of Common Stock was last publicly reported or confirmed to the Holder. The "Beneficial Ownership Limitation" shall initially be set at 19.9% for each Holder and its Attribution Parties and may be adjusted at the discretion of the Holder to a percentage between 4.9% and 19.9% of the number of shares of Common Stock outstanding immediately after giving effect to the issuance of shares of Common Stock pursuant to the Automatic Conversion or such Notice of Conversion (as applicable), to the extent permitted by this Section 6.3. The Corporation shall be entitled to rely on representations made to it by the Holder in any Notice of Conversion regarding its Beneficial Ownership Limitation. Notwithstanding the foregoing, by written notice to the Corporation, (i) which will not be effective until the sixty-first (61st) day after such written notice is delivered to the Corporation, the Holder may reset the Beneficial Ownership Limitation percentage to a higher percentage, not to exceed 19.9%, to the extent then applicable and (ii) which will be effective immediately after such notice is delivered to the Corporation, the Holder may reset the Beneficial Ownership Limitation percentage to a lower percentage (but in no event less than 4.9%) provided that such decrease shall not become effective until the later of (x) 5:00 p.m. Eastern time on the third Business Day after the date of the Stockholder Approval and (y) if Stockholder Approval is not obtained within six months after the initial issuance of the Series C Non-Voting Preferred Stock, the date that is three Business Days after the date that is six months after the initial issuance of the Series C Non-Voting Preferred Stock. Upon such a change by a Holder of the Beneficial Ownership Limitation, not to exceed 19.9%, the Beneficial Ownership Limitation may not be further amended by such Holder without first providing the minimum notice required by this Section 6.4. Notwithstanding the foregoing, at any time following notice of a Fundamental Transaction, the Holder may waive and/or change the Beneficial Ownership Limitation effective immediately upon written notice to the Corporation and may reinstitute a Beneficial Ownership Limitation at any time thereafter effective immediately upon written notice to the Corporation. The provisions of this Section 6.4 shall be construed, corrected and implemented in a manner so as to effectuate the intended Beneficial Ownership Limitation herein contained and the shares of Common Stock underlying the securities in excess of the Beneficial Ownership Limitation shall not be deemed to be beneficially owned by the Holder for any purpose including for purposes of Section 13(d) or Rule 16a-1(a) (1) of the Exchange Act.

6.5 Mechanics of Conversion.

6.5.1 <u>Delivery of Certificate or Electronic Issuance</u>. Upon Conversion not later than two (2) Trading Days after the applicable Conversion Date, or if the Holder requests the issuance of physical certificate(s), two (2) Trading Days after receipt by the Corporation of the original certificate(s) representing such shares of Series C Non-Voting Preferred Stock being converted, duly endorsed, and the accompanying Notice of Conversion (the "*Share Delivery Date*"), the Corporation shall either: (a) deliver, or cause to be delivered, to the converting Holder a physical certificate or certificates representing the number of Conversion Shares being acquired upon the conversion of shares of Series C Non-Voting Preferred Stock, or (b) in the case of a DWAC Delivery (if so requested by the Holder), electronically transfer such Conversion Shares by crediting the account of the Holder's prime broker with DTC through its DWAC system. If in the case of any Notice of Conversion such certificate or certificates for the Conversion Shares are not delivered to or as directed by or, in the case of a DWAC Delivery, such shares are not electronically delivered to or as directed by, the applicable Holder by the Share Delivery Date, the applicable Holder shall be entitled to elect to rescind such Notice of Conversion by written notice to the Corporation at any time on or before its receipt of such certificate or certificates for Conversion Shares or electronic receipt of such shares, as applicable, in which event the Corporation shall promptly return to such Holder any original Series C Non-Voting Preferred Stock certificate delivered to the Corporation and such Holder shall promptly return to the Corporation any Common Stock certificates or otherwise direct the return of any shares of Common Stock delivered to the Holder through the DWAC system, representing the shares of Series C Non-Voting Preferred Stock unsuccessfully tendered for conversion to the Corporation.

Obligation Absolute. Subject to Section 6.4 and subject to Holder's right to rescind a Notice of Conversion pursuant to 6.5.2 Section 6.5.1, the Corporation's obligation to issue and deliver the Conversion Shares upon conversion of Series C Non-Voting Preferred Stock in accordance with the terms hereof are absolute and unconditional, irrespective of any action or inaction by a Holder to enforce the same, any waiver or consent with respect to any provision hereof, the recovery of any judgment against any Person or any action to enforce the same, or any setoff, counterclaim, recoupment, limitation or termination, or any breach or alleged breach by such Holder or any other Person of any obligation to the Corporation or any violation or alleged violation of law by such Holder or any other Person, and irrespective of any other circumstance which might otherwise limit such obligation of the Corporation to such Holder in connection with the issuance of such Conversion Shares. Subject to Section 6.4 and subject to Holder's right to rescind a Notice of Conversion pursuant to Section 6.5.1, in the event a Holder shall elect to convert any or all of its Series C Non-Voting Preferred Stock, the Corporation may not refuse conversion based on any claim that such Holder or anyone associated or affiliated with such Holder has been engaged in any violation of law, agreement or for any other reason, unless an injunction from a court, on notice to Holder, restraining and/or enjoining conversion of all or part of the Series C Non-Voting Preferred Stock of such Holder shall have been sought and obtained by the Corporation, and the Corporation posts a surety bond for the benefit of such Holder in the amount of 150% of the value of the Conversion Shares into which would be converted the Series C Non-Voting Preferred Stock which is subject to such injunction, which bond shall remain in effect until the completion of arbitration/litigation of the underlying dispute and the proceeds of which shall be payable to such Holder to the extent it obtains judgment. In the absence of such injunction, the Corporation shall, subject to Section 6.4 and subject to Holder's right to rescind a Notice of Conversion pursuant to Section 6.5.1, issue Conversion Shares upon a properly noticed conversion.

6.5.3 Cash Settlement. If, at any time after the earlier of Stockholder Approval or six months after the initial issuance of the Series C Non-Voting Preferred Stock, the Corporation fails to deliver to a Holder such certificate or certificates, or electronically deliver (or cause its transfer agent to electronically deliver) such shares in the case of a DWAC Delivery, pursuant to Section 6.5.1 on or prior to the third (3rd) Trading Day after the Share Delivery Date applicable to such conversion (other than a failure caused by (i) materially incorrect or incomplete information provided by Holder to the Corporation or (ii) the application of the Beneficial Ownership Limitation), then, unless the Holder has rescinded the applicable Notice of Conversion pursuant to Section 6.5.1, the Corporation shall, at the request of the Holder, pay an amount equal to the Fair Value (as defined below) of such undelivered shares, with such payment to be made within two Business Days from the date of request by the Holder, whereupon the Corporation's obligations to deliver such shares underlying the Notice of Conversion shall be extinguished upon payment in full of the Fair Value of such undelivered shares; provided, however that such request shall be presumed to have been made by such Holder if Stockholder Approval shall not have been obtained prior to the date on which the Notice of Conversion is delivered to the Corporation. For purposes of this Section 6.5.3, the "Fair Value" of shares shall be fixed with reference to the last reported Closing Sale Price on the principal Trading Market on which the Common Stock is listed as of the Trading Day immediately prior to, in the case of the Automatic Conversion, the date of the Stockholder Approval, and in the case of an Optional Conversion, the Conversion Date. For the avoidance of doubt, the cash settlement provisions set forth in this Section 6.5.3 shall be available irrespective of the reason for the Corporation's failure to timely deliver Conversion Shares (other than a failure caused by (i) materially incorrect or incomplete information provided by Holder to the Corporation or (ii) the application of the Beneficial Ownership Limitation), including due to limitations set forth in Section 6.5.6, the lack of obtaining Stockholder Approval, or due to applicable Trading Market rules.

Buy-In on Failure to Timely Deliver Certificates. If the Corporation fails to deliver to a Holder the applicable certificate 6.5.4 or certificates or to effect a DWAC Delivery, as applicable, by the Share Delivery Date pursuant to Section 6.5.1 (other than a failure caused by materially incorrect or incomplete information provided by Holder to the Corporation or the application of the Beneficial Ownership Limitation), and if after such Share Delivery Date such Holder is required by its brokerage firm to purchase (in an open market transaction or otherwise), or the Holder's brokerage firm otherwise purchases, shares of Common Stock to deliver in satisfaction of a sale by such Holder of the Conversion Shares which such Holder was entitled to receive upon the conversion relating to such Share Delivery Date (a "Buy-In"), then the Corporation shall (A) pay in cash to such Holder (in addition to any other remedies available to or elected by such Holder) the amount by which (x) such Holder's total purchase price (including any brokerage commissions) for the shares of Common Stock so purchased exceeds (y) the product of (1) the aggregate number of shares of Common Stock that such Holder was entitled to receive from the conversion at issue multiplied by (2) the actual sale price at which the sell order giving rise to such purchase obligation was executed (including any brokerage commissions) and (B) at the option of such Holder, either reissue (if surrendered) the shares of Series C Non-Voting Preferred Stock equal to the number of shares of Series C Non-Voting Preferred Stock submitted for conversion or deliver to such Holder the number of shares of Common Stock that would have been issued if the Corporation had timely complied with its delivery requirements under Section 6.5.1. For example, if a Holder purchases shares of Common Stock having a total purchase price of \$11,000 to cover a Buy-In with respect to an attempted conversion of shares of Series C Non-Voting Preferred Stock with respect to which the actual sale price (including any brokerage commissions) giving rise to such purchase obligation was a total of \$10,000 under clause (A) of the immediately preceding sentence, the Corporation shall be required to pay such Holder \$1,000. The Holder shall provide the Corporation written notice, within three (3) Trading Days after the occurrence of a Buy-In, indicating the amounts payable to such Holder in respect of such Buy-In together with applicable confirmations and other evidence reasonably requested by the Corporation. Nothing herein shall limit a Holder's right to pursue any other remedies available to it hereunder, at law or in equity including, without limitation, a decree of specific performance and/or injunctive relief with respect to the Corporation's failure to timely deliver certificates representing shares of Common Stock upon conversion of the shares of Series C Non-Voting Preferred Stock as required pursuant to the terms hereof or the cash settlement remedy set forth in Section 6.5.3; provided, however, that the Holder shall not be entitled to both (i) require the reissuance of the shares of Series C Non-Voting Preferred Stock submitted for conversion for which such conversion was not timely honored and (ii) receive the number of shares of Common Stock that would have been issued if the Corporation had timely complied with its delivery requirements under Section 6.5.1.

6.5.5 <u>Reservation of Shares Issuable Upon Conversion</u>. The Corporation covenants that at all times it will reserve and keep available out of its authorized and unissued shares of Common Stock for the sole purpose of issuance upon conversion of the Series C Non-Voting Preferred Stock, free from preemptive rights or any other actual contingent purchase rights of Persons other than the Holders of the Series C Non-Voting Preferred Stock, not less than such aggregate number of shares of the Common Stock as shall be issuable (taking into account the adjustments of <u>Section 7</u>) upon the conversion of all outstanding shares of Series C Non-Voting Preferred Stock. The Corporation covenants that all shares of Common Stock that shall be so issuable shall, upon issue, be duly authorized, validly issued, fully paid and non-assessable.

6.5.6 <u>Fractional Shares</u>. No fractional shares of Common Stock shall be issued upon conversion of the Series C Non-Voting Preferred Stock, no certificates or scrip for any such fractional shares shall be issued and no cash shall be paid for any such fractional shares. Any fractional shares of Common Stock that a Holder of Series C Non-Voting Preferred Stock would otherwise be entitled to receive shall be aggregated with all fractional shares of Common Stock issuable to such Holder and any remaining fractional shares shall be rounded up to the nearest whole share. Whether or not fractional shares would be issuable upon such conversion shall be determined on the basis of the total number of shares of Series C Non-Voting Preferred Stock and the aggregate number of shares of Common Stock issuable upon such conversion.

6.5.7 <u>Transfer Taxes</u>. The issuance of certificates for shares of the Common Stock upon conversion of the Series C Non-Voting Preferred Stock shall be made without charge to any Holder for any documentary stamp or similar taxes that may be payable in respect of the issue or delivery of such certificates, provided that the Corporation shall not be required to pay any tax that may be payable in respect of any transfer involved in the issuance and delivery of any such certificate upon conversion in a name other than that of the registered Holder(s) of such shares of Series C Non-Voting Preferred Stock and the Corporation shall not be required to issue or deliver such certificates unless or until the Person or Persons requesting the issuance thereof shall have paid to the Corporation the amount of such tax or shall have established to the satisfaction of the Corporation that such tax has been paid.

6.6 <u>Status as Stockholder</u>. Upon each Conversion Date, (i) the shares of Series C Non-Voting Preferred Stock being converted shall be deemed converted into shares of Common Stock and (ii) the Holder's rights as a holder of such converted shares of Series C Non-Voting Preferred Stock shall cease and terminate, excepting only the right to receive certificates for such shares of Common Stock and to any remedies provided herein or otherwise available at law or in equity to such Holder because of a failure by the Corporation to comply with the terms of this Certificate of Designation. In all cases, the Holder shall retain all of its rights and remedies for the Corporation's failure to convert Series C Non-Voting Preferred Stock. In no event shall the Series C Non-Voting Preferred Stock convert into shares of Common Stock prior to the Stockholder Approval.

7. Certain Adjustments.

7.1 <u>Stock Dividends and Stock Splits</u>. If the Corporation, at any time while this Series C Non-Voting Preferred Stock is outstanding: (A) pays a stock dividend or otherwise makes a distribution or distributions payable in shares of Common Stock (which, for avoidance of doubt, shall not include any shares of Common Stock issued by the Corporation upon conversion of this Series C Non-Voting Preferred Stock) with respect to the then outstanding shares of Common Stock; (B) subdivides outstanding shares of Common Stock into a larger number of shares; or (C) combines (including by way of a reverse stock split) outstanding shares of Common Stock into a smaller number of shares, then the Conversion Ratio shall be multiplied by a fraction of which the numerator shall be the number of shares of Common Stock (excluding any treasury shares of the Corporation) outstanding immediately after such event and of which the denominator shall be the number of shares of Common Stock outstanding immediately before such event (excluding any treasury shares of the Corporation). Any adjustment made pursuant to this <u>Section 7.1</u> shall become effective immediately after the effective date in the case of a subdivision or combination.

Fundamental Transaction. If, at any time while this Series C Non-Voting Preferred Stock is outstanding, (A) the Corporation 7.2 effects any merger or consolidation of the Corporation with or into another Person or any stock sale to, or other business combination (including, without limitation, a reorganization, recapitalization, spin-off, share exchange or scheme of arrangement) with or into another Person (other than such a transaction in which the Corporation is the surviving or continuing entity and its Common Stock is not exchanged for or converted into other securities, cash or property), (B) the Corporation effects any sale, lease, transfer or exclusive license of all or substantially all of its assets in one transaction or a series of related transactions, (C) any tender offer or exchange offer (whether by the Corporation or another Person) is completed pursuant to which more than 50% of the Common Stock not held by the Corporation or such Person is exchanged for or converted into other securities, cash or property, or (D) the Corporation effects any reclassification of the Common Stock or any compulsory share exchange pursuant (other than as a result of a dividend, subdivision or combination covered by Section 7.1) to which the Common Stock is effectively converted into or exchanged for other securities, cash or property (in any such case, a "Fundamental Transaction"), then, upon any subsequent conversion of this Series C Non-Voting Preferred Stock the Holders shall have the right to receive, in lieu of the right to receive Conversion Shares, for each Conversion Share that would have been issuable upon such conversion immediately prior to the occurrence of such Fundamental Transaction, the same kind and amount of securities, cash or property as it would have been entitled to receive upon the occurrence of such Fundamental Transaction if it had been, immediately prior to such Fundamental Transaction, the holder of one share of Common Stock (the "Alternate Consideration"). For purposes of any such subsequent conversion, the determination of the Conversion Ratio shall be appropriately adjusted to apply to such Alternate Consideration based on the amount of Alternate Consideration issuable in respect of one share of Common Stock in such Fundamental Transaction, and the Corporation shall adjust the Conversion Ratio in a reasonable manner reflecting the relative value of any different components of the Alternate Consideration. If holders of Common Stock are given any choice as to the securities, cash or property to be received in a Fundamental Transaction, then the Holders shall be given the same choice as to the Alternate Consideration it receives upon any conversion of this Series C Non-Voting Preferred Stock following such Fundamental Transaction. To the extent necessary to effectuate the foregoing provisions, any successor to the Corporation or surviving entity in such Fundamental Transaction shall file a new certificate of designations with the same terms and conditions and issue to the Holders new preferred stock consistent with the foregoing provisions and evidencing the Holders' right to convert such preferred stock into Alternate Consideration. The terms of any agreement to which the Corporation is a party and pursuant to which a Fundamental Transaction is effected shall include terms requiring any such successor or surviving entity to comply with the provisions of this Section 7.2 and insuring that this Series C Non-Voting Preferred Stock (or any such replacement security) will be similarly adjusted upon any subsequent transaction analogous to a Fundamental Transaction. The Corporation shall cause to be delivered to each Holder, at its last address as it shall appear upon the stock books of the Corporation, written notice of any Fundamental Transaction at least 20 calendar days prior to the date on which such Fundamental Transaction is expected to become effective or close. Notwithstanding anything to the contrary herein, the Corporation's disposition of certain assets pursuant to the CVR Agreement shall not constitute a Fundamental Transaction.

7.3 <u>Calculations</u>. All calculations under this <u>Section 7</u> shall be made to the nearest cent or the nearest 1/100th of a share, as the case may be. For purposes of this <u>Section 7</u>, the number of shares of Common Stock deemed to be issued and outstanding as of a given date shall be the sum of the number of shares of Common Stock (excluding any treasury shares of the Corporation) issued and outstanding.

8. <u>Redemption</u>. The shares of Series C Non-Voting Preferred Stock shall not be redeemable; provided, however, that the foregoing shall not limit the ability of the Corporation to purchase or otherwise deal in such shares to the extent otherwise permitted hereby and by law, nor shall the foregoing limit the Holder's rights under <u>Section 6.5.3</u>.

9. <u>Transfer</u>. A Holder may transfer any shares of Series C Non-Voting Preferred Stock together with the accompanying rights set forth herein, held by such holder without the consent of the Corporation; provided that such transfer is in compliance with applicable securities laws. The Corporation shall in good faith (a) do and perform, or cause to be done and performed, all such further acts and things, and (b) execute and deliver all such other agreements, certificates, instruments and documents, in each case, as any holder of Series C Non-Voting Preferred Stock may reasonably request in order to carry out the intent and accomplish the purposes of this <u>Section 9</u>. The transferee of any shares of Series C Non-Voting Preferred Stock shall be subject to the Beneficial Ownership Limitation applicable to the transferor as of the time of such transfer.

10. Series C Non-Voting Preferred Stock Register. The Corporation shall maintain at its principal executive offices (or such other office or agency of the Corporation as it may designate by notice to the Holders in accordance with Section 11), a register for the Series C Non-Voting Preferred Stock, in which the Corporation shall record (a) the name, address, and electronic mail address of each holder in whose name the shares of Series C Non-Voting Preferred Stock. The Corporation may deem and treat the registered Holder of shares of Series C Non-Voting Preferred Stock as the absolute owner thereof for the purpose of any conversion thereof and for all other purposes. The Corporation shall keep the register open and available at all times during business hours for inspection by any holder of Series C Non-Voting Preferred Stock or his, her or its legal representatives.

11. <u>Notices</u>. Any notice required or permitted by the provisions of this Certificate of Designation to be given to a Holder of shares of Series C Non-Voting Preferred Stock shall be mailed, postage prepaid, to the post office address last shown on the records of the Corporation, or given by electronic communication in compliance with the provisions of the Delaware General Corporation Law, and shall be deemed sent upon such mailing or electronic transmission.

12. <u>Book-Entry; Certificates</u>. The Series C Non-Voting Preferred Stock will be issued in book-entry form; provided that, if a Holder requests that such Holder's shares of Series C Non-Voting Preferred Stock be issued in certificated form, the Corporation will instead issue a stock certificate to such Holder representing such Holder's shares of Series C Non-Voting Preferred Stock. To the extent that any shares of Series C Non-Voting Preferred Stock are issued in book-entry form, references herein to "certificates" shall instead refer to the book-entry notation relating to such shares.

13. <u>Waiver</u>. Any waiver by the Corporation or a Holder of a breach of any provision of this Certificate of Designation shall not operate as or be construed to be a waiver of any other breach of such provision or of any breach of any other provision of this Certificate of Designation or a waiver by any other Holders, other than as expressly set forth herein. The failure of the Corporation or a Holder to insist upon strict adherence to any term of this Certificate of Designation on one or more occasions shall not be considered a waiver or deprive that party (or any other Holder) of the right thereafter to insist upon strict adherence to that term or any other term of this Certificate of Designation. Any waiver by the Corporation or a Holder must be in writing. Notwithstanding any provision in this Certificate of Designation to the contrary, any provision contained herein and any right of the Holders of Series C Non-Voting Preferred Stock granted hereunder may be waived as to all shares of Series C Non-Voting Preferred Stock (and the Holders thereof) upon the written consent of the Holders of not less than a majority of the shares of Series C Non-Voting Preferred Stock then outstanding, provided, however, that the Beneficial Ownership Limitation applicable to a Holder, and any provisions contained herein that are related to such Beneficial Ownership Limitation, cannot be modified, waived or terminated without the consent of such Holder, provided further, that any proposed waiver that would, by its terms, have a disproportionate and materially adverse effect on any Holder shall require the consent of such Holder(s).

14. <u>Severability</u>. Whenever possible, each provision hereof shall be interpreted in a manner as to be effective and valid under applicable law, but if any provision hereof is held to be prohibited by or invalid under applicable law, then such provision shall be ineffective only to the extent of such prohibition or invalidity, without invalidating or otherwise adversely affecting the remaining provisions hereof.

15. <u>Status of Converted Series C Non-Voting Preferred Stock</u>. If any shares of Series C Non-Voting Preferred Stock shall be converted or redeemed by the Corporation, such shares shall, to the fullest extent permitted by applicable law, be retired and cancelled upon such acquisition, and shall not be reissued as a share of Series C Non-Voting Preferred Stock. Any share of Series C Non-Voting Preferred Stock so acquired shall, upon its retirement and cancellation, and upon the taking of any action required by applicable law, resume the status of authorized but unissued shares of preferred stock and shall no longer be designated as Series C Non-Voting Preferred Stock.

[Remainder of Page Intentionally Left Blank]

IN WITNESS WHEREOF, Onconova Therapeutics, Inc. has caused this Certificate of Designation of Preferences, Rights and Limitations of Series C Non-Voting Convertible Preferred Stock to be duly executed by its Chief Executive Officer on April 1, 2024.

ONCONOVA THERAPEUTICS, INC.

By: /s/ Steven M. Fruchtman

Name: Steven M. Fruchtman

Title: President and CEO

ANNEX A

NOTICE OF CONVERSION

(TO BE EXECUTED BY THE REGISTERED HOLDER IN ORDER TO CONVERT SHARES OF SERIES C NON-VOTING CONVERTIBLE PREFERRED STOCK)

The undersigned Holder hereby irrevocably elects to convert the number of shares of Series C Non-Voting Preferred Stock indicated below, represented in book-entry form, into shares of common stock, par value \$0.01 per share (the "*Common Stock*"), of Onconova Therapeutics, Inc., a Delaware corporation (the "*Corporation*"), as of the date written below. If securities are to be issued in the name of a Person other than the undersigned, the undersigned will pay all transfer taxes payable with respect thereto. Capitalized terms utilized but not defined herein shall have the meaning ascribed to such terms in that certain Certificate of Designation of Preferences, Rights and Limitations of Series C Non-Voting Convertible Preferred Stock (the "*Certificate of Designation*") filed by the Corporation with the Secretary of State of the State of Delaware on April 1, 2024.

As of the date hereof, the number of shares of Common Stock beneficially owned by the undersigned Holder (together with such Holder's Attribution Parties), including the number of shares of Common Stock issuable upon conversion of the Series C Non-Voting Preferred Stock subject to this Notice of Conversion, but excluding the number of shares of Common Stock which are issuable upon (A) conversion of the remaining, unconverted Series C Non-Voting Preferred Stock beneficially owned by such Holder or any of its Attribution Parties, and (B) exercise or conversion of the unexercised or unconverted portion of any other securities of the Corporation (including any warrants) beneficially owned by such Holder or any of its Attribution Parties that are subject to a limitation on conversion or exercise similar to the limitation contained in <u>Section 6.4</u> of the Certificate of Designation, is ______. For purposes hereof, beneficial ownership shall be calculated in accordance with Section 13(d) of the Exchange Act and the applicable regulations of the Commission. In addition, for purposes hereof, "group" has the meaning set forth in Section 13(d) of the Exchange Act and the applicable regulations of the Commission.

CONVERSION CALCULATIONS:

Date to Effect Conversion:	
Number of shares of Series C Non-Voting Preferred Stock owned prior to Conversion:	
Number of shares of Series C Non-Voting Preferred Stock to be Converted:	
Number of shares of Common Stock to be Issued:	
Address for delivery of physical certificates:	
For DWAC Delivery, please provide the following:	
Broker No.:	
Account No.:	

[HOLDER]

By:	
Name:	
Title:	

CERTIFICATE OF AMENDMENT TO TENTH AMENDED AND RESTATED CERTIFICATE OF INCORPORATION OF ONCONOVA THERAPEUTICS, INC.

Onconova Therapeutics, Inc. (the "Corporation"), a corporation organized and existing under and by virtue of the laws of the General Corporation Law of the State of Delaware (the "DGCL"),

DOES HEREBY CERTIFY:

- 1. The name of the Corporation is Onconova Therapeutics, Inc. The Tenth Amended and Restated Certificate of Incorporation was originally filed with the Secretary of State of the State of Delaware (the "Secretary of State") on July 30, 2013, and has been amended by a Certificate of Amendment to the Tenth Amended and Restated Certificate of Incorporation filed with the Secretary of State on May 31, 2016, a Certificate of Designation of Preference, Rights and Limitations of Series A Convertible Preferred Stock filed with the Secretary of State on February 8, 2018, a Certificate of Amendment to the Tenth Amended and Restated Certificate of Incorporation, as amended, filed with the Secretary of State on March 21, 2018, a Certificate of Amendment to the Tenth Amended and Restated Certificate of Incorporation, as amended, filed with the Secretary of State on June 7, 2018, a Certificate of Designation of Preference, Rights and Limitations of Series B Convertible Preferred Stock filed with the Secretary of State on April 27, 2018, a Certificate of Amendment to the Tenth Amended and Restated Certificate of Incorporation, as amended, filed with the Secretary of State on April 27, 2018, a Certificate of State on April 27, 2018, a Certificate of Amendment to the Tenth Amended and Restated Certificate of Incorporation, as amended, filed with the Secretary of State on April 27, 2018, a Certificate of State on September 25, 2018, and a Certificate of Designation of Preferences, Rights and Limitations of Series C Non-Voting Convertible Preferred Stock filed with the Secretary of State on April 1, 2024 (the Tenth Amended and Restated Certificate of Incorporation").
- 2. ARTICLE I of the Certificate of Incorporation of the Corporation is hereby amended and restated in its entirety to read as follows:

"The name of the corporation is Traws Pharma, Inc."

3. That the amendment was duly adopted in accordance with the provisions of Section 242 of the DGCL.

[Signature page follows]

IN WITNESS WHEREOF, the Corporation has caused this certificate to be duly adopted and executed in its corporate name and on its behalf by its duly authorized officer as of the 1st day of April, 2024.

By: /s/ Mark Guerin

Name: Mark Guerin Title: Chief Financial Officer

[Signature Page to Certificate of Amendment]

SECURITIES PURCHASE AGREEMENT

This SECURITIES PURCHASE AGREEMENT (this "*Agreement*") is dated as of April 1, 2024, by and among ONCONOVA THERAPEUTICS, INC., a Delaware corporation (the "*Company*"), and each purchaser identified on Annex A hereto (each, including its successors and assigns, a "*Purchaser*" and collectively, the "*Purchasers*").

RECITALS

A. The Company and each Purchaser is executing and delivering this Agreement in reliance upon the exemption from securities registration afforded by Section 4(a)(2) of the Securities Act of 1933, as amended, and the rules and regulations thereunder (the "*Securities Act*"), and Rule 506 of Regulation D ("*Regulation D*") as promulgated by the United States Securities and Exchange Commission (the "*Commission*") under the Securities Act.

B. Each Purchaser, severally and not jointly, wishes to purchase, and the Company wishes to issue and sell, upon the terms and conditions stated in this Agreement, an aggregate of (1) 496,935 shares of the Company's common stock, par value \$0.01 per share ("*Common Stock*") and (2) 1,578.21200 shares of Series C Non-Voting Convertible Preferred Stock, par value \$0.01 per share (and including any other class of securities into which the Series C Preferred Stock may hereafter be reclassified or changed into, the "*Series C Preferred Stock*" and, together with the Common Stock, the "*Securities*") of the Company, having the designation, preferences, conversion or other rights, voting powers, restrictions, limitations as to dividends, qualifications and terms and conditions as specified in the Certificate of Designation, in the form attached hereto as **Exhibit A** (the "*Certificate of Designation*"), which will be convertible into shares (the "*Conversion Shares*") of the Company's Common Stock, in accordance with the terms set forth in the Certificate of Designation.

C. Pursuant to the terms and conditions of the Certificate of Designation, the conversion of the Series C Preferred Stock shall be subject to receipt of the Requisite Stockholder Approval (as defined herein).

D. The Company has engaged Tungsten Advisors as its exclusive placement agent (the "*Placement Agent*") for the offering of the Securities on a "best efforts" basis.

E. Concurrently with the execution and delivery of this Agreement, the Company is entering into an Agreement and Plan of Merger by and among the Company, TRAWS MERGER SUB I, INC., a Delaware corporation and wholly owned subsidiary of Company ("*First Merger Sub*"), TRAWS MERGER SUB II, LLC, a Delaware limited liability company and wholly owned subsidiary of Company ("*Second Merger Sub*"), and Trawsfynydd Therapeutics, Inc. ("*Trawsfynydd*"), in substantially the form attached hereto as **Exhibit E** attached hereto (the "*Merger Agreement*"), pursuant to which the Company and Trawsfynydd intend to effect a merger of First Merger Sub with and into Trawsfynydd (the "*First Merger*"). Upon consummation of the First Merger, the First Merger Sub will cease to exist and Trawsfynydd will become a wholly-owned subsidiary of the Company. Immediately following the First Merger and as part of the same overall transaction as the First Merger, Trawsfynydd will merge with and into Second Merger Sub (the "*Second Merger*") and the First Merger, the First Merger, the First Merger, the First Merger, the "*Merger*"), with Second Merger Sub being the surviving entity of the Second Merger.

F. Prior to the Closing: (i) the parties hereto shall execute and deliver a Registration Rights Agreement, substantially in the form attached hereto as **Exhibit B** (the "*Registration Rights Agreement*"), pursuant to which, among other things, the Company will agree to provide certain registration rights with respect to the Conversion Shares under the Securities Act and the rules and regulations promulgated thereunder and applicable state securities laws and (ii) the Company shall file with the Delaware Secretary of State the Certificate of Designation, duly executed by an officer of the Company.

NOW, THEREFORE, in consideration of the mutual covenants contained in this Agreement, and for other good and valuable consideration, the receipt and adequacy of which are hereby acknowledged, the Company and each Purchaser, severally and not jointly, hereby agree as follows:

ARTICLE 1 DEFINITIONS

1.1 **Definitions**. In addition to the terms defined elsewhere in this Agreement, for all purposes of this Agreement, the following terms shall have the meanings indicated in this Section 1.1:

"Acquiring Person" has the meaning set forth in Section 4.5.

"Action" means any action, suit, inquiry, notice of violation, proceeding (including any partial proceeding such as a deposition) or investigation pending or, to the Company's Knowledge, threatened against the Company, its Subsidiaries or any of their respective properties, or any officer, director or employee of the Company or any of its Subsidiaries acting in his or her capacity as an officer, director or employee, before or by any federal, state, county, local or foreign court, arbitrator, governmental or administrative agency, regulatory authority, stock market, stock exchange or trading facility.

"Affiliate" means, with respect to any Person, any other Person that, directly or indirectly through one or more intermediaries, Controls, is controlled by or is under common control with such Person, as such terms are used in and construed under Rule 405 under the Securities Act.

"Agreement" has the meaning set forth in the Preamble.

"Board of Directors" means the board of directors of the Company.

"Business Day" means any day except Saturday, Sunday, any day which is a federal legal holiday in the United States or any day on which banking institutions in the State of New York are authorized or required by law or other governmental action to close.

"Certificate of Designation" has the meaning set forth in the Recitals.

"*Closing*" has the meaning set forth in Section 2.2(a).

"Closing Date" has the meaning set forth in Section 2.2(a).

"Code" means the Internal Revenue Code of 1986, as amended.

"*Commission*" has the meaning set forth in the Recitals.

"Common Stock" has the meaning set forth in the Recitals.

"Company" has the meaning set forth in the Preamble.

"Company Deliverables" has the meaning set forth in Section 2.3(a).

2

"Company's Knowledge" means with respect to any statement made to the Company's Knowledge, that the statement is based upon the actual knowledge, or knowledge that would have been acquired after reasonable inquiry, of the executive officers or directors of the Company having responsibility for the matter or matters that are the subject of the statement. With respect to any matters relating to Intellectual Property Rights, such awareness or reasonable expectation to have knowledge does not require any such individual to conduct or have conducted or obtain or have obtained any freedom to operate opinions of counsel or any Intellectual Property Rights clearance searches.

"Contract" means, with respect to any Person, any written or oral agreement, contract, subcontract, lease (whether for real or personal property), mortgage, license, or other legally binding commitment or undertaking of any nature to which such Person is a party or by which such Person or any of its assets are bound or affected under applicable Law.

"*Control*" (including the terms "controlling", "controlled by" or "under common control with") means the possession, direct or indirect, of the power to direct or cause the direction of the management and policies of a Person, whether through the ownership of voting securities, by Contract or otherwise.

"Delaware Courts" means Court of Chancery of the State of Delaware, the United States District Court for the District of Delaware, or the Superior Court of the State of Delaware.

"Effect" means any effect, change, event, circumstance, state of fact, occurrence or development.

"*Effective Date*" means the date on which the initial Registration Statement required by Section 2(a) of the Registration Rights Agreement is first declared effective by the Commission.

"Employee Plan" means any "employee benefit plan" as defined in Section 3(3) of ERISA and any other pension, retirement, deferred compensation, excess benefit, profit-sharing, bonus, incentive, equity or equity-based, phantom equity, employment, consulting, severance, change-of-control, retention, health, life, disability, group insurance, paid time off, holiday, welfare and fringe benefit plan, program, agreement, Contract, or arrangement (whether written or unwritten, qualified or nonqualified, funded or unfunded, subject or not subject to ERISA and including any that have been frozen) that the Company or any of its Subsidiaries (i) sponsors, maintains, administers, or contributes to, (ii) provides benefits under or through, (iii) has any obligation to contribute to or provide benefits under or through, (iv) with respect to which have any liability, or (v) utilizes to provide benefits to or other service provider of the Company or any of its Subsidiaries (or their spouses, dependents, or beneficiaries).

"*Encumbrance*" means any lien, pledge, hypothecation, charge, mortgage, security interest, lease, exclusive license, option, easement, reservation, servitude, adverse title, claim, infringement, interference, option, right of first refusal, preemptive right, community property interest or restriction or encumbrance of any nature (including any restriction on the voting of any security, any restriction on the transfer of any security or other asset, any restriction on the receipt of any income derived from any asset, any restriction on the use of any asset and any restriction on the possession, exercise or transfer of any other attribute of ownership of any asset).

"ERISA" means the Employee Retirement Income Security Act of 1974, as amended.

"*Evaluation Date*" has the meaning set forth in Section 3.1(s).

"Exchange Act" means the Securities Exchange Act of 1934, as amended, or any successor statute, and the rules and regulations promulgated thereunder.

3

"Extended Stockholder Approval Period" has the meaning set forth in Section 4.12.

"First Merger" has the meaning set forth in the Recitals.

"*FCPA*" has the meaning set forth in Section 3.1(bb).

"GAAP" means generally accepted accounting principles and practices in effect from time to time within the United States applied consistently throughout the period involved.

"Governmental Authority" means any: (a) nation, state, commonwealth, province, territory, county, municipality, district or other jurisdiction of any nature, (b) federal, state, local, municipal, foreign, supra-national or other government, (c) governmental or quasi-governmental authority of any nature (including any governmental division, department, agency, commission, bureau, instrumentality, official, ministry, fund, foundation, center, organization, unit, body or entity and any court or other tribunal, and any taxing authority) or (d) self-regulatory organization (including Nasdaq).

"Intellectual Property Rights" means Intellectual Property Rights (as defined in the Merger Agreement).

"Irrevocable Transfer Agent Instructions" means, with respect to the Company, the Irrevocable Transfer Agent Instructions, in substantially the form of **Exhibit C**, executed by the Company and delivered to and acknowledged in writing by the Transfer Agent.

"*Law*" means any federal, state, national, supra-national, foreign, local or municipal or other law, statute, constitution, principle of common law, resolution, ordinance, code, edict, decree, rule, regulation, ruling, order, judgment or requirement issued, enacted, adopted, promulgated, implemented or otherwise put into effect by or under the authority of any Governmental Authority (including under the authority of Nasdaq or the Financial Industry Regulatory Authority, Inc.).

"Material Adverse Effect" means any Effect, individually or together with any other Effect, that (a) has had, has, or would reasonably be expected to have a material adverse effect on the business, condition (financial or otherwise), general affairs, management, assets, liabilities, results of operations, earnings, prospects or properties of the Company or its Subsidiaries, taken as a whole; provided, however, that Effects arising or resulting from the following shall not be taken into account in determining whether there has been a Material Adverse Effect: (1) the announcement or disclosure of the sale of the Securities or other transactions contemplated by this Agreement, (2) the taking of any action, or the failure to take any action, by the Company that is required to comply with the terms of this Agreement, (3) any natural disaster or epidemics, pandemics or other force majeure events, or any act or threat of terrorism or war, any armed hostilities or terrorist activities (including any escalation or general worsening of any of the foregoing) anywhere in the world or any governmental or other response or reaction to any of the foregoing, (4) any change in GAAP or applicable Law or the interpretation thereof, (5) general economic or political conditions or conditions generally affecting the industries in which the Company and its Subsidiaries operate or (6) any change in the cash position of the Company and its Subsidiaries which results from operations in the ordinary course of business; except in each case with respect to clauses (3), (4) and (5), (x) to the extent disproportionately affecting the Company and its Subsidiaries, taken as a whole, relative to other similarly situated companies in the industries in which the Company and its Subsidiaries operate and (y) the underlying cause of such Effect may be considered except to the extent such underlying cause would otherwise be excluded in accordance with the foregoing; or (b) prevents, materially adversely delays or materially adversely impedes, or could reasonably be expected to prevent, materially adversely delay or materially adversely impede the performance by the Company of its obligations under this Agreement and the other Transaction Documents, including, without limitation, the issuance and sale of the Securities and the Conversion Shares.



"Merger" has the meaning set forth in the Recitals.

"Merger Agreement" has the meaning set forth in the Recitals.

"Money Laundering Laws" has the meaning set forth in Section 3.1(cc).

"Nasdaq" means The Nasdaq Stock Market.

"OFAC" has the meaning set forth in Section 3.1(dd).

"Outside Date" means the fifteenth (15th) day following the date of this Agreement.

"Permitted Encumbrances" means: (a) any Encumbrance for current taxes not yet due and payable or for taxes that are being contested in good faith and, in each case, for which adequate reserves have been made on the Audited Balance Sheet in accordance with GAAP; (b) minor liens that have arisen in the ordinary course of business and that do not (individually or in the aggregate) materially detract from the value of the assets or properties subject thereto or materially impair the operations of the Company or any of its Subsidiaries; (c) statutory liens to secure obligations to landlords, lessors or renters under leases or rental agreements; (d) deposits or pledges made in connection with, or to secure payment of, workers' compensation, unemployment insurance or similar programs mandated by Law; (e) non-exclusive licenses of Intellectual Property Rights granted by the Company or any of its Subsidiaries in the ordinary course of business and that do not (individually or in the aggregate) materially detract from the value of the Intellectual Property Rights subject thereto; and (f) statutory liens in favor of carriers, warehousemen, mechanics and materialmen, to secure claims for labor, materials or supplies.

"Person" means an individual or a corporation, partnership, trust, incorporated or unincorporated association, joint venture, limited liability company, joint stock company, government (or an agency or subdivision thereof) or other entity of any kind.

"Placement Agent" has the meaning set forth in the Recitals.

"Press Release" has the meaning set forth in Section 4.4.

"Principal Trading Market" means the Trading Market on which the Common Stock is primarily listed on and quoted for trading, which, as of the date of this Agreement and the Closing Date, shall be the Nasdaq Capital Market.

"*Proceeding*" means an action, claim, suit, investigation or proceeding (including, without limitation, an investigation or partial proceeding, such as a deposition), whether commenced or threatened.

"Purchaser" or "Purchasers" has the meaning set forth in the Preamble.

"Purchaser Deliverables" has the meaning set forth in Section 2.2(b).

"Registrable Securities" has the meaning set forth in the Registration Rights Agreement.

"Registration Rights Agreement" has the meaning set forth in the Recitals.

"*Registration Statement*" means a registration statement meeting the requirements set forth in the Registration Rights Agreement and covering the resale by the Purchasers of the Registrable Securities.

5

"*Regulation D*" has the meaning set forth in the Recitals.

"Regulatory Agencies" has the meaning set forth in Section 3.1(hh).

"Requisite Stockholder Approval" has the meaning set forth in Section 4.12.

"*Rule 144*" means Rule 144 promulgated by the Commission pursuant to the Securities Act, as such Rule may be amended from time to time, or any similar rule or regulation hereafter adopted by the Commission having substantially the same effect as such Rule.

"SEC Reports" has the meaning set forth in Section 3.1(h).

"Second Merger" has the meaning set forth in the Recitals.

"Securities" has the meaning set forth in the Recitals.

"Securities Act" has the meaning set forth in the Recitals.

"Series C Preferred Stock" has the meaning set forth in the Recitals, and also includes any other class of securities into which the Series C Preferred Stock may hereafter be reclassified or changed into.

"Short Sales" include, without limitation, (i) all "short sales" as defined in Rule 200 promulgated under Regulation SHO under the Exchange Act, whether or not against the box, and all types of direct and indirect stock pledges, forward sale contracts, options, puts, calls, short sales, swaps, "put equivalent positions" (as defined in Rule 16a-1(h) under the Exchange Act) and similar arrangements (including on a total return basis), and (ii) sales and other transactions through non-U.S. broker dealers or non-U.S. regulated brokers (but shall not be deemed to include the location and/or reservation of borrowable shares of Common Stock).

"Stockholder Meeting" has the meaning set forth in Section 4.12.

"Stockholder Meeting Deadline" has the meaning set forth in Section 4.12.

"Subscription Amount" means, with respect to each Purchaser, the aggregate amount to be paid for the Securities purchased hereunder as indicated on Annex A opposite such Purchaser's name, in United States dollars and in immediately available funds, which amount represents the number of Securities being purchased by such Purchaser multiplied by the per Security price of \$0.86.

"Subsidiary" means any subsidiary of the Company.

"Support Agreement" means each Parent Stockholder Support Agreement (as defined in the Merger Agreement).

"Trading Day" means a day on which the principal Trading Market is open for business.

"*Trading Market*" means any of the following markets or exchanges on which the Common Stock is listed or quoted for trading on the date in question: the NYSE American, the Nasdaq Capital Market, the Nasdaq Global Market, the Nasdaq Global Select Market, or the New York Stock Exchange (or any successors to any of the foregoing).

6

"Transaction Documents" means this Agreement, the schedules and exhibits attached hereto, the Registration Rights Agreement, the Certificate of Designation, the Irrevocable Transfer Agent Instructions and any other documents or agreements explicitly contemplated hereunder.

"Transfer Agent" means EQ Shareowner Services, the current transfer agent of the Company, or any successor transfer agent for the Company.

"Audited Balance Sheet" means the audited condensed consolidated balance sheets of the Company as of December 31, 2023 (the "Balance Sheet Date").

ARTICLE 2 PURCHASE AND SALE

2.1 **Purchase and Sale**. On the Closing Date, upon the terms and subject to the conditions set forth herein, the Company will issue and sell to the Purchasers, and the Purchasers will purchase, severally and not jointly, the number of Securities set forth opposite the names of each such Purchaser under the heading "*Number of Securities Purchased*" for the Subscription Amount set forth on Annex A attached hereto.

2.2 Closing.

(a) **Closing**. The closing of the purchase and sale of the Securities (the "*Closing*") shall take place remotely via the electronic exchange of documents and signatures simultaneously with the execution and delivery of this Agreement (the "*Closing Date*"), or at such other time and place as mutually agreed by the Company and Purchasers, but in no event later than the fifteenth (15th) Business Day following the date of this Agreement.

(b) **Payment**. Within two Business Days following the Closing Date, (i) each Purchaser shall deliver, or cause to be delivered, its Subscription Amount, in United States dollars and in immediately available funds, in the amount set forth in the "Aggregate Purchase Price (Subscription Amount)" column opposite each Purchaser's name in the table set forth on Annex A by wire transfer to an account designated in writing by the Company or by other means approved by the Company on or prior to the Closing Date; and (ii) the Company shall deliver, or cause to be delivered, to each Purchaser against payment therefor a book-entry statement dated the Closing Date from the Transfer Agent evidencing the number of Securities set forth opposite such Purchaser's name on Annex A, registered in the name of such Purchaser (or its nominee in accordance with its delivery instructions), free and clear of any liens or restrictions (other than those arising under state and federal securities laws and bearing the legend set forth in Section 4.1(b). Notwithstanding anything in this Agreement to the contrary and as may be agreed to among the Company and one or more Purchasers, a Purchaser that is a mutual fund and subject to regulations related to the timing of funding and the issuance of securities thereunder or a Purchaser that has internal policies and/or procedures relating to the timing of funding and issuance of securities thereafter shall not be required to wire its respective portion of the Subscription Amount as set forth on Annex A until it confirms receipt of a book-entry statement from the Transfer Agent evidencing the issuance of the Securities to such Purchaser on and as of the Closing Date.

2.3 Closing Deliverables.

(a) On or prior to the Closing, the Company shall issue, deliver or cause to be delivered to each Purchaser the following (the "*Company Deliverables*"):

(i) this Agreement, duly executed by the Company;

(ii) the Registration Rights Agreement, duly executed by the Company;

(iii) a copy of the duly executed Irrevocable Transfer Agent Instructions acknowledged in writing by the Transfer Agent instructing the Transfer Agent to deliver, on an expedited basis, of the issuance of the number of Securities set forth opposite the name of such Purchaser under the heading "*Number of Securities Purchased*" on Annex A attached hereto, registered in the name of such Purchaser (or its nominee, as directed by the Purchaser) as set forth on such Purchaser's signature page hereto;

(iv) the Compliance Certificate referred to in Section 5.1(h);

(v) a certificate of the Secretary of the Company, in a form reasonably acceptable to the Purchasers, dated as of the Closing Date, (A) certifying the resolutions adopted by the Board of Directors or a duly authorized committee thereof approving the transactions contemplated by this Agreement, the other Transaction Documents and the issuance of the Securities and the Conversion Shares, and the Merger, (B) certifying the current versions of the certificate of incorporation, as amended, and bylaws of the Company, and (C) certifying as to the signatures and authority of persons signing the Transaction Documents and related documents on behalf of the Company;

(vi) a certificate evidencing the formation and good standing of the Company issued by the Secretary of State of the State of Delaware, as of a date within three (3) Business Days of the Closing Date and a certificate evidencing the Company's qualification as a foreign corporation and good standing issued by the Secretary of State (or comparable office) of each jurisdiction in which the Company is qualified to do business as a foreign corporation, as of a date within three (3) Business Days of the Closing Date; and

- (vii) a certified copy of the Certificate of Designation, as filed with the Secretary of State of the State of Delaware.
- (b) On or prior to the Closing, each Purchaser shall deliver or cause to be delivered to the Company the following (the "*Purchaser Deliverables*"):
 - (i) this Agreement, duly executed by such Purchaser; and
 - (ii) the Registration Rights Agreement, duly executed by such Purchaser.

ARTICLE 3 REPRESENTATIONS AND WARRANTIES

3.1 **Representations and Warranties of the Company**. Except as previously disclosed in the SEC Reports, the Company hereby represents and warrants the following as of the date hereof and the Closing Date (except for the representations and warranties that speak as of a specific date, which shall be made as of such date) to each of the Purchasers and to the Placement Agent:

(a) **Due Organization; Subsidiaries**. Each of the Company and its Subsidiaries is a corporation or limited liability company duly incorporated or formed, validly existing and in good standing under the Laws of the jurisdiction of its incorporation or organization and has all necessary corporate power and authority: (i) to conduct its business in the manner in which its business is currently being conducted and as proposed to be conducted as described in the SEC Reports, (ii) to own or lease and use its property and assets in the manner in which its property and assets are currently owned or leased and used and (iii) to perform its obligations under all Contracts by which it is bound. Each of the Company and the Subsidiaries is licensed and qualified to do business, and is in good standing (to the extent applicable in such jurisdiction), under the Laws of all jurisdictions where the nature of its business or the manner in which its business is currently being conducted requires such licensing or qualification other than in jurisdictions where the failure to be so qualified individually or in the aggregate would not have or reasonably be expected to have a Material Adverse Effect.

(b) Authorization; Enforcement; Validity. The Company has the requisite corporate power and authority to enter into the Transaction Documents and to perform its obligations under and consummate the transactions contemplated hereby or thereby. All corporate action on the part of the Company, its directors and stockholders necessary for the authorization, execution, sale, issuance and delivery of the Securities and, subject to the Requisite Stockholder Approval, the Conversion Shares contemplated herein has been taken. Each of the Transaction Documents have been (or upon delivery will have been) duly executed and delivered by the Company and is, or when delivered in accordance with the terms hereof or thereof, will constitute the legal, valid and binding obligation of the Company enforceable against the Company in accordance with its respective terms, except (i) as such enforceability may be limited by applicable bankruptcy, examinership, insolvency, reorganization, moratorium, liquidation or similar laws relating to, or affecting generally the enforcement of, creditors' rights and remedies or by other equitable principles of general application, (ii) as limited by Laws relating to the availability of specific performance, injunctive relief or other equitable remedies and (iii) insofar as indemnification and contribution provisions may be limited by applicable law.

(c) No Conflicts. The execution, delivery and performance by the Company of the Transaction Documents and the issuance, sale and delivery of the securities to be sold by the Company under the Transaction Documents (including, subject to the Company obtaining Requisite Stockholder Approval, the issuance of Conversion Shares upon the conversion of the Securities), the performance by the Company of its obligations under the Transaction Documents and the consummation of the transactions contemplated hereby or thereby (including without limitation, the issuance of the Securities and the reservation for issuance of the Conversion Shares) do not and will not conflict with, result in the breach or violation of, or constitute (with or without the giving of notice or the passage of time or both) a violation of, or default under, (i) any bond, debenture, note or other evidence of indebtedness, or under any lease, license, franchise, permit, indenture, mortgage, deed of trust, loan agreement, joint venture or other Contract, agreement or instrument to which the Company or any of its Subsidiaries is a party or by which it or its properties may be bound or affected, (ii) the Company's restated certificate of incorporation, as amended (the "*Certificate of Incorporation*"), the Company's bylaws, as amended (the "*Bylaws*"), or the equivalent document with respect to any of the Company's Subsidiaries, as amended and as in effect on the date hereof, or (iii) subject to the Requisite Stockholder Approval, any statute or Law, judgment, decree, rule, regulation, ordinance or order of any court or governmental or regulatory body (including Nasdaq), governmental agency, arbitration panel or authority applicable to the Company, any of its subsidiaries or their respective properties, except in the case of clauses (i) and (iii) for such conflicts, breaches, violations or defaults that would not reasonably be expected to have, individually or in the aggregate, a Material Adverse Effect.

(d) **Filings, Consents and Approvals**. Except for any Current Report on Form 8-K or Notice of Exempt Offering of Securities on Form D to be filed by the Company in connection with the transaction contemplated hereby, any required filing with Nasdaq (other than the Listing of Additional Shares notification form for the listing of the Conversion Shares), the Requisite Stockholder Approval, the filing of the Certificate of Designation and the Registration Statement required to be filed by the Registration Rights Agreement, neither the Company nor any of its Subsidiaries is required to give any notice to, or make any filings with, or obtain any authorization, consent, or approval of any government or governmental agency in order to consummate the transactions contemplated by the Transaction Documents. Assuming the accuracy of the representations of the Purchasers in Section 3.2, no consent, approval, authorization or other order of, or registration, qualification or filing with, any court, regulatory body, administrative agency, self-regulatory organization, stock exchange or market (including Nasdaq), or other governmental body is required for the execution and delivery of the Transaction Documents, the valid issuance, sale and delivery of the Securities to be sold pursuant to the Transaction Documents (including, subject to the Company obtaining the Requisite Stockholder Approval, the issuance of Conversion Shares upon conversion of the Securities laws applicable to the offering of the Securities or the issuance of Conversion Shares upon conversion of the Securities laws applicable to the offering of the Securities or the issuance of any facts or circumstances that might prevent the Company from obtaining or effecting any of the registration, application or filings pursuant to this Section 3.1(d).

(e) **Issuance of the Securities and Conversion Shares**. The issuance of the Securities has been duly authorized, and the Securities, when issued and paid for in accordance with the terms of the Transaction Documents, will be duly and validly issued, fully paid and nonassessable and free and clear of any Encumbrances, preemptive rights or restrictions (other than as provided in the Certificate of Designation or any restrictions on transfer generally imposed under applicable securities laws). The issuance of the Conversion Shares has been duly authorized and the Conversion Shares, subject to receipt of the Requisite Stockholder Approval, when issued in accordance with the terms of the Certificate of Designation, will be duly authorized, validly issued, fully paid and non-assessable, and shall be free and clear of any Encumbrances, preemptive rights or restrictions (other than as provided in this Agreement or any restrictions on transfer generally imposed under applicable securities laws). The Company has reserved such number of shares of Common Stock sufficient to enable full conversion of all of the Securities to the extent allowable prior to receipt of the Requisite Stockholder Approval, the Company shall have reserved such number of shares of Common Stock sufficient to enable full conversion of all of the Securities to the extent allowable prior to receipt of the Requisite Stockholder Approval, the Company shall have reserved such number of shares of Common Stock sufficient to enable full conversion of all of the Securities to the extent allowable prior to receipt of the Requisite Stockholder Approval, the Company shall have reserved such number of shares of Common Stock sufficient to enable to enable to enable full conversion of all of the Securities.

(f) **Capitalization**. The Company represents and warrants to the Purchasers the information included in Section 3.6 of the Merger Agreement, which is hereby incorporated by reference in all respects, is true and correct in all respects. The Company intends to declare a distribution of one contingent value right for each outstanding share of Common Stock as of a record date shortly following the closing of the Merger, representing the right to receive contingent payments, payable in cash, upon the occurrence of certain events relating to the sale of the Company's legacy non-cash assets (the "*CVRs*").

(g) **Merger Agreement**. The Merger Agreement has been duly and validly authorized, executed and delivered by the Company, the First Merger Sub and the Second Merger Sub and, assuming due authorization, execution and delivery by the other parties thereto, constitutes a valid and binding agreement of the Company, the First Merger Sub and the Second Merger Sub enforceable against the Company, the First Merger Sub and the Second Merger Sub in accordance with its terms, except as enforceability may be limited by applicable bankruptcy, insolvency or similar laws affecting the enforcement of creditors' rights generally or by equitable principles relating to enforceability.

(h) **SEC Reports; Disclosure Materials**. The Company has filed or furnished, as applicable, on a timely basis all forms, statements, schedules, certifications, reports and other documents required to be filed or furnished by it with the Commission under the Exchange Act or the Securities Act since January 1, 2021 (collectively, and in each case including all exhibits and schedules thereto and documents incorporated by reference therein, the "*SEC Reports*"). As of the time it was filed with the Commission (or, if amended or superseded by a filing prior to the date of this Agreement, then on the date of such filing), each of the SEC Reports complied in all material respects with the applicable requirements of the Securities Act or the Exchange Act (as the case may be) and as of the time they were filed, none of the SEC Reports contained any untrue statement of a material fact or omitted to state a material fact required to be stated therein or necessary in order to make the statements therein, in the light of the circumstances under which they were made, not misleading. There are no material outstanding or unresolved comments in comment letters from the staff of the Division of Corporation Finance of the Commission with respect to any of the SEC Reports. The Company meets the requirements for use of Form S-3 under the Securities Act.

(i) **Financial Statements**. As of their respective filing dates, the financial statements (including any related notes) contained or incorporated by reference in the SEC Reports (i) complied as to form in all material respects with the Securities Act and the Exchange Act, as applicable, and the published rules and regulations of the Commission applicable thereto, (ii) were prepared in accordance with GAAP (except as may be indicated in the notes to such financial statements or, in the case of unaudited financial statements, as permitted by Form 10-Q of the Commission, and except that the unaudited financial statements may not contain footnotes and are subject to normal and recurring year-end adjustments that are not reasonably expected to be material in amount) applied on a consistent basis unless otherwise noted therein throughout the periods indicated and (iii) fairly present, in all material respects, the consolidated financial position of the Company as of the respective dates thereof and the results of operations and cash flows of the Company for the periods covered thereby. Other than as expressly disclosed in the SEC Reports filed at least one (1) Business Day prior to the date hereof, there has been no material change in the Company's accounting methods or principles that would be required to be disclosed in the SEC Reports filed at least one (1) Business Day prior to the date hereof, the Company has not incurred any liabilities, contingent or otherwise, except those incurred in the ordinary course of business, consistent (as to amount and nature) with past practices since the date of such financial statements, none of which, individually or in the aggregate, have had or would reasonably be expected to have a Material Adverse Effect. The books of account and other financial records of the Company and each of its Subsidiaries are true and complete in all material respects.

(j) **Independent Accountants**. Ernst & Young LLP, who have certified certain financial statements of the Company and delivered their report with respect to the audited financial statements included in the SEC Reports, have at all times since the date of enactment of the Sarbanes-Oxley Act been (i) a registered public accounting firm (as defined in Section 2(a)(12) of the Sarbanes-Oxley Act), (ii) to the Company's Knowledge, "independent" with respect to the Company within the meaning of Regulation S-X under the Exchange Act and (iii) to the Company's Knowledge, in compliance with subsections (g) through (l) of Section 10A of the Exchange Act and the rules and regulations promulgated by the Commission and the Public Accounting Oversight Board thereunder.

Absence of Certain Changes. Since the Balance Sheet Date, there has been (i) no material adverse change to, and no material (k) adverse development in, the assets, liabilities, business, properties, operations, condition (financial or otherwise), results of operations or prospects of the Company or its Subsidiaries, (ii) no Material Adverse Effect, (iii) no satisfaction or discharge of any material Encumbrance or payment of any obligation by the Company, except in the ordinary course of business and (iv) no waiver, not in the ordinary course of business, by the Company or any Subsidiary of a material right or of a material debt owed to it. Since Balance Sheet Date, neither the Company nor any of its Subsidiaries has (i) purchased any of its outstanding Common Stock (other than from its employees or other service providers in connection with the termination of their service pursuant to the terms of its equity compensation plans or agreements) or declared or paid any dividends or distributions, other than the CVRs, (ii) sold any material assets, individually or in the aggregate, outside of the ordinary course of business, (iii) made any material change or material amendment to, or waiver of any material right, or termination of, any material Contract, (iv) had material transaction entered into or material capital expenditures, individually or in the aggregate, outside of the ordinary course of business or (v) experienced any loss of services of any executive officer (as defined in Rule 405 under the Securities Act), other than as disclosed in the SEC Reports prior to the date hereof. Neither the Company nor any of its Subsidiaries has taken any steps to seek protection pursuant to any bankruptcy law, nor does the Company have any knowledge or reason to believe that its creditors (if any) intend to initiate involuntary bankruptcy proceedings or any actual knowledge of any fact that would reasonably lead any such creditor to do so. The Company and its Subsidiaries, individually and on a consolidated basis, are not as of the date hereof, and after giving effect to the transactions contemplated hereby to occur at the Closing, will not be Insolvent (as defined below). For purposes of this Section 3.1(k), "Insolvent" means, with respect to any Person, (i) the present fair saleable value of such Person's assets is less than the amount required to pay such Person's total indebtedness, (ii) such Person is unable to pay its debts and liabilities, subordinated, contingent or otherwise, as such debts and liabilities become absolute and matured, (iii) such Person intends to incur or believes that it will incur debts that would be beyond its ability to pay as such debts mature or (iv) such Person has unreasonably small capital with which to conduct the business in which it is engaged as such business is now conducted and is proposed to be conducted.

(1) Litigation. There is no Action pending or, to the Company's Knowledge, currently threatened in writing against the Company, any of its Subsidiaries or any of their respective directors and officers that questions the validity of the Transaction Documents or the right of the Company to enter into the Transaction Documents or to consummate the transactions contemplated hereby and thereby. There is no Action pending or, to the Company's Knowledge, currently threatened in writing against the Company or any Subsidiary or any of their respective directors and officers which would, if there were an unfavorable decision, have or reasonably be expected to have, either individually or in the aggregate, a Material Adverse Effect.

(m) Employment Matters. No material labor dispute exists or, to the Company's Knowledge, is imminent with respect to any of the employees of the Company or any of its Subsidiaries which would have or would reasonably be expected to result in a Material Adverse Effect. None of the Company's or any Subsidiary's employees is a member of a labor union that relates to such employee's relationship with the Company, and neither the Company nor any of its Subsidiaries is a party to a collective bargaining agreement. To the Company's Knowledge, no executive officer or key employee of the Company or any Subsidiary, is, or is now expected to be, in violation of any material term of any employment Contract, confidentiality, disclosure or proprietary information agreement or non-competition agreement, or any other Contract or agreement or any restrictive covenant in favor of any third party, and to the Company's Knowledge, the continued employment of each such executive officer or key employee does not subject the Company or any Subsidiary to any liability with respect to any of the foregoing matters, except, in each case, matters that, individually or in the aggregate, would not reasonably be expected to result in a Material Adverse Effect. The Company is in compliance in all material respects with all U.S. federal, state, local and foreign Laws and regulations relating to employment and employment practices, terms and conditions of employment and wages and hours. Any Employee Plans that are "employee pension benefit plans" within the meaning of Section 3(2) of ERISA and which are intended to meet the qualification requirements of Section 401(a) of the Code have received determination or opinion letters from the Internal Revenue Service on which they may currently rely to the effect that such plans are qualified under Section 401(a) of the Code and the related trusts are exempt from federal income taxes under Section 501(a) of the Code, respectively, and nothing has occurred that would reasonably be expected to materially adversely affect the qualification of such Employee Plan or the tax exempt status of the related trust.

(n) **Conduct of Business; Regulatory Permits**. Neither the Company nor any of its Subsidiaries is in violation of any term of or in default under its Certificate of Incorporation, any Certificate of Designation of any outstanding series of preferred stock of the Company or the Bylaws or their organizational charter or bylaws, respectively. Neither the Company nor any of its Subsidiaries (i) is in default of or in violation of, nor has the Company or any of its Subsidiaries received written notice of a claim that it is in default under or that it is in violation of, any Parent Material Contract (as defined in the Merger Agreement) (whether or not such default or violation has been waived), or (ii) is in violation of any judgment, decree or order or any statute, ordinance, rule or regulation applicable to the Company or its Subsidiaries, and neither the Company nor any of its Subsidiaries will conduct its business in violation of any of the foregoing, except for possible violations which would not, individually or in the aggregate, have or reasonably be expected to have a Material Adverse Effect. Without limiting the generality of the foregoing, the Company is not in violation of any of the rules, regulations or requirements of the Nasdaq and, to the Company's Knowledge, there exist no facts or circumstances that would reasonably lead to delisting or suspension of the Common Stock by Nasdaq in the foreseeable future. The Company and its Subsidiaries possess all certificates, authorizations and permits issued by the appropriate regulatory authorities necessary to conduct their respective businesses as currently conducted and as proposed to be conducted, except where the failure to possess such certificates, authorizations or permits would not reasonably be expected to have, individually or in the aggregate, a Material Adverse Effect, and neither the Company nor any such Subsidiary has received any written notice of Proceedings relating to the revocation or modification of any such certificate, authorization or per

(o) **Title to Properties and Assets**. None of the Company or its Subsidiaries owns, or has ever owned, any real property. The Company's or is Subsidiary's possession, occupancy, lease, use and/or operation of each such leased property conforms to all applicable Laws in all material respects, and the Company or its Subsidiary, as applicable, has exclusive possession of each such leased property and leasehold interest and has not granted any occupancy rights to tenants or licensees with respect to such leased property or leasehold interest. In addition, each such leased property and leasehold interest is free and clear of all Encumbrances other than Permitted Encumbrances. The Company and each of its Subsidiaries owns, and has good and marketable title to, or, in the case of leased properties and assets, valid leasehold interests in, all tangible properties or tangible assets and equipment used or held for use in their respective business or operations or purported to be owned by any of them, including: (i) all tangible assets reflected on the Audited Balance Sheet and (ii) all other tangible assets reflected in the books and records of the Company as being owned by the Company or any of its Subsidiaries. All of such assets are owned or, in the case of leased assets, leased by the Company or any of its Subsidiaries free and clear of any Encumbrances.

(p) **Intellectual Property Rights**. The Company represents and warrants to the Purchasers the information included in Section 3.12 of the Merger Agreement, which is hereby incorporated by reference in all respects, is true and correct in all respects.

(q) **Transactions with Affiliates and Employees**. Since the date of the Company's Definitive Proxy Statement on Schedule 14A filed with the Commission on June 7, 2023, no event has occurred that would be required to be reported by the Company pursuant to Item 404 of Regulation S-K promulgated by the SEC.

(r) **Company's Accounting System**. The Company and each of its Subsidiaries makes and keeps accurate books and records and maintains a system of internal control over financial reporting (as defined in Rules 13a-15 and 15d-15 under the Exchange Act) sufficient to provide reasonable assurance that: (i) transactions are executed in accordance with management's general or specific authorization; (ii) transactions are recorded as necessary to permit preparation of financial statements in conformity with generally accepted accounting principles as applied in the United States and to maintain accountability for assets; (iii) access to assets is permitted only in accordance with management's general or specific authorization; (iv) the recorded accountability for assets is compared with existing assets at reasonable intervals and appropriate action is taken with respect to any differences; and (v) the interactive data in eXtensible Business Reporting Language included or incorporated by reference in the SEC Reports fairly presents the information called for in all material respects and is prepared in accordance with the Commission's rules and guidelines applicable thereto, except as would not, individually or the in aggregate, be reasonably expected to have a Material Adverse Effect. Since January 1, 2021, (i) neither the Company nor any Subsidiary nor, to the Company's Knowledge, any director, officer, employee, auditor, accountant or representative of the Company or any Subsidiary has received or otherwise had or obtained knowledge of any material complaint, allegation, assertion or claim that the Company or any Subsidiary or their respective internal accounting or auditing practices, procedures, methodologies or methods of the Company or any Subsidiary or their respective internal accounting or auditing practices.

(s) Sarbanes-Oxley; Disclosure Controls. The Company is in compliance in all material respects with the applicable provisions of the Sarbanes-Oxley Act of 2002 and the rules and regulations promulgated thereunder. The Company maintains a system of internal accounting controls designed to ensure that (i) material information relating to the Company and its Subsidiaries, is made known to the Company's principal executive officer and its principal financial officer by others within those entities and (ii) that information required to be disclosed by the Company in reports that it file, furnishes or submits under the Exchange Act is recorded, processed, summarized and reported within the time periods specified in the Commission's rules and forms, including controls and procedures designed to ensure that such information is accumulated and communicated to the Company's management as appropriate to allow timely decisions regarding required disclosure. The Company has established and maintains disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) for the Company and designed such disclosure controls and procedures to ensure that material information relating to the Company is made known to the certifying officers by others within those entities, particularly during the period in which the Company's most recently filed periodic report under the Exchange Act, as the case may be, is being prepared. The Company has established internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with GAAP. The Company's certifying officers have evaluated the effectiveness of the Company's disclosure controls and procedures and the Company's internal control over financial reporting (collectively, "internal controls") as of the end of the period covered by the most recently filed periodic report under the Exchange Act (such date, the "Evaluation Date"). The Company presented in its most recently filed periodic report under the Exchange Act the conclusions of the certifying officers about the effectiveness of such internal controls based on their evaluations as of the Evaluation Date. Since the Evaluation Date, there have been no significant changes in the Company's internal controls or, to the Company's Knowledge, in other factors that could materially affect the Company's internal controls and there have been no material weaknesses in the Company's internal control over financial reporting (whether or not remediated). The Company maintains and will continue to maintain a standard system of accounting established and administered in accordance with GAAP and the applicable requirements of the Exchange Act.

(t) **Private Placement**. Assuming the accuracy of the Purchasers' representations and warranties set forth in Section 3.2 of this Agreement, no registration under the Securities Act is required for the offer and sale of the Securities by the Company to the Purchasers under the Transaction Documents (including, subject to the Company obtaining Requisite Stockholder Approval, the issuance of Conversion Shares upon the conversion of the Securities). The issuance and sale of the Securities hereunder (including, subject to the Company obtaining Requisite Stockholder Approval, the issuance of Conversion Shares upon the conversion of the Securities) does not contravene the rules and regulations of the Trading Market.

(u) **Company Not an "Investment Company."** The Company is not, and will not be, immediately after receipt of payment for the Securities, required to register as an "investment company" under the Investment Company Act of 1940, as amended.

(v) **Registration Rights**. Other than each of the Purchasers, no Person has any right to cause the Company to effect the registration under the Securities Act of the offer and sale of any securities of the Company other than those offers and sales which are currently registered on an effective registration statement on file with the Commission.

(w) **Listing and Maintenance Requirements**. The Company's Common Stock is registered pursuant to Section 12(b) or Section 12(g) of the Exchange Act, and the Company has taken no action designed to terminate the registration of the Common Stock under the Exchange Act, nor has the Company received any notification that the Commission or the Principal Trading Market is contemplating terminating such registration or listing. The Company is, and immediately following the Closing will be, in compliance with all applicable listing requirements of the Principal Trading Market. The Company has filed with Nasdaq an additional shares listing application covering Conversion Shares and has not received any objections from Nasdaq with respect to such application with respect to the transactions contemplated hereby or by the Merger Agreement.

(x) **Disclosure**. The Company confirms that it has not provided, and to the Company's Knowledge, none of its officers or directors nor any other Person acting on its or their behalf (including, without limitation, the Placement Agent) has provided, and it has not authorized the Placement Agent to provide, any Purchaser or its respective agents or counsel with any information that it believes constitutes material, non-public information except insofar as the existence, provisions and terms of the Transaction Documents, the Merger Agreement, and the proposed transactions hereunder and thereunder may constitute such information, all of which will be disclosed by the Company in the Press Release as contemplated by Section 4.4 hereof. The Company understands and confirms that the Purchasers will rely on the foregoing representations in effecting transactions in securities of the Company.

(y) **No Integrated Offering**. Assuming the accuracy of the Purchasers' representations and warranties set forth in Section 3.2, and except with respect to the capital stock to be issued pursuant to the Merger Agreement, none of the Company, its Subsidiaries nor, to the Company's Knowledge, any of its Affiliates or any Person acting on its behalf has, directly or indirectly, at any time within the past six (6) months, made any offers or sales of any Company security or solicited any offers to buy any security under circumstances that would (i) eliminate the availability of the exemption from registration under the Securities Act, including Regulation D, in connection with the offer and sale by the Company of the Securities as contemplated hereby or (ii) cause the offering of the Securities pursuant to the Transaction Documents to be integrated with prior offerings by the Company for purposes of any applicable law, regulation or stockholder approval provisions, including, without limitation, under the rules and regulations of any Trading Market on which any of the securities of the Company are listed or designated.

(z) **Tax Matters**. The Company and each of its Subsidiaries has timely filed all income tax returns and all other material tax returns that were required to be filed by or with respect to it under applicable Law. All such tax returns were correct and complete in all material respects and have been prepared in material compliance with all applicable Law. Subject to exceptions as would not be material, no claim has ever been made by a Governmental Authority in a jurisdiction where the Company or any of its Subsidiaries does not file tax returns that the Company or any of its Subsidiaries is subject to taxation by that jurisdiction. All material amounts of taxes due and owing by the Company and each of its Subsidiaries (whether or not shown on any tax return) have been timely paid. The unpaid taxes of the Company and each of its Subsidiaries for periods (or portions thereof) ending on or prior to the Balance Sheet Date do not materially exceed the accruals for current taxes set forth on the Audited Balance Sheet. Since the Balance Sheet Date, neither the Company nor any of its Subsidiaries has incurred any material liability for taxes outside the ordinary course of business or otherwise inconsistent with past custom and practice. The Company is classified as a Subchapter C corporation for U.S. federal tax purposes.

(aa) **No General Solicitation**. Neither the Company nor, to the Company's Knowledge, any person acting on behalf of the Company has, directly or indirectly, offered or sold any of the Securities or Conversion Shares, or solicited any offers to buy any Securities or Conversion Shares, under any circumstances that would require registration under the Securities Act of the Securities or the Conversion Shares, including by any form of general solicitation or general advertising.

(bb) Anti-Corruption and Anti-Bribery Laws. Neither the Company nor any of its Subsidiaries nor any director, officer, or employee of the Company or any of its Subsidiaries, nor to the Company's Knowledge, any agent, Affiliate or other person acting on behalf of the Company or any of its Subsidiaries has, in the course of its actions for, or on behalf of, the Company or any of its Subsidiaries (i) used any corporate funds for any unlawful contribution, gift, entertainment or other unlawful expenses relating to political activity; (ii) made or taken any act in furtherance of an offer, promise, or authorization of any direct or indirect unlawful payment or benefit to any non-U.S. or domestic government official or employee, including of any government-owned or controlled entity or public international organization, or any political party, party official, or candidate for political office; (iii) violated or is in violation of any provision of the U.S. Foreign Corrupt Practices Act of 1977, as amended (the "*FCPA*"), the UK Bribery Act 2010, or any other applicable anti-bribery or anti-corruption Law; or (iv) made, offered, authorized, requested, or taken an act in furtherance of any unlawful bribe, rebate, payoff, influence payment, kickback or other unlawful payment or benefit. The Company and its Subsidiaries and, to the Company's Knowledge, the Company's Affiliates have conducted their respective businesses in compliance with the FCPA and have instituted and maintain policies and procedures designed to ensure, and which are reasonably expected to continue to ensure, continued compliance therewith.

(cc) **Money Laundering Laws**. The operations of the Company and its Subsidiaries are and have been conducted at all times in compliance with applicable financial recordkeeping and reporting requirements of the USA Patriot Act, the Bank Secrecy Act of 1970, as amended, the money laundering statutes of all jurisdictions, the rules and regulations thereunder and any related or similar rules, regulations or guidelines, issued, administered or enforced by any governmental agency (collectively, the "*Money Laundering Laws*"); and no Action by or before any court or governmental agency, authority or body or any arbitrator or non-governmental authority involving the Company or its Subsidiaries with respect to the Money Laundering Laws is pending or, to the Company's Knowledge, threatened.

(dd) **OFAC**. Neither the Company nor its Subsidiaries nor any of their respective Affiliates, directors, officers, nor to the Company's Knowledge, any agent or employee of the Company or its Subsidiaries is subject to any sanctions administered or enforced by the Office of Foreign Assets Control ("**OFAC**") of the United States Treasury Department, the U.S. Department of State, the United Nations Security Council, the European Union, His Majesty's Treasury or any other relevant sanctions authority; and the Company will not directly or indirectly use the proceeds of the offering of the securities contemplated hereby, or lend, contribute or otherwise make available such proceeds to any Subsidiary, joint venture partner or other person or entity for the purpose of financing the activities of any person that is the target of sanctions administered or enforced by such authorities or in connection with any country or territory that is the target of country- or territory-wide OFAC sanctions (currently, Iran, Syria, Cuba, North Korea, the Crimea, so-called Donetsk People's Republic, and so-called Luhansk People's Republic regions of Ukraine).

(ee) **Off Balance Sheet Arrangements**. There is no transaction, arrangement, or other relationship between the Company (or any Subsidiary) and an unconsolidated or other off balance sheet entity that is required to be disclosed by the Company in SEC Reports and is not so disclosed and would have or reasonably be expected to result in a Material Adverse Effect.

(ff) Acknowledgment Regarding Purchaser's Purchase of Securities. The Company acknowledges and agrees that each Purchaser is acting solely in the capacity of an arm's length purchaser with respect to this Agreement and the other Transaction Documents and the transactions contemplated hereby and thereby, and that the obligations of each Purchaser under this Agreement and the other Transaction Documents are several and not joint. The Company further acknowledges that no Purchaser is acting as a financial advisor or fiduciary of the Company or any of its Subsidiaries (or in any similar capacity) with respect to this Agreement and the other Transaction Documents and thereby, and any advice given by a Purchaser or any of its representatives or agents in connection with this Agreement and the other Transaction Documents and the transactions contemplated hereby and thereby is merely incidental to such Purchaser's purchase of the Securities. The Company further represents to each Purchaser that the Company's decision to enter into the Transaction Documents has been based solely on the independent evaluation by the Company and its representatives.

(gg) No Price Stabilization or Manipulation; Compliance with Regulation M. Neither the Company nor any of its Subsidiaries has taken, directly or indirectly, any action designed to or that might cause or result in stabilization or manipulation of the price of any security of the Company to facilitate the sale or resule of the Securities or Conversion Shares or otherwise, and has taken no action which would directly or indirectly violate Regulation M under the Exchange Act.

(hh) **Clinical Data and Regulatory Compliance**. The preclinical tests and clinical trials, and other studies (collectively, "studies") that are described in, or the results of which are referred to in, the SEC Reports were and, if still pending, are being conducted in all material respects in accordance with the protocols, procedures and controls designed and approved for such studies and with standard medical and scientific research procedures. Each description of the results of such studies is accurate and complete in all material respects and fairly presents the data derived from such studies, and the Company and its Subsidiaries have no knowledge of any other studies the results of which are inconsistent with, or otherwise call into question, the results described or referred to in the SEC Reports. The Company and its Subsidiaries have made all such filings and obtained all such approvals as may be required by the Food and Drug Administration of the U.S. Department of Health and Human Services or any committee thereof or from any other U.S. or non-U.S. government or drug or medical device regulatory agency, or health care facility Institutional Review Board (collectively, the "**Regulatory Agencies**"). Neither the Company nor any of its Subsidiaries has received any notice of, or correspondence from, any Regulatory Agency requiring the termination, suspension or modification of any clinical trials that are described or referred to in the SEC Reports. The Company and its Subsidiaries have each operated and currently are in compliance in all material respects with all applicable rules, regulations and policies of the Regulatory Agencies.

(ii) **No Additional Agreements**. The Company does not have any agreement or understanding (including side letters) with any Purchaser with respect to the transactions contemplated by the Transaction Documents other than as specified in the Transaction Documents.

(jj) **No Disqualification Events**. No "bad actor" disqualifying event described in Rule 506(d)(1)(i)-(viii) of the Securities Act (a "*Disqualification Event*") is applicable to the Company or, to the Company's Knowledge, any Company Covered Person (as defined below), except for a Disqualification Event as to which Rule 506(d)(2)(ii)-(iv) or (d)(3), is applicable. "*Company Covered Person*" means, with respect to the Company as an "issuer" for purposes of Rule 506 promulgated under the Securities Act, any person listed in the first paragraph of Rule 506(d)(1). Other than the Placement Agent, the Company is not aware of any Person (other than any Company Covered Person) that has been or will be paid (directly or indirectly) remuneration for solicitation of purchasers in connection with the sale of the Securities or the Conversion Shares pursuant to this Agreement. The Company has complied, to the extent applicable, with its disclosure obligations under Rule 506(e), and has furnished to the Placement Agent a copy of any disclosures provided thereunder.

(kk) Shell Company Status. The Company is not, and has never been, an issuer identified in Rule 144(i)(1).

Security. Except as would not reasonably be expected to have a Material Adverse Effect, the Company and its Subsidiaries' (11) information technology assets and equipment, computers, systems, networks, hardware, software, websites, applications, and databases (collectively, "IT Systems") are adequate for, and operate and perform in all respects as required in connection with the operation of the business of the Company and its Subsidiaries as currently conducted, and are free and clear of all material Trojan horses, time bombs, malware and other malicious code. The Company and its Subsidiaries have implemented and maintained commercially reasonable physical, technical and administrative controls designed to maintain and protect the confidentiality, integrity, availability, privacy and security of all sensitive, confidential or regulated data ("Confidential Data") used or maintained in connection with their businesses and Personal Data, and the integrity, availability continuous operation, redundancy and security of all IT Systems. "Personal Data" means the following data used in connection with the Company's and its Subsidiaries' businesses and in their possession or control: (i) a natural person's name, street address, telephone number, e-mail address, photograph, social security number or other tax identification number, driver's license number, passport number, credit card number, bank information, or customer or account number; (ii) information that identifies, relates to, or may reasonably be used to identify an individual; (iii) any information regarding an individual's medical history, mental or physical condition, or medical treatment or diagnosis by a health care professional; (iv) an individual's health insurance policy number or subscriber identification number, any unique identifier used by a health insurer to identify the individual, or any information in an individual's application and claims history; (v) any information which would qualify as "protected health information" under the Health Insurance Portability and Accountability Act of 1996, as amended by the Health Information Technology for Economic and Clinical Health Act (collectively, "HIPAA"); (vi) any information which would qualify as "personal data," "personal information" (or similar term) under the Privacy Laws (as defined below); and (vii) any other piece of information that alone, or combined with other information, allows the identification of such natural person, or his or her family, or permits the collection or analysis of any data related to an identified person's health or sexual orientation. To the Company's Knowledge, there have been no breaches, outages or unauthorized uses of or accesses to the IT Systems, Confidential Data, and Personal Data. The Company and its Subsidiaries are presently, and at all prior times were, in material compliance with all applicable laws or statutes and all judgments and orders binding on the Company, applicable binding rules and regulations of any court or arbitrator or governmental or regulatory authority, and their internal policies and contractual obligations, each relating to the Processing, privacy and security of Personal Data and Confidential Data, the privacy and security of IT Systems and the protection of such IT Systems, Confidential Data, and Personal Data from unauthorized use, access, misappropriation or modification.

(mm) **Compliance with Data Privacy Laws**. The Company and its Subsidiaries are, and at all prior times to January 1, 2021 were, in material compliance with all applicable state and federal data privacy and security Laws and regulations regarding the collection, use, storage, retention, disclosure, transfer, disposal, or any other processing (collectively "*Processing*") of Personal Data, including HIPAA, the California Consumer Privacy Act, and the European Union General Data Protection Regulation (EU 2016/679) (collectively, the "*Privacy Laws*"). To ensure compliance with the Privacy Laws, the Company and its Subsidiaries have in place, comply with, and take all appropriate steps necessary to ensure compliance in all material respects with their policies and procedures relating to data privacy and security, and the Processing of Personal Data and Confidential Data (the "*Privacy Statements*"). The Company and its Subsidiaries have, except as would not reasonably be expected, individually or in the aggregate, to have a Material Adverse Effect, at all times since January 1, 2021 provided accurate notice of its Privacy Statements have been materially inaccurate, misleading, incomplete, or in material violation of any Privacy Laws. The Company further certifies that neither it nor any of its Subsidiaries: (i) has received notice of any actual or potential claim, complaint, proceeding, regulatory proceeding or liability under or relating to, or actual or potential violation of, any of the Privacy Laws, contracts related to the Processing of Personal Data, or Privacy Statements, and has no knowledge of any event or condition that would reasonably be expected to result in any such notice; (ii) is currently conducting or paying for, in whole or in part, any investigation, remediation, or other corrective action pursuant to any Privacy Law or contract; or (iii) is a party to any order, decree, or agreement that imposes any obligation or liability under any Privacy Law.

3.2 **Representations and Warranties of the Purchasers**. Each Purchaser hereby, for itself and for no other Purchaser, represents and warrants to the Company as follows:

(a) **Organization; Authority**. Such Purchaser is an entity duly organized, validly existing and in good standing under the Laws of the jurisdiction of its organization with the requisite corporate or, if such Purchaser is not a corporation, such partnership, limited liability company or other applicable power and authority to enter into and to consummate the transactions contemplated by the applicable Transaction Documents and otherwise to carry out its obligations hereunder and thereunder. The execution and delivery of this Agreement by such Purchaser and performance by such Purchaser of the transactions contemplated by this Agreement have been duly authorized by all necessary corporate or, if such Purchaser is not a corporation, such partnership, limited liability company or other applicable like action, on the part of such Purchaser. Each Transaction Document to which it is a party has been duly executed by such Purchaser, and when delivered by such Purchaser in accordance with the terms hereof, will constitute the valid and legally binding obligation of such Purchaser, enforceable against it in accordance with its terms, except (i) as such enforceability may be limited by applicable bankruptcy, examinership, insolvency, reorganization, moratorium, liquidation or similar laws relating to, or affecting generally the enforcement of, creditors' rights and remedies or by other equitable principles of general application, (ii) as limited by Laws relating to the availability of specific performance, injunctive relief or other equitable remedies and (iii) insofar as indemnification and contribution provisions may be limited by applicable Law.

(b) No Conflicts. The execution, delivery and performance by such Purchaser of this Agreement and the Registration Rights Agreement and the consummation by such Purchaser of the transactions contemplated hereby and thereby will not (i) result in a violation of the organizational documents of such Purchaser, (ii) conflict with, or constitute a default (or an event which with notice or lapse of time or both would become a default) under, or give to others any rights of termination, amendment, acceleration or cancellation of, any agreement, indenture or instrument to which such Purchaser is a party, or (iii) result in a violation of any Law, rule, regulation, order, judgment or decree (including U.S. federal and state securities laws) applicable to such Purchaser, except in the case of clauses (ii) and (iii) above, for such conflicts, defaults, rights or violations which would not, individually or in the aggregate, reasonably be expected to have a material adverse effect on the ability of such Purchaser to perform its obligations hereunder.



(c) **Investment Intent**. Such Purchaser understands that the Securities are (and the Conversion Shares will be) "restricted securities" and the offer and sale thereof have not been registered under the Securities Act or any applicable U.S. state securities law and is acquiring the Securities as principal for its own account and not with a view to, or for distributing or reselling such Securities (or the Conversion Shares) or any part thereof in violation of the Securities Act or any applicable U.S. state or other securities laws, *provided*, *however*, that by making the representations herein, such Purchaser does not agree to hold any of the Securities or Conversion Shares for any minimum period of time and reserves the right, subject to the provisions of this Agreement and the Registration Rights Agreement, at all times to sell or otherwise dispose of all or any part of such Securities or Conversion Shares pursuant to an effective registration statement under the Securities Act or under an exemption from such registration and in compliance with applicable U.S. federal, state and other securities laws. Such Purchaser is acquiring the Securities hereunder in the ordinary course of its business.

Such Purchaser does not presently have any agreement, plan or understanding, directly or indirectly, with any Person to distribute or effect any distribution of any of the Securities (or any securities which are derivatives thereof) to or through any person or entity in violation of federal securities law; such Purchaser is not a registered broker-dealer under Section 15 of the Exchange Act or an entity engaged in a business that would require it to be so registered as a broker-dealer.

(d) **Purchaser Status**. At the time such Purchaser was offered the Securities, it was, and at the date hereof it is, an "accredited investor" as defined in Rule 501(a) under the Securities Act.

(e) **General Solicitation**. Such Purchaser is not purchasing the Securities as a result of any advertisement, article, notice or other communication regarding the Securities published in any newspaper, magazine or similar media or broadcast over television or radio or presented at any seminar or any other general advertisement. The purchase of the Securities by such Purchaser has not been solicited by or through anyone other than the Company or, on the Company's behalf, the Placement Agent.

(f) **Experience of Such Purchaser**. Such Purchaser, either alone or together with its representatives, has such knowledge, sophistication and experience in business and financial matters so as to be capable of evaluating the merits and risks of the prospective investment in the Securities, and has so evaluated the merits and risks of such investment. Such Purchaser is able to bear the economic risk of an investment in the Securities.

(g) Access to Information. Such Purchaser acknowledges that it has had the opportunity to review the SEC Reports and has been afforded (i) the opportunity to ask such questions as it has deemed necessary of, and to receive answers from, representatives of the Company concerning the terms and conditions of the offering of the Securities and the merits and risks of investing in the Securities; (ii) access to information about the Company and the Subsidiaries and their respective financial condition, results of operations, business, properties, management and prospects sufficient to enable it to evaluate its investment; and (iii) the opportunity to obtain such additional information that the Company possesses or can acquire without unreasonable effort or expense that is necessary to make an informed investment decision with respect to the investment. Neither such inquiries nor any other investigation conducted by or on behalf of such Purchaser or its representatives or counsel shall modify, amend or affect such Purchaser's right to rely on the truth, accuracy and completeness of the SEC Reports and the Company's representations and warranties contained in the Transaction Documents. Such Purchaser has sought such accounting, legal and tax advice as it has considered necessary to make an informed decision with respect to its acquisition of the Securities.

Certain Trading Activities. Other than with respect to the transactions contemplated herein, since the time that such Purchaser (h) was first contacted by the Company, the Placement Agent or any other Person regarding the transactions contemplated hereby, the Purchaser has not. directly or indirectly, effected or agreed to effect any Short Sales. Notwithstanding the foregoing, (i) in the case of a Purchaser that is a multi-managed investment vehicle whereby separate portfolio managers manage separate portions of such Purchaser's assets, the foregoing representation shall only apply with respect to the portion of assets managed by the portfolio manager that made the investment decision to purchase the Securities covered by this Agreement and (ii) and in the case of a Purchaser whose investment adviser utilized an information barrier with respect to the information regarding the transactions contemplated hereunder after first being contacted by the Company or such other Person representing the Company, the representation set forth above shall only apply after the point in time when the portfolio manager who manages such Purchaser's assets was informed of the information regarding the transactions contemplated hereunder and, with respect to the Purchaser's investment adviser, the representation set forth above shall only apply with respect to any purchases or sales, including Short Sales, of the securities of the Company on behalf of other funds or investment vehicles for which the Purchaser's investment adviser is also an investment adviser or sub-adviser after the point in time when the portfolio manager who manages the assets of such other funds or investment vehicles for which the Purchaser's investment adviser is also an investment adviser or sub-adviser was informed of the information regarding the transactions contemplated hereunder. Other than to other Persons party to this Agreement and to the Purchaser's representatives or agents, including, but not limited to, the Purchaser's legal, tax and investment advisors, such Purchaser has maintained the confidentiality of all disclosures made to it in connection with this transaction (including the existence and terms of this transaction). Notwithstanding the foregoing, for avoidance of doubt, nothing contained herein shall constitute a representation or warranty, or preclude any actions, with respect to the identification of the availability of, or securing of, available shares to borrow in order to effect Short Sales or similar transactions in the future.

(i) **Brokers and Finders**. No Person will have, as a result of the transactions contemplated by this Agreement, any valid right, interest or claim against or upon the Purchaser for any commission, fee or other compensation pursuant to any agreement, arrangement or understanding entered into by or on behalf of the Purchaser. No Purchaser shall have any obligation with respect to any fees, or with respect to any claims made by or on behalf of other Persons for fees, in each case of the type contemplated by this Section 3.2(i) that may be due in connection with the transactions contemplated by this Agreement or the Transaction Documents.

(j) Independent Investment Decision. Such Purchaser has independently evaluated the merits of its decision to purchase Securities pursuant to the Transaction Documents, and such Purchaser confirms that it has not relied on the advice of any other Purchaser's business and/or legal counsel in making such decision. Such Purchaser understands that nothing in this Agreement or any other materials presented by or on behalf of the Company to the Purchaser in connection with the purchase of the Securities constitutes legal, tax or investment advice. Such Purchaser has consulted such legal, tax and investment advisors as it, in its sole discretion, has deemed necessary or appropriate in connection with its purchase of the Securities. Such Purchaser understands that the Placement Agent has acted solely as the agent of the Company in this placement of the Securities and such Purchaser has not relied on the business or legal advice of the Placement Agent or any of their agents, counsel or Affiliates in making its investment decision hereunder, and confirms that none of such Persons has made any representations or warranties to such Purchaser in connection with the transactions contemplated by the Transaction Documents.

(k) **Reliance on Exemptions**. Such Purchaser understands that the Securities are being offered and sold to it in reliance on specific exemptions from the registration requirements of United States federal and state securities laws and that the Company is relying in part upon the truth and accuracy of, and such Purchaser's compliance with, the representations, warranties, agreements, acknowledgements and understandings of such Purchaser set forth herein in order to determine the availability of such exemptions and the eligibility of such Purchaser to acquire the Securities.

(1) **No Governmental Review**. Such Purchaser understands that no United States federal or state agency or any other government or governmental agency has passed on or made any recommendation or endorsement of the Securities or the fairness or suitability of the investment in the Securities nor have such authorities passed upon or endorsed the merits of the offering of the Securities.

(m) **Regulation M.** Such Purchaser is aware that the anti-manipulation rules of Regulation M under the Exchange Act may apply to sales of Securities and other activities with respect to the Securities by the Purchasers.

(n) **Beneficial Ownership**. The purchase by such Purchaser of the Securities issuable to it at the Closing will not result in such Purchaser (individually or together with any other Person with whom such Purchaser has identified, or will have identified, itself as part of a "group" in a public filing made with the Commission involving the Company's securities) acquiring, or obtaining the right to acquire, beneficial ownership in excess of 19.999% of the outstanding shares of Common Stock or the voting power of the Company on a post transaction basis that assumes that such Closing shall have occurred. Such Purchaser does not presently intend to, alone or together with others, make a public filing with the Commission to disclose that it has (or that it together with such other Persons have) acquired, or obtained the right to acquire, as a result of such Closing when added to any other securities of the Company that it or they then own or have the right to acquire, beneficial ownership in excess of 19.999% of the outstanding shares of Common Stock or the voting power of the Company that it exceeded to any other securities of the to acquire, beneficial ownership in excess of 19.999% of the outstanding shares of Common Stock or the voting power of the Company on a post transaction basis that assumes that each Closing shall have occurred.

(o) **Residency**. Such Purchaser's residence (if an individual) or offices in which its investment decision with respect to the Securities was made (if an entity) are located at the address set forth under such Purchaser's name on its signature page hereto or as otherwise specified below its address on its signature page hereto.

The Company and each of the Purchasers acknowledge and agree that no party to this Agreement has made or makes any representations or warranties with respect to the transactions contemplated hereby other than those specifically set forth in this Article 3 and the Transaction Documents.

ARTICLE 4 OTHER AGREEMENTS OF THE PARTIES

4.1 Transfer Restrictions.

(a) **Compliance with Laws**. Notwithstanding any other provision of this Article 4, each Purchaser covenants that the Securities and Conversion Shares may be disposed of only pursuant to an effective registration statement under, and in compliance with the requirements of, the Securities Act, or pursuant to an available exemption from, or in a transaction not subject to, the registration requirements of the Securities Act, and in compliance with any applicable U.S. state and federal securities laws. In connection with any transfer of the Securities other than (i) pursuant to an effective registration statement, (ii) to the Company, (iii) pursuant to Rule 144 (provided that the Purchaser provides the Company with reasonable assurances (in the form of seller and, if applicable, broker representation letters) that the securities may be sold pursuant to such rule), or (iv) in connection with a bona fide pledge as contemplated in Section 4.1(b), the Company may require the transferor thereof to provide to the Company an opinion of counsel selected by the transferor and reasonably acceptable to the Company, the form and substance of which opinion shall be reasonably satisfactory to the Company, to the effect that such transfer does not require registration of such transferred Securities or Conversion Shares under the Securities Act and, as a condition of transfer, any such transferee shall agree in writing to be bound by the terms of this Agreement and the Registration Rights Agreement and shall have the rights of a Purchaser under this Agreement and the Registration Rights Agreement and the Registration Rights Agreement with respect to such transferred Securities or Conversion Shares.

(b) Legends. Certificates and book-entry statements evidencing the Securities and any Conversion Shares shall bear any legend as required by the "blue sky" Laws of any state and a restrictive legend in substantially the following form:

THE OFFER AND SALE OF THE SECURITIES REPRESENTED HEREBY TO WHICH THIS CONFIRMATION RELATES HAVE NOT BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933, AS AMENDED, OR THE SECURITIES LAWS OF ANY STATE OF THE UNITED STATES. THE SECURITIES MAY NOT BE SOLD, OFFERED FOR SALE, PLEDGED, HYPOTHECATED, TRANSFERRED OR ASSIGNED IN THE ABSENCE OF AN EFFECTIVE REGISTRATION STATEMENT FOR THE SECURITIES UNDER APPLICABLE SECURITIES LAWS, OR UNLESS OFFERED, SOLD, PLEDGED, HYPOTHECATED OR TRANSFERRED PURSUANT TO AN AVAILABLE EXEMPTION FROM THE REGISTRATION REQUIREMENTS OF THOSE LAWS. THE COMPANY AND ITS TRANSFER AGENT SHALL BE ENTITLED TO REQUIRE AN OPINION OF COUNSEL SATISFACTORY TO THE COMPANY AND THE TRANSFER AGENT THAT SUCH REGISTRATION IS NOT REQUIRED. NOTWITHSTANDING THE FOREGOING, THE SECURITIES MAY BE PLEDGED IN CONNECTION WITH A BONA FIDE MARGIN ACCOUNT OR OTHER LOAN OR FINANCING ARRANGEMENT SECURED BY THE SECURITIES.

The Company acknowledges and agrees that a Purchaser may from time to time pledge, and/or grant a security interest in, some or all of the legended Securities or Conversion Shares in connection with applicable securities laws, pursuant to a bona fide margin agreement in compliance with a bona fide margin loan. Such a pledge would not be subject to approval or consent of the Company and no legal opinion of legal counsel to the pledgee, secured party or pledgor shall be required in connection with the pledge, but such legal opinion shall be required in connection with a subsequent transfer or foreclosure following default by the Purchaser transfere of the pledge. No notice shall be required of such pledge, but Purchaser's transferee shall promptly notify the Company of any such subsequent transfer or foreclosure. Each Purchaser acknowledges that the Company shall not be responsible for any pledges relating to, or the grant of any security interest in, any of the Securities or Conversion Shares or for any agreement, understanding or arrangement between any Purchaser and its pledgee or secured party. At the appropriate Purchaser's expense, the Company will execute and deliver such reasonable documentation as a pledge or secured party of Securities or Conversion Shares may reasonably request in connection with a pledge or transfer of the Securities Act or other applicable provision of the Securities Act to appropriately amend the list of selling stockholders thereunder. Each Purchaser acknowledges and agrees that, except as otherwise provided in Section 4.1(c), any Securities or Conversion Shares subject to a pledge or security interest as contemplated by this Section 4.1(b) shall continue to bear the legend set forth in this Section 4.1(b) and be subject to the restrictions on transfer set forth in Section 4.1(b).

(c) **Removal of Legends**. Once a Registration Statement covering the resale of the Conversion Shares is declared effective, the Company shall remove all restrictive legends, including the legend set forth in Section 4.1(b) above (or, in the event that Conversion Shares are issued upon conversion after the Registration Statement is declared effective, the Conversion Shares shall be issued without restrictive legends). Further, the Company shall remove all restrictive legends, including the legend set forth in Section 4.1(b) above, (i) following any sale of such Securities or Conversion Shares are eligible for resale under Rule 144(b)(1) or any successor provision (or, in the event that Conversion Shares are issued upon conversion after the conditions set forth in clauses (i) and (ii) above, the Conversion Shares shall be issued without restrictive legends). Without limiting the foregoing, either (i) upon request of the Purchaser, upon receipt by the Company of an opinion of counsel reasonably satisfactory to the Company to the effect that such legend is no longer required under the Securities Act and applicable state securities laws or (ii) as contemplated by the Irrevocable Transfer Agent Instructions, the Company shall promptly cause the legend to be removed from any certificate for any Securities or Conversion Shares that are free from all restrictive and other legends or, at the request of such Purchaser, via DWAC transfer to such Purchaser's account.

(d) **Irrevocable Transfer Agent Instructions**. The Company shall issue the Irrevocable Transfer Agent Instructions. The Company represents and warrants that no instruction other than the Irrevocable Transfer Agent Instructions referred to in this Section 4.1(d) (or instructions that are consistent therewith) will be given by the Company to its Transfer Agent in connection with this Agreement, and that the Securities and Conversion Shares shall otherwise be freely transferable on the books and records of the Company as and to the extent provided in this Agreement and the other Transaction Documents and applicable law. The Company acknowledges that a breach by it of its obligations under this Section 4.1(d) will cause irreparable harm to a Purchaser. Accordingly, the Company acknowledges that the remedy at law for a breach of its obligations under this Section 4.1(d) may be inadequate and agrees, in the event of a breach or threatened breach by the Company of the provisions of this Section 4.1(d) that a Purchaser shall be entitled, in addition to all other available remedies, to an order and/or injunction restraining any breach and requiring immediate issuance and transfer, without the necessity of showing irreparable harm or economic loss and without any bond or other security being required.

(e) Acknowledgement. Each Purchaser hereunder acknowledges its primary responsibilities under the Securities Act and accordingly will not sell or otherwise transfer the Securities or Conversion Shares or any interest therein without complying with the requirements of the Securities Act.

4.2 **Furnishing of Information**. In order to enable the Purchasers to sell the Securities and Conversion Shares under Rule 144, until such time as Purchaser may sell the Securities and Conversion Shares without limitation under Rule 144, the Company shall use its commercially reasonable efforts to timely file (or obtain extensions in respect thereof and file within the applicable grace period) all reports required to be filed by the Company after the date hereof pursuant to the Exchange Act and, if during such period, the Company is not required to file reports pursuant to the Exchange Act, it will prepare and furnish to the Purchasers and make publicly available in accordance with Rule 144(c) such information as is required for the Purchasers to sell the Securities and Conversion Shares under Rule 144.

4.3 **Integration**. The Company shall not, and shall use its commercially reasonable efforts to ensure that no Affiliate of the Company shall, sell, offer for sale or solicit offers to buy or otherwise negotiate in respect of any security (as defined in Section 2 of the Securities Act) that will be integrated with the offer or sale of the Securities in a manner that would require the registration under the Securities Act of the sale of the Securities to the Purchasers, or that will be integrated with the offer or sale of the Securities for purposes of the rules and regulations of any Trading Market such that it would require stockholder approval prior to the closing of such other transaction unless stockholder approval is obtained before the closing of such subsequent transaction; *provided, however*, that this Section 4.3 shall not limit the Company's right to issue shares of capital stock pursuant to the Merger Agreement.

Securities Laws Disclosure; Publicity. By no later than 9:00 A.M., New York City time, on the Trading Day immediately following the 4.4 date hereof (provided that, if this Agreement is executed between midnight and 9:00 A.M., New York City time on any Trading Day, no later than 9:01 A.M. on the date hereof), the Company shall (a) issue a press release (the "Press Release") reasonably acceptable to the Placement Agent disclosing all material terms of the transactions contemplated hereby and (b) file a Current Report on Form 8-K with the Commission describing the terms of the Transaction Documents (and including as exhibits to such Current Report on Form 8-K the material Transaction Documents (including, without limitation, this Agreement and the Registration Rights Agreement); provided that the Press Release shall not publicly disclose the name of any Purchaser or investment adviser of any Purchaser, or include the name of any Purchaser or an Affiliate of any Purchaser without the prior written consent of such Purchaser. In addition, notwithstanding the foregoing, the Company shall not publicly disclose the name of any Purchaser or investment adviser of any Purchaser, or include the name of any Purchaser or an Affiliate of any Purchaser without the prior written consent of such Purchaser (i) in any press release or marketing materials or (ii) in any filing with the Commission or any regulatory agency or Trading Market, except as required by U.S. federal securities law (A) in connection with any registration statement contemplated by the Registration Rights Agreement (which shall be subject to review and comment of the Purchasers pursuant to the terms of the Registration Rights Agreement) or the filing of final Transaction Documents (including signature pages thereto) with the Commission and (B) to the extent such disclosure is required by law, request of the Commission's staff or Trading Market regulations, in which case the Company shall provide the Purchasers with prior written notice of such disclosure permitted under this subclause (ii). From and after the issuance of the Press Release, no Purchaser shall be in possession of any material, non-public information received from the Company, any Subsidiary or any of their respective officers, directors, employees or agents, that is not disclosed in the Press Release. Each Purchaser, severally and not jointly with the other Purchasers, covenants that until such time as the transactions contemplated by this Agreement are publicly disclosed by the Company pursuant to the Press Release, such Purchaser will maintain the confidentiality of the existence and terms of this transaction and the information provided in connection therewith; provided, however, that any disclosure may be made by the Purchaser to the Purchaser's representatives or agents, including, but not limited to, the Purchaser's legal, tax and investment advisors.

4.5 **Shareholder Rights Plan**. No claim will be made or enforced by the Company or, with the consent of the Company, any other Person, that any Purchaser is an "*Acquiring Person*" under any control share acquisition, business combination, poison pill (including any distribution under a rights agreement) or similar anti-takeover plan or arrangement or Law (including Section 203 of the Delaware General Corporation Law) in effect or hereafter adopted by the Company, or that any Purchaser could be deemed to trigger the provisions of any such plan or arrangement, in either case solely by virtue of receiving Securities or Conversion Shares under the Transaction Documents.

4.6 **Non-Public Information**. Except with respect to the material terms and conditions of the transactions contemplated by the Transaction Documents, including this Agreement, or as expressly required by any applicable securities law, the Company covenants and agrees that neither it, nor any other Person acting on its behalf, will provide any Purchaser or its agents or counsel with any information regarding the Company that the Company believes constitutes material non-public information without the express written (email being sufficient) consent of such Purchaser, (i) unless prior thereto such Purchaser shall have committed to customary obligations regarding the confidentiality and use of such information and (ii) except in the case of material, nonpublic information provided to an observer of the Company's or Trawsfynydd's board of directors or member of the Company's or Trawsfynydd's board of directors who is affiliated with such Purchaser. The Company understands and confirms that each Purchaser shall be relying on the foregoing covenant in effecting transactions in securities of the Company.

4.7 Use of Proceeds. The Company shall use the net proceeds from the sale of the Securities hereunder for working capital and general corporate purposes.

4.8 **Principal Trading Market Listing**. The Company shall use its reasonable best efforts to take all steps necessary to cause the Conversion Shares to be approved for listing on the Principal Trading Market as promptly as possible.

4.9 **Form D; Blue Sky**. The Company agrees to timely file a Form D with respect to the Securities as required under Regulation D and to provide a copy thereof, promptly upon the written request of any Purchaser. The Company, on or before the Closing Date, shall take such action as the Company shall reasonably determine is necessary in order to obtain an exemption for or to qualify the Securities for sale to the Purchasers under applicable securities or "blue sky" Laws of the states of the United States (or to obtain an exemption from such qualification) and shall provide evidence of such actions promptly upon the written request of any Purchaser.

4.10 **Short Sales After the Date Hereof**. Such Purchaser shall not engage, directly or indirectly, in any transactions in the Company's securities (including, without limitation, any Short Sales involving the Company's securities) during the period from the date hereof until the earlier of such time as (i) the transactions contemplated by this Agreement are first publicly announced as required by and described in Section 4.4 or (ii) this Agreement is terminated in full pursuant to Section 6.17.

Notwithstanding the foregoing, no Purchaser makes any representation, warranty or covenant hereby that it will not engage in Short Sales in the securities of the Company after the time that the transactions contemplated by this Agreement are first publicly announced as described in Section 4.4.

4.11 Beneficial Ownership Limitation. Notwithstanding anything to the contrary set forth in the Certificate of Designation, the Company shall not effect any conversion of any share of Series C Preferred Stock, and a Purchaser shall not have the right to convert any portion of its Series C Preferred Stock, to the extent that, after giving effect to such attempted conversion set forth on an applicable Notice of Conversion (as defined in the Certificate of Designation) with respect to the Series C Preferred Stock, such Purchaser (or any of such Purchaser's Affiliates or any other Person who would be a beneficial owner of Common Stock beneficially owned by the Purchaser for purposes of Section 13(d) or Section 16 of the Exchange Act and the applicable rules and regulations of the Commission, including any "group" of which the Purchaser is a member (the foregoing, "Attribution Parties")) would beneficially own a number of shares of Common Stock in excess of the Beneficial Ownership Limitation (as defined below). For purposes of the foregoing sentence, the aggregate number of shares of Common Stock beneficially owned by such Purchaser and its Attribution Parties shall include the number of shares of Common Stock issuable upon conversion of the Series C Preferred Stock subject to the Notice of Conversion or the Automatic Conversion (as defined in the Certificate of Designation), as applicable, with respect to which such determination is being made, but shall exclude the number of shares of Common Stock which are issuable upon (i) conversion of the remaining, unconverted Series C Preferred Stock beneficially owned by such Purchaser or any of its Attribution Parties, and (ii) exercise or conversion of the unexercised or unconverted portion of any other securities of the Company beneficially owned by such Purchaser or any of its Attribution Parties that are subject to and would exceed a limitation on conversion or exercise similar to the limitation contained herein. Except as set forth in the preceding sentence, for purposes of this Section 4.11, beneficial ownership shall be calculated in accordance with Section 13(d) of the Exchange Act and the applicable rules and regulations of the Commission, and the terms "beneficial ownership" and "beneficially own" have the meanings ascribed to such terms therein. In addition, for purposes hereof, "group" has the meaning set forth in Section 13(d) of the Exchange Act and the applicable rules and regulations of the Commission. For purposes of this Section 4.11, in determining the number of outstanding shares of Common Stock, a Purchaser may rely on the number of outstanding shares of Common Stock as stated in the most recent of the following: (i) the Company's most recent periodic or annual filing with the Commission, as the case may be, (ii) a more recent public announcement by the Company that is filed with the Commission, or (iii) a more recent notice by the Company or the Transfer Agent to the Purchaser setting forth the number of shares of Common Stock then outstanding. For any reason at any time, upon the written request of a Purchaser (which may be by e-mail), the Company shall, within two (2) Trading Days of such request, confirm in writing to such Purchaser (which may be by e-mail) the number of shares of Common Stock then outstanding. In any case, the number of outstanding shares of Common Stock shall be determined after giving effect to any actual conversion or exercise of securities of the Company, including Series C Preferred Stock, by such Purchaser or its Attribution Parties since the date as of which such number of outstanding shares of Common Stock was last publicly reported or confirmed to the Purchaser. The "Beneficial Ownership Limitation" shall initially be set at the discretion of each Purchaser to a percentage between 0% and 19.9% of the number of shares of the Common Stock outstanding or deemed to be outstanding as of the applicable measurement date, and such percentage shall be set at 19.9% for any Purchaser that does not make such designation on the signature page hereto. The Company shall be entitled to rely on representations made to it by any Purchaser in any Notice of Conversion regarding its Beneficial Ownership Limitation. Notwithstanding the foregoing, by written notice to the Company, (i) any Purchaser may reset the Beneficial Ownership Limitation percentage to a higher percentage, not to exceed 19.9%, which increase will not be effective until the sixty-first (61st) day after such written notice is delivered to the Company, and (ii) any Purchaser may reset the Beneficial Ownership Limitation percentage to a lower percentage provided that such decrease shall not become effective until the later of (x) 5:00 p.m. Eastern time on the third Business Day after the date of the Requisite Stockholder Approval, and if Requisite Stockholder Approval is not obtained within six months after the initial issuance of the Series C Preferred Stock, the date that is three Business Days after the date that is six months after the initial issuance of the Series C Preferred Stock. Upon such a change by a Purchaser of the Beneficial Ownership Limitation, not to exceed 19.9%, the Beneficial Ownership Limitation may not be further amended by such Purchaser without first providing the minimum notice required by this Section 4.11. Notwithstanding the foregoing, at any time following notice of a Fundamental Transaction (as defined in the Certificate of Designation), the Purchaser may waive and/or change the Beneficial Ownership Limitation effective immediately upon written notice to the Company and may reinstitute a Beneficial Ownership Limitation at any time thereafter effective immediately upon written notice to the Company. The provisions of this Section 4.11 shall be construed, corrected and implemented in a manner so as to effectuate the intended Beneficial Ownership Limitation herein contained and the shares of Common Stock underlying the Securities in excess of the Beneficial Ownership Limitation shall not be deemed to be beneficially owned by the Purchaser for any purpose including for purposes of Section 13(d) or Rule 16a-1(a)(1) of the Exchange Act.

4.12 **Requisite Stockholder Approval.** The Company shall hold a special meeting of stockholders (a "*Stockholder Meeting*") within 75 days from the Closing (the "*Stockholder Meeting Deadline*") for the purpose of obtaining stockholder approval of the conversion of all issued and outstanding Series C Preferred Stock into shares of Common Stock in accordance with the Nasdaq Stock Market Rules (the "*Requisite Stockholder Approval*"). The Company shall use its best efforts to solicit its stockholders' approval of such resolution and to cause the Board of Directors to recommend to the stockholders that they approve such resolution. If the Requisite Stockholder Approval is not obtained on or prior to the Stockholder *Approval Period*"). If the Requisite Stockholder Meeting to be held within 90 days from the prior meeting (the "*Extended Stockholder Approval Period*"). If the Requisite Stockholder Approval is not obtained within the Extended Stockholder Approval Period, then the Company shall convene additional stockholder meetings every 90 days thereafter until the Requisite Stockholder Approval is obtained. The Company shall enforce the terms of each Support Agreement, and shall not amend or waive any provision of any Support Agreement.

4.13 **Conversion and Exercise Procedures**. The form of Notice of Conversion included in the Certificate of Designation sets forth the totality of the procedures required of the Purchasers in order to convert the Securities. Without limiting the preceding sentence, no ink-original Notice of Conversion shall be required, nor shall any medallion guarantee (or other type of guarantee or notarization) of any Notice of Conversion form be required in order for the registered holder thereof to convert the Securities. No additional legal opinion, other information or instructions shall be required of a Purchaser to convert its Securities. The Company shall honor conversions of the Securities and shall deliver Conversion Shares in accordance with the terms, conditions and time periods set forth in the Transaction Documents.

4.14 **Lock-Up Agreements**. The Company shall not consent or agree to amend, alter, waive or otherwise modify the terms of any of the Lock-Up Agreements (as defined in the Merger Agreement) without the consent of the Placement Agent.

4.15 Indemnification of Purchasers. Subject to the provisions of this Section 4.15, the Company will indemnify and hold each Purchaser and its directors, officers, shareholders, members, partners, employees, investment advisers and agents (and any other Persons with a functionally equivalent role of a Person holding such titles notwithstanding a lack of such title or any other title), each Person who controls such Purchaser (within the meaning of Section 15 of the Securities Act and Section 20 of the Exchange Act), and the directors, officers, shareholders, agents, members, partners, investment advisers or employees (and any other Persons with a functionally equivalent role of a Person holding such titles notwithstanding a lack of such title or any other title) of such controlling persons (each, a "Purchaser Party") harmless from any and all losses, liabilities, obligations, claims, contingencies, damages, costs and expenses, including all judgments, amounts paid in settlements, court costs and reasonable attorneys' fees and costs of investigation that any such Purchaser Party may suffer or incur as a result of or relating to (i) any breach of any of the representations, warranties, covenants or agreements made by the Company in this Agreement or in the other Transaction Documents or (ii) any Action instituted against a Purchaser in any capacity, or any Purchaser Party, by any stockholder of the Company who is not an Affiliate of such Purchaser seeking indemnification, with respect to any of the transactions contemplated by the Transaction Documents (unless such Action is based upon a breach of such Purchaser's representations, warranties or covenants under the Transaction Documents). Promptly after receipt by any such Person (the "Indemnified Person") of notice of any demand, claim or circumstances that would or may give rise to a claim or the commencement of any Proceeding or investigation in respect of which indemnity may be sought pursuant to this Section 4.15, such Indemnified Person shall promptly notify the Company in writing and the Company shall assume the defense thereof, including the employment of counsel reasonably satisfactory to such Indemnified Person, and shall assume the payment of all fees and expenses relating to such Proceeding or investigation; provided, however, that the failure of any Indemnified Person so to notify the Company shall not relieve the Company of its obligations hereunder except to the extent that the Company is actually and materially prejudiced by such failure to notify. In any such proceeding, any Indemnified Person shall have the right to retain its own counsel, but the fees and expenses of such counsel shall be at the expense of such Indemnified Person unless: (i) the Company and the Indemnified Person shall have mutually agreed to the retention of such counsel; (ii) the Company shall have failed promptly to assume the defense of such proceeding and to employ counsel reasonably satisfactory to such Indemnified Person in such proceeding; or (iii) in the reasonable judgment of counsel to such Indemnified Person, representation of both parties by the same counsel would be inappropriate due to actual or potential differing interests between them. In the event of the circumstances described in the foregoing clause (iii), if the Indemnified Person notifies the Company in writing that such Indemnified Person elects to employ separate counsel at the expense of the Company, then the Company shall not have the right to assume the defense of such claim on behalf of such Indemnified Person. The Company shall not be liable for any settlement of any proceeding effected without its prior written consent, which consent shall not be unreasonably withheld, delayed or conditioned or to the extent fees or costs incurred pursuant to this Section 4.15 are attributable to the Indemnified Person's breach of any of the representations, warranties, covenants or agreements made by the Purchasers in this Agreement or the other Transaction Documents. Without the prior written consent of the Indemnified Person, which consent shall not be unreasonably withheld, delayed or conditioned, the Company shall not effect any settlement of any pending or threatened proceeding in respect of which any Indemnified Person is or could have been a party and indemnity could have been sought hereunder by such Indemnified Person, unless such settlement includes an unconditional release of such Indemnified Person from all liability arising out of such proceeding.

4.16 **Equal Treatment of Purchasers**. No consideration shall be offered or paid to any Purchaser to amend or consent to a waiver or modification of any provision of this Agreement unless the same consideration is also offered to all of the Purchasers. For clarification purposes, this provision constitutes a separate right granted to each Purchaser by the Company and negotiated separately by each Purchaser and shall not in any way be construed as the Purchasers acting in concert or as a group with respect to the purchase, disposition or voting of shares of Common Stock or otherwise.

ARTICLE 5 CONDITIONS PRECEDENT TO CLOSING

5.1 **Conditions Precedent to the Obligations of the Purchasers to Purchase Securities**. The obligation of each Purchaser to acquire Securities at the Closing is subject to the fulfillment to such Purchaser's satisfaction, on or prior to the Closing Date, of each of the following conditions, any of which may be waived by such Purchaser (as to itself only):

(a) **Representations and Warranties**. The representations and warranties of the Company contained herein shall be true and correct in all respects (as of the date when made and as of the Closing Date, as though made on and as of such date, except for such representations and warranties that speak as of a specific date, which shall be true and correct in all respects as of such date.

(b) **Performance**. The Company shall have performed, satisfied and complied in all material respects with all covenants, agreements and conditions required by the Transaction Documents to be performed, satisfied or complied with by it at or prior to the Closing.

(c) **No Injunction**. No statute, rule, regulation, order, executive order, decree, judgment, writ, order, ruling or injunction shall have been enacted, entered, promulgated, issued or endorsed by any court of competent jurisdiction or any Governmental Authority that enjoins, prevents or prohibits the consummation of any of the transactions contemplated by the Transaction Documents.

(d) **Consents**. The Company shall have obtained in a timely fashion any and all consents, permits, approvals, registrations and waivers necessary for consummation of the purchase and sale of the Securities (except for the Requisite Stockholder Approval), all of which shall be and remain so long as necessary in full force and effect.

(e) Adverse Changes. Since the date of execution of this Agreement, no event or series of events shall have occurred that has had or would reasonably be expected to have a Material Adverse Effect.

(f) **No Suspensions of Trading in Common Stock; No Stop Orders; Listing**. The Common Stock shall not have been suspended by the Commission or the Principal Trading Market from trading on the Principal Trading Market nor, except as previously disclosed in the SEC Reports, shall suspension by the Commission or the Principal Trading Market have been threatened, either (i) in writing by the Commission or the Principal Trading Market or (ii) by falling below the minimum listing maintenance requirements of the Principal Trading Market. No stop order shall have been imposed by Principal Trading Market, the Commission or any other Governmental Authority or regulatory body with respect to public trading in the Common Stock. Nasdaq shall have raised no objection to the consummation of the transactions contemplated by the Transaction Documents or the Merger Agreement.

(g) Company Deliverables. The Company shall have delivered the Company Deliverables in accordance with Section 2.3(a).

(h) **Compliance Certificate**. The Company shall have delivered to each Purchaser a certificate, dated as of the Closing Date and signed by its Chief Executive Officer and its Chief Financial Officer, dated as of the Closing Date, certifying to the fulfillment of the conditions specified in Sections 5.1(a), (b), (e) (f) and (i) in the form attached hereto as **Exhibit D**.

- (i) Merger. The Merger shall have been consummated in accordance with the Merger Agreement.
- (j) Termination. This Agreement shall not have been terminated as to such Purchaser in accordance with Section 6.17 herein.

5.2 **Conditions Precedent to the Obligations of the Company to issue Securities**. The Company's obligation to issue the Securities at the Closing to each Purchaser is subject to the fulfillment to the satisfaction of the Company on or prior to the Closing Date of the following conditions, any of which may be waived by the Company:

(a) **Representations and Warranties**. The representations and warranties made by each Purchaser in Section 3.2 hereof shall be true and correct in all material respects (except for those representations and warranties which are qualified as to materiality, in which case such representations and warranties shall be true and correct in all respects) as of the date when made, and as of the Closing Date as though made on and as of such date, except for representations and warranties that speak as of a specific date, which shall be true and correct in all material respects (except for those representations and warranties which are qualified as to materiality or Material Adverse Effect, which representations and warranties shall be true and correct in all respects) as of such date.

(b) **Performance**. Such Purchaser shall have performed, satisfied and complied in all material respects with all covenants, agreements and conditions required by the Transaction Documents to be performed, satisfied or complied with by such Purchaser at or prior to the Closing Date.

(c) **No Injunction**. No statute, rule, regulation, order, executive order, decree, judgment, writ, order, ruling or injunction shall have been enacted, entered, promulgated, issued or endorsed by any court of competent jurisdiction or any Governmental Authority that enjoins, prevents or prohibits the consummation of any of the transactions contemplated by the Transaction Documents.

(d) Purchasers Deliverables. Such Purchaser shall have delivered its Purchaser Deliverables in accordance with Section 2.3(b).

(e) **Merger**. The Merger shall have been consummated in accordance with the Merger Agreement (which shall not have been amended in any manner that is materially and adversely affects the Purchasers).

(f) Termination. This Agreement shall not have been terminated as to such Purchaser in accordance with Section 6.17 herein.

ARTICLE 6 MISCELLANEOUS

6.1 **Fees and Expenses.** The Company and the Purchasers shall each pay the fees and expenses of their respective advisers, counsel, accountants and other experts, if any, and all other expenses incurred by such party in connection with the negotiation, preparation, execution, delivery and performance of this Agreement. The Company shall pay all Transfer Agent fees, stamp taxes and other taxes and duties levied in connection with the issuance and sale of the Securities to the Purchasers.

6.2 **Entire Agreement**. The Transaction Documents, together with the exhibits and schedules thereto, contain the entire understanding of the parties with respect to the subject matter hereof and thereof and supersede all prior agreements, understandings, discussions and representations, oral or written, with respect to such matters, which the parties acknowledge have been merged into such documents, exhibits and schedules. Before or at the Closing, the Company and the Purchasers will execute and deliver to the other such further documents as may be reasonably requested in order to give practical effect to the intention of the parties under the Transaction Documents.

6.3 **Notices**. Any and all notices or other communications or deliveries required or permitted to be provided hereunder shall be in writing and shall be deemed given and effective on the earliest of (a) the date of transmission, if such notice or communication is delivered via electronic mail at the e-mail address specified in this Section 6.3 prior to 5:00 P.M., New York City time, on a Trading Day, (b) the next Trading Day after the date of transmission, if such notice or communication is delivered via electronic mail at the e-mail address or facsimile number specified in this Section 6.3 on a day that is not a Trading Day or later than 5:00 P.M., New York City time, on any Trading Day, (c) the Trading Day following the date of mailing, if sent by U.S. nationally recognized overnight courier service with next day delivery specified, or (d) upon actual receipt by the party to whom such notice is required to be given if delivered personally or if sent by U.S. certified or registered mail, return receipt requested; provided, in the case of clauses (a) and (b), that notice shall not be deemed given or effective if the sender receives an automatic system-generated response that such electronic mail was undeliverable. The address for such notices and communications shall be as follows:

If to the Company:

Onconova Therapeutics, Inc. 12 Penns Trail Newtown, PA 18940 Attention: Steven M. Fruchtman; Mark Guerin Email Address: sfruchtman@onconova.us; mguerin@onconova.us

with a copy to (which shall not constitute notice):

Orrick, Herrington & Sutcliffe LLP Columbia Center 1152 15th Street, N.W. Washington, D.C. 20005-1706 United States Attention: David Schulman Email: dschulman@orrick.com

Snell & Wilmer L.L.P. Executive Center 12230 El Camino Real, Suite 300 San Diego, CA 92130 United States Attention: Bardia Moayedi Email: bmoayedi@swlaw.com

If to a Purchaser:

To the address set forth under such Purchaser's name on its signature page hereto; or such other address as may be designated in writing hereafter, in the same manner, by such Person.

6.4 **Amendments; Waivers; No Additional Consideration**. No provision of this Agreement may be waived, modified, supplemented or amended except in a written instrument signed, in the case of an amendment, by the Company and the Purchasers of at least a majority in interest of the Securities still held by Purchasers, provided that (i) no amendment to Section 4.4, Section 4.6, Section 4.10, Section 4.11 Section 4.12, Section 4.15, Section 4.16, this Section 6.4 or Section 6.17 may be made without the consent of each Purchaser, or, in the case of a waiver, by the party against whom enforcement of any such waiver is sought and (ii) any proposed amendment that would, by its terms, have a disproportionate and materially adverse effect on any Purchaser shall require the consent of such Purchaser(s). No waiver of any default with respect to any provision, condition or requirement of this Agreement shall be deemed to be a continuing waiver in the future or a waiver of any subsequent default or a waiver of any other provision, condition or requirement hereof, nor shall any delay or omission of either party to exercise any right hereunder in any manner impair the exercise of any such right. No consideration shall be offered or paid to any Purchaser to amend or consent to a waiver or modification of any provision of any Transaction Document unless the same consideration is also offered to all Purchasers who then hold Securities.

6.5 **Construction**. The headings herein are for convenience only, do not constitute a part of this Agreement and shall not be deemed to limit or affect any of the provisions hereof. The language used in this Agreement will be deemed to be the language chosen by the parties to express their mutual intent, and no rules of strict construction will be applied against any party. This Agreement shall be construed as if drafted jointly by the parties, and no presumption or burden of proof shall arise favoring or disfavoring any party by virtue of the authorship of any provisions of this Agreement or any of the Transaction Documents. As used in this Agreement, the words "include" and "including," and variations thereof, shall not be deemed to be terms of limitation, but rather shall be deemed to be followed by the words "without limitation."

6.6 **Successors and Assigns**. The provisions of this Agreement shall inure to the benefit of and be binding upon the parties and their successors and permitted assigns. This Agreement, or any rights or obligations hereunder, may not be assigned by the Company without the prior written consent of each Purchaser. Any Purchaser may assign its rights hereunder in whole or in part to any Person to whom such Purchaser assigns or transfers any Securities in compliance with the Transaction Documents and applicable law, provided such transferee shall agree in writing to be bound, with respect to the transferred Securities, by the terms and conditions of this Agreement that apply to the Purchasers.

6.7 **No Third-Party Beneficiaries**. This Agreement is intended for the benefit of the parties hereto and their respective successors and permitted assigns and is not for the benefit of, nor may any provision hereof be enforced by, any other Person, except (a) as set forth in Section 4.15 and (b) the Placement Agent is an intended third-party beneficiary of the representations and warranties in Article 3 and Article 4, and of this Section 6.7 and Section 6.18.

6.8 **Governing Law**. All questions concerning the construction, validity, enforcement and interpretation of this Agreement shall be governed by and construed and enforced in accordance with the internal laws of the State of Delaware, without regard to the principles of conflicts of law thereof. Each party agrees that all Proceedings concerning the interpretations, enforcement and defense of the transactions contemplated by this Agreement and any other Transaction Documents (whether brought against a party hereto or its respective Affiliates, employees or agents) shall be commenced exclusively in the Delaware Courts. Each party hereto hereby irrevocably submits to the exclusive jurisdiction of the Delaware Courts for the adjudication of any dispute hereunder or in connection herewith or with any transaction contemplated hereby or discussed herein (including with respect to the enforcement of any of the Transaction Documents), and hereby irrevocably waives, and agrees not to assert in any Proceeding, any claim that it is not personally subject to the jurisdiction of any such Delaware Court, or that such Proceeding has been commenced in an improper or inconvenient forum. Each party hereto hereby irrevocably waives personal service of process and consents to process being served in any such Proceeding by mailing a copy thereof via registered or certified mail or overnight delivery (with evidence of delivery) to such party at the address in effect for notices to it under this Agreement and agrees that such service shall constitute good and sufficient service of process and notice thereof. Nothing contained herein shall be deemed to limit in any way any right to serve process in any manner permitted by law.

EACH PARTY HERETO HEREBY IRREVOCABLY WAIVES, TO THE FULLEST EXTENT PERMITTED BY APPLICABLE LAW, ANY AND ALL RIGHT TO TRIAL BY JURY IN ANY LEGAL PROCEEDING ARISING OUT OF OR RELATING TO THIS AGREEMENT OR THE TRANSACTIONS CONTEMPLATED HEREBY.

6.9 **Survival**. Subject to applicable statute of limitations, the representations, warranties, agreements and covenants contained herein shall survive the Closing and the delivery of the Securities.

6.10 **Execution**. This Agreement may be executed in two or more counterparts, all of which when taken together shall be considered one and the same agreement and shall become effective when counterparts have been signed by each party and delivered to the other party, it being understood that both parties need not sign the same counterpart. In the event that any signature is delivered by facsimile transmission, or by e-mail delivery of a ".pdf" format data file, or by any electronic signature complying with the U.S. ESIGN Act of 2000, such signature shall create a legally valid and binding obligation of the party executing (or on whose behalf such signature is executed) with the same force and effect as if such facsimile or ".pdf" signature page were an original thereof.

6.11 **Severability**. If any provision of this Agreement is held to be invalid or unenforceable in any respect, the validity and enforceability of the remaining terms and provisions of this Agreement shall not in any way be affected or impaired thereby and the parties will attempt to agree upon a valid and enforceable provision that is a reasonable substitute therefor, and upon so agreeing, shall incorporate such substitute provision in this Agreement.

6.12 **Rescission and Withdrawal Right**. Notwithstanding anything to the contrary contained in (and without limiting any similar provisions of) the Transaction Documents, whenever any Purchaser exercises a right, election, demand or option under a Transaction Document and the Company does not timely perform its related obligations within the periods therein provided, then such Purchaser may rescind or withdraw, in its sole discretion from time to time upon written notice to the Company, any relevant notice, demand or election in whole or in part without prejudice to its future actions and rights.

6.13 **Replacement of Securities**. If any certificate or instrument evidencing any Securities or Conversion Shares is mutilated, lost, stolen or destroyed, the Company may issue or cause to be issued in exchange and substitution for and upon cancellation thereof, or in lieu of and substitution therefor, a new certificate or instrument, but only upon receipt of evidence reasonably satisfactory to the Company and the Transfer Agent of such loss, theft or destruction and the execution by the holder thereof of a customary lost certificate affidavit of that fact and an agreement to indemnify and hold harmless the Company and the Transfer Agent for any losses in connection therewith or, if required by the Transfer Agent, a bond in such form and amount as is required by the Transfer Agent. The applicants for a new certificate or instrument under such circumstances shall also pay any reasonable third-party costs associated with the issuance of such replacement Securities or Conversion Shares. If a replacement certificate or instrument as a condition precedent to any issuance of a replacement.

6.14 **Remedies**. In addition to being entitled to exercise all rights provided herein or granted by Law, including recovery of damages, each of the Purchasers and the Company will be entitled to specific performance under the Transaction Documents, without the requirement of posting a bond. The parties agree that monetary damages may not be adequate compensation for any loss incurred by reason of any breach of obligations described in the foregoing sentence and hereby agree to waive in any action for specific performance of any such obligation (other than in connection with any action for a temporary restraining order) the defense that a remedy at law would be adequate.

6.15 **Payment Set Aside**. To the extent that the Company makes a payment or payments to any Purchaser pursuant to any Transaction Document or a Purchaser enforces or exercises its rights thereunder, and such payment or payments or the proceeds of such enforcement or exercise or any part thereof are subsequently invalidated, declared to be fraudulent or preferential, set aside, recovered from, disgorged by or are required to be refunded, repaid or otherwise restored to the Company, a trustee, receiver or any other person under any Law (including, without limitation, any bankruptcy law, state or federal law, common law or equitable cause of action), then to the extent of any such restoration the obligation or part thereof originally intended to be satisfied shall be revived and continued in full force and effect as if such payment had not been made or such enforcement or setoff had not occurred.

6.16 Adjustments in Share Numbers and Prices. In the event of any stock split, subdivision, dividend or distribution payable in shares of Common Stock (or other securities or rights convertible into, or entitling the holder thereof to receive directly or indirectly shares of Common Stock), combination or other similar recapitalization or event occurring after the date hereof and prior to the Closing, each reference in any Transaction Document to a number of shares or a price per share shall be deemed to be amended to appropriately account for such event.

6.17 **Independent Nature of Purchasers' Obligations and Rights**. The obligations of each Purchaser under any Transaction Document are several and not joint with the obligations of any other Purchaser, and no Purchaser shall be responsible in any way for the performance of the obligations of any other Purchaser under any Transaction Document. The decision of each Purchaser to purchase Securities pursuant to the Transaction Documents has been made by such Purchaser independently of any other Purchaser and independently of any information, materials, statements or opinions as to the business, affairs, operations, assets, properties, liabilities, results of operations, condition (financial or otherwise) or prospects of the Company or any Subsidiary which may have been made or given by any other Purchaser or by any agent or employee of any other Purchaser, and no Purchaser and any of its agents or employees shall have any liability to any other Purchaser (or any other Person) relating to or arising from any such information, materials, statement or opinions. Nothing contained herein or in any Transaction Document, and no action taken by any Purchaser pursuant thereto, shall be deemed to constitute the Purchasers as a partnership, an association, a joint venture or any other kind of entity, or create a presumption that the Purchasers are in any way acting in concert or as a group (including, without limitation, a "group" within the meaning of Section 13(d)(3) of the Exchange Act) with respect to such obligations or the transactions contemplated by the Transaction Documents. Each Purchaser acknowledges that no other Purchaser has acted as agent for such Purchaser in connection with making its investment hereunder and that no Purchaser will be acting as agent of such Purchaser in connection with monitoring its investment in the Securities or enforcing its rights under the Transaction Documents. Each Purchaser shall be entitled to independently protect and enforce its rights, including without limitation the rights arising out of this Agreement or out of the other Transaction Documents, and it shall not be necessary for any other Purchaser to be joined as an additional party in any proceeding for such purpose. Each Purchaser has been represented by its own separate legal counsel in its review and negotiation of the Transaction Documents. The Company has elected to provide all Purchasers with the same terms and Transaction Documents for the convenience of the Company and not because it was required or requested to do so by any Purchaser. It is expressly understood that each provision contained in this Agreement is between the Company and a Purchaser, solely, and not between the Company and the Purchasers collectively and not between and among the Purchasers.

6.18 **Termination**. This Agreement may be terminated and the sale and purchase of the Securities abandoned at any time prior to the Closing (i) automatically if the Closing has not been consummated on or prior to 5:00 P.M., New York City time, on Outside Date, (ii) by a Purchaser (with respect to itself) if any of the conditions set forth in Section 5.1 shall have become incapable of fulfillment, and shall not have been waived by such Purchaser, or (iii) automatically if the Merger Agreement is terminated in accordance with its terms; *provided, however*, that the right to terminate this Agreement under clause (ii) shall not be available to any Person whose failure to comply with its obligations under this Agreement has been the cause of or resulted in the failure of the Closing to occur on or before such time. Nothing in this Section 6.17 shall be deemed to release any party from any liability for any breach by such party of the terms and provisions of this Agreement or the other Transaction Documents or to impair the right of any party to compel specific performance by any other party of its obligations under this Agreement or the other Transaction Documents. In the event of a termination pursuant to this Section 6.17, the Company shall promptly notify all non-terminating Purchasers and if a Purchaser has delivered the Subscription Amount prior to such termination, the Company shall promptly (but not later than one (1) Business Day after such termination) return the Subscription Amount to such Purchaser by wire transfer of United States dollars in immediately available funds to the account specified by such Purchaser. Upon a termination in accordance with this Section 6.17, the Company and the terminating Purchaser(s) shall not have any further obligation or liability (including arising from such termination) to the other, and no Purchaser will have any liability to any other Purchaser under the Transaction Documents as a result therefrom.

6.19 **Exculpation of the Placement Agent**. Each party hereto agrees for the express benefit of the Placement Agent, their Affiliates and their representatives that:

(a) The Placement Agent is acting as placement agent for the Company solely in connection with the sale of the Securities and is not acting in any other capacity and is not and shall not be construed as a fiduciary for any Purchaser, or any other person or entity in connection with the sale of Securities.

(b) Neither the Placement Agent nor any of its Affiliates or any of its representatives (i) shall be liable for any improper payment made in accordance with the information provided by the Company; (ii) has made or will make any representation or warranty, express or implied, of any kind or character, and has not provided any recommendation in connection with the purchase or sale of the Securities; (iii) has any responsibilities as to the validity, accuracy, completeness, value or genuineness, as of any date, of any information, certificates or documentation delivered by or on behalf of the Company pursuant to this Agreement, the other Transaction Documents or the Merger Agreement, or in connection with any of the transactions contemplated by such agreements; (iv) shall be liable or have any obligation (including, without limitation, for or with respect to any losses, claims, damages, obligations, penalties, judgments, awards, liabilities, costs, expenses or disbursements incurred by any Purchaser, the Company or any other Person or entity), whether in contract, tort or otherwise to any Purchaser or to any person claiming through such Purchaser, (A) for any action taken, suffered or omitted by any of them in good faith and reasonably believed to be authorized or within the discretion or rights or powers conferred upon it by this Agreement, any other Transaction Document or the Merger Agreement, or (C) for anything otherwise in connection with the purchase and sale of the Securities or the issuance of the Conversion Shares, except in each case for such party's own gross negligence or willful misconduct.

6.20 *Arm's Length Transaction*. The Company acknowledges and agrees that (a) the transactions described in this Agreement are an arm's-length commercial transaction between the parties, (b) the Purchasers have not assumed nor will they assume an advisory or fiduciary responsibility in the Company's favor with respect to any of the transactions contemplated by this Agreement or the process leading thereto, and the Purchasers have no obligation to the Company with respect to the transactions contemplated by this Agreement except those obligations expressly set forth in this Agreement or the other Transaction Documents to which they are a party, and (c) the Company's decision to enter into the Transaction Documents and the Merger Agreement has been based solely on the independent evaluation by the Company and its representatives.

[REMAINDER OF PAGE INTENTIONALLY LEFT BLANK]

IN WITNESS WHEREOF, the parties hereto have caused this Securities Purchase Agreement to be duly executed by their respective authorized signatories as of the date first indicated above.

ONCONOVA THERAPEUTICS, INC.

By: /s/ Steven M. Fruchtman

Name:Steven M. FruchtmanTitle:President and CEO

IN WITNESS WHEREOF, the parties hereto have caused this Securities Purchase Agreement to be duly executed by their respective authorized signatories as of the date first indicated above.

PURCHASER:

ORBIMED PRIVATE INVESTMENTS VIII, LP

By: OrbiMed Capital GP VIII LLC, its General Partner

By: OrbiMed Advisors LLC, its Managing Member

By: <u>/s/ Carl Gordon</u> Name: Carl Gordon Title: Member

Beneficial Ownership Limitation: 9.99%

IN WITNESS WHEREOF, the parties hereto have caused this Securities Purchase Agreement to be duly executed by their respective authorized signatories as of the date first indicated above.

TPAV, LLC

By: /s/ Nikolay Savchuk

Name: Nikolay Savchuk

Title: Managing Member

<u>Annex A</u>

Purchaser	Aggregate Purchase Price (Subscription Amount)	Number of Securities Purchased
TPAV, LLC	\$9,499,995.16	337,206 shares of Common Stock
		and
		1,070.93000 shares of Series C Preferred Stock
OrbiMed Private Investments VIII, LP	\$4,499,992.14	159,729 shares of Common Stock
		and
		507.28200 shares of Series C Preferred Stock

REGISTRATION RIGHTS AGREEMENT

This Registration Rights Agreement (this "Agreement") is dated as of April 1, 2024, by and among Onconova Therapeutics, Inc., a Delaware corporation (the "Company"), and the several purchasers signatory hereto (each, including its successors and assigns, a "Purchaser" and collectively, the "Purchasers").

This Agreement is made pursuant to the Securities Purchase Agreement, dated as of the date hereof, between the Company and each Purchaser (the "Purchase Agreement").

NOW, THEREFORE, IN CONSIDERATION of the mutual covenants contained in this Agreement, and for other good and valuable consideration, the receipt and adequacy of which are hereby acknowledged, the Company and each of the Purchasers agree as follows:

1. <u>Definitions</u>. Capitalized terms used and not otherwise defined herein that are defined in the Purchase Agreement shall have the meanings given such terms in the Purchase Agreement. As used in this Agreement, the following terms shall have the following meanings:

"Advice" has the meaning set forth in Section 6(d).

"Affiliate" means any Person that, directly or indirectly through one or more intermediaries, controls or is controlled by or is under common control with a Person, as such terms are used in and construed under Rule 405 of the Securities Act.

"Agreement" has the meaning set forth in the Preamble.

"Business Day" means any day except any Saturday, any Sunday, any day which is a federal legal holiday in the United States or any day on which banking institutions in the State of New York are authorized or required by law or other governmental action to close.

"Commission" means the Securities and Exchange Commission.

"Common Stock" means the Company's common stock, par value \$0.01 per share, and stock of any other class of securities into which such securities may hereafter be reclassified or changed.

"Company" has the meaning set forth in the Preamble.

"Demand Notice" has the meaning set forth in Section 3(g).

"Demand Registration" has the meaning set forth in Section 3(g).

"Effective Date" means the date that the Registration Statement filed pursuant to Section 2(a) is first declared effective by the Commission.

"Effectiveness Deadline" means, with respect to the Initial Registration Statement or the New Registration Statement, the thirtieth (30th) calendar day following the Filing Deadline (or, in the event the Commission reviews and has written comments to the Initial Registration Statement or the New Registration Statement, the sixtieth (60th) calendar day following the Filing Deadline); provided, however, that if the Company is notified by the Commission that the Initial Registration Statement or the New Registration Statement will not be reviewed or is no longer subject to further review and comments, the Effectiveness Deadline as to such Registration Statement shall be the fifth (5th) Trading Day following the date on which the Company is so notified if such date precedes the dates otherwise required above; provided, further, that if the Effectiveness Deadline falls on a Saturday, Sunday or other day that the Commission is closed for business, the Effectiveness Deadline shall be extended to the next Business Day on which the Commission is open for business.

"Effectiveness Period" has the meaning set forth in Section 2(b).

"Exchange Act" means the Securities Exchange Act of 1934, as amended, and the rules and regulations promulgated thereunder.

"Filing Deadline" means, with respect to the Initial Registration Statement required to be filed pursuant to <u>Section 2(a)</u>, the ninetieth (90th) calendar day following the Closing Date, provided, however, that if the Filing Deadline falls on a Saturday, Sunday or other day that the Commission is closed for business, the Filing Deadline shall be extended to the next Business Day on which the Commission is open for business.

"FINRA" has the meaning set forth in Section 3(i).

"Holder" or "Holders" means the holder or holders, as the case may be, from time to time of Registrable Securities.

"Indemnified Party" has the meaning set forth in Section 5(c).

"Indemnifying Party" has the meaning set forth in Section 5(c).

"Initial Registration Statement" has the meaning set forth in Section 2(a).

"Losses" has the meaning set forth in Section 5(a).

"New Registration Statement" has the meaning set forth in Section 2(a).

"Person" means an individual or a corporation, partnership, trust, incorporated or unincorporated association, joint venture, limited liability company, joint stock company, government (or an agency or subdivision thereof) or other entity of any kind.

"Principal Market" means the Trading Market on which the Common Stock are primarily listed on and quoted for trading, which, as of the Closing Date, shall be the Nasdaq Stock Market LLC.

"Proceeding" means an action, claim, suit, investigation or proceeding (including, without limitation, an investigation or partial proceeding, such as a deposition), whether commenced or threatened.

"Prospectus" means the prospectus included in a Registration Statement (including, without limitation, a prospectus that includes any information previously omitted from a prospectus filed as part of an effective registration statement in reliance upon Rule 430B promulgated under the Securities Act), as amended or supplemented by any prospectus supplement, with respect to the terms of the offering of any portion of the Registrable Securities covered by a Registration Statement, and all other amendments and supplements to the Prospectus, including post-effective amendments, and all material incorporated by reference or deemed to be incorporated by reference in such Prospectus.

"Purchase Agreement" has the meaning set forth in the Recitals.

"Purchaser" or "Purchasers" has the meaning set forth in the Preamble.

"Registrable Securities" means all of (i) the Shares, (ii) the Series C Preferred Stock issued pursuant to the Purchase Agreement, (iii) any Common Stock issued on or around the date hereof pursuant to that certain Agreement and Plan of Merger, dated as of the date hereof, by and among the Company, Traws Merger Sub I, Inc., a Delaware corporation and wholly owned subsidiary of the Company, Traws Merger Sub II, LLC, a Delaware limited liability company and wholly owned subsidiary of the Company, and Trawsfynydd Therapeutics, Inc. (the *"Merger Agreement"*), (iv) the Series C Preferred Stock issued pursuant to the Merger Agreement, and (v) any securities issued or issuable upon any stock split, dividend or other distribution, recapitalization or similar event with respect to the foregoing, provided that, with respect to a particular Holder, such Holder's Common Stock or Series C Preferred Stock referenced in the preceding clauses (i) to (v) shall cease to be Registrable Securities upon the earlier to occur of the following: (A) a sale pursuant to a Registration Statement or Rule 144 under the Securities Act (in which case, only such security sold by the Holder shall cease to be a Registrable Security); and (B) such Common Stock or Series C Preferred Stock become eligible for resale by the Holder under Rule 144 without the requirement for the Company to be in compliance with the current public information required thereunder and without volume or manner-of-sale restrictions, pursuant to a written opinion letter of counsel for the Company to such effect, addressed, delivered and reasonably acceptable to the Transfer Agreent.

"Registration Statement(s)" means any one or more registration statements of the Company filed under the Securities Act that cover the resale of any of the Registrable Securities pursuant to the provisions of this Agreement (including without limitation the Initial Registration Statement, any New Registration Statement and any Remainder Registration Statements), amendments and supplements to such Registration Statements, including post-effective amendments, all exhibits and all material incorporated by reference or deemed to be incorporated by reference in such Registration Statements.

"Remainder Registration Statement" has the meaning set forth in Section 2(a).

"Rule 144" means Rule 144 promulgated by the Commission pursuant to the Securities Act, as such Rule may be amended from time to time, or any similar rule or regulation hereafter adopted by the Commission having substantially the same effect as such Rule.

"Rule 172" means Rule 172 promulgated by the Commission pursuant to the Securities Act, as such Rule may be amended from time to time, or any similar rule or regulation hereafter adopted by the Commission having substantially the same effect as such Rule.

"Rule 415" means Rule 415 promulgated by the Commission pursuant to the Securities Act, as such Rule may be amended from time to time, or any similar rule or regulation hereafter adopted by the Commission having substantially the same effect as such Rule.

"*Rule 424*" means Rule 424 promulgated by the Commission pursuant to the Securities Act, as such Rule may be amended from time to time, or any similar rule or regulation hereafter adopted by the Commission having substantially the same effect as such Rule.

"Rule 461" means Rule 461 promulgated by the Commission pursuant to the Securities Act, as such Rule may be amended from time to time, or any similar rule or regulation hereafter adopted by the Commission having substantially the same effect as such Rule.

"SEC Guidance" means (i) any publicly-available written or oral guidance, comments, requirements or requests of the Commission staff; provided, that any such oral guidance, comments, requirements or requests are reduced to writing by the Commission and (ii) the Securities Act.

"Securities Act" means the Securities Act of 1933, as amended, and the rules and regulations promulgated thereunder.

"Selling Shareholder Questionnaire" means a questionnaire in the form attached as Annex B hereto, or such other form of questionnaire or information provided to the Company in connection with the preparation of a Registration Statement hereunder.

"Series C Preferred Stock" means the Series C Non-Voting Convertible Preferred Stock, par value \$0.01 per share, of the Company.

"Shares" means (a) the shares of Common Stock issued to the Purchasers pursuant to the Purchase Agreement and (b) the shares of Common Stock which may be issued upon conversion of the Series C Preferred Stock issued to the Purchasers pursuant to the Purchase Agreement.

"Shelf Registration Statement" has the meaning set forth in Section 3(f).

"Takedown Request" has the meaning set forth in Section 3(f).

"Trading Day" means a day on which the Principal Market is open for business.

"Trading Market" means any of the following markets or exchanges on which the Common Stock is listed or quoted for trading on the date in question: the NYSE American, the Nasdaq Capital Market, the Nasdaq Global Market, the Nasdaq Global Select Market, the Nasdaq Stock Market LLC or the New York Stock Exchange (or any successors to any of the foregoing).

"Underwriter" means, with respect any Underwritten Offering, a securities dealer who purchases any Registrable Securities as a principal in connection with a distribution of such Registrable Securities.

"Underwritten Offering" means a public offering of securities registered under the Securities Act in which an Underwriter participates in the distribution of such securities, including on a firm commitment basis for reoffer and resale to the public, including any such offering that is a "bought deal" or a block trade.

2. Registration Rights.

On or prior to the Filing Deadline, the Company shall prepare and file with the Commission a Registration Statement covering the (a) resale of all of the Registrable Securities for an offering to be made on a continuous basis pursuant to Rule 415 (the "Initial Registration Statement"). The Initial Registration Statement shall be on Form S-3 (except if the Company is then ineligible to register for resale the Registrable Securities on Form S-3, in which case such registration shall be on such other form available to register for resale the Registrable Securities as a secondary offering) subject to the provisions of Section 2(c) and shall contain (except if otherwise required pursuant to written comments received from the Commission upon a review of such Registration Statement) the "Plan of Distribution" section substantially in the form attached hereto as Annex A (which may be modified to respond to comments, if any, provided by the Commission). Notwithstanding the registration obligations set forth in this Section 2, in the event the Commission informs the Company that all of the Registrable Securities cannot, as a result of the application of Rule 415, be registered for resale as a secondary offering on a single registration statement, the Company agrees to promptly (i) inform each of the Holders thereof and use its commercially reasonable efforts to file amendments to the Initial Registration Statement as required by the Commission and/or (ii) withdraw the Initial Registration Statement and file a new registration statement (a "New Registration Statement"), in either case covering the maximum number of Registrable Securities permitted to be registered by the Commission, on Form S-3 or, if the Company is ineligible to register the Registrable Securities on Form S-3, such other form available to register for resale the Registrable Securities as a secondary offering; provided, however, that prior to filing such amendment or New Registration Statement, the Company shall be obligated to use its commercially reasonable efforts to advocate with the Commission for the registration of all of the Registrable Securities in accordance with the SEC Guidance, including without limitation, the Securities Act Rules Compliance and Disclosure Interpretations Question 612.09. Notwithstanding any other provision of this Agreement, if the Commission or any SEC Guidance sets forth a limitation of the number of Registrable Securities permitted to be registered on a particular Registration Statement as a secondary offering (and notwithstanding that the Company used diligent efforts to advocate with the Commission for the registration of all or a greater number of Registrable Securities), unless otherwise directed in writing by a Holder as to its Registrable Securities, the number of Registrable Securities to be registered on such Registration Statement will first be reduced by Registrable Securities not acquired pursuant to the Purchase Agreement (whether pursuant to registration rights or otherwise), and second by Registrable Securities represented by Shares applied to the Holders on a pro rata basis based on the total number of Shares held by such Holders, subject to a determination by the Commission that certain Holders must be reduced first based on the number of Shares held by such Holders. In the event of a cutback hereunder, the Company shall give the Holder at least one (1) Trading Day prior notice along with the calculations as to such Holder's allotment. In the event the Company amends the Initial Registration Statement or files a New Registration Statement, as the case may be, in accordance with the foregoing, the Company will use its commercially reasonable efforts to file with the Commission, as promptly as allowed by Commission or SEC Guidance provided to the Company or to registrants of securities in general, one or more registration statements on Form S-3 or such other form available to register for resale those Registrable Securities that were not registered for resale on the Initial Registration Statement, as amended, or the New Registration Statement (the "Remainder Registration Statements"). No Holder shall be named as an "underwriter" in any Registration Statement without such Holder's prior written consent.

The Company shall use its commercially reasonable efforts to cause each Registration Statement to be declared effective by the (b) Commission as soon as practicable and, with respect to the Initial Registration Statement or the New Registration Statement, as applicable, no later than the Effectiveness Deadline (including filing with the Commission a request for acceleration of effectiveness in accordance with Rule 461 promulgated under the Securities Act), and shall use its commercially reasonable efforts to keep each Registration Statement continuously effective under the Securities Act until the earlier of (i) such time as all of the Registrable Securities covered by such Registration Statement have been publicly sold by the Holders; or (ii) the date that all Registrable Securities covered by such Registration Statement may be sold by non-affiliates without volume or manner-of-sale restrictions pursuant to Rule 144, without the requirement for the Company to be in compliance with the current public information requirement under Rule 144 (the "Effectiveness Period"). The Company shall request effectiveness of a Registration Statement as of 4:00 P.M. New York City time on a Trading Day. The Company shall promptly notify the Holders via e-mail of the effectiveness of a Registration Statement or any post-effective amendment thereto on the same Trading Day that the Company telephonically confirms effectiveness with the Commission, which date of confirmation shall initially be the date requested for effectiveness of such Registration Statement. The Company shall, by 9:30 A.M. New York City time on the first Trading Day after the Effective Date, file a final Prospectus with the Commission, as required by Rule 424(b) and shall provide the Purchasers with copies of the final Prospectus to be used in connection with the sale or other disposition of the securities covered thereby. The Company shall promptly inform each Holder in writing if, at any time during the Effectiveness Period, the Company does not satisfy the conditions specified in Rule 172 and, as a result thereof, the Holder is required to deliver a Prospectus in connection with any disposition of Registrable Securities.

(c) Each Holder of Registrable Securities to be sold agrees to furnish to the Company a completed Selling Shareholder Questionnaire not more than five (5) Trading Days following the date of this Agreement. At least 10 Trading Days prior to the first anticipated filing date of a Registration Statement for any registration under this Agreement, the Company will notify each Holder of the information the Company requires from that Holder for inclusion in the Registration Statement other than the information contained in the Selling Shareholder Questionnaire, if any, which shall be completed and delivered to the Company promptly upon request and, in any event, within three (3) Trading Days prior to the applicable anticipated filing date. Each Holder further agrees that it shall not be entitled to be named as a selling securityholder in the Registration Statement or use the Prospectus for offers and resales of Registrable Securities at any time, unless such Holder has provided such information to the Company and responded to any reasonable requests for further information as described in the previous sentence. Each Holder acknowledges and agrees that the information in the Selling Shareholder Questionnaire or request for further information as described in this Section 2(c) will be used by the Company in the preparation of the Registration Statement and hereby consents to the inclusion of such information in the Registration Statement (subject to such Holder's right to timely review the Registration Statement as set forth herein).

(d) In the event that Form S-3 is not available for the registration of the resale of Registrable Securities hereunder, the Company shall (i) register the resale of the Registrable Securities on another appropriate form reasonably acceptable to the Holders and (ii) undertake to register the Registrable Securities on Form S-3 promptly after such form is available, provided that the Company shall maintain the effectiveness of the Registration Statement then in effect until such time as a Registration Statement on Form S-3 covering the Registrable Securities has been declared effective by the Commission.

If (i) a Registration Statement covering the Registrable Securities is not filed with the Commission on or prior to the Filing (e) Deadline, the Company will make pro rata payments to each Purchaser then holding Registrable Securities, or (ii) following receipt of the Requisite Stockholder Approval, (A) a Registration Statement covering the Registrable Securities is not declared effective by the Commission prior to the Effectiveness Deadline or (B) after a Registration Statement has been declared effective by the Commission, (1) such Registration Statement ceases for any reason (including, without limitation, by reason of a stop order, or the Company's failure to update the Registration Statement) to remain continuously effective as to sell all Registrable Securities for which it is required to be effective, (2) the Purchasers are not permitted to utilize the Prospectus therein to resell such Registrable Securities (other than during permitted suspension under Section 6(d)), or (3) a permitted suspension of the Registration Statement under Section 6(d) exceeds the length set forth therein, (each of the foregoing events in clauses (i) and (ii), a "Registration Failure"), then, in addition to any other rights the Purchasers may have hereunder or under applicable law, the Company will make pro rata payments to each Purchaser of then outstanding Registrable Securities, as liquidated damages and not as a penalty (the "Liquidated Damages"), in an amount equal to one percent (1.0%) of the aggregate amount invested by such Purchaser for the Registrable Securities then held by such Purchaser for the initial day of a Registration Failure and for each thirty (30) day period (or pro rata portion thereof with respect to a final period, if any) thereafter until the Registration Failure is cured. The Liquidated Damages shall be paid monthly within ten (10) Business Days of the date of such Registration Failure and the end of each subsequent thirty (30) day period (or portion thereof with respect to a final period, if any) thereafter until the Registration Failure is cured. Such payments shall be made in cash to each Purchaser then holding Registrable Securities. Interest shall accrue at the rate of one percent (1%) per month on any such liquidated damages payments that shall not be paid by the applicable payment date until such amount is paid in full. Notwithstanding the foregoing, (I) no Liquidated Damages shall be payable with respect to any period after the expiration of the Effectiveness Period (as defined below) (it being understood that this sentence shall not relieve the Company of any Liquidated Damages accruing prior to the expiration of the Effectiveness Period), (II) in no event shall the aggregate amount of Liquidated Damages payable to a Purchaser exceed, in the aggregate, five percent (5.0%) of the aggregate purchase price paid by such Purchaser pursuant to the Purchase Agreement and (III) no Liquidated Damages shall accrue or be payable with respect to any reduction in the number or Registrable Securities to be included in a Registration Statement due to the application of Rule 415 as set forth in Section 2(a).

(f) At any time and from time to time, if a Registration Statement is effective that is on Form S-3 (a "Shelf Registration Statement"), then upon written request of any Holder (a "Takedown Request"), the Company will use reasonable best efforts to cooperate with the Holder and any Underwriter in effecting an Underwritten Offering pursuant to the Shelf Registration Statement as promptly as reasonably practicable following receipt of such request, provided that Holders shall not be entitled to make more than two (2) requests (whether for an Underwritten Offering or a Demand Registration) in any twelve (12)-month period. Each Takedown Request will specify the number of Registrable Securities proposed by the Holders to be included in such Underwritten Offering, the intended method of distribution and the estimated gross proceeds of such Underwritten Offering, which may not be less than \$5,000,000, other than an Underwritten Offering for all remaining Registrable Securities held by all Purchasers.

(g) If, at any time while there still remain Registrable Securities, the Company is no longer eligible to use or, notwithstanding its obligations under Sections 2(a), a Purchaser are not permitted to use the Registration Statement filed pursuant to Section 2(a) for resales, a Purchaser may deliver to the Company a written request to register a primary offering of all or a specified amount of the Registrable Securities under the Securities Act (a "Demand Notice"), and the Company shall use its commercially reasonable efforts to file, within thirty (30) Trading Days after receipt of the Demand Notice, a Registration Statement, on an appropriate form which the Company is then eligible to use, to register the offering of such Registrable Securities, which Registration Statement will (if specified in the Demand Notice) contemplate the ability of the Purchasers to effect an Underwritten Offering in accordance with Section 2(f) (each such registration, a "Demand Registration"); provided, however, that the Purchasers shall not be entitled to request more than two (2) requests (whether for an Underwritten Offering or for a Demand Registration) in any twelve-(12) month period. Each Demand Notice will specify the number of Registrable Securities proposed to be offered for sale, the intended method of distribution thereof and the estimated gross proceeds of such Demand Registration, which may not be less than \$5,000,000, other than a Demand Registration for all remaining Registrable Securities.

(h) In the event an Underwritten Offering of Registrable Securities involves one or more Underwriters, such Underwriters will be selected by the Purchasers; provided, that any Underwriter must be approved by the Company, which approval will not be unreasonably withheld. The Company shall use commercially reasonable efforts to take such actions as the lead Underwriter(s) may reasonably request in their efforts to sell Registrable Securities pursuant to such Registration Statement.

(i) Notwithstanding any other provision of this Section 2, but subject to the other terms of this Agreement, at any time and from time to time when an effective Shelf Registration Statement is on file with the Commission, if a Purchaser wishes to engage in an Underwritten Offering not involving a "roadshow," an offer commonly known as a "block trade" (a "Block Trade"), with a total offering price reasonably expected to exceed, in the aggregate, either (x) \$5,000,000.00 or (y) all remaining Registrable Securities held by the Purchaser, then such Purchaser only need to notify the Company of the Block Trade at least five (5) Trading Days prior to the day such offering is to commence and the Company shall as expeditiously as possible use its commercially reasonable efforts to facilitate such Block Trade; provided that the Purchaser wishing to engage in the Block Trade shall use commercially reasonable efforts to work with the Company and any Underwriters prior to making such request in order to facilitate preparation of the registration statement, prospectus and other offering documentation related to the Block Trade. Prior to the filing of the applicable "red herring" prospectus or prospectus supplement used in connection with a Block Trade, any Purchaser initiating such Block Trade. Notwithstanding anything to the contrary in this Agreement, the Company shall be responsible for the registration expenses (in accordance with Section 5) incurred in connection with a Block Trade prior to its withdrawal under this subsection.

3. <u>Registration Procedures</u>

In connection with the Company's registration obligations hereunder, the Company shall:

(a) Not less than five (5) Trading Days prior to the filing of each Registration Statement and not less than two (2) Trading Days prior to the filing of any related Prospectus or any amendment or supplement thereto (except for Annual Reports on Form 10-K, and Quarterly Reports on Form 10-Q and Current Reports on Form 8-K and any similar or successor reports), (i) furnish to each Holder copies of such Registration Statement, Prospectus or amendment or supplement thereto, as proposed to be filed, which documents will be subject to the review of such Holder (it being acknowledged and agreed that if a Holder does not object to or comment on the aforementioned documents within such five (5) Trading Day or two (2) Trading Day period, as the case may be, then the Holder shall be deemed to have consented to and approved the use of such documents) and to the Underwriters, if any, (ii) to the Underwriters, if any, and to the extent that a Holder is identified in the Registration Statement as an "underwriter" (as defined in the Securities Act), use commercially reasonable efforts to cause its officers and directors, counsel and independent registered public accountants to respond to such inquiries as shall be necessary, in the reasonable opinion of respective counsel to each Holder and Underwriter, as applicable, to conduct a reasonable investigation within the meaning of the Securities Act. The Company shall not file any Registration Statement or supplement thereto in a form to which a Holder reasonably objects in good faith, provided that, such Holder notifies the Company of such objection in writing within the five (5) Trading Day or two (2) Trading Day periods described above, as applicable.

(b) (i) Prepare and file with the Commission such amendments (including post-effective amendments) and supplements, to each Registration Statement and the Prospectus used in connection therewith as may be necessary to keep such Registration Statement continuously effective as to the applicable Registrable Securities for its Effectiveness Period and prepare and file with the Commission such additional Registration Statements in order to register for resale under the Securities Act all of the Registrable Securities (and before filing with the Commission furnish to the Underwriters, if any, and to the Holders, copies of such documents) (ii) cause the related Prospectus to be amended or supplemented by any required Prospectus supplement (subject to the terms of this Agreement), and, as so supplemented or amended, to be filed pursuant to Rule 424; (iii) respond as promptly as reasonably practicable to any comments received from the Commission with respect to each Registration Statement or any amendment thereto and, as promptly as reasonably possible, provide the Holders true and complete copies of all correspondence from and to the Commission relating to such Registration Statement that pertains to the Holders as "Selling Stockholders" but not any comments that would result in the disclosure to the Holders of material and non-public information concerning the Company (unless such Holder consents to receive such material and non-public information); and (iv) comply with the provisions of the Securities Act and the Exchange Act with respect to the disposition of all Registrable Securities covered by a Registration Statement until such time as all of such Registrable Securities shall have been disposed of (subject to the terms of this Agreement) in accordance with the intended methods of disposition by the Holders thereof as set forth in such Registration Statement as so amended or in such Prospectus as so supplemented; provided, however, that each Purchaser shall be responsible for the delivery of the Prospectus to the Persons to whom such Purchaser sells any of the Registrable Securities (including in accordance with Rule 172 under the Securities Act) to the extent required under the Securities Act, and each Purchaser agrees to dispose of Registrable Securities in compliance with the "Plan of Distribution" described in the Registration Statement and otherwise in compliance with applicable federal and state securities laws. In the case of amendments and supplements to a Registration Statement which are required to be filed pursuant to this Agreement (including pursuant to this Section 3(b)) by reason of the Company filing a report on Form 10-K, Form 10-Q or Form 8-K or any analogous report under the Exchange Act, the Company shall have incorporated such report by reference into such Registration Statement, if applicable, or shall file such amendments or supplements with the Commission on the same day on which the Exchange Act report which created the requirement for the Company to amend or supplement such Registration Statement was filed.

Notify the Underwriters, if any, and the Holders of Registrable Securities to be sold (which notice shall, if given pursuant to (c) clauses (iii) through (vi) hereof, be accompanied by an instruction to suspend the use of the Prospectus until the requisite changes have been made. provided that the Company shall omit any material, non-public information relating to the Company and/or any of its subsidiaries) as promptly as reasonably practicable (and, in the case of (i)(A) below, not less than one (1) Trading Day prior to such filing) and (if requested by any such Person) confirm such notice in writing no later than one (1) Trading Day following the day: (i)(A) when a Prospectus or any Prospectus supplement or posteffective amendment to a Registration Statement is proposed to be filed; (B) when the Commission notifies the Company whether there will be a "review" of such Registration Statement and whenever the Commission comments in writing on any Registration Statement (in which case the Company shall provide to each of the Holders and Underwriters, if any, true and complete copies of all comments that pertain to the Holders as a "Selling Stockholder" or to the "Plan of Distribution" and all written responses thereto, but not information that the Company believes would constitute material and non-public information); and (C) with respect to each Registration Statement or any post-effective amendment, when the same has become effective; (ii) of any request by the Commission or any other federal or state governmental authority for amendments or supplements to a Registration Statement or Prospectus or for additional information that pertains to the Holders as "Selling Stockholders" or the "Plan of Distribution"; (iii) of the issuance by the Commission or any other federal or state governmental authority of any stop order suspending the effectiveness of a Registration Statement covering any or all of the Registrable Securities or the initiation of any Proceedings for that purpose; (iv) of the receipt by the Company of any notification with respect to the suspension of the qualification or exemption from qualification of any of the Registrable Securities for sale in any jurisdiction, or the initiation or threatening of any Proceeding for such purpose; (v) of the occurrence of any event or passage of time that makes the financial statements included or incorporated by reference in a Registration Statement ineligible for inclusion or incorporation by reference therein or any statement made in such Registration Statement or Prospectus or any document incorporated or deemed to be incorporated therein by reference untrue in any material respect or that requires any revisions to such Registration Statement, Prospectus or other documents so that, in the case of such Registration Statement or the Prospectus, as the case may be, it will not contain any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary to make the statements therein (in the case of any Prospectus, form of prospectus or supplement thereto, in light of the circumstances under which they were made), not misleading and (vi) of the occurrence or existence of any pending corporate development with respect to the Company that, upon the advice of legal counsel, the Company's board of directors reasonably believes may be material and that, in the reasonable determination of the Company's board of directors, makes it not in the best interest of the Company to allow continued availability of a Registration Statement or Prospectus, provided that, in no event shall any such notice contain any information that would constitute material, non-public information regarding the Company or any of its subsidiaries and the Company agrees that the Holders shall not have any duty of confidentiality to the Company or any of its subsidiaries and shall not have any duty to the Company or any of its subsidiaries not to trade on the basis of such information.

(d) Use commercially reasonable efforts to avoid the issuance of, or, if issued, obtain the withdrawal of (i) any order stopping or suspending the effectiveness of a Registration Statement, or (ii) any suspension of the qualification (or exemption from qualification) of any of the Registrable Securities for sale in any jurisdiction, as soon as practicable.

(e) If requested by a Holder or an Underwriter (if any), furnish to such Holder or Underwriter, without charge, at least one conformed copy of each Registration Statement and each amendment thereto, including financial statements and schedules, all documents incorporated or deemed to be incorporated therein by reference to the extent requested by such Person, and all exhibits to the extent requested by such Person (including those previously furnished or incorporated by reference) promptly after the filing of such documents with the Commission; provided, that the Company shall have no obligation to provide any document pursuant to this clause that is available on the Commission's EDGAR system.

(f) Prior to any resale of Registrable Securities by a Holder, use its commercially reasonable efforts to register or qualify or cooperate with the selling Holders and Underwriter, if any, in connection with the registration or qualification (or exemption from the registration or qualification) of the resale of such Registrable Securities (or, in the case of qualification, of such Registrable Securities for the resale) by the Holder under the securities or blue sky laws of such jurisdictions within the United States as any Holder reasonably requests in writing, to keep each registration or qualification (or exemption therefrom) effective during the Effectiveness Period and to do any and all other acts or things reasonably necessary to enable the disposition in such jurisdictions of the Registrable Securities covered by each Registration Statement; provided, that the Company shall not be required to qualify generally to do business in any jurisdiction where it is not then so qualified, subject the Company to any material tax in any such jurisdiction where it is not then so subject or file a general consent to service of process in any such jurisdiction.

(g) Cooperate with such Holder or Underwriter, if any, to facilitate the timely preparation and delivery of certificates or book entry statements, as applicable, representing Registrable Securities to be delivered to a transferee pursuant to the Registration Statement, which certificates or statements shall be free, to the extent permitted by the Purchase Agreement and under law, of all restrictive legends, and to enable such Registrable Securities to be in such denominations and registered in such names as any such Holders or Underwriter may reasonably request.

Following the occurrence of any event contemplated by Section 3(c), as promptly as reasonably practicable (taking into account (h) the Company's good faith assessment of any adverse consequences to the Company and its shareholders of the premature disclosure of such event), prepare a supplement or amendment, including a post-effective amendment, to the affected Registration Statements or a supplement to the related Prospectus or any document incorporated or deemed to be incorporated therein by reference, and file any other required document so that, as thereafter delivered, no Registration Statement nor any Prospectus will contain an untrue statement of a material fact or omit to state a material fact required to be stated therein or necessary to make the statements therein (in the case of any Prospectus, form of prospectus or supplement thereto, in light of the circumstances under which they were made), not misleading. If the Company notifies the Holders and Underwriter, if any, in accordance with clauses (iii) through (vi) of Section 3(c) above to suspend the use of any Prospectus until the requisite changes to such Prospectus have been made, then the Holders shall suspend use of such Prospectus. The Company will use its commercially reasonable efforts to ensure that the use of the Prospectus may be resumed as promptly as is practicable. The Company shall be entitled to exercise its right under this Section 3(h) to suspend the availability of a Registration Statement and Prospectus, subject to the payment of Liquidated Damages otherwise required pursuant to Section 2(e), in accordance with the time periods set forth in Section 6(d), which may be extended only in accordance with Section 6(f). For the avoidance of doubt, the Company's rights under this Section 3(h) shall include suspensions of availability arising from the filing of a post-effective amendment to a Registration Statement to update the Prospectus therein to include the information contained in the Company's Annual Report on Form 10-K, which suspensions may extend for the amount of time reasonably required to respond to any comments of the staff of the Commission on such amendment.

(i) The Company may require each selling Holder to furnish to the Company a certified statement as to (i) the number of shares of Common Stock beneficially owned by such Holder and any Affiliate thereof, (ii) any Financial Industry Regulatory Authority (*"FINRA"*) affiliations, (iii) any natural persons who have the power to vote or dispose of the Common Stock and (iv) any other information as may be requested by the Commission, FINRA or any state securities commission.

(j) The Company shall cooperate with any registered broker through which a Holder proposes to resell its Registrable Securities in effecting a filing with FINRA pursuant to FINRA Rule 5110 as requested by any such Holder and the Company shall pay the filing fee required for the first such filing within two (2) Business Days of the request therefor.

(k) If at any time the Commission takes the position that the offering of some or all of the Registrable Securities in a Registration Statement is not eligible to be made on a delayed or continuous basis under the provisions of Rule 415 under the Securities Act or requires any Holder to be named as an "underwriter," the Company shall use commercially reasonable efforts to persuade the Commission that the offering contemplated by such Registration Statement is a valid secondary offering and not an offering "by or on behalf of the issuer" as defined in Rule 415 and that none of the Holders is an "underwriter."

(1) With respect to an Underwritten Offering, obtain from the Company's independent public accountants any comfort letter reasonably requested by any Underwriter and furnish to each Holder of Registrable Securities included in any Registration Statement a signed counterpart of such comfort letter.

4 Registration Expenses. All fees and expenses incident to the Company's performance of or compliance with its obligations under this Agreement (excluding any underwriting discounts and selling commissions and all legal fees and expenses of legal counsel for any Holder) shall be borne by the Company whether or not any Registrable Securities are sold pursuant to a Registration Statement. The fees and expenses referred to in the foregoing sentence shall include, without limitation, (i) all registration and filing fees (including, without limitation, fees and expenses of the Company's counsel and independent registered public accountants (A) with respect to filings required to be made with any Trading Market on which the Common Stock are then listed for trading, (B) with respect to compliance with applicable state securities or blue sky laws (including, without limitation, fees and disbursements of counsel for the Company in connection with blue sky qualifications or exemptions of the Registrable Securities and determination of the eligibility of the Registrable Securities for investment under the laws of such jurisdictions as requested by the Holders) and (C) if not previously paid by the Company in connection with Section 3(j) above, with respect to any filing that may be required to be made by any broker through which a Holder intends to make sales of Registrable Securities with FINRA pursuant to the FINRA Rule 5110, so long as the broker is receiving no more than a customary brokerage commission in connection with such sale), (ii) printing expenses (including, without limitation, expenses of printing certificates for Registrable Securities and of printing prospectuses if the printing of prospectuses is reasonably requested by the Holders of a majority of the Registrable Securities included in the Registration Statement), (iii) messenger, telephone and delivery expenses, (iv) fees, expenses and disbursements of counsel for the Company, (v) Securities Act liability insurance, if the Company so desires such insurance, and (vi) fees and expenses of all other Persons retained by the Company in connection with the registrations and consummation of the transactions contemplated by this Agreement. In addition, the Company shall be responsible for all of its internal expenses incurred in connection with the consummation of the transactions contemplated by this Agreement (including, without limitation, all salaries and expenses of its officers and employees performing legal or accounting duties), the expense of any annual audit and the fees and expenses incurred in connection with the listing of the Registrable Securities on any securities exchange as required hereunder. In no event shall the Company be responsible for any underwriting, broker or similar fees or commissions of any Holder or, except to the extent provided for in the Transaction Documents, any legal fees or other costs of the Holders.

5. Indemnification.

Indemnification by the Company. The Company shall, notwithstanding any termination of this Agreement, indemnify, defend and (a) hold harmless each Holder and each of their respective officers, directors, agents, partners, members, managers, stockholders, Affiliates, investment advisers, brokers (including brokers who offer and sell Registrable Securities as principal as a result of a pledge or any failure to perform under a margin call of Common Stock), and employees (and any other Persons with a functionally equivalent role of a Person holding such titles, notwithstanding a lack of such title or any other title) of each of them, each Person who controls any such Holder (within the meaning of Section 15 of the Securities Act or Section 20 of the Exchange Act) and the officers, directors, partners, members, managers, stockholders, agents, investment advisers and employees (and any Persons with a functionally equivalent role of a Person holding such titles, notwithstanding a lack of such title or any other title) of each such controlling Person, to the fullest extent permitted by applicable law, from and against any and all losses, claims, damages, liabilities, costs (including, without limitation, reasonable costs of preparation and investigation and reasonable attorneys' fees), expenses and disbursements (collectively, "Losses"), as incurred, that arise out of or relate to (i) any untrue or alleged untrue statement of a material fact contained in any Registration Statement, any Prospectus or any form of prospectus or in any amendment or supplement thereto or in any preliminary prospectus, or arising out of or relating to any omission or alleged omission to state a material fact required to be stated therein or necessary to make the statements therein (in the case of any Prospectus or form of prospectus or supplement thereto, in light of the circumstances under which they were made) not misleading, or (ii) any violation or alleged violation by the Company or its agents of the Securities Act, the Exchange Act or any state securities law or any rule or regulation thereunder, in connection with the performance of its obligations under this Agreement or any action or inaction required of the Company in connection with any registration, except to the extent, but only to the extent, that (A) such untrue statements, alleged untrue statements, omissions or alleged omissions are based solely upon information regarding such Holder furnished in writing to the Company by such Holder expressly for use therein, or to the extent that such information relates to such Holder or such Holder's proposed method of distribution of Registrable Securities and was reviewed and expressly approved in writing by such Holder expressly for use in the Registration Statement, such Prospectus or such form of Prospectus or in any amendment or supplement thereto (it being understood that each Holder has approved Annex A hereto for this purpose), (B) in the case of an occurrence of an event of the type specified in Section 3(c)(iii)-(vi), related to the use by a Holder of an outdated or defective Prospectus after the Company has notified such Holder in writing that the Prospectus is outdated or defective and prior to the receipt by such Holder of the Advice contemplated and defined in Section 6(d) below, to the extent that following the receipt of the Advice the misstatement or omission giving rise to such Loss would have been corrected or (C) to the extent that any such Losses arise out of the Purchaser's (or any other indemnified Person's) failure to send or give a copy of the Prospectus or supplement (as then amended or supplemented), if required, pursuant to Rule 172 under the Securities Act (or any successor rule) to the Persons asserting an untrue statement or alleged untrue statement or alleged untrue statement or omission or alleged omission at or prior to the written confirmation of the sale of Registrable Securities to such Person if such statement or omission was corrected in such Prospectus or supplement. The Company shall notify the Holders promptly of the institution, threat or assertion of any Proceeding arising from or in connection with the transactions contemplated by this Agreement of which the Company is aware. Such indemnity shall remain in full force and effect regardless of any investigation made by or on behalf of an Indemnified Party (as defined in Section 5(c)) and shall survive the transfer of the Registrable Securities by the Holders.

Indemnification by Holders. Each Holder shall, severally and not jointly, indemnify and hold harmless the Company, its directors, (b) officers, agents and employees, each Person who controls the Company (within the meaning of Section 15 of the Securities Act and Section 20 of the Exchange Act), and the directors, officers, agents or employees of such controlling Persons (and any other Persons with a functionally equivalent role of a Person holding such titles, notwithstanding a lack of such title or any other title), to the fullest extent permitted by applicable law, from and against all Losses, as incurred, arising out of or are based solely upon any untrue or alleged untrue statement of a material fact contained in any Registration Statement, any Prospectus, or any form of prospectus, or in any amendment or supplement thereto or in any preliminary prospectus, or arising out of or relating to any omission or alleged omission of a material fact required to be stated therein or necessary to make the statements therein (in the case of any Prospectus, or any form of prospectus or supplement thereto, in light of the circumstances under which they were made) not misleading (i) to the extent that such untrue statements or omissions are based solely upon information regarding such Holder furnished in writing to the Company by such Holder expressly for use therein or (ii) to the extent that such information relates to such Holder or such Holder's proposed method of distribution of Registrable Securities and was reviewed and approved in writing by such Holder expressly for use in a Registration Statement (it being understood that the Holder has approved Annex A hereto for this purpose), such Prospectus or such form of Prospectus or in any amendment or supplement thereto or (iii) in the case of an occurrence of an event of the type specified in Section 3(c)(iii)-(vi), to the extent related to the use by such Holder of an outdated or defective Prospectus after the Company has notified such Holder in writing that the Prospectus is outdated or defective and prior to the receipt by such Holder of the Advice contemplated in Section 6(d). In no event shall the liability of any selling Holder hereunder be greater in amount than the dollar amount of the net proceeds received by such Holder upon the sale of the Registrable Securities giving rise to such indemnification obligation.

(c) <u>Conduct of Indemnification Proceedings</u>. If any Proceeding shall be brought or asserted against any Person entitled to indemnity hereunder (an "*Indemnified Party*"), such Indemnified Party shall promptly notify the Person from whom indemnity is sought (the "*Indemnifying Party*") in writing, and the Indemnifying Party shall have the right to assume the defense thereof, including the employment of counsel reasonably satisfactory to the Indemnified Party and the payment of all reasonable fees, expenses and disbursements incurred in connection with defense thereof; provided, that the failure of any Indemnified Party to give such notice shall not relieve the Indemnifying Party of its obligations or liabilities pursuant to this Agreement, except (and only) to the extent that it shall be finally determined by a court of competent jurisdiction (which determination is not subject to appeal or further review) that such failure shall have materially and adversely prejudiced the Indemnifying Party.

An Indemnified Party shall have the right to employ separate counsel in any such Proceeding and to participate in the defense thereof, but the fees, expenses and disbursements of such counsel shall be at the expense of such Indemnified Party or Indemnified Parties unless: (1) the Indemnifying Party has agreed in writing to pay such fees, expenses and amounts; (2) the Indemnifying Party shall have failed promptly to assume the defense of such Proceeding and to employ counsel reasonably satisfactory to such Indemnified Party in any such Proceeding; or (3) the named parties to any such Proceeding (including any impleaded parties) include both such Indemnified Party and the Indemnifying Party, and such Indemnified Party shall have been advised by counsel that a conflict of interest exists if the same counsel were to represent such Indemnified Party and the Indemnifying Party (in which case, if such Indemnifying Party shall not have the right to assume the defense thereof and such counsel shall be at the expense of the Indemnifying Party); provided, that the Indemnifying Party shall not be liable for the fees, expenses and disbursements of more than one separate firm of attorneys at any time for all Indemnified Parties. The Indemnifying Party shall not be liable for any settlement of any such Proceeding effected without its written consent, which consent shall not be unreasonably withheld, delayed or conditioned. No Indemnifying Party shall, without the prior written consent of the Indemnified Party, effect any settlement of any pending Proceeding in respect of which any Indemnified Party is a party, unless such settlement includes an unconditional release of such Indemnified Party from all liability on claims that are the subject matter of such Proceeding.

Subject to the terms of this Agreement, all fees, expenses and disbursements of the Indemnified Party (including reasonable fees, expenses and disbursements to the extent incurred in connection with investigating or preparing to defend such Proceeding in a manner not inconsistent with this <u>Section 5</u>) shall be paid to the Indemnified Party, as incurred, within 20 Trading Days of written notice thereof to the Indemnifying Party; provided, that the Indemnified Party shall promptly reimburse the Indemnifying Party for that portion of such fees, expenses and disbursements applicable to such actions for which such Indemnified Party is finally judicially determined to not be entitled to indemnification hereunder. The failure to deliver written notice to the Indemnifying Party within a reasonable time of the commencement of any such action shall not relieve such Indemnifying Party of any liability to the Indemnified Party under this <u>Section 5</u>, except to the extent that the Indemnifying Party is materially and adversely prejudiced in its ability to defend such action.

(d) <u>Contribution</u>. If a claim for indemnification under Section 5(a) or 5(b) is unavailable to an Indemnified Party or insufficient to hold an Indemnified Party harmless for any Losses, then each Indemnifying Party, in lieu of indemnifying such Indemnified Party, shall contribute to the amount paid or payable by such Indemnified Party as a result of such Losses, in such proportion as is appropriate to reflect the relative fault of the Indemnifying Party and Indemnified Party in connection with the actions, statements or omissions that resulted in such Losses as well as any other relevant equitable considerations. The relative fault of such Indemnifying Party and Indemnified Party shall be determined by reference to, among other things, whether any action in question, including any untrue or alleged untrue statement of a material fact or omission or alleged omission of a material fact, has been taken or made by, or relates to information supplied by, such Indemnifying Party or Indemnified Party, and the parties' relative intent, knowledge, access to information and opportunity to correct or prevent such action, statement or omission. The amount paid or payable by a party as a result of any Losses shall be determed to include, subject to the limitations set forth in this Agreement, any reasonable attorneys' or other reasonable fees, expenses or disbursements incurred by such party in connection with any Proceeding to the extent such party would have been indemnified for such fees, expenses or disbursements if the indemnification provided for in this Section 5 was available to such party in accordance with its terms.

The parties hereto agree that it would not be just and equitable if contribution pursuant to this <u>Section 5(d)</u> were determined by pro rata allocation or by any other method of allocation that does not take into account the equitable considerations referred to in the immediately preceding paragraph. Notwithstanding the provisions of this <u>Section 5(d)</u>, (A) no Holder shall be required to contribute, in the aggregate, any amount in excess of the amount by which the net proceeds actually received by such Holder from the sale of the Registrable Securities subject to the Proceeding exceeds the amount of any damages that such Holder has otherwise been required to pay by reason of such untrue or alleged untrue statement or omission or alleged omission and (B) no contribution will be made under circumstances where the maker of such contribution would not have been required to indemnify the Indemnified Party under the fault standards set forth in this <u>Section 5</u>. No person guilty of fraudulent misrepresentation (within the meaning of Section 11(f) of the Securities Act) shall be entitled to contribution from any Person who was not guilty of such fraudulent misrepresentation.

The indemnity and contribution agreements contained in this <u>Section 5</u> are in addition to any liability that the Indemnifying Parties may have to the Indemnified Parties and are not in diminution or limitation of the indemnification provisions under the Purchase Agreement.

6. Miscellaneous.

(a) <u>Remedies</u>. In the event of a breach by the Company or by a Holder of any of their obligations under this Agreement, each Holder or the Company, as the case may be, in addition to being entitled to exercise all rights granted by law and under this Agreement, including recovery of damages, will be entitled to seek specific performance of its rights under this Agreement, without the requirement of posting a bond. The Company and each Holder agree that monetary damages would not provide adequate compensation for any losses incurred by reason of a breach by it of any of the provisions of this Agreement and hereby further agrees that, in the event of any action for specific performance in respect of such breach, it shall waive the defense that a remedy at law would be adequate.

(b) <u>No Piggyback on Registrations; Prohibition on Filing Other Registration Statements</u>. Neither the Company nor any of its security holders (other than the Holders in such capacity pursuant hereto) may include securities of the Company in a Registration Statement other than the Registrable Securities and the Company shall not prior to the Effective Date enter into any agreement providing any such right to any of its security holders. For the avoidance of doubt, the provisions of this Agreement shall not impact the terms of any lock-up agreement entered into by Purchaser for the benefit of the Company on or about the date hereof.

(c) <u>Compliance</u>. Each Holder covenants and agrees that it will comply with the prospectus delivery requirements of the Securities Act as applicable to it (unless an exemption therefrom is available) in connection with sales of Registrable Securities pursuant to the Registration Statement and shall sell the Registrable Securities only in accordance with a method of distribution described in the Registration Statement, or pursuant to an exemption from, or in a transaction not subject to, the registration requirements of the Securities Act.

(d) <u>Discontinued Disposition</u>. By its acquisition of Registrable Securities, each Holder agrees that, upon receipt of a notice from the Company of the occurrence of any event of the kind described in <u>Section 3(c)(iii)-(vi)</u>, such Holder will forthwith discontinue disposition of such Registrable Securities under a Registration Statement until it is advised in writing (the "*Advice*") by the Company that the use of the applicable Prospectus (as it may have been supplemented or amended) may be resumed. The Company will use its commercially reasonable efforts to ensure that the use of the Prospectus may be resumed as promptly as is practicable. Notwithstanding anything herein to the contrary, no Holder shall be required to discontinue disposition of Registrable Securities under a Registration Statement by virtue of the delivery by the Company of a notice of the occurrence of any event of the kind described in <u>Section 3(c)(vi)</u> on more than two occasions or for more than 90 total calendar days, in each case during any twelve-month period, or for more than 45 calendar days during any 90-day period.

(e) <u>Piggy-Back Registrations</u>. If, at any time during the Effectiveness Period, there is not an effective Registration Statement covering all of the Registrable Securities and the Company shall determine to prepare and file with the Commission a registration statement relating to an offering for its own account or the account of others under the Securities Act of any of its equity securities, other than on Form S-4 or Form S-8 (each as promulgated under the Securities Act) or their then equivalents relating to equity securities to be issued solely in connection with any acquisition of any entity or business or equity securities issuable in connection with the Company's stock option or other employee benefit plans, then the Company shall deliver to each Holder a written notice of such determination and, if within fifteen days after the date of the delivery of such notice, any such Holder shall so request in writing, the Company shall include in such registration statement all or any part of such Registrable Securities such Holder requests to be registered; provided, however, that the Company shall not be required to register any Registrable Securities pursuant to this <u>Section 6(e)</u> that are eligible for resale pursuant to Rule 144 (without volume restrictions or current public information requirements) promulgated by the Commission pursuant to the Securities Act or that are the subject of a then effective Registration Statement.

(f) <u>No Inconsistent Agreements</u>. Neither the Company nor any of its subsidiaries has entered, as of the date hereof, nor shall the Company or any of its subsidiaries, on or after the date hereof, enter into any agreement with respect to its securities, that would have the effect of impairing the rights granted to the Holders in this Agreement or otherwise conflicts with the provisions hereof.

(g) <u>Amendments and Waivers</u>. The provisions of this Agreement, including the provisions of this sentence, may not be amended, modified or supplemented, or waived unless the same shall be in writing and signed by the Company and Holders holding no less than a majority of the then outstanding Registrable Securities, provided that (i) any party may give a waiver as to itself and (ii) any proposed amendment that would, by its terms, have a disproportionate and materially adverse effect on any Holder shall require the consent of such Holder(s). Notwithstanding the foregoing, (1) a waiver or consent to depart from the provisions hereof with respect to a matter that relates exclusively to the rights of Holders and that does not directly or indirectly affect the rights of other Holders may be given by Holders of all of the Registrable Securities to which such waiver or consent relates and (2) none of the definitions of Filing Deadline, Effectiveness Deadline or Effectiveness Period, Section 2(e), Section 3(c), Section 5, Section 6(d), or the provisions of this sentence, may be amended, modified, or supplemented except with the consent of each Holder.

(h) <u>Notices</u>. Any and all notices or other communications or deliveries required or permitted to be provided hereunder shall be delivered as set forth in the Purchase Agreement.

(i) <u>Successors and Assigns</u>. This Agreement shall inure to the benefit of and be binding upon the successors and permitted assigns of each of the parties and shall inure to the benefit of each Holder. Nothing in this Agreement, express or implied, is intended to confer upon any party other than the parties hereto or their respective successors and assigns any rights, remedies, obligations, or liabilities under or by reason of this Agreement, except as expressly provided in this Agreement. The Company may not assign its rights (except by merger or in connection with another entity acquiring all or substantially all of the Company's assets) or obligations hereunder without the prior written consent of all the Holders of the then outstanding Registrable Securities. Each Holder may assign its respective rights hereunder in the manner and to the Persons as permitted under the Purchase Agreement; provided in each case that (i) the Holder agrees in writing with the transferee or assignee to assign such rights and related obligations under this Agreement, and for the transferee or assignee to assume such obligations, and a copy of such agreement is furnished to the Company within a reasonable time after such transfere or assignee and the securities with respect to which such registration rights are being transferred or assigned, (iii) at or before the time the Company received the written notice contemplated by clause (ii) of this sentence, the transferee or assignee agrees in writing with the Company to be bound by all of the provisions contained herein and (iv) the transferee is an "accredited investor," as that term is defined in Rule 501 of Regulation D.

(j) <u>Execution and Counterparts</u>. This Agreement may be executed in two or more counterparts, each of which when so executed shall be deemed to be an original and, all of which taken together shall constitute one and the same Agreement and shall become effective when counterparts have been signed by each party and delivered to the other party, it being understood that both parties need not sign the same counterpart. Counterparts may be delivered via facsimile, electronic mail (including without limitation any electronic signature covered by the U.S. federal ESIGN Act of 2000) or other transmission method and any counterpart so delivered shall be deemed to have been duly and validly delivered and be valid and effective for all purposes.

(k) <u>Governing Law</u>. All questions concerning the construction, validity, enforcement and interpretation of this Agreement shall be determined in accordance with the provisions of the Purchase Agreement.

(1) <u>Cumulative Remedies</u>. The remedies provided herein are cumulative and not exclusive of any other remedies provided by law.

(m) <u>Severability</u>. If any term, provision, covenant or restriction of this Agreement is held by a court of competent jurisdiction to be invalid, illegal, void or unenforceable, the remainder of the terms, provisions, covenants and restrictions set forth herein shall remain in full force and effect and shall in no way be affected, impaired or invalidated, and the parties hereto shall use their good faith reasonable efforts to find and employ an alternative means to achieve the same or substantially the same result as that contemplated by such term, provision, covenant or restriction. It is hereby stipulated and declared to be the intention of the parties that they would have executed the remaining terms, provisions, covenants and restrictions without including any of such that may be hereafter declared invalid, illegal, void or unenforceable.

(n) <u>Headings</u>. The headings in this Agreement are for convenience only and shall not limit or otherwise affect the meaning hereof.

(0) Independent Nature of Purchasers' Obligations and Rights. The obligations of each Purchaser under this Agreement are several and not joint with the obligations of any other Purchaser hereunder, and no Purchaser shall be responsible in any way for the performance of the obligations of any other Purchaser hereunder. The decision of each Purchaser to purchase Securities pursuant to the Transaction Documents has been made by such Purchaser independently of any other Purchaser and independently of any information, materials, statements or opinions as to the business, affairs, operations, assets, properties, liabilities, results of operations, condition (financial or otherwise) or prospects of the Company or any Subsidiary which may have been made or given by any other Purchaser or by any agent or employee of any other Purchaser, and no Purchaser and any of its agents or employees shall have any liability to any other Purchaser (or any other Person) relating to or arising from any such information, materials, statement or opinions. Nothing contained herein or in any Transaction Document, and no action taken by any Purchaser pursuant hereto or thereto, shall be deemed to constitute the Purchasers as a partnership, an association, a joint venture or any other kind of entity, or create a presumption that the Purchasers are in any way acting in concert or as a group (including, without limitation, a "group" within the meaning of Section 13(d)(3) of the Exchange Act) with respect to such obligations or the transactions contemplated by the Transaction Documents. Each Purchaser acknowledges that no other Purchaser has acted as agent for such Purchaser in connection with making its investment hereunder and that no Purchaser will be acting as agent of such Purchaser in connection with monitoring its investment in the securities or enforcing its rights under the Transaction Documents. Each Purchaser shall be entitled to independently protect and enforce its rights, including, without limitation, the rights arising out of this Agreement or out of the other Transaction Documents, and it shall not be necessary for any other Purchaser to be joined as an additional party in any Proceeding for such purpose. The Company acknowledges that each of the Purchasers has been provided with the same Registration Rights Agreement for the purpose of closing a transaction with multiple Purchasers and not because it was required or requested to do so by any Purchaser. It is expressly understood that each provision contained in this Agreement is between the Company and a Purchaser, solely, and not between the Company and the Purchasers collectively and not between and among the Purchasers.

(p) <u>Current Public Information</u>. With a view to making available to the Holders the benefits of Rule 144 (or its successor rule) and any other rule or regulation of the Commission that may at any time permit the Holders to sell shares of Common Stock to the public without registration, the Company covenants and agrees to use commercially reasonable efforts to: (i) make and keep adequate current public information available, as those terms are understood and defined in Rule 144, until the earlier of (A) six months after such date as all of the Registrable Securities may be sold without restriction by the holders thereof pursuant to Rule 144 or any other rule of similar effect or (B) such date as there are no longer Registrable Securities; and (ii) file with the SEC in a timely manner all reports and other documents required of the Company under the Exchange Act.

IN WITNESS WHEREOF, the parties have executed this Registration Rights Agreement as of the date first written above.

ONCONOVA THERAPEUTICS, INC.

By: /s/ Steven M. Fruchtman

Name: Steven M. Fruchtman Title: President and CEO IN WITNESS WHEREOF, the parties have executed this Registration Rights Agreement as of the date first written above.

ORBIMED PRIVATE INVESTMENTS VIII, LP

By: OrbiMed Capital GP VIII LLC, its General Partner

By: OrbiMed Advisors LLC, its Managing Member

By: /s/ Carl Gordon Name: Carl Gordon

Title: Member

IN WITNESS WHEREOF, the parties have executed this Registration Rights Agreement as of the date first written above.

TPAV, LLC

By: /s/ Nikolay Savchuk

Name: Nikolay Savchuk Title: Managing Member

PLAN OF DISTRIBUTION¹

We are registering the shares of common stock of Onconova Therapeutics, Inc., par value of \$0.01 per share, or the Common Stock, which we refer to herein as "Shares," issued to the selling stockholders to permit the sale, transfer or other disposition of the Shares by the selling stockholders or their donees, pledgees, transferees or other successors-in-interest from time to time after the date of this prospectus. We will not receive any of the proceeds from the sale by the selling stockholders of the Shares. We will, or will procure to, bear all fees and expenses incident to our obligation to register the Shares.

The selling stockholders may sell all or a portion of the Shares beneficially owned by them and offered hereby from time to time directly or through one or more underwriters, broker-dealers or agents. If the Shares are sold through underwriters or broker-dealers, the selling stockholders will be responsible for underwriting discounts (it being understood that the selling stockholders shall not be deemed to be underwriters solely as a result of their participation in this offering) or commissions or agent's commissions. The Shares may be sold on any national securities exchange or quotation service on which the securities may be listed or quoted at the time of sale, in the over-the-counter market or in transactions otherwise than on these exchanges or systems or in the over-the-counter market and in one or more transactions at fixed prices, at prevailing market prices at the time of the sale, at varying prices determined at the time of sale, or at negotiated prices. These sales may be effected in transactions, which may involve crosses or block transactions. The selling stockholders may use any one or more of the following methods when selling Shares:

- ordinary brokerage transactions and transactions in which the broker-dealer solicits purchasers;
- block trades in which the broker-dealer will attempt to sell the shares as agent but may position and resell a portion of the block as principal to facilitate the transaction;
- to or through underwriters or purchases by a broker-dealer as principal and resale by the broker-dealer for its account;
- an exchange distribution in accordance with the rules of the applicable exchange;
- · privately negotiated transactions;
- settlement of short sales entered into after the effective date of the registration statement of which this prospectus is a part;
- broker-dealers may agree with the selling stockholders to sell a specified number of such Shares at a stipulated price per Share;
- through the writing or settlement of options or other hedging transactions, whether such options are listed on an options exchange or otherwise;
- a combination of any such methods of sale; and
- any other method permitted pursuant to applicable law.

¹ Note to Draft: Definitions to be conformed to resale S-3.

The selling stockholders also may resell all or a portion of the Shares in open market transactions in reliance upon Rule 144 under the Securities Act, as amended, or the Securities Act, as permitted by that rule, or Section 4(a)(1) under the Securities Act, if available, rather than under this prospectus, provided that they meet the criteria and conform to the requirements of those provisions.

Broker-dealers engaged by the selling stockholders may arrange for other broker-dealers to participate in sales. If the selling stockholders effect such transactions by selling Shares to or through underwriters, broker-dealers or agents, such underwriters, broker-dealers or agents may receive commissions in the form of discounts, concessions or commissions from the selling stockholders or commissions from purchasers of the Shares for whom they may act as agent or to whom they may sell as principal. Such commissions will be in amounts to be negotiated, but, except as set forth in a supplement to this Prospectus, in the case of an agency transaction will not be in excess of a customary brokerage commission in compliance with FINRA Rule 2121; and in the case of a principal transaction a markup or markdown in compliance with FINRA IM-2121.01.

In connection with sales of the Shares or otherwise, the selling stockholders may enter into hedging transactions with broker-dealers or other financial institutions, which may in turn engage in short sales of the Shares in the course of hedging in positions they assume. The selling stockholders may also sell Shares short and if such short sale takes place after the date that this Registration Statement is declared effective by the Commission, the selling stockholders may deliver Shares covered by this prospectus to close out short positions and to return borrowed Shares in connection with such short sales. The selling stockholders may also loan or pledge Shares to broker-dealers that in turn may sell such Shares, to the extent permitted by applicable law. The selling stockholders may also enter into option or other transactions with broker-dealers or other financial institutions or the creation of one or more derivative securities which require the delivery to such broker-dealer or other financial institution of shares offered by this prospectus, which shares such broker-dealer or other financial institution may resell pursuant to this prospectus (as supplemented or amended to reflect such transaction). Notwithstanding the foregoing, the selling stockholders have been advised that they may not use Shares the resale of which has been registered on this registration statement to cover short sales of our Common Stock made prior to the date the registration statement, of which this prospectus forms a part, has been declared effective by the SEC.

The selling stockholders may, from time to time, pledge or grant a security interest in some or all of the Shares owned by them and, if they default in the performance of their secured obligations, the pledgees or secured parties may offer and sell the Shares from time to time pursuant to this prospectus or any amendment to this prospectus under Rule 424(b)(3) or other applicable provision of the Securities Act of 1933, amending, if necessary, the list of selling stockholders to include the pledgee, transferee or other successors in interest as selling stockholders under this prospectus. The selling stockholders also may transfer and donate the Shares in other circumstances in which case the transferees, donees, pledgees or other successors in interest will be the selling beneficial owners for purposes of this prospectus.

The selling stockholders and any broker-dealer or agents participating in the distribution of the Shares may be deemed to be "underwriters" within the meaning of Section 2(11) of the Securities Act in connection with such sales. In such event, any commissions paid, or any discounts or concessions allowed to, any such broker-dealer or agent and any profit on the resale of the shares purchased by them may be deemed to be underwriting commissions or discounts under the Securities Act. Selling Stockholders who are "underwriters" within the meaning of Section 2(11) of the Securities Act will be subject to the applicable prospectus delivery requirements of the Securities Act including Rule 172 thereunder and may be subject to certain statutory liabilities of, including but not limited to, Sections 11, 12 and 17 of the Securities Act and Rule 10b-5 under the Securities Exchange Act of 1934, as amended, or the Exchange Act. Each selling stockholder has informed the Company that it is not a registered broker-dealer and does not have any written or oral agreement or understanding, directly or indirectly, with any person to distribute the Shares. Upon the Company being notified in writing by a selling stockholder that any material arrangement has been entered into with a broker-dealer for the sale of Common Stock through a block trade, special offering, exchange distribution or secondary distribution or a purchase by a broker or dealer, a supplement to this prospectus will be filed, if required, pursuant to Rule 424(b) under the Securities Act, disclosing (i) the name of each such selling stockholder and of the participating broker-dealer(s), (ii) the number of Shares involved, (iii) the price at which such the Shares were sold, (iv) the commissions paid or discounts or concessions allowed to such broker-dealer(s), where applicable, (v) that such broker-dealer(s) did not conduct any investigation to verify the information set out or incorporated by reference in this prospectus, and (vi) other facts material to the transaction. In no event shall any broker-dealer receive fees, commissions and markups, which, in the aggregate, would exceed eight percent (8.0%).

Under the securities laws of some U.S. states, the Shares may be sold in such states only through registered or licensed brokers or dealers. In addition, in some U.S. states the Shares may not be sold unless such shares have been registered or qualified for sale in such state or an exemption from registration or qualification is available and is complied with.

There can be no assurance that any selling stockholder will sell any or all of the Shares registered pursuant to the shelf registration statement, of which this prospectus forms a part.

Each selling stockholder and any other person participating in such distribution will be subject to applicable provisions of the Exchange Act and the rules and regulations thereunder, including, without limitation, to the extent applicable, Regulation M of the Exchange Act, which may limit the timing of purchases and sales of any of the Shares by the selling stockholder and any other participating person. To the extent applicable, Regulation M may also restrict the ability of any person engaged in the distribution of the Shares to engage in market-making activities with respect to the Shares. All of the foregoing may affect the marketability of the Shares and the ability of any person or entity to engage in market-making activities with respect to the Shares.

We will pay all expenses of the registration of the Shares pursuant to the registration rights agreement, including, without limitation, Securities and Exchange Commission filing fees and expenses of compliance with state securities or "blue sky" laws; provided, however, that each selling stockholder will pay all underwriting discounts and selling commissions, if any and any related legal expenses incurred by it. We will indemnify the selling stockholders against certain liabilities, including some liabilities under the Securities Act, in accordance with the registration rights agreement, or the selling stockholders will be entitled to contribution. We may be indemnified by the selling stockholders against certain civil liabilities under the Securities Act, that may arise from any written information furnished to us by the selling stockholders specifically for use in this prospectus, in accordance with the registration rights agreements, or we may be entitled to contribution.

SELLING STOCKHOLDER NOTICE AND QUESTIONNAIRE

The undersigned holder of shares of the (i) common stock, par value \$0.01 per share, of Onconova Therapeutics, Inc. (the "*Company*") and/or (ii) Series C Non-Voting Convertible Preferred Shares, par value \$0.01 per share, of the Company issued pursuant to a certain Securities Purchase Agreement by and among the Company and the Purchasers named therein, dated as of April 1, 2024 (the "*Agreement*"), understands that the Company intends to file with the Securities and Exchange Commission a registration statement on Form S-3 (the "*Resale Registration Statement*") for the registration and the resale under Rule 415 of the Securities Act of 1933, as amended (the "*Securities Act*"), of the Registrable Securities (as defined in the Registration Rights Agreement) in accordance with the terms of the Agreement. All capitalized terms not otherwise defined herein shall have the meanings ascribed thereto in the Agreement.

In order to sell or otherwise dispose of any Registrable Securities pursuant to the Resale Registration Statement, a holder of Registrable Securities generally will be required to be named as a selling stockholder in the related prospectus or a supplement thereto (as so supplemented, the "*Prospectus*"), deliver the Prospectus to purchasers of Registrable Securities (including pursuant to Rule 172 under the Securities Act) and be bound by the provisions of the Agreement (including certain indemnification provisions, as described below). Holders must complete and deliver this Notice and Questionnaire in order to be named as selling stockholders in the Prospectus. Holders of Registrable Securities who do not complete, execute and return this Notice and Questionnaire within five (5) Trading Days following the date of the Agreement (1) will not be named as selling stockholders in the Resale Registration Statement or the Prospectus and (2) may not use the Prospectus for resales of Registrable Securities.

Certain legal consequences arise from being named as a selling stockholder in the Resale Registration Statement and the Prospectus. Holders of Registrable Securities are advised to consult their own securities law counsel regarding the consequences of being named or not named as a selling stockholder in the Resale Registration Statement and the Prospectus.

NOTICE

The undersigned holder (the "Selling Stockholder") of Registrable Securities hereby gives notice to the Company of its intention to sell or otherwise dispose of Registrable Securities owned by it and listed below in Item (3), unless otherwise specified in Item (3), pursuant to the Resale Registration Statement. The undersigned, by signing and returning this Notice and Questionnaire, understands and agrees that it will be bound by the terms and conditions of this Notice and Questionnaire and the Agreement.

The undersigned hereby provides the following information to the Company and represents and warrants that such information is accurate and complete:

QUESTIONNAIRE

1. Name.

- (a) Full Legal Name of Selling Stockholder:
- (b) Full Legal Name of Registered Holder (if not the same as (a) above) through which Registrable Securities Listed in Item 3 below are held:
- (c) Full Legal Name of Natural Control Person (which means a natural person who directly or indirectly alone or with others has power to vote or dispose of the securities covered by the questionnaire):

2. Address for Notices to Selling Stockholder:

Telephone: Fax: Contact Person: E-mail address of Contact Person:

3. Beneficial Ownership of Registrable Securities Issuable Pursuant to the Agreement:

- (a) Type and Number of Registrable Securities beneficially owned and issued pursuant to the Agreement:
- (b) Number of Registrable Securities to be registered pursuant to this Notice for resale:

4. Broker-Dealer Status:

(a) Are you a broker-dealer?

Yes 🗆 No 🗆

(b) If "yes" to Section 4(a), did you receive your Registrable Securities as compensation for investment banking services to the Company?

Yes \square No \square

Note: If no, the Commission's staff has indicated that you should be identified as an underwriter in the Registration Statement.

(c) Are you an affiliate of a broker-dealer?

Yes \Box No \Box

Note: If yes, provide a narrative explanation below:

(d) If you are an affiliate of a broker-dealer, do you certify that you bought the Registrable Securities in the ordinary course of business, and at the time of the purchase of the Registrable Securities to be resold, you had no agreements or understandings, directly or indirectly, with any person to distribute the Registrable Securities?

Yes 🗆 No 🗆

Note: If no, the Commission's staff has indicated that you should be identified as an underwriter in the Registration Statement.

5. Beneficial Ownership of Other Securities of the Company Owned by the Selling Stockholder.

Except as set forth below in this Item 5, the undersigned is not the beneficial or registered owner of any securities of the Company other than the Registrable Securities listed above in Item 3.

Type and amount of other securities beneficially owned:

6. Relationships with the Company:

Except as set forth below, neither the undersigned nor any of its affiliates, officers, directors or principal equity holders (owners of 5% of more of the equity securities of the undersigned) has held any position or office or has had any other material relationship with the Company (or its predecessors or affiliates) during the past three years.

State any exceptions here:

7. Plan of Distribution:

The undersigned has reviewed the form of Plan of Distribution attached as Annex A to the Registration Rights Agreement, and hereby confirms that, except as set forth below, the information contained therein regarding the undersigned and its plan of distribution is correct and complete.

State any exceptions here:

The undersigned agrees to promptly notify the Company of any inaccuracies or changes in the information provided herein that may occur subsequent to the date hereof and prior to the effective date of any applicable Resale Registration Statement. All notices hereunder and pursuant to the Agreement shall be made in writing, by hand delivery, confirmed or facsimile transmission, first-class mail or air courier guaranteeing overnight delivery at the address set forth below. In the absence of any such notification, the Company shall be entitled to continue to rely on the accuracy of the information in this Notice and Questionnaire.

By signing below, the undersigned consents to the disclosure of the information contained herein in its answers to Items (1) through (7) above and the inclusion of such information in the Resale Registration Statement and the Prospectus. The undersigned understands that such information will be relied upon by the Company in connection with the preparation or amendment of any such Registration Statement and the Prospectus.

By signing below, the undersigned acknowledges that it understands its obligation to comply, and agrees that it will comply, with the provisions of the Exchange Act and the rules and regulations thereunder, particularly Regulation M in connection with any offering of Registrable Securities pursuant to the Resale Registration Statement. The undersigned also acknowledges that it understands that the answers to this Questionnaire are furnished for use in connection with Registration Statements filed pursuant to the Registration Rights Agreement and any amendments or supplements thereto filed with the Commission pursuant to the Securities Act.

The undersigned hereby acknowledges and is advised of the following Question 239.10 of the Securities Act Rules Compliance and Disclosure Interpretations regarding short selling:

"An Issuer filed a Form S-3 registration statement for a secondary offering of common stock which is not yet effective. One of the selling stockholders wanted to do a short sale of common stock "against the box" and cover the short sale with registered shares after the effective date. The issuer was advised that the short sale could not be made before the registration statement become effective, because the shares underlying the short sale are deemed to be sold at the time such sale is made. There would, therefore, be a violation of Section 5 if the shares were effectively sold prior to the effective date."

By returning this Questionnaire, the undersigned will be deemed to be aware of the foregoing interpretation.

I confirm that, to the best of my knowledge and belief, the foregoing statements (including without limitation the answers to this Questionnaire) are correct.

IN WITNESS WHEREOF the undersigned, by authority duly given, has caused this Questionnaire to be executed and delivered either in person or by its duly authorized agent.

Dated:

Beneficial Owner:

By:

Name: Title:

PLEASE FAX A COPY OF THE COMPLETED AND EXECUTED NOTICE AND QUESTIONNAIRE, AND RETURN THE ORIGINAL BY OVERNIGHT MAIL, TO:

Bryan S. Keighery One Federal Street, Boston, MA 02110 Telephone No.: (617) 341-7269 Attention: Bryan S. Keighery E-mail: bryan.keighery@morganlewis.com

ONCONOVA THERAPEUTICS, INC.

EMPLOYMENT AGREEMENT

This Employment Agreement (the "<u>Agreement</u>") is entered into by and between Onconova Therapeutics, Inc., a Delaware corporation (the "<u>Company</u>") and Werner Cautreels ("<u>Employee</u>"). This Agreement shall become effective upon the Closing (as defined in that certain Agreement and Plan of Merger, by and among the Company and Trawsfynydd Therapeutics, Inc., a Delaware corporation. and the other signatories thereto (the "<u>Merger Agreement</u>")). The date on which this Agreement becomes effective is referred to herein as the "<u>Effective Date</u>." In the event that the Closing does not occur, this Agreement shall be null and void *ab initio* and neither party hereto shall have any liability or obligation hereunder.

WHEREAS, the Company desires to employ Employee and Employee desires to be so employed by the Company upon the terms and conditions hereinafter set forth.

NOW, THEREFORE, in consideration of the mutual promises and undertakings herein contained and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, and intending to be legally bound hereby, the parties hereto enter into this Agreement as follows:

1. <u>Term</u>. The initial term of this Agreement shall begin on the Effective Date, and shall continue for one year, unless sooner terminated by either party as set forth below, or until the termination of Employee's employment, if earlier. The term of this Agreement shall automatically renew for periods of one-year unless either party gives the other party written notice at least 90 days prior to the end of the then existing term or at least 90 days prior to the end of any one-year renewal period that the term of this Agreement shall not be further extended. The period commencing on the Effective Date and ending on the date on which the term of this Agreement terminates is referred to herein as the "<u>Term</u>." The Company's failure to renew this Agreement at the end of the Term shall not, by itself, constitute termination without Cause (as defined below) or Good Reason (as defined below).

2. <u>Position; Duties</u>. Subject to all the terms and conditions hereof, the Company shall employ Employee, and Employee shall serve the Company as the Chief Executive Officer, reporting to the Board of Directors of the Company (the "<u>Board</u>"). Employee shall perform all duties and accept all responsibilities incident to such position as may be reasonably assigned to Employee by the Board. Employee represents to the Company that Employee is not subject to or a party to any employment agreement, noncompetition covenant, or other agreement that would be breached by, or prohibit Employee from, executing this Agreement and performing fully Employee's duties and responsibilities hereunder.

3. <u>Best Efforts</u>. As Employee's position is a full-time position, Employee agrees to devote Employee's full-time effort, attention, and energies, from such location to be mutually agreed upon between Employee and the Board, to Employee's duties hereunder. Employee will not render any professional services or engage in any activity that might be competitive with, adverse to the best interest of, or create the appearance of a conflict of interest with the Company. Employee agrees to abide by the policies, rules and regulations of the Company as they may be amended from time to time. Employee may not engage in outside employment or consulting without first obtaining prior express permission from the Board. The foregoing shall not be construed as preventing Employee from (a) serving on civic, educational, philanthropic or charitable boards or committees, or, with the prior written consent of the Board, in its sole discretion, on corporate boards, and (b) managing personal investments, so long as such activities are permitted under the Company's code of conduct and employment policies and do not violate the provisions of this Agreement. Employee will provide to the Board a list of Employee's existing board seats of both Non-Profit organizations and board seats of private biotechnology companies for review and approval or where appropriate resignation.

4. <u>Compensation and Other Benefits</u>.

(a) <u>Salary</u>. For all services rendered by Employee under this Agreement, the Company agrees to pay Employee a base salary at an initial annualized rate of \$610,000 (the "<u>Base Salary</u>"), in installments in accordance with the Company's normal payroll cycle, less authorized deductions. Employee's Base Salary shall be reviewed annually by the Compensation Committee of the Board (the "<u>Compensation Committee</u>") or the Board pursuant to the normal performance review policies for senior-level executives and may be adjusted from time to time as determined in the sole discretion of the Board or the Compensation Committee.

(b) <u>Discretionary Annual Bonus</u>. During the Term, Employee shall be eligible to receive a discretionary annual bonus, as determined by the Board or the Compensation Committee in its sole discretion ("<u>Annual Bonus</u>"). The target of this annual bonus will be 50% of the Base Salary, and will be evaluated on the basis of pre-set annual bonus goals. Any Annual Bonus shall be paid after the end of the fiscal year to which it relates, at the same time and under the same terms and conditions as the bonuses for other executives of the Company; provided that in no event shall Employee's Annual Bonus, if any, be paid later than two and a half months after the last day of the fiscal year to which the Annual Bonus relates. Any Annual Bonus may be paid to Employee in the form of cash, stock options, shares of the Company's common stock ("<u>Common Stock</u>"), or a combination thereof, at the Board's or the Compensation Committee's discretion.

(c) <u>Equity Awards</u>. Subject to the approval of the Board or the Compensation Committee, which has already been obtained contingent on Employee's commencement of employment, as an inducement for Employee to join the Company in the role of Chief Executive Officer and agree to the restrictive covenants set forth below, Employee will be granted 200,000 restricted stock units, subject to the terms of the restricted stock unit agreement for inducement grants provided by the Company ("<u>Restricted Stock Units</u>"). The Restricted Stock Units will vest as to 25% on the first anniversary of the Effective Date and the remainder will vest in substantially equal annual installments for three years thereafter. The Restricted Stock Units are intended to be inducement awards under Rule 5635(c)(4) of the Nasdaq Stock Market Listing Rules and will be granted outside of the Onconova Therapeutics, Inc. 2021 Incentive Compensation Plan (the "<u>Plan</u>"). Although granted as inducement awards outside of the Plan, the Restricted Stock Units shall be subject to the terms of the Plan as if issued thereunder.

-2-

(d) <u>Employee Benefits</u>. During the Term, Employee shall be eligible to participate in any employee benefit plans or programs of the Company that are made generally available from time to time by the Company to similarly situated employees, including, but not limited to, health insurance, a flexible spending account, and 401(k) participation. Nothing in this Agreement shall preclude the Company or any affiliate of the Company from terminating or amending any employee benefit plan or program from time to time after the Effective Date.

(e) <u>Vacation and Holidays</u>. During Employee's employment hereunder, Employee shall be entitled each year to four weeks of vacation, and holiday and sick leave at levels commensurate with those provided to other senior executives of the Company, in accordance with the Company's vacation, holiday and other pay-for-time-not-worked policies. Vacation shall be taken by Employee at such time or times as are mutually convenient to Employee and the Company.

(f) <u>Reimbursement of Expenses</u>. The Company shall reimburse Employee for all reasonable expenses incurred by Employee in connection with Employee's employment hereunder provided, however, that such expenses were incurred in conformance with the policies of the Company, as established from time to time, and that Employee submits detailed vouchers and other records reasonably required by the Company in support of the amount and nature of such expense.

5. <u>Termination of Employment</u>.

(a) <u>Death of Employee</u>. If Employee dies during the term of this Agreement, this Agreement shall terminate immediately and the Company shall pay to Employee's then-current spouse, if such spouse survives Employee, or if not, to Employee's estate, the balance of any accrued and unpaid Base Salary, unreimbursed expenses, and unused accrued vacation time through the termination date.

(b) <u>Disability of Employee</u>. If Employee incurs a Disability during the term of this Agreement, the Company may terminate Employee's employment on or after the date of Disability. If Employee's employment terminates on account of Disability, the Company shall pay to Employee's the balance of any accrued and unpaid Base Salary, unreimbursed expenses, and unused, accrued vacation time through the termination date. For purposes of this Agreement, the term "<u>Disability</u>" shall mean Employee is eligible to receive long-term disability benefits under the Company's long-term disability plan.

(c) <u>Termination for Cause.</u> If Employee's employment is terminated by the Company for Cause (as defined below), the Company shall pay Employee only the balance of Employee's accrued, but unpaid Base Salary, unreimbursed expenses, and unused, accrued vacation time through the termination date. Employee's employment may be terminated for "Cause" at any time upon delivery of written notice to Employee.

-3-

(d) <u>Termination by the Company without Cause or by Employee for Good Reason</u>. If Employee's employment by the Company ceases due to a termination by the Company without Cause or a resignation by Employee for Good Reason (as defined below), the Company shall:

(i) pay to Employee the balance of Employee's accrued, but unpaid Base Salary, unreimbursed expenses, and unused, accrued vacation time through the termination date;

(ii) to the extent then approved, accrued and unpaid, pay to Employee the Annual Bonus (if any) with respect to the fiscal year ended immediately prior to the cessation of Employee's employment, which such Annual Bonus shall be paid at the time such Annual Bonus would have otherwise been paid absent Employee's cessation of employment;

(iii) pay to Employee,

(A) in the event Employee's employment by the Company ceases due to a termination by the Company without Cause or by Employee for Good Reason other than during the Change in Control Protection Period (as defined below), monthly severance payments equal to one-twelfth of the sum of (1) Employee's then current Base Salary, plus (2) an amount equal to the Target Bonus for the fiscal year during which Employee's employment by the Company ceases, which severance payments shall be paid for the duration of the Severance Period (as defined below) in accordance with the Company's usual payroll practices, commencing within 60 days following the termination date, and the first payment shall include any unpaid installments from the termination date until the first payment date; or

(B) in the event Employee's employment by the Company ceases due to a termination by the Company without Cause or by Employee for Good Reason during the Change in Control Protection Period, a severance payment amount equal to the sum of one and one-half times (1) Employee's then current Base Salary plus (2) an amount equal to the Target Bonus for the fiscal year during which Employee's employment by the Company ceases, which severance payment shall be paid in a lump sum payment within 60 days following the termination date; provided that such payment shall be made in installments as set forth in Section 5(d)(iii)(A) above if the Change in Control is not a "change in control event" as defined under Section 409A of the Internal Revenue Code of 1986, as amended (the "Code") and the severance payments hereunder are not exempt from Section 409A of the Code;

(iv) cause any outstanding unvested equity awards on Common Stock that vest based on continued service previously awarded to Employee to become fully vested as of the date of Employee's termination of employment pursuant to this Section 5(d); and

(v) provided that Employee is eligible for and timely elects continuation coverage under COBRA, the Company will pay on Employee's behalf the COBRA premiums for continued health care coverage under the Company's group health plans for Employee and Employee's eligible dependents ("<u>COBRA Payments</u>"). The Company will pay the COBRA Payments for the period from Employee's termination date until the earliest to occur of (i) the end of the Severance Period; (ii) the date Employee becomes eligible for group health insurance coverage through a subsequent employer; or (iii) the date Employee ceases to be eligible for COBRA coverage for any reason (each of the events described in (ii) or (iii) in this Section 5(d) shall be referred to herein as a "<u>Disqualifying Event</u>"). Employee is required to notify the Company within five days of becoming aware that a Disqualifying Event has occurred or will occur. The COBRA health care continuation coverage period under Section 4980B of the Internal Revenue Code of 1986, as amended (the "<u>Code</u>"), shall run concurrently with the period during which the Company pays the COBRA Payments.

Except as otherwise provided in this Section 5, all compensation and benefits will cease at the time of Employee's cessation of employment and the Company will have no further liability or obligation by reason of such cessation of employment. The payments and benefits set forth in Section 5(d)(iii), (iv) and (v) shall only be paid if Employee signs and does not revoke a release and waiver of claims in a form approved by the Company, and such release becomes effective and irrevocable within 60 days of Employee's cessation of employment, and Employee continues to comply with Sections 7, 8 and 9 in this Agreement.

(e) <u>Voluntary Resignation other than for Good Reason</u>. Employee may voluntarily resign from employment with the Company at any time. In the event Employee voluntarily resigns from Employee's employment with the Company other than for Good Reason, Employee shall provide the Company with at least 30 days' notice of Employee's intent to resign. The Company shall pay Employee only the balance of Employee's accrued, but unpaid Base Salary, unreimbursed expenses, and unused, accrued vacation time through the termination date.

(f) <u>Resignation of Positions</u>. Effective as of the date of any termination of employment, Employee will resign from all Companyrelated positions, including as an officer and director of the Company and its parents, subsidiaries and affiliates.

(g) <u>No Other Severance</u>. Upon termination of Employee's employment for any reason, the Company will have no severance obligations under this Agreement other than as provided in this Section 5, which shall supersede any prior or contemporaneous oral or written severance plan, policy, program, or other arrangement maintained by the Company to the extent such benefits would provide for duplication of benefits to Employee.

(h) <u>Definitions</u>. For purposes of this Agreement:

(i) "Cause" shall mean the occurrence of any of the following events: (1) any gross failure on the part of Employee (other than by reason of Disability as provided in Section 5(b)) to faithfully and professionally carry out Employee's duties or to comply with any other material provision of this Agreement, which failure continues after written notice thereof by the Board, provided that the Company shall not be required to provide such notice in the event that such failure (A) is not susceptible to remedy or (B) relates to the same type of acts or omissions as to which such notice has been given on a prior occasion; (2) Employee's dishonesty (which shall include without limitation any misuse or misappropriation of the Company's assets), or other willful misconduct (including without limitation any conduct on the part of Employee intended to or likely to injure the business of the Company); (3) Employee's conviction for any felony or for any other crime involving moral turpitude, whether or not relating to Employee's employment; (4) in accordance with applicable federal, state or local laws, Employee's insobriety or use of illegal drugs, chemicals or controlled substances either (A) in the course of performing Employee's duties and responsibilities under this Agreement, or (B) otherwise affecting the ability of Employee to perform the same; (5) Employee's failure to comply with a lawful written direction of the Board; (6) Employee's failure to comply with the Company's code of conduct or employment policies; (7) any wanton and willful dereliction of duties by Employee; or (8) Employee's breach of this Agreement. The existence of any of the foregoing events or conditions shall be determined by the Board in the exercise of its reasonable judgment.

-5-

- (ii) "Change in Control" shall have the meaning ascribed to it under the Plan or its successor then in effect.
- (iii) "Change in Control Protection Period" shall mean the 12-month period following a Change in Control.

(iv) "<u>Good Reason</u>" shall mean: (1) the breach by the Company of any material provision of this Agreement (provided, however, that a reduction in Employee's Base Salary by less than 20% in and for any 12-month period shall not be a material breach by the Company if it is made in connection with a reduction in base salaries imposed on a majority of other senior executives of the Company and Employee's Base Salary is not reduced by a percentage that is greater than the percentage by which the base salaries of a majority of other senior executives of the Company is reduced in and for that same 12-month period); or (2) following a Change in Control (A) a material diminution in Employee's Base Salary or Target Bonus or (B) a material diminution in Employee's authorities, duties and responsibilities. None of the foregoing events or conditions will constitute Good Reason unless Employee provides the Company with written objection to the event or condition within 30 days of receiving that written objection, and Employee resigns Employee's employment within 30 days following the expiration of that cure period.

(v) "<u>Severance Period</u>" shall mean (1) the 12-month period immediately following the date Employee's employment with the Company ceases due to a termination by the Company without Cause or by Employee for Good Reason other than during the Change in Control Protection Period or (2) the 18-month period immediately following the date Employee's employment with the Company ceases due to a termination by the Company without Cause or by Employee for Good Reason other than during the Change in Control Protection Period.

6. <u>Code Section 280G</u>. In the event of a change in ownership or control under Section 280G of the Code, if it shall be determined that any payment or distribution in the nature of compensation (within the meaning of Section 280G(b)(2) of the Code) to or for the benefit of Employee, whether paid or payable or distributed or distributable pursuant to the terms of this Agreement or otherwise (a "<u>Payment</u>"), would constitute an "excess parachute payment" within the meaning of Section 280G of the Code, the aggregate present value of the Payments under the Agreement shall be reduced (but not below zero) to the Reduced Amount (defined below) if and only if the Accounting Firm (described below) determines that the reduction will provide Employee with a greater net after-tax benefit than would no reduction. No reduction shall be made unless the reduction would provide Employee with a greater net after-tax benefit. The determinations under this Section shall be made as follows:

-6-

(a) The "<u>Reduced Amount</u>" shall be an amount expressed in present value which maximizes the aggregate present value of Payments under this Agreement without causing any Payment under this Agreement to be subject to the Excise Tax (defined below), determined in accordance with Section 280G(d)(4) of the Code. The term "<u>Excise Tax</u>" means the excise tax imposed under Section 4999 of the Code, together with any interest or penalties imposed with respect to such excise tax.

(b) Payments under this Agreement shall be reduced on a nondiscretionary basis in such a way as to minimize the reduction in the economic value deliverable to Employee. Where more than one payment has the same value for this purpose and they are payable at different times, they will be reduced on a pro rata basis.

(c) All determinations to be made under this Section shall be made by an independent certified public accounting firm selected by the Company and agreed to by Employee immediately prior to the change-in-ownership or -control transaction (the "<u>Accounting Firm</u>"). The Accounting Firm shall provide its determinations and any supporting calculations both to the Company and Employee within 10 days of the transaction. Any such determination by the Accounting Firm shall be binding upon the Company and Employee. All of the fees and expenses of the Accounting Firm in performing the determinations referred to in this Section shall be borne solely by the Company.

7. Non-Competition.

(a) For purposes of this Agreement, "<u>Competitor</u>" shall mean any person, company, or entity whose primary business at the time is or whose then-current business plan contemplates engaging in activities, which may be competitive with products and services that were or were being designed, conceived, marketed, sold, distributed and/or developed by the Company during Employee's employment by the Company or at the time of termination of Employee's employment by the Company.

(b) Employee agrees that so long as Employee is employed by the Company, and for a period of 12 months after the termination of Employee's employment for any reason whatsoever, Employee will not, directly or indirectly, whether for compensation or not, own, manage, operate, join, control, work for, or participate in, or be connected as a stockholder, officer, employee, partner, creditor, guarantor, advisor or otherwise, with a Competitor. The foregoing shall not be construed, however, as preventing Employee from investing Employee's assets in such form or manner as will not require services on the part of Employee in the operations of the businesses in which such investments are made, provided that any such business is publicly owned and the interest of Employee therein is solely that of an investor owning not more than 5% of the outstanding equity securities of any such business. Should Employee breach the provisions of this Section, the Company shall, in addition to any equitable or legal relief to which it is otherwise entitled, be entitled to cease all payments and benefits under the terms of this Agreement and shall be entitled to pursue all remedies it might have including, but not limited to, those contained in this Agreement.

(c) For the period of 12 months after the termination of this Agreement for any reason whatsoever, Employee shall not hire, retain or engage as a director, officer, employee, agent or in any other capacity any person or persons who are employed by the Company or who were at any time (within a period of six months immediately prior to the date of Employee's termination) employed by the Company or otherwise interfere with the relationship between such persons and the Company.

-7-

(d) If the period of time or area herein specified should be adjudged unreasonable in any court proceeding, then the period of time shall be reduced by such number of months or the area shall be reduced by elimination of such portion thereof as deemed unreasonable, so that this covenant may be enforced during such period of time and in such area as is adjudged to be reasonable.

8. <u>Confidential Information</u>.

(a) Subject to Section 8(f), at all times during Employee's employment and thereafter, Employee will hold in strictest confidence and will not disclose, use, lecture upon or publish any of the Company's Proprietary Information (defined below), except as such use may be required in connection with Employee's work for the Company, or unless an officer of the Company expressly authorizes such disclosure in writing. Employee will obtain the Company's written approval before publishing or submitting for publication any material (written, verbal, or otherwise) that relates to Employee's work for the Company and/or incorporates any Proprietary Information. Employee hereby assigns to the Company any rights Employee may have or acquire in such Proprietary Information and recognizes that all Proprietary Information shall be the sole property of the Company and its assigns.

(b) The term "<u>Proprietary Information</u>" shall mean any and all confidential and/or proprietary knowledge, data or information of the Company, whether acquired by Employee while employed by the Company, during Employee's prior service as a consultant to the Company, or otherwise. By way of illustration but not limitation, "<u>Proprietary Information</u>" includes, but is not limited to (i) trade secrets, inventions, mask works, ideas, methods, processes, formulas, chemical structures and methods for chemical synthesis, structure-activity relationships, assay methodologies, characteristics, equipment and equipment designs, results, formulations and biological, pharmacological, toxicological and clinical data, physical, chemical or biological materials, source and object codes, data, programs, other works of authorship, know-how, improvements, discoveries, developments, compilations, shop practices, supplier lists, designs and techniques (hereinafter collectively referred to as "<u>Inventions</u>"); and (ii) information regarding plans for research, development, new products, marketing and selling, business plans, budgets and unpublished financial statements, licenses, prices and costs, suppliers and customers; and (iii) information regarding the skills and compensation of other employees of the Company. Notwithstanding the foregoing, it is understood that, at all times, Employee is free to use information which is generally known in the trade or industry, which is not gained as a result of a breach of this Agreement, and which is acquired as a result of Employee's own skill, knowledge, know-how and experience.

(c) Employee understands, in addition, that the Company has received and in the future will receive from third parties confidential or proprietary information ("<u>Third Party Information</u>") subject to a duty on the Company's part to maintain the confidentiality of such information and to use it only for certain limited purposes. Subject to Section 8(f), during the period of Employee's employment and thereafter, Employee will hold Third Party Information in the strictest confidence and will not disclose to anyone (other than Company personnel who need to know such information in connection with their work for the Company) or use, except in connection with Employee's work for the Company, Third Party Information unless expressly authorized by an officer of the Company in writing.

-8-

(d) During Employee's employment by the Company, Employee will not improperly use or disclose any confidential information or trade secrets, if any, of any of Employee's former employers or any other person to whom Employee has an obligation of confidentiality, and Employee will not bring onto the premises of the Company any unpublished documents or any property belonging to any former employer or any other person to whom Employee has an obligation of confidentiality, unless such action is consented to in writing by all persons to whom the relevant obligation of confidentiality is owed. Employee shall not work on Company projects on the grounds of, or using the equipment of, any third party, unless such work is agreed to by the Company in writing.

(e) Upon termination of Employee's employment, Employee shall return to the Company all Proprietary Information in any tangible form in Employee's possession, including copies thereof.

(f) Nothing in this Agreement shall prohibit or restrict Employee from initiating communications directly with, responding to any inquiries from, providing testimony before, providing confidential information to, reporting possible violations of law or regulation to, or from filing a claim or assisting with an investigation directly with a self-regulatory authority or a government agency or entity, including the U.S. Equal Employment Opportunity Commission, the Department of Labor, the National Labor Relations Board, the Department of Justice, the Securities and Exchange Commission, Congress, and any agency Inspector General (collectively, the "<u>Regulators</u>"), or from making other disclosures that are protected under the whistleblower provisions of state or federal law or regulation. Employee does not need the prior authorization of the Company to engage in such communications, respond to such inquiries, provide confidential information or documents to the Regulators, or make any such reports or disclosures to the Regulators. Employee is not required to notify the Company that Employee has engaged in such communications with the Regulators. If Employee is required by law to disclose Proprietary Information, other than to Regulators as described above, Employee shall give prompt written notice to the Company so as to permit the Company to protect its interests in confidentiality to the extent possible. Federal law provides criminal and civil immunity to federal and state claims for trade secret misappropriation to individuals who disclose a trade secret to their attorney, a court, or a government official in certain, confidential circumstances that are set forth at 18 U.S.C. §§ 1833(b)(1) and 1833(b)(2), related to the reporting or investigation of a suspected violation of the law, or in connection with a lawsuit for retaliation for reporting a suspected violation of the law.

-9-

9. Intellectual Property.

(a) Employee has provided on <u>Appendix A</u> (Previous Inventions) attached hereto a complete list of all Inventions that Employee has, alone or jointly with others, conceived, developed or reduced to practice or caused to be conceived, developed or reduced to practice prior to the commencement of Employee's employment with the Company, that Employee considers to be Employee's property or the property of third parties, and that Employee wishes to have excluded from the scope of the assignment under Section 9(b) (collectively referred to as "Prior Inventions"). If disclosure of any such Prior Invention would cause Employee to violate any prior confidentiality agreement, Employee understands that Employee shall not list such Prior Inventions in <u>Appendix A</u> but shall only disclose a cursory name for each such invention (bearing in mind that where necessary the naming shall not be so specific as to violate the confidentiality obligation), a listing of the party(ies) to whom the invention belongs, and the fact that full disclosure as to such invention has not been made for that reason. Space is provided on <u>Appendix A</u> for this purpose. Notwithstanding the foregoing, Employee agrees that Employee will not incorporate, or permit to be incorporated, Prior Inventions in any Company Inventions without the Company's prior written consent and, furthermore, Employee shall not incorporate a Prior Invention into a Company product, process or machine without having the ability to make the grant set forth in the foregoing. If, in the course of Employee's employment with the Company, Employee incorporates or uses a Prior Invention into or with a Company product, process or machine, the Company is hereby granted and shall have a nonexclusive, fully-paid up, royalty-free, irrevocable, perpetual, worldwide license (with rights to sublicense through multiple tiers of sublicensees) to internally and externally reproduce, display, perform, distribute, make, have made, modify, use, import, sell and offer to sell such Prior Invention as

(b) To the extent not vested in the Company by operation of law under the United States copyright law, subject to Section 9(j) below, and to the maximum extent permitted by law, Employee agrees to assign and hereby does assign to the Company all of Employee's rights, title and interest in and to any and all Inventions, whether or not patentable or registerable under patent, intellectual property, copyright or similar statutes, made or conceived or reduced to practice or learned by Employee, either alone or jointly with others, (i) within the scope of or resulting from employment with the Company (whether or not during normal business hours or on or off Company premises), (ii) during the period of Employee's employment with the Company, including in the future (e.g., when any such Inventions are first reduced to practice or a description thereof first fixed in a tangible medium, as applicable) relating to the Company's actual or anticipated business, operations, research, products, services, activities, investigations or development, or (iii) using, or resulting from any use of, the Company's equipment, supplies, facilities, property, time, data or Proprietary Information (collectively, "<u>Work Product</u>") as well as all copyrights, know-how, trade secrets, trademarks, patents, the right of priority, the right of publication, trademarks and all other intellectual property or proprietary rights in or to the Work Product, "<u>Company Inventions</u>") free and clear of all liens and encumbrances and without further consideration.

(c) During the period of Employee's employment, Employee will promptly disclose to the Company fully and in writing all Inventions authored, conceived or reduced to practice by Employee, either alone or jointly with others. In addition, Employee will promptly disclose to the Company all patent applications filed by Employee or on Employee's behalf during Employee's employment and within one year after termination of employment. At the time of each such disclosure, Employee will advise the Company in writing of any Inventions that Employee believes qualify for exclusion from Employee's obligation to assign hereunder; and Employee will at that time provide to the Company in writing all evidence necessary to substantiate that belief.

-10-

(d) As directed by the Company, Employee agrees to assign all Employee's rights, title and interest in and to any particular Company Invention to a third party, including without limitation the United States.

(e) Employee acknowledges that all original works of authorship which are prepared by Employee (solely or jointly with others) within the scope of Employee's employment and which are protectable by copyright are "works made for hire," pursuant to United States Copyright Act (17 U.S.C. § 101).

(f) Employee will assist the Company in every proper way to obtain, and from time to time enforce, United States and foreign trade secret, patent, copyright, mask work and other intellectual property rights ("<u>Proprietary Rights</u>") relating to Company Inventions in any and all countries. To that end, Employee will execute, verify and deliver such documents and perform such other acts (including appearances as a witness) as the Company may reasonably request for use in applying for, obtaining, perfecting, evidencing, sustaining and enforcing such Proprietary Rights and the assignment thereof. In addition, Employee will execute, verify and deliver assignments of such Proprietary Rights to the Company, its successor in interest, or its designee. Employee's obligation to assist the Company with respect to Proprietary Rights relating to such Company Inventions in any and all countries shall continue beyond the termination of Employee's employment.

In the event the Company is unable for any reason, after reasonable effort, to secure Employee's signature on any document needed in connection with the actions specified in this Section 9(f), Employee hereby irrevocably designates and appoints the Company and its duly authorized officers and agents as Employee's agent and attorney-in-fact, which appointment is coupled with an interest, to act for and on Employee's behalf to execute, verify and file any such documents and to do all other lawfully permitted acts to further the purposes of the preceding paragraph with the same legal force and effect as if executed by Employee.

(g) Employee agrees to keep and maintain adequate and current records (in the form of notes, sketches, drawings and in any other form that may be required by the Company) of all Proprietary Information developed by Employee and all Inventions made by Employee during the period of Employee's employment at the Company, which records shall be available to and remain the sole property of the Company at all times.

(h) Employee represents that Employee's performance of all the terms of this Agreement and as an employee of the Company does not and will not breach any agreement to keep in confidence information acquired by Employee in confidence or in trust prior to Employee's employment by the Company. Employee has not entered into, and Employee agrees that Employee will not enter into, any agreement either written or oral in conflict herewith.

(i) To the maximum extent permitted by law, the Company, its affiliates and its/their designees may create photographs and recordings of Employee in connection with any Company Invention or any products, services, operation or business of the Company or its affiliate and may reproduce, use, modify, display, publish and distribute Employee's names, voice, signature, photographs, images, substantially similar imitation, likeness and the like captured, produced or provided in the course of engagement or employment with the Company as well as their derivative works and copies (collectively, "<u>Likenesses</u>"), publicly and perpetually, in connection with any products, services, operation or business of the Company or its affiliate. Employee waives any attribution, integrity and moral rights in relation to any Company Invention or the Company's products, services, operation or business. To the maximum extent permitted by law, Employee waives and agrees not to assert any and all rights and claims Employee may have, before or after the date hereof, with respect to any Company Inventions, any Prior Inventions or any Likenesses.

-11-

(j) The provisions in Section 9 are applicable only to the maximum extent permitted by applicable law. This Agreement does not apply to an invention to the extent applicable to Employee and excluded by applicable law. 19 Delaware Code Section 805 provides in part: "Any provision in an employment agreement which provides that the employee shall assign or offer to assign any of the employee's rights in an invention to the employee developed entirely on the employee's own time without using the employer's equipment, supplies, facility or trade secret information, except for those inventions that: (1) Relate to the employer's business or actual or demonstrably anticipated research or development; or (2) Result from any work performed by the employee for the employer."

10. <u>Remedies</u>. Because Employee's services are personal and unique and because Employee may have access to and become acquainted with the Proprietary Information of the Company, the Company shall have the right to enforce this Agreement and any of its provisions by injunction, or other equitable relief, without bond (if allowed by applicable law), and without prejudice to any other rights and remedies that the Company may have for a breach of this Agreement. In the event that Employee performs services for other entities while employed by the Company or leaves the employ of the Company, Employee hereby consents to the notification of Employee's new employer of Employee's rights and obligations under this Agreement.

11. <u>Arbitration</u>. Any and all disputes between the parties (except actions to enforce the provisions of Sections 7, 8 or 9 of this Agreement), arising under or relating to this Agreement or any other dispute arising between the parties, including claims arising under any employment discrimination laws, shall be adjudicated and resolved exclusively through binding arbitration before the American Arbitration Association pursuant to the American Arbitration Association's then-in-effect National Rules for the Resolution of Employment Disputes (hereafter "Rules"). The initiation and conduct of any arbitration hereunder shall be in accordance with the Rules and each side shall bear its own costs and counsel fees in such arbitration. Any arbitration hereunder shall be conducted in Philadelphia, Pennsylvania, and any arbitration award shall be final and binding on the Parties. The arbitrator shall have no authority to depart from, modify, or add to the written terms of this Agreement. The arbitration provisions of this Section 11 shall be interpreted according to, and governed by, the Federal Arbitration Act, 9 U.S.C. § 1 *et seq.*, and any action pursuant to such Act to enforce any rights hereunder shall be brought exclusively in the United States District Court for the Eastern District of Pennsylvania. The parties consent to the jurisdiction of (and the laying of venue in) such court.

-12-

12. <u>General Indemnification</u>. In the event Employee is made, or threatened to be made, a party to any legal action or proceeding, whether civil or criminal, including any governmental or regulatory proceedings or investigations, by reason of the fact that Employee is or was a director or officer of the Company or its affiliates, Employee shall be indemnified by the Company, and the Company shall pay Employee's related expenses when and as incurred, to the fullest extent permitted by the laws of the state of Delaware and the Company's articles of incorporation and bylaws. During Employee's employment with the Company and its affiliates and after termination of employment for any reason, the Company shall cover Employee under the Company's directors and officers insurance applicable to other officers and directors.

13. <u>Taxes and Withholding</u>. All compensation payable and other benefits provided under this Agreement shall be subject to customary and legally required withholding for income and other employment taxes. Employee shall bear all expense of, and be solely responsible for, all federal, state and local taxes due with respect to any payment received under this Agreement.

14. <u>Severability</u>. The terms of this Agreement and each Section hereof shall be considered severable and the invalidity or unenforceability of any part thereof shall not affect the validity or enforceability of the remaining portions or provisions hereof.

15. <u>Attorneys' Fees</u>. The Company shall pay or reimburse Employee up to \$10,000 for reasonable attorneys' fees incurred by Employee in connection with the review, negotiation and documentation of this Agreement, within 30 days following presentation of appropriate receipts for such fees.

16. <u>Notices</u>. Any notice required or permitted to be given under this Agreement shall be sufficient, if in writing and delivered by registered or certified mail or overnight delivery service to Employee's residence in the case of Employee, or to its principal office in the case of the Company.

17. <u>Assignment</u>. The rights and obligations of the Company under this Agreement shall inure to the benefit of and be binding upon its successors and assigns. Neither this Agreement nor any rights or interests herein or created hereby may be assigned or otherwise transferred voluntarily or involuntarily by Employee.

18. <u>Waiver</u>. The waiver by the Company or Employee of a breach of any provision of this Agreement by the other shall not operate or be construed as a waiver of any subsequent breach.

19. <u>Applicable Law</u>. This Agreement shall be interpreted and construed under the laws of the Commonwealth of Pennsylvania without regard to rules governing conflicts of law.

20. <u>Entire Agreement and Interpretation</u>. This instrument contains the entire agreement of the parties with respect to the subject matter hereof and supersedes any and all prior or contemporaneous agreements, oral or written, concerning the subject matter contained herein, including without limitation any prior agreements between the Company and Employee. It may not be changed or altered, except by an agreement in writing signed by the party against whom enforcement of any waiver, change, modification, extension or discharge is sought. Headings are used for reference only and shall not be considered when interpreting this Agreement. Each of the terms "include," "includes," and "including" is deemed to be followed by "without limitation".

-13-

21. Section 409A of the Code.

(a) This Agreement is intended to comply with Section 409A of the Code, and its corresponding regulations, or an exemption thereto, and payments may only be made under this Agreement upon an event and in a manner permitted by Section 409A of the Code, to the extent applicable. Severance benefits under this Agreement are intended to be exempt from Section 409A of the Code under the "short-term deferral" exception, to the maximum extent applicable, and then under the "separation pay" exception, to the maximum extent applicable. Notwithstanding anything in this Agreement to the contrary, if required by Section 409A of the Code, if Employee is considered a "specified employee" for purposes of Section 409A of the Code and if payment of any amounts under this Agreement is required to be delayed for a period of six months after separation from service pursuant to Section 409A of the Code, payment of such amounts shall be delayed as required by Section 409A of the Code, and the accumulated amounts shall be paid in a lump-sum payment within 10 days after the end of the six-month period. If Employee dies during the postponement period prior to the payment of benefits, the amounts withheld on account of Section 409A of the Code shall be paid to the personal representative of Employee estate within 60 days after the date of Employee's death.

(b) All payments to be made upon a termination of employment under this Agreement may only be made upon a "separation from service" under Section 409A of the Code. For purposes of Section 409A of the Code, each payment hereunder shall be treated as a separate payment, and the right to a series of installment payments under this Agreement shall be treated as a right to a series of separate payments. In no event may Employee, directly or indirectly, designate the fiscal year of a payment. Notwithstanding any provision of this Agreement to the contrary, in no event shall the timing of Employee execution of the Release, directly or indirectly, result in Employee designating the fiscal year of payment of any amounts of deferred compensation subject to Section 409A of the Code, and if a payment that is subject to execution of the Release could be made in more than one taxable year, payment shall be made in the later taxable year.

(c) All reimbursements and in-kind benefits provided under this Agreement shall be made or provided in accordance with the requirements of Section 409A of the Code, including, where applicable, the requirement that (i) any reimbursement be for expenses incurred during the period specified in this Agreement, (ii) the amount of expenses eligible for reimbursement, or in-kind benefits provided, during a fiscal year not affect the expenses eligible for reimbursement, or in-kind benefits to be provided, in any other fiscal year, (iii) the reimbursement of an eligible expense be made no later than the last day of the fiscal year following the year in which the expense is incurred, and (iv) the right to reimbursement or in-kind benefits not be subject to liquidation or exchange for another benefit.

-14-

22. <u>Clawback Policy</u>. Employee acknowledges and agrees that to the extent permitted under applicable law, all amounts payable under this Agreement are subject to the terms of any applicable Clawback Policy(as defined below) and, to the extent permitted by applicable law, including without limitation Section 409A of the Code, all amounts payable under this Agreement are subject to offset in the event that Employee has an outstanding clawback, recoupment or forfeiture obligation to the Company under the terms of any applicable Clawback Policy. In the event of a clawback, recoupment or forfeiture event under an applicable Clawback Policy, the amount required to be clawed back, recouped or forfeited pursuant to such policy shall be deemed not to have been earned under the terms of this Agreement or otherwise, and the Company shall be entitled to recover from Employee the amount specified under the policy to be clawed back, recouped or forfeiture policy approved by the Board (or a committee thereof), as in effect from time to time, whether approved before or after the Effective Date. Employee acknowledges and agrees that Employee will be bound by the terms of any such Clawback Policy as if it were set forth in this Agreement.

23. <u>Counterparts</u>. This Amendment may be executed in one or more counterparts, each of which shall for all purposes be deemed to be an original and all of which shall constitute the same instrument. Any and all counterparts may be executed by facsimile.

[signature page follows]



IN WITNESS WHEREOF, the parties hereto have executed this Agreement as of the day and year first above written.

ONCONOVA THERAPEUTICS, INC.

By: /s/ Mark Guerin

Mark Guerin CFO

EMPLOYEE: WERNER CAUTREELS

> /s/ Werner Cautreels (Signature)

[Signature Page to Onconova Therapeutics, Inc. Employment Agreement]

Appendix A

TO:	
FROM:	Werner Cautreels
DATE:	April 1, 2024
SUBJECT:	PREVIOUS INVENTIONS

1. Except as listed in Section 2 below, the following is a complete list of all inventions or improvements relevant to the subject matter of my employment by Onconova Therapeutics, Inc. (the "<u>Company</u>") that have been made or conceived or first reduced to practice by me alone or jointly with others prior to my engagement by the Company:

- \boxtimes No inventions or improvements.
- \Box See below:
- \Box Additional sheet(s) attached.
- 2. Due to a prior confidentiality agreement, I cannot complete the disclosure under Section 1 above with respect to inventions or improvements generally listed below, the proprietary rights and duty of confidentiality with respect to which I owe to the following party(ies):

 \Box Additional sheet(s) attached.

[Date]

[Name of Prospective Employee] [Street Address of Prospective Employee] [City, State Zip of Prospective Employee]

Dear [Name of Prospective Employee],

On behalf of Onconova Therapeutics, Inc. (the "Company"), we are pleased to present you with this offer of at-will employment. This letter agreement ("Offer Letter") shall become effective upon the Closing (as defined in that certain Agreement and Plan of Merger, by and among the Company and Trawsfynydd Therapeutics, Inc., a Delaware corporation, and the other signatories thereto). The date on which this Offer Letter becomes effective is referred to herein as the "Effective Date." In the event that the Closing does not occur, this Offer Letter shall be null and void *ab initio*, and neither party hereto shall have any liability or obligation hereunder.

The details of your offer are as follows:

Title:

Reports to: [Chief Executive Officer of the Company]

Job Classification: [Full-Time/Part-Time]

FLSA Classification: Exempt

Base Salary: You will receive a base salary of \$[Salary] per year, which will be paid in accordance with the Company's payroll practices.

Discretionary Bonus: You shall be eligible to receive a discretionary annual bonus, as determined by the Board of Directors of the Company ("Board") or the Compensation Committee of the Board in its sole discretion ("Annual Bonus"). Any Annual Bonus shall be paid after the end of the fiscal year to which it relates, at the same time and under the same terms and conditions as the bonuses for other executives of the Company; provided that in no event will your Annual Bonus, if any, be paid later than two and a half months after the last day of the fiscal year to which the Annual Bonus relates. Any Annual Bonus may be paid in the form of cash, stock options, shares of the Company's common stock, or a combination thereof, at the Board's or the Compensation Committee's discretion.

Equity Awards. Subject to the approval of the Compensation Committee, as an inducement for you to join the Company in the role of [Title] and agree to the restrictive covenants set forth below, you will be granted [Number] restricted stock units, subject to the terms of the restricted stock unit agreement for inducement grants provided by the Company ("<u>Restricted Stock Units</u>"). The Restricted Stock Units will vest as to 25% on the first anniversary of the Effective Date and the remainder will vest in substantially equal annual installments for three years thereafter. The Restricted Stock Units are intended to be inducement awards under Rule 5635(c)(4) of the Nasdaq Stock Market Listing Rules and will be granted outside of the Onconova Therapeutics, Inc. 2021 Incentive Compensation Plan (the "<u>Plan</u>"). Although granted as inducement awards outside of the Plan, the Restricted Stock Units shall be subject to the terms of the Plan as if issued thereunder. In addition, you will be eligible to receive annual equity awards as part of the Company's overall bonus program.

Employee Benefits. During your employment, you may be eligible to participate in any employee benefit plans or programs of the Company that are made generally available from time to time by the Company to similarly situated employees, in accordance with the terms of the applicable plans and programs. Nothing in this Offer letter shall preclude the Company or any affiliate of the Company from terminating or amending any employee benefit plan or program from time to time after your start date.

All compensation payable and other benefits provided under this Offer Letter shall be subject to customary and legally required withholding for income and other employment taxes. You shall bear all expense of, and be solely responsible for, all federal, state and local taxes due with respect to any payment received under this Offer Letter.

Your employment with the Company would be at-will, which means that it may be terminated at any time by either you or the Company, and will be subject to the Company's applicable employment requirements and other policies, procedures and guidelines of the Company.

If you have not already done so, you must disclose to the Company any and all agreements relating to your prior employment that may affect your eligibility to be employed by the Company or limit the manner in which you may be employed. It is the Company's understanding that any such agreements will not prevent you from performing the duties of your position and by accepting this offer of employment, you represent that you are not under any obligation to a third party that would prevent you from performing the duties of your position with the Company. You further acknowledge and agree that you are not being asked to, and you agree not to bring any third-party confidential or proprietary information or trade secrets to the Company, including that of any former employer, or put such information onto Company systems, and that in performing your duties for the Company you will not in any way utilize such information.

[The Company's expectation is that you will devote your full business time and attention to your role with the Company, and that you will not maintain any employment or work, including as a consultant or independent contractor for any third party (including the continuation of any consulting work that predates your hire by the Company), outside of the Company without written approval of an authorized representative of the Company.]

This job offer is also contingent upon your agreement to the [Intellectual Property And Non-Disclosure Agreement], attached hereto as <u>Exhibit A</u>. As a condition of employment, you will be required to sign the [Intellectual Property And Non-Disclosure Agreement] before you start work. In addition, you are expected to comply with all applicable employment requirements and other policies, procedures and guidelines of the Company.

This Offer Letter and the compensation payable hereunder shall be subject to any applicable clawback or recoupment policies, share trading policies, and other policies that may be implemented by the Board from time to time with respect to officers of the Company.

On your first day of work with the Company, you will need to complete a standard employment application (if you have not yet completed one), IRS Form W-4, and relevant paperwork (Form I-9) to satisfy U.S. Department of Homeland Security requirements. You will need to supply documentation of both your identity and your eligibility for employment in the United States.

As an indication of your acceptance, please sign and return this letter to [Name], [Title] on or before [Date].

We are excited to have you join our team!

Accepted:

[Name of Prospective Employee]

Dated:

Onconova Therapeutics, Inc. and Trawsfynydd Therapeutics, Inc. Announce Business Combination to Form Traws Pharma, Inc, a Best-in-Class Virology and Oncology Company

- Closing cash balance of approximately \$28 million expected from concurrent private placement led by OrbiMed and Torrey Pines
- Funding expected to advance development of three potential best-in-class/class leading assets: viroksavir, a cap-dependent endonuclease inhibitor for influenza; travaltrelvir, a protease inhibitor for COVID19; and narazaciclib, a next generation CDK4/6 inhibitor for low grade endometriod endometrial cancer (LGEEC)
- Multiple near-term catalysts, for viroksavir and travaltrevir in 2024, with first top-line read-outs expected in H2 2024
- Traws Pharma to be led by incoming CEO, Werner Cautreels, Ph.D.
- Combined board to be led by Executive Chairman Iain Dukes DPhil (OrbiMed) along with Nikolay Savchuk, Ph.D. (Torrey Pines)
- Companies to host joint webcast, April 2, 2024 at 8:30 a.m. ET

NEWTOWN, PA and Rockville, MD – April 2, 2024 (GLOBE NEWSWIRE) – Onconova Therapeutics, Inc. (NASDAQ: ONTX) ("Onconova"), and Trawsfynydd Therapeutics, Inc. ("Trawsfynydd"), a privately-held biotechnology company developing next-generation, best-in-class antivirals for influenza, COVID and other infectious diseases, today announced that the companies have entered into a definitive merger agreement to combine in an all-stock transaction (the "Merger"). Under the terms of the agreement, Onconova acquired 100% of Trawsfynydd's outstanding equity interests. In connection with the transaction and concurrent with the Merger, the combined company which has been renamed "Traws Pharma, Inc." ("Traws") will trade on NASDAQ under the new ticker symbol "TRAW", commencing prior to the opening of trading Wednesday, April 3, 2024.

In connection with the Merger, Traws announced that it will raise \$14 million in a committed private placement financing by OrbiMed and Torrey Pines, expected to close on April 3, 2024. Upon closing of the private placement, Traws expects to have in excess of \$28 million of cash and cash equivalents from the proceeds of the private placement and cash from both companies. These proceeds will be used to advance the Traws' programs through multiple clinical data catalysts and complete the dose ranging study for narazaciclib.

"I am pleased to announce the combination of Onconova and Trawsfynydd at this important time, as Trawsfynydd readies to initiate Phase 2 studies in H2 2024 for its lead antiviral programs for influenza and COVID19, supported by advisors with unparalleled expertise in viral disease, and Onconova is preparing to finalize the recommended Phase 2 dose (RP2D) for narazaciclib," said Dr. Cautreels incoming Chief Executive Officer of the combined company.

Commented Steven Fruchtman, M.D., President and Chief Executive Officer of Onconova and President and CSO, Oncology of the combined company, "Trawsfynydd has a differentiated pipeline and an accomplished leadership team poised to advance their lead programs. With a shared focus on developing best-in-class medicines for patients with unmet needs, we look forward to Traws' continued progress with its anti-viral programs and narazaciclib."

Traws Proprietary Portfolio:

TRX100 (viroksavir): a cap-dependent endonuclease inhibitor for influenza: Phase 1

- Targets the cap-dependent endonuclease of influenza and is a potent inhibitor of influenza virus replication including A and B strains
- Preclinical data showed that TRX100 inhibits viral replication of pandemic-potential influenza viruses circulating in nature during 2022, and importantly, also in oseltamivir and baloxavir-resistant viruses
- · Completed a first Phase 1 study that demonstrated safety and tolerability in healthy volunteers. The study also provided pharmacokinetics and pharmacodynamics (PK/PD) data to support the potential use of a single oral dose administration for either treatment or prophylaxis

Next milestones: H2 2024

• Phase 1 dose extension will evaluate two additional, higher doses prior to the initiation of Phase 2 studies in H2 2024. Topline data from the Phase 2 study are expected in H1 2025

TRX01 (travaltrevir): Mpro protease inhibitor for COVID19: Phase 1

- Potent oral inhibitor of SARS-CoV-2 Mpro (3CL protease), effective against the original, delta, and omicron variants of SARS-CoV-2, with potentially superior properties to nirmatrelvir (Pfizer's Mpro inhibitor, PAXLOVID™)
- Does not require co-administration with a human cytochrome P450 (CYP) inhibitor such as ritonavir, avoiding potential significant drug:drug interactions, with the opportunity to expand the number of eligible patients
- Safe in GLP toxicology studies with no adverse events (AEs) in the expected human dose range. The drug candidate's pharmacokinetic (PK) profile may enable a once-daily 10 day treatment regimen, to reduce the likelihood of viral rebound

Next milestones:

- Phase 1 first-in-human single ascending dose/multiple ascending dose (SAD/MAD) study in normal volunteers initiated screening in Q1 2024. Topline data are expected H2 2024
- Phase 2 study planned to be initiated in H2 2024. The study will enroll people with moderate to severe COVID19. Topline data are expected H1 2025

Narazaciclib: CDK 4/6 inhibitor for LGEEC: Phase 1/2

- Narazaciclib's mechanism of action in LGEEC has been validated by Phase 2 studies with other approved CDK4/6 inhibitors: palbociclib (Pfizer), ribociclib (Novartis), and abemaciclib (Lilly). Available preclinical and clinical data suggest that narazaciclib has the potential to provide a better efficacy/safety ratio compared to approved products with respect to reduced gastrointestinal (GI) and hematological toxicities. These characteristics may permit daily administration with no need for the drug holidays employed by other approved agents to manage severe bone marrow suppression.
- In pre-clinical studies, narazaciclib demonstrated reduced neutropenia compared to palbociclib and inhibited the growth of cancer cell lines resistant to
 palbociclib
- · Currently in Phase 1/2a study to define the RP2D

Next milestone:

· Define the RP2D and development strategy for LGEEC/other indications

Management and Organization

Traws will be led by incoming Chief Executive Officer, Werner Cautreels, Ph.D.; President and Chief Scientific Officer, Oncology, Steven Fruchtman M.D., (Onconova); Chief Financial Officer, Mark Guerin (Onconova), Chief Medical Officer, Robert Redfield, M.D., (Trawsfynydd), Chief Scientific Officer, Virology, C. David Pauza, Ph.D., (Trawsfynydd) and Chief Operating Officer, Nikolay Savchuk, Ph.D., (Trawsfynydd/Torrey Pines), as well as several other members of the Onconova and Trawsfynydd teams.

Traws' Board of Directors will be comprised of Trawsfynydd's Chairman Iain Dukes, DPhil (Venture Partner at OrbiMed), Executive Chairman, Werner Cautreels, Nikolay Savchuk, Ph.D. (General Partner of Torrey Pines) as well as existing Onconova Directors Trafford Clarke, Ph.D, James Marino, J.D. and M. Theresa Shoemaker and Jack E. Stover.

About the Merger and Private Financing

Onconova issued the following in the transactions: in connection with the merger, the stockholders of Traswfynydd received an aggregate of 3,549,538 shares of common stock and 10,359.0916 shares of newly issued Series C non-voting convertible preferred stock (with a conversion ratio of preferred to common at 1:10,000) (the "Series C preferred stock"), and in connection with the private financing, OrbiMed and Torrey Pines received an aggregate of 496,935 shares of common stock and 1,578.2120 shares of Series C preferred stock. This represents, on a fully diluted basis, 75.7% for Trawsfynydd, 13.7% for Onconova and 10.6% for new investors with a combined fully diluted equity value of \$132 million (excluding transaction fees). In connection with the transactions, a non-transferrable contingent value right (a "CVR") will be distributed to Onconova stockholders of record as of the close of business on April 15, 2024. Holders of the CVR will be entitled to receive certain proceeds received by Onconova, if any, related to the disposition, net sales or monetization of narazaciclib and rigosertib.

The shares of common stock issuable upon conversion of the Series C preferred stock issued in the Merger and the private financing shall be subject to stockholder approval in compliance with the rules of the NASDAQ Stock Market.

Tungsten Advisors served as the exclusive financial advisor and placement agent to Onconova. Orrick, Herrington & Sutcliffe, LLP and Morgan, Lewis & Bockius LLP are serving as legal counsel to Onconova. Snell & Wilmer L.L.P. is serving as legal counsel to Trawsfynydd.

Webcast Presentation

The companies will host a webcast presentation to discuss the proposed transaction tomorrow, April 2 at 8:30 a.m. ET.

Dial-in details are:

- · Investors Dial-in: 1-877-407-0784
- · International Investors Dial-in: 1-201-689-8560
- · Conference ID: 13745512

Call me^{TM} : Participants can use Guest dial-in #s above and be answered by an operator OR click the Call me^{TM} link for instant telephone access to the event.

- https://callme.viavid.com/viavid/?callme=true&passcode=13745512&h=true&info=company-email&r=true&B=6
- · Call me[™] link will be made active 15 minutes prior to scheduled start time.

Webcast: https://viavid.webcasts.com/starthere.jsp?ei=1663864&tp_key=8103bfd962:

A replay of the webcast will also be available via Onconova's investor website approximately two hours after the call's conclusion.

About Onconova Therapeutics, Inc.

Onconova is a clinical-stage biopharmaceutical company focused on discovering and developing novel products for patients with cancer. Onconova has been focused on developing targeted anti-cancer agents designed to disrupt specific cellular pathways that are important for cancer cell proliferation including the novel, multi-kinase inhibitor narazaciclib, under evaluation for low grade endometroid endometrial cancer (LGEEC).

About Trawsfynydd Therapeutics, Inc.

Trawsfynydd is an antiviral drug development company committed to reducing disease and death among people with respiratory viral diseases including Influenza and COVID19. Trawsfynydd combines excellence in medicinal chemistry with the tools of AI and machine learning, to guide and accelerate the drug development process. We are especially committed to developing best-in-class treatments for the elderly, immunocompromised and other vulnerable populations who are at increased risk for severe disease and death from respiratory virus infections.

About Traws Pharma, Inc.

Traws is developing next-generation, best-in-class antivirals for influenza, COVID and other respiratory infections and narazaciclib. Traws was formed from the business combination of Onconova Therapeutics, Inc. and Trawsfynydd Therapeutics, Inc. and is headquartered in Newtown, PA and will trade on NASDAQ as "TRAW" starting on April 3, 2024.

Forward-Looking Statements

This communication contains "forward-looking statements" within the meaning of the "safe harbor" provisions of the Private Securities Litigation Reform Act of 1995, including but not limited to, express or implied statements regarding the structure, the combined company's listing on Nasdaq; expectations regarding the ownership structure of the combined company; expectations regarding the issuance and value of CVRs; expectations regarding the conversion of the Series C preferred stock and related stockholder approval; expectations regarding the structure, timing and funding of the private placement financing, expected proceeds and impact on ownership structure; each company's and the combined company's expected cash position at the closing of the Merger and the expected cash runway of the combined company following the Merger and private financing; the nature, strategy and focus of the combined company; the development and commercial potential and potential benefits of any product candidates of the combined company; anticipated clinical drug development activities, related timelines and expected milestones; and other statements that are not historical fact. All statements other than statements of historical fact contained in this communication are forward-looking statements. These forward-looking statements are made as of the date they were first issued, and were based on the then-current expectations, estimates, forecasts, and projections, as well as the beliefs and assumptions of management. There can be no assurance that future developments affecting Traws, the Merger or the private placement financing will be those that have been anticipated. Forward-looking statements are subject to a number of risks and uncertainties, many of which involve factors or circumstances that are beyond Traws' control. Traws' actual results could differ materially from those stated or implied in forward-looking statements due to a number of factors, including but not limited to (i) failure to timely obtain stockholder approval for the transaction, if at all; (ii) uncertainties as to the timing and the funding of the private financing; (iii) risks related to Traws' ability to manage its operating expenses and its expenses associated with the Merger; (iv) unexpected costs, charges or expenses resulting from the transaction; (v) potential adverse reactions or changes to business relationships resulting from the announcement or completion of the Merger; (vi) the uncertainties associated with Traws' product candidates, as well as risks associated with the clinical development and regulatory approval of product candidates, including potential delays in the commencement and completion of clinical trials, studies and evaluations: (vii) risks related to the inability of the combined company to obtain sufficient additional capital to continue to advance these or other product candidates; (viii) uncertainties in obtaining successful clinical results for product candidates and unexpected costs that may result therefrom; (ix) risks related to the failure to realize any value from product candidates currently being developed and anticipated to be developed in light of inherent risks and difficulties involved in successfully bringing product candidates to market; and (x) risks associated with the possible failure to realize certain anticipated benefits of the Merger, including with respect to future financial and operating results. Actual results and the timing of events could differ materially from those anticipated in such forward-looking statements as a result of these risks and uncertainties. These and other risks and uncertainties are more fully described in periodic filings with the U.S. Securities and Exchange Commission (the "SEC"), including the factors described in the section titled "Risk Factors" in Onconova's Annual Report on Form 10-K for the year ended December 31, 2023, which was filed with the SEC on April 1 2024, subsequent Quarterly Reports on Form 10-Q filed with the SEC, and in other filings that Traws makes and will make with the SEC. You should not place undue reliance on these forward-looking statements, which are made only as of the date hereof or as of the dates indicated in the forward-looking statements. Traws expressly disclaims any obligation or undertaking to release publicly any updates or revisions to any forward-looking statements contained herein to reflect any change in its expectations with regard thereto or any change in events, conditions or circumstances on which any such statements are based. This communication does not purport to summarize all of the conditions, risks and other attributes of an investment in Traws.

Company Contacts:

Mark Guerin Onconova Therapeutics, Inc. 267-759-3680 <u>ir@onconova.us</u> <u>https://www.onconova.com/contact/</u>

Investor Contact:

Bruce Mackle LifeSci Advisors, LLC 646-889-1200 bmackle@lifesciadvisors.com